

30th ANNUAL WILLEM C. VIS INTERNATIONAL COMMERCIAL ARBITRATION MOOT 31 MARCH – 6 APRIL, 2023, VIENNA

RULES BOOKLET



ភ្នំផ្

HI HI H



30th ANNUAL WILLEM C. VIS INTERNATIONAL COMMERCIAL ARBITRATION MOOT 31 MARCH – 6 APRIL, 2023, VIENNA

RULES BOOKLET

TABLE OF CONTENTS

United Nations Convention on Contracts for the International Sale of Goods
Unidroit Principles of International Commercial Contracts 201656
Permanent Court of Arbitration Arbitration Rules 201291
UNCITRAL Model Law on International Commercial Arbitration
Convention on the Recognition and Enforcement of Foreign Arbitral Awards 165
United Nations Convention against Corruption
UNCITRAL Arbitration Rules (2021) UNCITRAL Expedited Arbitration Rules UNCITRAL Rules on Transparency in Treaty-based Investor-State Arbitration

Win

4 *2-1



UNITED NATIONS CONVENTION ON CONTRACTS FOR THE INTERNATIONAL SALE OF GOODS

//**ភិ**ជា || || || ខ

United Nations Convention on Contracts for the International Sale of Goods





UNITED NATIONS

Further information may be obtained from:

UNCITRAL secretariat, Vienna International Centre, P.O. Box 500, 1400 Vienna, Austria

Telephone: (+43-1) 26060-4060 Internet: uncitral.un.org Telefax: (+43-1) 26060-5813 Email: uncitral@un.org

United Nations Convention on Contracts for the International Sale of Goods



UNITED NATIONS New York, 2010

NOTE

Symbols of United Nations documents are composed of capital letters combined with figures. Mention of such a symbol indicates a reference to a United Nations document.

UNITED NATIONS PUBLICATION Sales No. E.10.V.14 ISBN 978-92-1-133699-3

© United Nations, November 2010. All rights reserved.

The designations employed and the presentation of material in this publication do not imply the expression of any opinion whatsoever on the part of the Secretariat of the United Nations concerning the legal status of any country, territory, city or area, or of its authorities, or concerning the delimitation of its frontiers or boundaries.

Publishing production: English, Publishing and Library Section, United Nations Office at Vienna.

This publication has not been formally edited.

Contents

I.	United Nations Convention on Contracts for the International Sale of Goods	1
	PREAMBLE	1
	Part I. Sphere of application and general provisions	1
	CHAPTER I. Sphere of application	1
	Article 1	1
	Article 2	2
	Article 3	2
	Article 4	2
	Article 5	3
	Article 6	3
		_
	CHAPTER II. General provisions	3
	Article 7	3
	Article 8	3
	Article 9	4
	Article 10	4
	Article 11	4
	Article 12	4
	Article 13	5
	Part II. Formation of the contract	5
	Article 14	5
	Article 15	5
	Article 16	5
	Article 17	6
	Article 18	6
	Article 19	6
	Article 20	7
	Article 21	7
	Article 22	7
	Article 23	7
	Article 24	7

Part III. Sale of goods	8
CHAPTER I. General provisions	8
Article 25	8
Article 26	8
Article 27	8
Article 28	8
Article 29	8
CHAPTER II. Obligations of the seller	9
Article 30	9
Section I. Delivery of the goods and handing over of documents	9
Article 31	9
Article 32	9
Article 33	10
Article 34	10
Section II. Conformity of the goods and third-party claims	10
Article 35	10
Article 36	11
Article 37	11
Article 38	11
Article 39	12
Article 40	12
Article 41	12
Article 42	12
Article 43	13
Article 44	13
Section III. Remedies for breach of contract by the seller	13
Article 45	13
Article 46	14
Article 47	14
Article 48	14
Article 49	15
Article 50	16
Article 51	16
Article 52	16
CHAPTER III. Obligations of the buyer	16
Article 53	16

Section I. Payment of the price	17
Article 54	17
Article 55	17
Article 56	17
Article 57	17
Article 58	17
Article 59	18
Section II. Taking delivery	18
Article 60	
Section III. Remedies for breach of contract by the buyer	18
Article 62	
Article 63	
Article 64	
Article 65	
Alticle 05	20
CHAPTER IV. Passing of risk	20
Article 66	20
Article 67	20
Article 68	21
Article 69	21
Article 70	21
CHAPTER V. Provisions common to the obligations of the seller and of the buyer	01
and of the buyer	21
Section I. Anticipatory breach and instalment contracts	21
Article 71	21
Article 72	22
Article 73	22
Section II. Damages	23
Article 74	23
Article 75	23
Article 76	23
Article 77	24
Section III. Interest	24
Section III. Interest Article 78	24 24
Article 78 Section IV. Exemptions	24
Article 78	24 24 24

	Section V. Effects of avoidance	25
	Article 81	25
	Article 82	25
	Article 83	26
	Article 84	26
	Section VI. Preservation of the goods	26
	Article 85	26
	Article 86	26
	Article 87	27
	Article 88	27
	Part IV. Final provisions	27
	Article 89	27
	Article 90	27
	Article 91	
	Article 92	
	Article 93	
	Article 94	
	Article 95	
	Article 96	
	Article 97	
	Article 98	
	Article 99	
	Article 100	
	Article 101	32
II.	Explanatory note by the UNCITRAL Secretariat on the United Nations Convention on Contracts for the International Sale of Goods	33
	Introduction	33
	Part One. Scope of application and general provisions	34
	A. Scope of application	34
		35
	C. Interpretation of the Convention	36
	D. Interpretation of the contract; usages	
	E. Form of the contract	30
	Part Two. Formation of the contract	37

Par	t Three. Sale of goods	38
A.	Obligations of the seller	38
B.	Obligations of the buyer	38
C.	Remedies for breach of contract	39
D.	Passing of risk	39
E.	Suspension of performance and anticipatory breach	40
F.	Exemption from liability to pay damages	40
G.	Preservation of the goods	40
Par	t Four. Final clauses	41
Con	nplementary texts	41

1. United Nations Convention on Contracts for the International Sale of Goods

PREAMBLE

The States Parties to this Convention,

Bearing in mind the broad objectives in the resolutions adopted by the sixth special session of the General Assembly of the United Nations on the establishment of a New International Economic Order,

Considering that the development of international trade on the basis of equality and mutual benefit is an important element in promoting friendly relations among States,

Being of the opinion that the adoption of uniform rules which govern contracts for the international sale of goods and take into account the different social, economic and legal systems would contribute to the removal of legal barriers in international trade and promote the development of international trade,

Have agreed as follows:

Part I. Sphere of application and general provisions

CHAPTER I. SPHERE OF APPLICATION

Article 1

(1) This Convention applies to contracts of sale of goods between parties whose places of business are in different States:

(a) when the States are Contracting States; or

(b) when the rules of private international law lead to the application of the law of a Contracting State.

(2) The fact that the parties have their places of business in different States is to be disregarded whenever this fact does not appear either from the contract or from any dealings between, or from information disclosed by, the parties at any time before or at the conclusion of the contract.

(3) Neither the nationality of the parties nor the civil or commercial character of the parties or of the contract is to be taken into consideration in determining the application of this Convention.

Article 2

This Convention does not apply to sales:

(*a*) of goods bought for personal, family or household use, unless the seller, at any time before or at the conclusion of the contract, neither knew nor ought to have known that the goods were bought for any such use;

(b) by auction;

(c) on execution or otherwise by authority of law;

(d) of stocks, shares, investment securities, negotiable instruments or money;

(e) of ships, vessels, hovercraft or aircraft;

(f) of electricity.

Article 3

(1) Contracts for the supply of goods to be manufactured or produced are to be considered sales unless the party who orders the goods undertakes to supply a substantial part of the materials necessary for such manufacture or production.

(2) This Convention does not apply to contracts in which the preponderant part of the obligations of the party who furnishes the goods consists in the supply of labour or other services.

Article 4

This Convention governs only the formation of the contract of sale and the rights and obligations of the seller and the buyer arising from such a contract. In particular, except as otherwise expressly provided in this Convention, it is not concerned with: (a) the validity of the contract or of any of its provisions or of any usage;

(b) the effect which the contract may have on the property in the goods sold.

Article 5

This Convention does not apply to the liability of the seller for death or personal injury caused by the goods to any person.

Article 6

The parties may exclude the application of this Convention or, subject to article 12, derogate from or vary the effect of any of its provisions.

CHAPTER II. GENERAL PROVISIONS

Article 7

(1) In the interpretation of this Convention, regard is to be had to its international character and to the need to promote uniformity in its application and the observance of good faith in international trade.

(2) Questions concerning matters governed by this Convention which are not expressly settled in it are to be settled in conformity with the general principles on which it is based or, in the absence of such principles, in conformity with the law applicable by virtue of the rules of private international law.

Article 8

(1) For the purposes of this Convention statements made by and other conduct of a party are to be interpreted according to his intent where the other party knew or could not have been unaware what that intent was.

(2) If the preceding paragraph is not applicable, statements made by and other conduct of a party are to be interpreted according to the understanding that a reasonable person of the same kind as the other party would have had in the same circumstances. (3) In determining the intent of a party or the understanding a reasonable person would have had, due consideration is to be given to all relevant circumstances of the case including the negotiations, any practices which the parties have established between themselves, usages and any subsequent conduct of the parties.

Article 9

(1) The parties are bound by any usage to which they have agreed and by any practices which they have established between themselves.

(2) The parties are considered, unless otherwise agreed, to have impliedly made applicable to their contract or its formation a usage of which the parties knew or ought to have known and which in international trade is widely known to, and regularly observed by, parties to contracts of the type involved in the particular trade concerned.

Article 10

For the purposes of this Convention:

(*a*) if a party has more than one place of business, the place of business is that which has the closest relationship to the contract and its performance, having regard to the circumstances known to or contemplated by the parties at any time before or at the conclusion of the contract;

(b) if a party does not have a place of business, reference is to be made to his habitual residence.

Article 11

A contract of sale need not be concluded in or evidenced by writing and is not subject to any other requirement as to form. It may be proved by any means, including witnesses.

Article 12

Any provision of article 11, article 29 or Part II of this Convention that allows a contract of sale or its modification or termination by agreement or any offer, acceptance or other indication of intention to be made in any form other than in writing does not apply where any party has his place of business in a Contracting State which has made a declaration under article 96 of this Convention. The parties may not derogate from or vary the effect of this article.

Article 13

For the purposes of this Convention "writing" includes telegram and telex.

Part II. Formation of the contract

Article 14

(1) A proposal for concluding a contract addressed to one or more specific persons constitutes an offer if it is sufficiently definite and indicates the intention of the offeror to be bound in case of acceptance. A proposal is sufficiently definite if it indicates the goods and expressly or implicitly fixes or makes provision for determining the quantity and the price.

(2) A proposal other than one addressed to one or more specific persons is to be considered merely as an invitation to make offers, unless the contrary is clearly indicated by the person making the proposal.

Article 15

(1) An offer becomes effective when it reaches the offeree.

(2) An offer, even if it is irrevocable, may be withdrawn if the withdrawal reaches the offeree before or at the same time as the offer.

Article 16

(1) Until a contract is concluded an offer may be revoked if the revocation reaches the offeree before he has dispatched an acceptance.

(2) However, an offer cannot be revoked:

(a) if it indicates, whether by stating a fixed time for acceptance or otherwise, that it is irrevocable; or

(b) if it was reasonable for the offeree to rely on the offer as being irrevocable and the offeree has acted in reliance on the offer.

Article 17

An offer, even if it is irrevocable, is terminated when a rejection reaches the offeror.

Article 18

(1) A statement made by or other conduct of the offeree indicating assent to an offer is an acceptance. Silence or inactivity does not in itself amount to acceptance.

(2) An acceptance of an offer becomes effective at the moment the indication of assent reaches the offeror. An acceptance is not effective if the indication of assent does not reach the offeror within the time he has fixed or, if no time is fixed, within a reasonable time, due account being taken of the circumstances of the transaction, including the rapidity of the means of communication employed by the offeror. An oral offer must be accepted immediately unless the circumstances indicate otherwise.

(3) However, if, by virtue of the offer or as a result of practices which the parties have established between themselves or of usage, the offeree may indicate assent by performing an act, such as one relating to the dispatch of the goods or payment of the price, without notice to the offeror, the acceptance is effective at the moment the act is performed, provided that the act is performed within the period of time laid down in the preceding paragraph.

Article 19

(1) A reply to an offer which purports to be an acceptance but contains additions, limitations or other modifications is a rejection of the offer and constitutes a counter-offer.

(2) However, a reply to an offer which purports to be an acceptance but contains additional or different terms which do not materially alter the terms of the offer constitutes an acceptance, unless the offeror, without undue delay, objects orally to the discrepancy or dispatches a notice to that effect. If he does not so object, the terms of the contract are the terms of the offer with the modifications contained in the acceptance.

(3) Additional or different terms relating, among other things, to the price, payment, quality and quantity of the goods, place and time of delivery, extent of one party's liability to the other or the settlement of disputes are considered to alter the terms of the offer materially.

Article 20

(1) A period of time for acceptance fixed by the offeror in a telegram or a letter begins to run from the moment the telegram is handed in for dispatch or from the date shown on the letter or, if no such date is shown, from the date shown on the envelope. A period of time for acceptance fixed by the offeror by telephone, telex or other means of instantaneous communication, begins to run from the moment that the offer reaches the offeree.

(2) Official holidays or non-business days occurring during the period for acceptance are included in calculating the period. However, if a notice of acceptance cannot be delivered at the address of the offeror on the last day of the period because that day falls on an official holiday or a non-business day at the place of business of the offeror, the period is extended until the first business day which follows.

Article 21

(1) A late acceptance is nevertheless effective as an acceptance if without delay the offeror orally so informs the offeree or dispatches a notice to that effect.

(2) If a letter or other writing containing a late acceptance shows that it has been sent in such circumstances that if its transmission had been normal it would have reached the offeror in due time, the late acceptance is effective as an acceptance unless, without delay, the offeror orally informs the offeree that he considers his offer as having lapsed or dispatches a notice to that effect.

Article 22

An acceptance may be withdrawn if the withdrawal reaches the offeror before or at the same time as the acceptance would have become effective.

Article 23

A contract is concluded at the moment when an acceptance of an offer becomes effective in accordance with the provisions of this Convention.

Article 24

For the purposes of this Part of the Convention, an offer, declaration of acceptance or any other indication of intention "reaches" the addressee when it is made orally to him or delivered by any other means to him personally, to his place of business or mailing address or, if he does not have a place of business or mailing address, to his habitual residence.

Part III. Sale of goods

CHAPTER I. GENERAL PROVISIONS

Article 25

A breach of contract committed by one of the parties is fundamental if it results in such detriment to the other party as substantially to deprive him of what he is entitled to expect under the contract, unless the party in breach did not foresee and a reasonable person of the same kind in the same circumstances would not have foreseen such a result.

Article 26

A declaration of avoidance of the contract is effective only if made by notice to the other party.

Article 27

Unless otherwise expressly provided in this Part of the Convention, if any notice, request or other communication is given or made by a party in accordance with this Part and by means appropriate in the circumstances, a delay or error in the transmission of the communication or its failure to arrive does not deprive that party of the right to rely on the communication.

Article 28

If, in accordance with the provisions of this Convention, one party is entitled to require performance of any obligation by the other party, a court is not bound to enter a judgement for specific performance unless the court would do so under its own law in respect of similar contracts of sale not governed by this Convention.

Article 29

(1) A contract may be modified or terminated by the mere agreement of the parties.

(2) A contract in writing which contains a provision requiring any modification or termination by agreement to be in writing may not be otherwise modified or terminated by agreement. However, a party may be precluded by his conduct from asserting such a provision to the extent that the other party has relied on that conduct.

CHAPTER II. OBLIGATIONS OF THE SELLER

Article 30

The seller must deliver the goods, hand over any documents relating to them and transfer the property in the goods, as required by the contract and this Convention.

Section I. Delivery of the goods and handing over of documents

Article 31

If the seller is not bound to deliver the goods at any other particular place, his obligation to deliver consists:

(a) if the contract of sale involves carriage of the goods—in handing the goods over to the first carrier for transmission to the buyer;

(b) if, in cases not within the preceding subparagraph, the contract relates to specific goods, or unidentified goods to be drawn from a specific stock or to be manufactured or produced, and at the time of the conclusion of the contract the parties knew that the goods were at, or were to be manufactured or produced at, a particular place—in placing the goods at the buyer's disposal at that place;

(c) in other cases—in placing the goods at the buyer's disposal at the place where the seller had his place of business at the time of the conclusion of the contract.

Article 32

(1) If the seller, in accordance with the contract or this Convention, hands the goods over to a carrier and if the goods are not clearly identified to the contract by markings on the goods, by shipping documents or otherwise, the seller must give the buyer notice of the consignment specifying the goods. (2) If the seller is bound to arrange for carriage of the goods, he must make such contracts as are necessary for carriage to the place fixed by means of transportation appropriate in the circumstances and according to the usual terms for such transportation.

(3) If the seller is not bound to effect insurance in respect of the carriage of the goods, he must, at the buyer's request, provide him with all available information necessary to enable him to effect such insurance.

Article 33

The seller must deliver the goods:

(a) if a date is fixed by or determinable from the contract, on that date;

(b) if a period of time is fixed by or determinable from the contract, at any time within that period unless circumstances indicate that the buyer is to choose a date; or

(c) in any other case, within a reasonable time after the conclusion of the contract.

Article 34

If the seller is bound to hand over documents relating to the goods, he must hand them over at the time and place and in the form required by the contract. If the seller has handed over documents before that time, he may, up to that time, cure any lack of conformity in the documents, if the exercise of this right does not cause the buyer unreasonable inconvenience or unreasonable expense. However, the buyer retains any right to claim damages as provided for in this Convention.

Section II. Conformity of the goods and third-party claims

Article 35

(1) The seller must deliver goods which are of the quantity, quality and description required by the contract and which are contained or packaged in the manner required by the contract.

(2) Except where the parties have agreed otherwise, the goods do not conform with the contract unless they:

(a) are fit for the purposes for which goods of the same description would ordinarily be used;

(b) are fit for any particular purpose expressly or impliedly made known to the seller at the time of the conclusion of the contract, except where the circumstances show that the buyer did not rely, or that it was unreasonable for him to rely, on the seller's skill and judgement;

(c) possess the qualities of goods which the seller has held out to the buyer as a sample or model;

(d) are contained or packaged in the manner usual for such goods or, where there is no such manner, in a manner adequate to preserve and protect the goods.

(3) The seller is not liable under subparagraphs (a) to (d) of the preceding paragraph for any lack of conformity of the goods if, at the time of the conclusion of the contract, the buyer knew or could not have been unaware of such lack of conformity.

Article 36

(1) The seller is liable in accordance with the contract and this Convention for any lack of conformity which exists at the time when the risk passes to the buyer, even though the lack of conformity becomes apparent only after that time.

(2) The seller is also liable for any lack of conformity which occurs after the time indicated in the preceding paragraph and which is due to a breach of any of his obligations, including a breach of any guarantee that for a period of time the goods will remain fit for their ordinary purpose or for some particular purpose or will retain specified qualities or characteristics.

Article 37

If the seller has delivered goods before the date for delivery, he may, up to that date, deliver any missing part or make up any deficiency in the quantity of the goods delivered, or deliver goods in replacement of any non-conforming goods delivered or remedy any lack of conformity in the goods delivered, provided that the exercise of this right does not cause the buyer unreasonable inconvenience or unreasonable expense. However, the buyer retains any right to claim damages as provided for in this Convention.

Article 38

(1) The buyer must examine the goods, or cause them to be examined, within as short a period as is practicable in the circumstances.

(2) If the contract involves carriage of the goods, examination may be deferred until after the goods have arrived at their destination.

(3) If the goods are redirected in transit or redispatched by the buyer without a reasonable opportunity for examination by him and at the time of the conclusion of the contract the seller knew or ought to have known of the possibility of such redirection or redispatch, examination may be deferred until after the goods have arrived at the new destination.

Article 39

(1) The buyer loses the right to rely on a lack of conformity of the goods if he does not give notice to the seller specifying the nature of the lack of conformity within a reasonable time after he has discovered it or ought to have discovered it.

(2) In any event, the buyer loses the right to rely on a lack of conformity of the goods if he does not give the seller notice thereof at the latest within a period of two years from the date on which the goods were actually handed over to the buyer, unless this time limit is inconsistent with a contractual period of guarantee.

Article 40

The seller is not entitled to rely on the provisions of articles 38 and 39 if the lack of conformity relates to facts of which he knew or could not have been unaware and which he did not disclose to the buyer.

Article 41

The seller must deliver goods which are free from any right or claim of a third party, unless the buyer agreed to take the goods subject to that right or claim. However, if such right or claim is based on industrial property or other intellectual property, the seller's obligation is governed by article 42.

Article 42

(1) The seller must deliver goods which are free from any right or claim of a third party based on industrial property or other intellectual property, of which at the time of the conclusion of the contract the seller knew or could not have been unaware, provided that the right or claim is based on industrial property or other intellectual property: (a) under the law of the State where the goods will be resold or otherwise used, if it was contemplated by the parties at the time of the conclusion of the contract that the goods would be resold or otherwise used in that State; or

(b) in any other case, under the law of the State where the buyer has his place of business.

(2) The obligation of the seller under the preceding paragraph does not extend to cases where:

(a) at the time of the conclusion of the contract the buyer knew or could not have been unaware of the right or claim; or

(b) the right or claim results from the seller's compliance with technical drawings, designs, formulae or other such specifications furnished by the buyer.

Article 43

(1) The buyer loses the right to rely on the provisions of article 41 or article 42 if he does not give notice to the seller specifying the nature of the right or claim of the third party within a reasonable time after he has become aware or ought to have become aware of the right or claim.

(2) The seller is not entitled to rely on the provisions of the preceding paragraph if he knew of the right or claim of the third party and the nature of it.

Article 44

Notwithstanding the provisions of paragraph (1) of article 39 and paragraph (1) of article 43, the buyer may reduce the price in accordance with article 50 or claim damages, except for loss of profit, if he has a reasonable excuse for his failure to give the required notice.

Section III. Remedies for breach of contract by the seller

Article 45

(1) If the seller fails to perform any of his obligations under the contract or this Convention, the buyer may:

- (a) exercise the rights provided in articles 46 to 52;
- (b) claim damages as provided in articles 74 to 77.

(2) The buyer is not deprived of any right he may have to claim damages by exercising his right to other remedies.

(3) No period of grace may be granted to the seller by a court or arbitral tribunal when the buyer resorts to a remedy for breach of contract.

Article 46

(1) The buyer may require performance by the seller of his obligations unless the buyer has resorted to a remedy which is inconsistent with this requirement.

(2) If the goods do not conform with the contract, the buyer may require delivery of substitute goods only if the lack of conformity constitutes a fundamental breach of contract and a request for substitute goods is made either in conjunction with notice given under article 39 or within a reasonable time thereafter.

(3) If the goods do not conform with the contract, the buyer may require the seller to remedy the lack of conformity by repair, unless this is unreasonable having regard to all the circumstances. A request for repair must be made either in conjunction with notice given under article 39 or within a reasonable time thereafter.

Article 47

(1) The buyer may fix an additional period of time of reasonable length for performance by the seller of his obligations.

(2) Unless the buyer has received notice from the seller that he will not perform within the period so fixed, the buyer may not, during that period, resort to any remedy for breach of contract. However, the buyer is not deprived thereby of any right he may have to claim damages for delay in performance.

Article 48

(1) Subject to article 49, the seller may, even after the date for delivery, remedy at his own expense any failure to perform his obligations, if he can do so without unreasonable delay and without causing the buyer unreasonable inconvenience or uncertainty of reimbursement by the seller of expenses

advanced by the buyer. However, the buyer retains any right to claim damages as provided for in this Convention.

(2) If the seller requests the buyer to make known whether he will accept performance and the buyer does not comply with the request within a reasonable time, the seller may perform within the time indicated in his request. The buyer may not, during that period of time, resort to any remedy which is inconsistent with performance by the seller.

(3) A notice by the seller that he will perform within a specified period of time is assumed to include a request, under the preceding paragraph, that the buyer make known his decision.

(4) A request or notice by the seller under paragraph (2) or (3) of this article is not effective unless received by the buyer.

Article 49

(1) The buyer may declare the contract avoided:

(a) if the failure by the seller to perform any of his obligations under the contract or this Convention amounts to a fundamental breach of contract; or

(b) in case of non-delivery, if the seller does not deliver the goods within the additional period of time fixed by the buyer in accordance with paragraph (1) of article 47 or declares that he will not deliver within the period so fixed.

(2) However, in cases where the seller has delivered the goods, the buyer loses the right to declare the contract avoided unless he does so:

(a) in respect of late delivery, within a reasonable time after he has become aware that delivery has been made;

(b) in respect of any breach other than late delivery, within a reasonable time:

(i) after he knew or ought to have known of the breach;

(ii) after the expiration of any additional period of time fixed by the buyer in accordance with paragraph (1) of article 47, or after the seller has declared that he will not perform his obligations within such an additional period; or

(iii) after the expiration of any additional period of time indicated by the seller in accordance with paragraph (2) of article 48, or after the buyer has declared that he will not accept performance.

Article 50

If the goods do not conform with the contract and whether or not the price has already been paid, the buyer may reduce the price in the same proportion as the value that the goods actually delivered had at the time of the delivery bears to the value that conforming goods would have had at that time. However, if the seller remedies any failure to perform his obligations in accordance with article 37 or article 48 or if the buyer refuses to accept performance by the seller in accordance with those articles, the buyer may not reduce the price.

Article 51

(1) If the seller delivers only a part of the goods or if only a part of the goods delivered is in conformity with the contract, articles 46 to 50 apply in respect of the part which is missing or which does not conform.

(2) The buyer may declare the contract avoided in its entirety only if the failure to make delivery completely or in conformity with the contract amounts to a fundamental breach of the contract.

Article 52

(1) If the seller delivers the goods before the date fixed, the buyer may take delivery or refuse to take delivery.

(2) If the seller delivers a quantity of goods greater than that provided for in the contract, the buyer may take delivery or refuse to take delivery of the excess quantity. If the buyer takes delivery of all or part of the excess quantity, he must pay for it at the contract rate.

CHAPTER III. OBLIGATIONS OF THE BUYER

Article 53

The buyer must pay the price for the goods and take delivery of them as required by the contract and this Convention.

Section I. Payment of the price

Article 54

The buyer's obligation to pay the price includes taking such steps and complying with such formalities as may be required under the contract or any laws and regulations to enable payment to be made.

Article 55

Where a contract has been validly concluded but does not expressly or implicitly fix or make provision for determining the price, the parties are considered, in the absence of any indication to the contrary, to have impliedly made reference to the price generally charged at the time of the conclusion of the contract for such goods sold under comparable circumstances in the trade concerned.

Article 56

If the price is fixed according to the weight of the goods, in case of doubt it is to be determined by the net weight.

Article 57

(1) If the buyer is not bound to pay the price at any other particular place, he must pay it to the seller:

(a) at the seller's place of business; or

(b) if the payment is to be made against the handing over of the goods or of documents, at the place where the handing over takes place.

(2) The seller must bear any increase in the expenses incidental to payment which is caused by a change in his place of business subsequent to the conclusion of the contract.

Article 58

(1) If the buyer is not bound to pay the price at any other specific time, he must pay it when the seller places either the goods or documents controlling their disposition at the buyer's disposal in accordance with the

contract and this Convention. The seller may make such payment a condition for handing over the goods or documents.

(2) If the contract involves carriage of the goods, the seller may dispatch the goods on terms whereby the goods, or documents controlling their disposition, will not be handed over to the buyer except against payment of the price.

(3) The buyer is not bound to pay the price until he has had an opportunity to examine the goods, unless the procedures for delivery or payment agreed upon by the parties are inconsistent with his having such an opportunity.

Article 59

The buyer must pay the price on the date fixed by or determinable from the contract and this Convention without the need for any request or compliance with any formality on the part of the seller.

Section II. Taking delivery

Article 60

The buyer's obligation to take delivery consists:

(a) in doing all the acts which could reasonably be expected of him in order to enable the seller to make delivery; and

(b) in taking over the goods.

Section III. Remedies for breach of contract by the buyer

Article 61

(1) If the buyer fails to perform any of his obligations under the contract or this Convention, the seller may:

(a) exercise the rights provided in articles 62 to 65;

(b) claim damages as provided in articles 74 to 77.

(2) The seller is not deprived of any right he may have to claim damages by exercising his right to other remedies.

(3) No period of grace may be granted to the buyer by a court or arbitral tribunal when the seller resorts to a remedy for breach of contract.

Article 62

The seller may require the buyer to pay the price, take delivery or perform his other obligations, unless the seller has resorted to a remedy which is inconsistent with this requirement.

Article 63

(1) The seller may fix an additional period of time of reasonable length for performance by the buyer of his obligations.

(2) Unless the seller has received notice from the buyer that he will not perform within the period so fixed, the seller may not, during that period, resort to any remedy for breach of contract. However, the seller is not deprived thereby of any right he may have to claim damages for delay in performance.

Article 64

(1) The seller may declare the contract avoided:

(*a*) if the failure by the buyer to perform any of his obligations under the contract or this Convention amounts to a fundamental breach of contract; or

(b) if the buyer does not, within the additional period of time fixed by the seller in accordance with paragraph (1) of article 63, perform his obligation to pay the price or take delivery of the goods, or if he declares that he will not do so within the period so fixed.

(2) However, in cases where the buyer has paid the price, the seller loses the right to declare the contract avoided unless he does so:

(a) in respect of late performance by the buyer, before the seller has become aware that performance has been rendered; or

(b) in respect of any breach other than late performance by the buyer, within a reasonable time:

(i) after the seller knew or ought to have known of the breach; or

(ii) after the expiration of any additional period of time fixed by the seller in accordance with paragraph (1) of article 63, or after the buyer has declared that he will not perform his obligations within such an additional period.

Article 65

(1) If under the contract the buyer is to specify the form, measurement or other features of the goods and he fails to make such specification either on the date agreed upon or within a reasonable time after receipt of a request from the seller, the seller may, without prejudice to any other rights he may have, make the specification himself in accordance with the requirements of the buyer that may be known to him.

(2) If the seller makes the specification himself, he must inform the buyer of the details thereof and must fix a reasonable time within which the buyer may make a different specification. If, after receipt of such a communication, the buyer fails to do so within the time so fixed, the specification made by the seller is binding.

CHAPTER IV. PASSING OF RISK

Article 66

Loss of or damage to the goods after the risk has passed to the buyer does not discharge him from his obligation to pay the price, unless the loss or damage is due to an act or omission of the seller.

Article 67

(1) If the contract of sale involves carriage of the goods and the seller is not bound to hand them over at a particular place, the risk passes to the buyer when the goods are handed over to the first carrier for transmission to the buyer in accordance with the contract of sale. If the seller is bound to hand the goods over to a carrier at a particular place, the risk does not pass to the buyer until the goods are handed over to the carrier at that place. The fact that the seller is authorized to retain documents controlling the disposition of the goods does not affect the passage of the risk.

(2) Nevertheless, the risk does not pass to the buyer until the goods are clearly identified to the contract, whether by markings on the goods, by shipping documents, by notice given to the buyer or otherwise.

Article 68

The risk in respect of goods sold in transit passes to the buyer from the time of the conclusion of the contract. However, if the circumstances so indicate, the risk is assumed by the buyer from the time the goods were handed over to the carrier who issued the documents embodying the contract of carriage. Nevertheless, if at the time of the conclusion of the contract of sale the seller knew or ought to have known that the goods had been lost or damaged and did not disclose this to the buyer, the loss or damage is at the risk of the seller.

Article 69

(1) In cases not within articles 67 and 68, the risk passes to the buyer when he takes over the goods or, if he does not do so in due time, from the time when the goods are placed at his disposal and he commits a breach of contract by failing to take delivery.

(2) However, if the buyer is bound to take over the goods at a place other than a place of business of the seller, the risk passes when delivery is due and the buyer is aware of the fact that the goods are placed at his disposal at that place.

(3) If the contract relates to goods not then identified, the goods are considered not to be placed at the disposal of the buyer until they are clearly identified to the contract.

Article 70

If the seller has committed a fundamental breach of contract, articles 67, 68 and 69 do not impair the remedies available to the buyer on account of the breach.

CHAPTER V. PROVISIONS COMMON TO THE OBLIGATIONS OF THE SELLER AND OF THE BUYER

Section I. Anticipatory breach and instalment contracts

Article 71

(1) A party may suspend the performance of his obligations if, after the conclusion of the contract, it becomes apparent that the other party will not perform a substantial part of his obligations as a result of: (a) a serious deficiency in his ability to perform or in his credit-worthiness; or

(b) his conduct in preparing to perform or in performing the contract.

(2) If the seller has already dispatched the goods before the grounds described in the preceding paragraph become evident, he may prevent the handing over of the goods to the buyer even though the buyer holds a document which entitles him to obtain them. The present paragraph relates only to the rights in the goods as between the buyer and the seller.

(3) A party suspending performance, whether before or after dispatch of the goods, must immediately give notice of the suspension to the other party and must continue with performance if the other party provides adequate assurance of his performance.

Article 72

(1) If prior to the date for performance of the contract it is clear that one of the parties will commit a fundamental breach of contract, the other party may declare the contract avoided.

(2) If time allows, the party intending to declare the contract avoided must give reasonable notice to the other party in order to permit him to provide adequate assurance of his performance.

(3) The requirements of the preceding paragraph do not apply if the other party has declared that he will not perform his obligations.

Article 73

(1) In the case of a contract for delivery of goods by instalments, if the failure of one party to perform any of his obligations in respect of any instalment constitutes a fundamental breach of contract with respect to that instalment, the other party may declare the contract avoided with respect to that instalment.

(2) If one party's failure to perform any of his obligations in respect of any instalment gives the other party good grounds to conclude that a fundamental breach of contract will occur with respect to future instalments, he may declare the contract avoided for the future, provided that he does so within a reasonable time. (3) A buyer who declares the contract avoided in respect of any delivery may, at the same time, declare it avoided in respect of deliveries already made or of future deliveries if, by reason of their interdependence, those deliveries could not be used for the purpose contemplated by the parties at the time of the conclusion of the contract.

Section II. Damages

Article 74

Damages for breach of contract by one party consist of a sum equal to the loss, including loss of profit, suffered by the other party as a consequence of the breach. Such damages may not exceed the loss which the party in breach foresaw or ought to have foreseen at the time of the conclusion of the contract, in the light of the facts and matters of which he then knew or ought to have known, as a possible consequence of the breach of contract.

Article 75

If the contract is avoided and if, in a reasonable manner and within a reasonable time after avoidance, the buyer has bought goods in replacement or the seller has resold the goods, the party claiming damages may recover the difference between the contract price and the price in the substitute transaction as well as any further damages recoverable under article 74.

Article 76

(1) If the contract is avoided and there is a current price for the goods, the party claiming damages may, if he has not made a purchase or resale under article 75, recover the difference between the price fixed by the contract and the current price at the time of avoidance as well as any further damages recoverable under article 74. If, however, the party claiming damages has avoided the contract after taking over the goods, the current price at the time of such taking over shall be applied instead of the current price at the time of avoidance.

(2) For the purposes of the preceding paragraph, the current price is the price prevailing at the place where delivery of the goods should have been made or, if there is no current price at that place, the price at such other place as serves as a reasonable substitute, making due allowance for differences in the cost of transporting the goods.

Article 77

A party who relies on a breach of contract must take such measures as are reasonable in the circumstances to mitigate the loss, including loss of profit, resulting from the breach. If he fails to take such measures, the party in breach may claim a reduction in the damages in the amount by which the loss should have been mitigated.

Section III. Interest

Article 78

If a party fails to pay the price or any other sum that is in arrears, the other party is entitled to interest on it, without prejudice to any claim for damages recoverable under article 74.

Section IV. Exemptions

Article 79

(1) A party is not liable for a failure to perform any of his obligations if he proves that the failure was due to an impediment beyond his control and that he could not reasonably be expected to have taken the impediment into account at the time of the conclusion of the contract or to have avoided or overcome it, or its consequences.

(2) If the party's failure is due to the failure by a third person whom he has engaged to perform the whole or a part of the contract, that party is exempt from liability only if:

(a) he is exempt under the preceding paragraph; and

(b) the person whom he has so engaged would be so exempt if the provisions of that paragraph were applied to him.

(3) The exemption provided by this article has effect for the period during which the impediment exists.

(4) The party who fails to perform must give notice to the other party of the impediment and its effect on his ability to perform. If the notice is not received by the other party within a reasonable time after the party who fails to perform knew or ought to have known of the impediment, he is liable for damages resulting from such non-receipt. (5) Nothing in this article prevents either party from exercising any right other than to claim damages under this Convention.

Article 80

A party may not rely on a failure of the other party to perform, to the extent that such failure was caused by the first party's act or omission.

Section V. Effects of avoidance

Article 81

(1) Avoidance of the contract releases both parties from their obligations under it, subject to any damages which may be due. Avoidance does not affect any provision of the contract for the settlement of disputes or any other provision of the contract governing the rights and obligations of the parties consequent upon the avoidance of the contract.

(2) A party who has performed the contract either wholly or in part may claim restitution from the other party of whatever the first party has supplied or paid under the contract. If both parties are bound to make restitution, they must do so concurrently.

Article 82

(1) The buyer loses the right to declare the contract avoided or to require the seller to deliver substitute goods if it is impossible for him to make restitution of the goods substantially in the condition in which he received them.

(2) The preceding paragraph does not apply:

(*a*) if the impossibility of making restitution of the goods or of making restitution of the goods substantially in the condition in which the buyer received them is not due to his act or omission;

(b) if the goods or part of the goods have perished or deteriorated as a result of the examination provided for in article 38; or

(c) if the goods or part of the goods have been sold in the normal course of business or have been consumed or transformed by the buyer in the course of normal use before he discovered or ought to have discovered the lack of conformity.

Article 83

A buyer who has lost the right to declare the contract avoided or to require the seller to deliver substitute goods in accordance with article 82 retains all other remedies under the contract and this Convention.

Article 84

(1) If the seller is bound to refund the price, he must also pay interest on it, from the date on which the price was paid.

(2) The buyer must account to the seller for all benefits which he has derived from the goods or part of them:

(a) if he must make restitution of the goods or part of them; or

(b) if it is impossible for him to make restitution of all or part of the goods or to make restitution of all or part of the goods substantially in the condition in which he received them, but he has nevertheless declared the contract avoided or required the seller to deliver substitute goods.

Section VI. Preservation of the goods

Article 85

If the buyer is in delay in taking delivery of the goods or, where payment of the price and delivery of the goods are to be made concurrently, if he fails to pay the price, and the seller is either in possession of the goods or otherwise able to control their disposition, the seller must take such steps as are reasonable in the circumstances to preserve them. He is entitled to retain them until he has been reimbursed his reasonable expenses by the buyer.

Article 86

(1) If the buyer has received the goods and intends to exercise any right under the contract or this Convention to reject them, he must take such steps to preserve them as are reasonable in the circumstances. He is entitled to retain them until he has been reimbursed his reasonable expenses by the seller.

(2) If goods dispatched to the buyer have been placed at his disposal at their destination and he exercises the right to reject them, he must take

possession of them on behalf of the seller, provided that this can be done without payment of the price and without unreasonable inconvenience or unreasonable expense. This provision does not apply if the seller or a person authorized to take charge of the goods on his behalf is present at the destination. If the buyer takes possession of the goods under this paragraph, his rights and obligations are governed by the preceding paragraph.

Article 87

A party who is bound to take steps to preserve the goods may deposit them in a warehouse of a third person at the expense of the other party provided that the expense incurred is not unreasonable.

Article 88

(1) A party who is bound to preserve the goods in accordance with article 85 or 86 may sell them by any appropriate means if there has been an unreasonable delay by the other party in taking possession of the goods or in taking them back or in paying the price or the cost of preservation, provided that reasonable notice of the intention to sell has been given to the other party.

(2) If the goods are subject to rapid deterioration or their preservation would involve unreasonable expense, a party who is bound to preserve the goods in accordance with article 85 or 86 must take reasonable measures to sell them. To the extent possible he must give notice to the other party of his intention to sell.

(3) A party selling the goods has the right to retain out of the proceeds of sale an amount equal to the reasonable expenses of preserving the goods and of selling them. He must account to the other party for the balance.

PART IV. FINAL PROVISIONS

Article 89

The Secretary-General of the United Nations is hereby designated as the depositary for this Convention.

Article 90

This Convention does not prevail over any international agreement which has already been or may be entered into and which contains provisions concerning the matters governed by this Convention, provided that the parties have their places of business in States parties to such agreement.

Article 91

(1) This Convention is open for signature at the concluding meeting of the United Nations Conference on Contracts for the International Sale of Goods and will remain open for signature by all States at the Headquarters of the United Nations, New York until 30 September 1981.

(2) This Convention is subject to ratification, acceptance or approval by the signatory States.

(3) This Convention is open for accession by all States which are not signatory States as from the date it is open for signature.

(4) Instruments of ratification, acceptance, approval and accession are to be deposited with the Secretary-General of the United Nations.

Article 92

(1) A Contracting State may declare at the time of signature, ratification, acceptance, approval or accession that it will not be bound by Part II of this Convention or that it will not be bound by Part III of this Convention.

(2) A Contracting State which makes a declaration in accordance with the preceding paragraph in respect of Part II or Part III of this Convention is not to be considered a Contracting State within paragraph (1) of article 1 of this Convention in respect of matters governed by the Part to which the declaration applies.

Article 93

(1) If a Contracting State has two or more territorial units in which, according to its constitution, different systems of law are applicable in relation to the matters dealt with in this Convention, it may, at the time of signature, ratification, acceptance, approval or accession, declare that this Convention is to extend to all its territorial units or only to one or more of them, and may amend its declaration by submitting another declaration at any time.

(2) These declarations are to be notified to the depositary and are to state expressly the territorial units to which the Convention extends.

(3) If, by virtue of a declaration under this article, this Convention extends to one or more but not all of the territorial units of a Contracting State, and if the place of business of a party is located in that State, this place of business, for the purposes of this Convention, is considered not to be in a Contracting State, unless it is in a territorial unit to which the Convention extends.

(4) If a Contracting State makes no declaration under paragraph (1) of this article, the Convention is to extend to all territorial units of that State.

Article 94

(1) Two or more Contracting States which have the same or closely related legal rules on matters governed by this Convention may at any time declare that the Convention is not to apply to contracts of sale or to their formation where the parties have their places of business in those States. Such declarations may be made jointly or by reciprocal unilateral declarations.

(2) A Contracting State which has the same or closely related legal rules on matters governed by this Convention as one or more non-Contracting States may at any time declare that the Convention is not to apply to contracts of sale or to their formation where the parties have their places of business in those States.

(3) If a State which is the object of a declaration under the preceding paragraph subsequently becomes a Contracting State, the declaration made will, as from the date on which the Convention enters into force in respect of the new Contracting State, have the effect of a declaration made under paragraph (1), provided that the new Contracting State joins in such declaration or makes a reciprocal unilateral declaration.

Article 95

Any State may declare at the time of the deposit of its instrument of ratification, acceptance, approval or accession that it will not be bound by subparagraph (1)(b) of article 1 of this Convention.

Article 96

A Contracting State whose legislation requires contracts of sale to be concluded in or evidenced by writing may at any time make a declaration in accordance with article 12 that any provision of article 11, article 29, or Part II of this Convention, that allows a contract of sale or its modification or termination by agreement or any offer, acceptance, or other indication of intention to be made in any form other than in writing, does not apply where any party has his place of business in that State.

Article 97

(1) Declarations made under this Convention at the time of signature are subject to confirmation upon ratification, acceptance or approval.

(2) Declarations and confirmations of declarations are to be in writing and be formally notified to the depositary.

(3) A declaration takes effect simultaneously with the entry into force of this Convention in respect of the State concerned. However, a declaration of which the depositary receives formal notification after such entry into force takes effect on the first day of the month following the expiration of six months after the date of its receipt by the depositary. Reciprocal unilateral declarations under article 94 take effect on the first day of the month following the expiration of six months after the receipt of the latest declaration by the depositary.

(4) Any State which makes a declaration under this Convention may withdraw it at any time by a formal notification in writing addressed to the depositary. Such withdrawal is to take effect on the first day of the month following the expiration of six months after the date of the receipt of the notification by the depositary.

(5) A withdrawal of a declaration made under article 94 renders inoperative, as from the date on which the withdrawal takes effect, any reciprocal declaration made by another State under that article.

Article 98

No reservations are permitted except those expressly authorized in this Convention.

Article 99

(1) This Convention enters into force, subject to the provisions of paragraph (6) of this article, on the first day of the month following the expiration of twelve months after the date of deposit of the tenth instrument of ratification, acceptance, approval or accession, including an instrument which contains a declaration made under article 92.

(2) When a State ratifies, accepts, approves or accedes to this Convention after the deposit of the tenth instrument of ratification, acceptance, approval or accession, this Convention, with the exception of the Part excluded, enters into force in respect of that State, subject to the provisions of paragraph (6) of this article, on the first day of the month following the expiration of twelve months after the date of the deposit of its instrument of ratification, acceptance, approval or accession.

(3) A State which ratifies, accepts, approves or accedes to this Convention and is a party to either or both the Convention relating to a Uniform Law on the Formation of Contracts for the International Sale of Goods done at The Hague on 1 July 1964 (1964 Hague Formation Convention) and the Convention relating to a Uniform Law on the International Sale of Goods done at The Hague on 1 July 1964 (1964 Hague Sales Convention) shall at the same time denounce, as the case may be, either or both the 1964 Hague Sales Convention and the 1964 Hague Formation Convention by notifying the Government of the Netherlands to that effect.

(4) A State party to the 1964 Hague Sales Convention which ratifies, accepts, approves or accedes to the present Convention and declares or has declared under article 92 that it will not be bound by Part II of this Convention shall at the time of ratification, acceptance, approval or accession denounce the 1964 Hague Sales Convention by notifying the Government of the Netherlands to that effect.

(5) A State party to the 1964 Hague Formation Convention which ratifies, accepts, approves or accedes to the present Convention and declares or has declared under article 92 that it will not be bound by Part III of this Convention shall at the time of ratification, acceptance, approval or accession denounce the 1964 Hague Formation Convention by notifying the Government of the Netherlands to that effect.

(6) For the purpose of this article, ratifications, acceptances, approvals and accessions in respect of this Convention by States parties to the 1964 Hague Formation Convention or to the 1964 Hague Sales Convention shall not be effective until such denunciations as may be required on the part of those States in respect of the latter two Conventions have themselves become effective. The depositary of this Convention shall consult with the Government of the Netherlands, as the depositary of the 1964 Conventions, so as to ensure necessary coordination in this respect.

Article 100

(1) This Convention applies to the formation of a contract only when the proposal for concluding the contract is made on or after the date when the Convention enters into force in respect of the Contracting States referred to in subparagraph (1)(a) or the Contracting State referred to in subparagraph (1)(b) of article 1.

(2) This Convention applies only to contracts concluded on or after the date when the Convention enters into force in respect of the Contracting States referred to in subparagraph (1)(a) or the Contracting State referred to in subparagraph (1)(b) of article 1.

Article 101

(1) A Contracting State may denounce this Convention, or Part II or Part III of the Convention, by a formal notification in writing addressed to the depositary.

(2) The denunciation takes effect on the first day of the month following the expiration of twelve months after the notification is received by the depositary. Where a longer period for the denunciation to take effect is specified in the notification, the denunciation takes effect upon the expiration of such longer period after the notification is received by the depositary.

DONE at Vienna, this day of eleventh day of April, one thousand nine hundred and eighty, in a single original, of which the Arabic, Chinese, English, French, Russian and Spanish texts are equally authentic.

IN WITNESS WHEREOF the undersigned plenipotentiaries, being duly authorized by their respective Governments, have signed this Convention.

II. Explanatory Note by the UNCITRAL Secretariat on the United Nations Convention on Contracts for the International Sale of Goods

This note has been prepared by the Secretariat of the United Nations Commission on International Trade Law for informational purposes; it is not an official commentary on the Convention.

Introduction

1. The United Nations Convention on Contracts for the International Sale of Goods provides a uniform text of law for international sales of goods. The Convention was prepared by the United Nations Commission on International Trade Law (UNCITRAL) and adopted by a diplomatic conference on 11 April 1980.

2. Preparation of a uniform law for the international sale of goods began in 1930 at the International Institute for the Unification of Private Law (UNIDROIT) in Rome. After a long interruption in the work as a result of the Second World War, the draft was submitted to a diplomatic conference in The Hague in 1964, which adopted two conventions, one on the international sale of goods and the other on the formation of contracts for the international sale of goods.

3. Almost immediately upon the adoption of the two conventions there was widespread criticism of their provisions as reflecting primarily the legal traditions and economic realities of continental Western Europe, which was the region that had most actively contributed to their preparation. As a result, one of the first tasks undertaken by UNCITRAL on its organization in 1968 was to enquire of States whether or not they intended to adhere to those conventions and the reasons for their positions. In the light of the responses received, UNCITRAL decided to study the two conventions to ascertain which modifications might render them capable of wider acceptance by countries of different legal, social and economic systems. The result of this study was the adoption by diplomatic conference on 11 April 1980 of the

United Nations Convention on Contracts for the International Sale of Goods, which combines the subject matter of the two prior conventions.

4. UNCITRAL's success in preparing a Convention with wider acceptability is evidenced by the fact that the original eleven States for which the Convention came into force on 1 January 1988 included States from every geographical region, every stage of economic development and every major legal, social and economic system. The original eleven States were: Argentina, China, Egypt, France, Hungary, Italy, Lesotho, Syria, United States, Yugoslavia and Zambia.

5. As of 1 September 2010, 76 States are parties to the Convention. The current updated status of the Convention is available on the UNCITRAL website.¹ Authoritative information on the status of the Convention, as well as on related declarations, including with respect to territorial application and succession of States, may be found on the United Nations Treaty Collection on the Internet.²

6. The Convention is divided into four parts. Part One deals with the scope of application of the Convention and the general provisions. Part Two contains the rules governing the formation of contracts for the international sale of goods. Part Three deals with the substantive rights and obligations of buyer and seller arising from the contract. Part Four contains the final clauses of the Convention concerning such matters as how and when it comes into force, the reservations and declarations that are permitted and the application of the Convention to international sales where both States concerned have the same or similar law on the subject.

Part One. Scope of application and general provisions

A. Scope of application

7. The articles on scope of application indicate both what is covered by the Convention and what is not covered. The Convention applies to contracts of sale of goods between parties whose places of business are in different States and either both of those States are Contracting States or the rules of private international law lead to the law of a Contracting State. A few States have availed themselves of the authorization in article 95 to declare that they would apply the Convention only in the former and not in the latter of these two situations. As the Convention becomes more widely adopted, the

¹www.uncitral.org/uncitral/en/uncitral_texts/sale_goods/1980CISG_status.html. ²http://treaties.un.org/.

practical significance of such a declaration will diminish. Finally, the Convention may also apply as the law applicable to the contract if so chosen by the parties. In that case, the operation of the Convention will be subject to any limits on contractual stipulations set by the otherwise applicable law.

8. The final clauses make two additional restrictions on the territorial scope of application that will be relevant to a few States. One applies only if a State is a party to another international agreement that contains provisions concerning matters governed by this Convention; the other permits States that have the same or similar domestic law of sales to declare that the Convention does not apply between them.

9. Contracts of sale are distinguished from contracts for services in two respects by article 3. A contract for the supply of goods to be manufactured or produced is considered to be a sale unless the party who orders the goods undertakes to supply a substantial part of the materials necessary for their manufacture or production. When the preponderant part of the obligations of the party who furnishes the goods consists in the supply of labour or other services, the Convention does not apply.

10. The Convention contains a list of types of sales that are excluded from the Convention, either because of the purpose of the sale (goods bought for personal, family or household use), the nature of the sale (sale by auction, on execution or otherwise by law) or the nature of the goods (stocks, shares, investment securities, negotiable instruments, money, ships, vessels, hovercraft, aircraft or electricity). In many States some or all of such sales are governed by special rules reflecting their special nature.

11. Several articles make clear that the subject matter of the Convention is restricted to formation of the contract and the rights and duties of the buyer and seller arising from such a contract. In particular, the Convention is not concerned with the validity of the contract, the effect which the contract may have on the property in the goods sold or the liability of the seller for death or personal injury caused by the goods to any person.

B. Party autonomy

12. The basic principle of contractual freedom in the international sale of goods is recognized by the provision that permits the parties to exclude the application of this Convention or derogate from or vary the effect of any of its provisions. This exclusion will occur, for example, if parties choose the law of a non-contracting State or the substantive domestic law of a contracting State as the law applicable to the contract. Derogation from the Convention

will occur whenever a provision in the contract provides a different rule from that found in the Convention.

C. Interpretation of the Convention

13. This Convention for the unification of the law governing the international sale of goods will better fulfil its purpose if it is interpreted in a consistent manner in all legal systems. Great care was taken in its preparation to make it as clear and easy to understand as possible. Nevertheless, disputes will arise as to its meaning and application. When this occurs, all parties, including domestic courts and arbitral tribunals, are admonished to observe its international character and to promote uniformity in its application and the observance of good faith in international trade. In particular, when a question concerning a matter governed by this Convention is not expressly settled in it, the question is to be settled in conformity with the general principles should the matter be settled in conformity with the law applicable by virtue of the rules of private international law.

D. Interpretation of the contract; usages

14. The Convention contains provisions on the manner in which statements and conduct of a party are to be interpreted in the context of the formation of the contract or its implementation. Usages agreed to by the parties, practices they have established between themselves and usages of which the parties knew or ought to have known and which are widely known to, and regularly observed by, parties to contracts of the type involved in the particular trade concerned may all be binding on the parties to the contract of sale.

E. Form of the contract

15. The Convention does not subject the contract of sale to any requirement as to form. In particular, article 11 provides that no written agreement is necessary for the conclusion of the contract. However, if the contract is in writing and it contains a provision requiring any modification or termination by agreement to be in writing, article 29 provides that the contract may not be otherwise modified or terminated by agreement. The only exception is that a party may be precluded by his conduct from asserting such a provision to the extent that the other person has relied on that conduct.

16. In order to accommodate those States whose legislation requires contracts of sale to be concluded in or evidenced by writing, article 96 entitles those

States to declare that neither article 11 not the exception to article 29 applies where any party to the contract has his place of business in that State.

Part Two. Formation of the contract

17. Part Two of the Convention deals with a number of questions that arise in the formation of the contract by the exchange of an offer and an acceptance. When the formation of the contract takes place in this manner, the contract is concluded when the acceptance of the offer becomes effective.

18. In order for a proposal for concluding a contract to constitute an offer, it must be addressed to one or more specific persons and it must be sufficiently definite. For the proposal to be sufficiently definite, it must indicate the goods and expressly or implicitly fix or make provisions for determining the quantity and the price.

19. The Convention takes a middle position between the doctrine of the revocability of the offer until acceptance and its general irrevocability for some period of time. The general rule is that an offer may be revoked. However, the revocation must reach the offeree before he has dispatched an acceptance. Moreover, an offer cannot be revoked if it indicates that it is irrevocable, which it may do by stating a fixed time for acceptance or otherwise. Furthermore, an offer may not be revoked if it was reasonable for the offeree to rely on the offer as being irrevocable and the offeree has acted in reliance on the offer.

20. Acceptance of an offer may be made by means of a statement or other conduct of the offeree indicating assent to the offer that is communicated to the offeror. However, in some cases the acceptance may consist of performing an act, such as dispatch of the goods or payment of the price. Such an act would normally be effective as an acceptance the moment the act was performed.

21. A frequent problem in contract formation, perhaps especially in regard to contracts of sale of goods, arises out of a reply to an offer that purports to be an acceptance but contains additional or different terms. Under the Convention, if the additional or different terms do not materially alter the terms of the offer, the reply constitutes an acceptance, unless the offeror without undue delay objects to those terms. If he does not object, the terms of the contract are the terms of the offer with the modifications contained in the acceptance.

22. If the additional or different terms do materially alter the terms of the contract, the reply constitutes a counter-offer that must in turn be accepted

for a contract to be concluded. Additional or different terms relating, among other things, to the price, payment, quality and quantity of the goods, place and time of delivery, extent of one party's liability to the other or settlement of disputes are considered to alter the terms of the offer materially.

Part Three. Sale of goods

A. Obligations of the seller

23. The general obligations of the seller are to deliver the goods, hand over any documents relating to them and transfer the property in the goods, as required by the contract and this Convention. The Convention provides supplementary rules for use in the absence of contractual agreement as to when, where and how the seller must perform these obligations.

24. The Convention provides a number of rules that implement the seller's obligations in respect of the quality of the goods. In general, the seller must deliver goods that are of the quantity, quality and description required by the contract and that are contained or packaged in the manner required by the contract. One set of rules of particular importance in international sales of goods involves the seller's obligation to deliver goods that are free from any right or claim of a third party, including rights based on industrial property or other intellectual property.

25. In connection with the seller's obligations in regard to the quality of the goods, the Convention contains provisions on the buyer's obligation to inspect the goods. He must give notice of any lack of conformity with the contract within a reasonable time after he has discovered it or ought to have discovered it, and at the latest two years from the date on which the goods were actually handed over to the buyer, unless this time limit is inconsistent with a contractual period of guarantee.

B. Obligations of the buyer

26. The general obligations of the buyer are to pay the price for the goods and take delivery of them as required by the contract and the Convention. The Convention provides supplementary rules for use in the absence of contractual agreement as to how the price is to be determined and where and when the buyer should perform his obligations to pay the price.

C. Remedies for breach of contract

27. The remedies of the buyer for breach of contract by the seller are set forth in connection with the obligations of the seller and the remedies of the seller are set forth in connection with the obligations of the buyer. This makes it easier to use and understand the Convention.

28. The general pattern of remedies is the same in both cases. If all the required conditions are fulfilled, the aggrieved party may require performance of the other party's obligations, claim damages or avoid the contract. The buyer also has the right to reduce the price where the goods delivered do not conform with the contract.

29. Among the more important limitations on the right of an aggrieved party to claim a remedy is the concept of fundamental breach. For a breach of contract to be fundamental, it must result in such detriment to the other party as substantially to deprive him of what he is entitled to expect under the contract, unless the result was neither foreseen by the party in breach nor foreseeable by a reasonable person of the same kind in the same circumstances. A buyer can require the delivery of substitute goods only if the goods delivered were not in conformity with the contract and the lack of conformity constituted a fundamental breach of contract. The existence of a fundamental breach is one of the two circumstances that justifies a declaration of avoidance of a contract by the aggrieved party; the other circumstance being that, in the case of non-delivery of the goods by the seller or non-payment of the price or failure to take delivery by the buyer, the party in breach fails to perform within a reasonable period of time fixed by the aggrieved party.

30. Other remedies may be restricted by special circumstances. For example, if the goods do not conform with the contract, the buyer may require the seller to remedy the lack of conformity by repair, unless this is unreasonable having regard to all the circumstances. A party cannot recover damages that he could have mitigated by taking the proper measures. A party may be exempted from paying damages by virtue of an impediment beyond his control.

D. Passing of risk

31. Determining the exact moment when the risk of loss or damage to the goods passes from the seller to the buyer is of great importance in contracts for the international sale of goods. Parties may regulate the issue in their contract either by an express provision or by the use of a trade term such as, for example, an INCOTERM. The effect of the choice of such a term

would be to amend the corresponding provisions of the CISG accordingly. However, for the frequent case where the contract does not contain such a provision, the Convention sets forth a complete set of rules.

32. The two special situations contemplated by the Convention are when the contract of sale involves carriage of the goods and when the goods are sold while in transit. In all other cases the risk passes to the buyer when he takes over the goods or from the time when the goods are placed at his disposal and he commits a breach of contract by failing to take delivery, whichever comes first. In the frequent case when the contract relates to goods that are not then identified, they must be identified to the contract before they can be considered to be placed at the disposal of the buyer and the risk of their loss can be considered to have passed to him.

E. Suspension of performance and anticipatory breach

33. The Convention contains special rules for the situation in which, prior to the date on which performance is due, it becomes apparent that one of the parties will not perform a substantial part of his obligations or will commit a fundamental breach of contract. A distinction is drawn between those cases in which the other party may suspend his own performance of the contract but the contract remains in existence awaiting future events and those cases in which he may declare the contract avoided.

F. Exemption from liability to pay damages

34. When a party fails to perform any of his obligations due to an impediment beyond his control that he could not reasonably have been expected to take into account at the time of the conclusion of the contract and that he could not have avoided or overcome, he is exempted from the consequences of his failure to perform, including the payment of damages. This exemption may also apply if the failure is due to the failure of a third person whom he has engaged to perform the whole or a part of the contract. However, he is subject to any other remedy, including reduction of the price, if the goods were defective in some way.

G. Preservation of the goods

35. The Convention imposes on both parties the duty to preserve any goods in their possession belonging to the other party. Such a duty is of even greater importance in an international sale of goods where the other party is from a foreign country and may not have agents in the country where the goods are located. Under certain circumstances the party in possession of the goods may sell them, or may even be required to sell them. A party selling the goods has the right to retain out of the proceeds of sale an amount equal to the reasonable expenses of preserving the goods and of selling them and must account to the other party for the balance.

Part Four. Final clauses

36. The final clauses contain the usual provisions relating to the Secretary-General as depositary and providing that the Convention is subject to ratification, acceptance or approval by those States that signed it by 30 September 1981, that it is open to accession by all States that are not signatory States and that the text is equally authentic in Arabic, Chinese, English, French, Russian and Spanish.

37. The Convention permits a certain number of declarations. Those relative to scope of application and the requirement as to a written contract have been mentioned above. There is a special declaration for States that have different systems of law governing contracts of sale in different parts of their territory. Finally, a State may declare that it will not be bound by Part II on formation of contracts or Part III on the rights and obligations of the buyer and seller. This latter declaration was included as part of the decision to combine into one convention the subject matter of the two 1964 Hague Conventions.

Complementary texts

38. The United Nations Convention on Contracts for the International Sale of Goods is complemented by the United Nations Convention on the Limitation Period in the International Sale of Goods, 1974, as amended by a Protocol in 1980 (the Limitation Convention). The Limitation Convention establishes uniform rules governing the period of time within which a party under a contract for the international sale of goods must commence legal proceedings against another party to assert a claim arising from the contract or relating to its breach, termination or validity. The amending Protocol of 1980 ensures that the scope of application of the Limitation Convention is identical to the one of the United Nations Convention on Contracts for the International Sale of Goods.

39. The United Nations Convention on Contracts for the International Sale of Goods is also complemented, with respect to the use of electronic communications,

by the United Nations Convention on the Use of Electronic Communications in International Contracts, 2005 (the Electronic Communications Convention). The Electronic Communications Convention aims at facilitating the use of electronic communications in international trade by assuring that contracts concluded and other communications exchanged electronically are as valid and enforceable as their traditional paper-based equivalents. The Electronic Communications Convention may help to avoid misinterpretation of the CISG that might occur, for example, when a State has lodged a declaration mandating the use of the traditional written form for contracts for the international sale of goods. It may also promote the understanding that the "communication" and/or "writing" under the CISG should be construed so as to include electronic communications. The Electronic Communications Convention is an enabling treaty whose effect is to remove those formal obstacles by establishing the requirements for functional equivalence between electronic and traditional written form.

Further information can be obtained from:

UNCITRAL Secretariat P.O. Box 500 Vienna International Centre 1400 Vienna Austria Telephone: (+43-1) 26060-4060 Telefax: (+43-1) 26060-5813 Email: uncitral@un.org Internet: uncitral.un.org



UNIDROIT PRINCIPLES OF INTERNATIONAL COMMERCIAL CONTRACTS 2016

//#¶ || || || || ||

UNIDROIT PRINCIPLES OF INTERNATIONAL COMMERCIAL CONTRACTS 2016

PREAMBLE

(Purpose of the Principles)

These Principles set forth general rules for international commercial contracts.

They shall be applied when the parties have agreed that their contract be governed by them.(*)

They may be applied when the parties have agreed that their contract be governed by general principles of law, the *lex mercatoria* or the like.

They may be applied when the parties have not chosen any law to govern their contract.

They may be used to interpret or supplement international uniform law instruments. They may be used to interpret or supplement domestic law.

They may serve as a model for national and international legislators.

CHAPTER 1 — GENERAL PROVISIONS

ARTICLE 1.1

(Freedom of contract)

The parties are free to enter into a contract and to determine its content.

ARTICLE 1.2

(No form required)

Nothing in these Principles requires a contract, statement or any other act to be made in or evidenced by a particular form. It may be proved by any means, including witnesses.

ARTICLE 1.3

(Binding character of contract)

A contract validly entered into is binding upon the parties. It can only be modified or terminated in accordance with its terms or by agreement or as otherwise provided in these Principles.

ARTICLE 1.4

(Mandatory rules)

Nothing in these Principles shall restrict the application of mandatory rules, whether of national, international or supranational origin, which are applicable in accordance with the relevant rules of private international law.

ARTICLE 1.5

(Exclusion or modification by the parties)

The parties may exclude the application of these Principles or derogate from or vary the effect of any of their provisions, except as otherwise provided in the Principles.

^(*) Parties wishing to provide that their agreement be governed by the Principles might use one of the *Model Clauses for the use of the UNIDROIT Principles of International Commercial Contracts* (see http://www.unidroit.org/instruments/commercial-contracts/upicc-model-clauses).

ARTICLE 1.6

(Interpretation and supplementation of the Principles)

(1) In the interpretation of these Principles, regard is to be had to their international character and to their purposes including the need to promote uniformity in their application.

(2) Issues within the scope of these Principles but not expressly settled by them are as far as possible to be settled in accordance with their underlying general principles.

ARTICLE 1.7

(Good faith and fair dealing)

(1) Each party must act in accordance with good faith and fair dealing in international trade.

(2) The parties may not exclude or limit this duty.

ARTICLE 1.8

(Inconsistent behaviour)

A party cannot act inconsistently with an understanding it has caused the other party to have and upon which that other party reasonably has acted in reliance to its detriment.

ARTICLE 1.9

(Usages and practices)

(1) The parties are bound by any usage to which they have agreed and by any practices which they have established between themselves.

(2) The parties are bound by a usage that is widely known to and regularly observed in international trade by parties in the particular trade concerned except where the application of such a usage would be unreasonable.

ARTICLE 1.10

(Notice)

(1) Where notice is required it may be given by any means appropriate to the circumstances.

(2) A notice is effective when it reaches the person to whom it is given.

(3) For the purpose of paragraph (2) a notice "reaches" a person when given to that person orally or delivered at that person's place of business or mailing address.

(4) For the purpose of this Article "notice" includes a declaration, demand, request or any other communication of intention.

ARTICLE 1.11

(Definitions)

In these Principles

- "court" includes an arbitral tribunal;

- where a party has more than one place of business the relevant "place of business" is that which has the closest relationship to the contract and its performance, having regard to the circumstances known to or contemplated by the parties at any time before or at the conclusion of the contract;

- "long-term contract" refers to a contract which is to be performed over a period of time and which normally involves, to a varying degree, complexity of the transaction and an ongoing relationship between the parties;

- "obligor" refers to the party who is to perform an obligation and "obligee" refers to the party who is entitled to performance of that obligation;

- "writing" means any mode of communication that preserves a record of the information contained therein and is capable of being reproduced in tangible form.

ARTICLE 1.12

(Computation of time set by parties)

(1) Official holidays or non-business days occurring during a period set by parties for an act to be performed are included in calculating the period.

(2) However, if the last day of the period is an official holiday or a non-business day at the place of business of the party to perform the act, the period is extended until the first business day which follows, unless the circumstances indicate otherwise.

(3) The relevant time zone is that of the place of business of the party setting the time, unless the circumstances indicate otherwise.

CHAPTER 2 — FORMATION AND AUTHORITY OF AGENTS

SECTION 1: FORMATION

ARTICLE 2.1.1

(Manner of formation)

A contract may be concluded either by the acceptance of an offer or by conduct of the parties that is sufficient to show agreement.

ARTICLE 2.1.2

(Definition of offer)

A proposal for concluding a contract constitutes an offer if it is sufficiently definite and indicates the intention of the offeror to be bound in case of acceptance.

ARTICLE 2.1.3

(Withdrawal of offer)

(1) An offer becomes effective when it reaches the offeree.

(2) An offer, even if it is irrevocable, may be withdrawn if the withdrawal reaches the offeree before or at the same time as the offer.

ARTICLE 2.1.4

(Revocation of offer)

(1) Until a contract is concluded an offer may be revoked if the revocation reaches the offeree before it has dispatched an acceptance.

(2) However, an offer cannot be revoked

(a) if it indicates, whether by stating a fixed time for acceptance or otherwise, that it is irrevocable; or

(b) if it was reasonable for the offeree to rely on the offer as being irrevocable and the offeree has acted in reliance on the offer.

ARTICLE 2.1.5

(Rejection of offer)

An offer is terminated when a rejection reaches the offeror.

ARTICLE 2.1.6

(Mode of acceptance)

(1) A statement made by or other conduct of the offeree indicating assent to an offer is an acceptance. Silence or inactivity does not in itself amount to acceptance.

(2) An acceptance of an offer becomes effective when the indication of assent reaches the offeror.

(3) However, if, by virtue of the offer or as a result of practices which the parties have established between themselves or of usage, the offeree may indicate assent by

UNIDROIT Principles 2016

performing an act without notice to the offeror, the acceptance is effective when the act is performed.

ARTICLE 2.1.7

(Time of acceptance)

An offer must be accepted within the time the offeror has fixed or, if no time is fixed, within a reasonable time having regard to the circumstances, including the rapidity of the means of communication employed by the offeror. An oral offer must be accepted immediately unless the circumstances indicate otherwise.

ARTICLE 2.1.8

(Acceptance within a fixed period of time)

A period of acceptance fixed by the offeror begins to run from the time that the offer is dispatched. A time indicated in the offer is deemed to be the time of dispatch unless the circumstances indicate otherwise.

ARTICLE 2.1.9

(Late acceptance. Delay in transmission)

(1) A late acceptance is nevertheless effective as an acceptance if without undue delay the offeror so informs the offeree or gives notice to that effect.

(2) If a communication containing a late acceptance shows that it has been sent in such circumstances that if its transmission had been normal it would have reached the offeror in due time, the late acceptance is effective as an acceptance unless, without undue delay, the offeror informs the offere that it considers the offer as having lapsed.

ARTICLE 2.1.10

(Withdrawal of acceptance)

An acceptance may be withdrawn if the withdrawal reaches the offeror before or at the same time as the acceptance would have become effective.

ARTICLE 2.1.11

(Modified acceptance)

(1) A reply to an offer which purports to be an acceptance but contains additions, limitations or other modifications is a rejection of the offer and constitutes a counter-offer.

(2) However, a reply to an offer which purports to be an acceptance but contains additional or different terms which do not materially alter the terms of the offer constitutes an acceptance, unless the offeror, without undue delay, objects to the discrepancy. If the offeror does not object, the terms of the contract are the terms of the offer with the modifications contained in the acceptance.

ARTICLE 2.1.12

(Writings in confirmation)

If a writing which is sent within a reasonable time after the conclusion of the contract and which purports to be a confirmation of the contract contains additional or different terms, such terms become part of the contract, unless they materially alter the contract or the recipient, without undue delay, objects to the discrepancy.

ARTICLE 2.1.13

(Conclusion of contract dependent on agreement on specific matters or in a particular form)

Where in the course of negotiations one of the parties insists that the contract is not concluded until there is agreement on specific matters or in a particular form, no contract is concluded before agreement is reached on those matters or in that form.

ARTICLE 2.1.14

(Contract with terms deliberately left open)

(1) If the parties intend to conclude a contract, the fact that they intentionally leave a term to be agreed upon in further negotiations or to be determined by one of the parties or by a third person does not prevent a contract from coming into existence.

- (2) The existence of the contract is not affected by the fact that subsequently
- (a) the parties reach no agreement on the term;
- (b) the party who is to determine the term does not do so; or
- (c) the third person does not determine the term,

provided that there is an alternative means of rendering the term definite that is reasonable in the circumstances, having regard to the intention of the parties.

ARTICLE 2.1.15

(Negotiations in bad faith)

(1) A party is free to negotiate and is not liable for failure to reach an agreement.

(2) However, a party who negotiates or breaks off negotiations in bad faith is liable for the losses caused to the other party.

(3) It is bad faith, in particular, for a party to enter into or continue negotiations when intending not to reach an agreement with the other party.

ARTICLE 2.1.16

(Duty of confidentiality)

Where information is given as confidential by one party in the course of negotiations, the other party is under a duty not to disclose that information or to use it improperly for its own purposes, whether or not a contract is subsequently concluded. Where appropriate, the remedy for breach of that duty may include compensation based on the benefit received by the other party.

ARTICLE 2.1.17

(Merger clauses)

A contract in writing which contains a clause indicating that the writing completely embodies the terms on which the parties have agreed cannot be contradicted or supplemented by evidence of prior statements or agreements. However, such statements or agreements may be used to interpret the writing.

ARTICLE 2.1.18

(Modification in a particular form)

A contract in writing which contains a clause requiring any modification or termination by agreement to be in a particular form may not be otherwise modified or terminated. However, a party may be precluded by its conduct from asserting such a clause to the extent that the other party has reasonably acted in reliance on that conduct.

ARTICLE 2.1.19

(Contracting under standard terms)

(1) Where one party or both parties use standard terms in concluding a contract, the general rules on formation apply, subject to Articles 2.1.20 - 2.1.22.

(2) Standard terms are provisions which are prepared in advance for general and repeated use by one party and which are actually used without negotiation with the other party.

ARTICLE 2.1.20

(Surprising terms)

(1) No term contained in standard terms which is of such a character that the other party could not reasonably have expected it, is effective unless it has been expressly accepted by that party.

(2) In determining whether a term is of such a character regard shall be had to its content, language and presentation.

ARTICLE 2.1.21

(Conflict between standard terms and non-standard terms)

In case of conflict between a standard term and a term which is not a standard term the latter prevails.

ARTICLE 2.1.22

(Battle of forms)

Where both parties use standard terms and reach agreement except on those terms, a contract is concluded on the basis of the agreed terms and of any standard terms which are common in substance unless one party clearly indicates in advance, or later and without undue delay informs the other party, that it does not intend to be bound by such a contract.

SECTION 2: AUTHORITY OF AGENTS

ARTICLE 2.2.1

(Scope of the Section)

(1) This Section governs the authority of a person ("the agent") to affect the legal relations of another person ("the principal") by or with respect to a contract with a third party, whether the agent acts in its own name or in that of the principal.

(2) It governs only the relations between the principal or the agent on the one hand, and the third party on the other.

(3) It does not govern an agent's authority conferred by law or the authority of an agent appointed by a public or judicial authority.

ARTICLE 2.2.2

(Establishment and scope of the authority of the agent)

(1) The principal's grant of authority to an agent may be express or implied.

(2) The agent has authority to perform all acts necessary in the circumstances to achieve the purposes for which the authority was granted.

ARTICLE 2.2.3

(Agency disclosed)

(1) Where an agent acts within the scope of its authority and the third party knew or ought to have known that the agent was acting as an agent, the acts of the agent shall directly affect the legal relations between the principal and the third party and no legal relation is created between the agent and the third party.

(2) However, the acts of the agent shall affect only the relations between the agent and the third party, where the agent with the consent of the principal undertakes to become the party to the contract.

ARTICLE 2.2.4

(Agency undisclosed)

(1) Where an agent acts within the scope of its authority and the third party neither knew nor ought to have known that the agent was acting as an agent, the acts of the agent shall affect only the relations between the agent and the third party.

(2) However, where such an agent, when contracting with the third party on behalf of a business, represents itself to be the owner of that business, the third party, upon discovery of the real owner of the business, may exercise also against the latter the rights it has against the agent.

ARTICLE 2.2.5

(Agent acting without or exceeding its authority)

(1) Where an agent acts without authority or exceeds its authority, its acts do not affect the legal relations between the principal and the third party.

(2) However, where the principal causes the third party reasonably to believe that the agent has authority to act on behalf of the principal and that the agent is acting within the scope of that authority, the principal may not invoke against the third party the lack of authority of the agent.

ARTICLE 2.2.6

(Liability of agent acting without or exceeding its authority)

(1) An agent that acts without authority or exceeds its authority is, failing ratification by the principal, liable for damages that will place the third party in the same position as if the agent had acted with authority and not exceeded its authority.

(2) However, the agent is not liable if the third party knew or ought to have known that the agent had no authority or was exceeding its authority.

ARTICLE 2.2.7

(Conflict of interests)

(1) If a contract concluded by an agent involves the agent in a conflict of interests with the principal of which the third party knew or ought to have known, the principal may avoid the contract. The right to avoid is subject to Articles 3.2.9 and 3.2.11 to 3.2.15.

(2) However, the principal may not avoid the contract

(a) if the principal had consented to, or knew or ought to have known of, the agent's involvement in the conflict of interests; or

(b) if the agent had disclosed the conflict of interests to the principal and the latter had not objected within a reasonable time.

ARTICLE 2.2.8

(Sub-agency)

An agent has implied authority to appoint a sub-agent to perform acts which it is not reasonable to expect the agent to perform itself. The rules of this Section apply to the sub-agency.

ARTICLE 2.2.9

(Ratification)

(1) An act by an agent that acts without authority or exceeds its authority may be ratified by the principal. On ratification the act produces the same effects as if it had initially been carried out with authority.

(2) The third party may by notice to the principal specify a reasonable period of time for ratification. If the principal does not ratify within that period of time it can no longer do so.

(3) If, at the time of the agent's act, the third party neither knew nor ought to have known of the lack of authority, it may, at any time before ratification, by notice to the principal indicate its refusal to become bound by a ratification.

ARTICLE 2.2.10

(Termination of authority)

(1) Termination of authority is not effective in relation to the third party unless the third party knew or ought to have known of it.

(2) Notwithstanding the termination of its authority, an agent remains authorised to perform the acts that are necessary to prevent harm to the principal's interests.

CHAPTER 3—VALIDITY

SECTION 1: GENERAL PROVISIONS

ARTICLE 3.1.1

(Matters not covered)

This Chapter does not deal with lack of capacity.

ARTICLE 3.1.2

(Validity of mere agreement)

A contract is concluded, modified or terminated by the mere agreement of the parties, without any further requirement.

ARTICLE 3.1.3

(Initial impossibility)

(1) The mere fact that at the time of the conclusion of the contract the performance of the obligation assumed was impossible does not affect the validity of the contract.

(2) The mere fact that at the time of the conclusion of the contract a party was not entitled to dispose of the assets to which the contract relates does not affect the validity of the contract.

ARTICLE 3.1.4

(Mandatory character of the provisions)

The provisions on fraud, threat, gross disparity and illegality contained in this Chapter are mandatory.

SECTION 2: GROUNDS FOR AVOIDANCE

ARTICLE 3.2.1

(Definition of mistake)

Mistake is an erroneous assumption relating to facts or to law existing when the contract was concluded.

ARTICLE 3.2.2

(Relevant mistake)

(1) A party may only avoid the contract for mistake if, when the contract was concluded, the mistake was of such importance that a reasonable person in the same situation as the party in error would only have concluded the contract on materially different terms or would not have concluded it at all if the true state of affairs had been known, and

(a) the other party made the same mistake, or caused the mistake, or knew or ought to have known of the mistake and it was contrary to reasonable commercial standards of fair dealing to leave the mistaken party in error; or

(b) the other party had not at the time of avoidance reasonably acted in reliance on the contract.

(2) However, a party may not avoid the contract if

(a) it was grossly negligent in committing the mistake; or

(b) the mistake relates to a matter in regard to which the risk of mistake was assumed or, having regard to the circumstances, should be borne by the mistaken party.

ARTICLE 3.2.3

(Error in expression or transmission)

An error occurring in the expression or transmission of a declaration is considered to be a mistake of the person from whom the declaration emanated.

ARTICLE 3.2.4

(Remedies for non-performance)

A party is not entitled to avoid the contract on the ground of mistake if the circumstances on which that party relies afford, or could have afforded, a remedy for non-performance.

ARTICLE 3.2.5

(Fraud)

A party may avoid the contract when it has been led to conclude the contract by the other party's fraudulent representation, including language or practices, or fraudulent non-disclosure of circumstances which, according to reasonable commercial standards of fair dealing, the latter party should have disclosed.

ARTICLE 3.2.6

(Threat)

A party may avoid the contract when it has been led to conclude the contract by the other party's unjustified threat which, having regard to the circumstances, is so imminent and serious as to leave the first party no reasonable alternative. In particular, a threat is unjustified if the act or omission with which a party has been threatened is wrongful in itself, or it is wrongful to use it as a means to obtain the conclusion of the contract.

ARTICLE 3.2.7

(Gross disparity)

(1) A party may avoid the contract or an individual term of it if, at the time of the conclusion of the contract, the contract or term unjustifiably gave the other party an excessive advantage. Regard is to be had, among other factors, to

(a) the fact that the other party has taken unfair advantage of the first party's dependence, economic distress or urgent needs, or of its improvidence, ignorance, inexperience or lack of bargaining skill, and

(b) the nature and purpose of the contract.

(2) Upon the request of the party entitled to avoidance, a court may adapt the contract or term in order to make it accord with reasonable commercial standards of fair dealing.

(3) A court may also adapt the contract or term upon the request of the party receiving notice of avoidance, provided that that party informs the other party of its request promptly after receiving such notice and before the other party has reasonably acted in reliance on it. Article 3.2.10(2) applies accordingly.

ARTICLE 3.2.8

(Third persons)

(1) Where fraud, threat, gross disparity or a party's mistake is imputable to, or is known or ought to be known by, a third person for whose acts the other party is responsible, the contract may be avoided under the same conditions as if the behaviour or knowledge had been that of the party itself.

(2) Where fraud, threat or gross disparity is imputable to a third person for whose acts the other party is not responsible, the contract may be avoided if that party knew or ought to have known of the fraud, threat or disparity, or has not at the time of avoidance reasonably acted in reliance on the contract.

ARTICLE 3.2.9

(Confirmation)

If the party entitled to avoid the contract expressly or impliedly confirms the contract after the period of time for giving notice of avoidance has begun to run, avoidance of the contract is excluded.

ARTICLE 3.2.10

(Loss of right to avoid)

(1) If a party is entitled to avoid the contract for mistake but the other party declares itself willing to perform or performs the contract as it was understood by the party entitled to avoidance, the contract is considered to have been concluded as the latter party understood it. The other party must make such a declaration or render such performance promptly after having been informed of the manner in which the party entitled to avoidance had understood the contract and before that party has reasonably acted in reliance on a notice of avoidance.

(2) After such a declaration or performance the right to avoidance is lost and any earlier notice of avoidance is ineffective.

ARTICLE 3.2.11

(Notice of avoidance)

The right of a party to avoid the contract is exercised by notice to the other party.

ARTICLE 3.2.12 (Time limits)

(1) Notice of avoidance shall be given within a reasonable time, having regard to the circumstances, after the avoiding party knew or could not have been unaware of the relevant facts or became capable of acting freely.

(2) Where an individual term of the contract may be avoided by a party under Article 3.2.7, the period of time for giving notice of avoidance begins to run when that term is asserted by the other party.

ARTICLE 3.2.13

(Partial avoidance)

Where a ground of avoidance affects only individual terms of the contract, the effect of avoidance is limited to those terms unless, having regard to the circumstances, it is unreasonable to uphold the remaining contract.

ARTICLE 3.2.14

(Retroactive effect of avoidance)

Avoidance takes effect retroactively.

ARTICLE 3.2.15 (Restitution)

(1) On avoidance either party may claim restitution of whatever it has supplied under the contract, or the part of it avoided, provided that the party concurrently makes restitution of whatever it has received under the contract, or the part of it avoided.

(2) If restitution in kind is not possible or appropriate, an allowance has to be made in money whenever reasonable.

(3) The recipient of the performance does not have to make an allowance in money if the impossibility to make restitution in kind is attributable to the other party.

(4) Compensation may be claimed for expenses reasonably required to preserve or maintain the performance received.

ARTICLE 3.2.16

(Damages)

Irrespective of whether or not the contract has been avoided, the party who knew or ought to have known of the ground for avoidance is liable for damages so as to put the other party in the same position in which it would have been if it had not concluded the contract.

ARTICLE 3.2.17

(Unilateral declarations)

The provisions of this Chapter apply with appropriate adaptations to any communication of intention addressed by one party to the other.

SECTION 3: ILLEGALITY

ARTICLE 3.3.1

(Contracts infringing mandatory rules)

(1) Where a contract infringes a mandatory rule, whether of national, international or supranational origin, applicable under Article 1.4 of these Principles, the effects of that infringement upon the contract are the effects, if any, expressly prescribed by that mandatory rule.

(2) Where the mandatory rule does not expressly prescribe the effects of an infringement upon a contract, the parties have the right to exercise such remedies under the contract as in the circumstances are reasonable.

- (3) In determining what is reasonable regard is to be had in particular to:
- (a) the purpose of the rule which has been infringed;
- (b) the category of persons for whose protection the rule exists;
- (c) any sanction that may be imposed under the rule infringed;
- (d) the seriousness of the infringement;
- (e) whether one or both parties knew or ought to have known of the infringement;
- (f) whether the performance of the contract necessitates the infringement; and
- (g) the parties' reasonable expectations.

ARTICLE 3.3.2

(Restitution)

(1) Where there has been performance under a contract infringing a mandatory rule under Article 3.3.1, restitution may be granted where this would be reasonable in the circumstances.

(2) In determining what is reasonable, regard is to be had, with the appropriate adaptations, to the criteria referred to in Article 3.3.1(3).

(3) If restitution is granted, the rules set out in Article 3.2.15 apply with appropriate adaptations.

CHAPTER 4—INTERPRETATION

ARTICLE 4.1

(Intention of the parties)

(1) A contract shall be interpreted according to the common intention of the parties.

(2) If such an intention cannot be established, the contract shall be interpreted according to the meaning that reasonable persons of the same kind as the parties would give to it in the same circumstances.

ARTICLE 4.2

(Interpretation of statements and other conduct)

(1) The statements and other conduct of a party shall be interpreted according to that party's intention if the other party knew or could not have been unaware of that intention.

(2) If the preceding paragraph is not applicable, such statements and other conduct shall be interpreted according to the meaning that a reasonable person of the same kind as the other party would give to it in the same circumstances.

ARTICLE 4.3

(Relevant circumstances)

In applying Articles 4.1 and 4.2, regard shall be had to all the circumstances, including

- (a) preliminary negotiations between the parties;
- (b) practices which the parties have established between themselves;
- (c) the conduct of the parties subsequent to the conclusion of the contract;
- (d) the nature and purpose of the contract;
- (e) the meaning commonly given to terms and expressions in the trade concerned;
- (f) usages.

ARTICLE 4.4

(Reference to contract or statement as a whole)

Terms and expressions shall be interpreted in the light of the whole contract or statement in which they appear.

ARTICLE 4.5

(All terms to be given effect)

Contract terms shall be interpreted so as to give effect to all the terms rather than to deprive some of them of effect.

ARTICLE 4.6

(Contra proferentem rule)

If contract terms supplied by one party are unclear, an interpretation against that party is preferred.

ARTICLE 4.7

(Linguistic discrepancies)

Where a contract is drawn up in two or more language versions which are equally authoritative there is, in case of discrepancy between the versions, a preference for the interpretation according to a version in which the contract was originally drawn up.

ARTICLE 4.8

(Supplying an omitted term)

(1) Where the parties to a contract have not agreed with respect to a term which is important for a determination of their rights and duties, a term which is appropriate in the circumstances shall be supplied.

(2) In determining what is an appropriate term regard shall be had, among other factors, to

(a) the intention of the parties;

(b) the nature and purpose of the contract;

(c) good faith and fair dealing;

(d) reasonableness.

CHAPTER 5— CONTENT AND THIRD PARTY RIGHTS

SECTION 1: CONTENT

ARTICLE 5.1.1

(Express and implied obligations)

The contractual obligations of the parties may be express or implied.

ARTICLE 5.1.2

(Implied obligations)

Implied obligations stem from

- (a) the nature and purpose of the contract;
- (b) practices established between the parties and usages;
- (c) good faith and fair dealing;
- (d) reasonableness.

ARTICLE 5.1.3

(Co-operation between the parties)

Each party shall cooperate with the other party when such co-operation may reasonably be expected for the performance of that party's obligations.

ARTICLE 5.1.4

(Duty to achieve a specific result.

Duty of best efforts)

(1) To the extent that an obligation of a party involves a duty to achieve a specific result, that party is bound to achieve that result.

(2) To the extent that an obligation of a party involves a duty of best efforts in the performance of an activity, that party is bound to make such efforts as would be made by a reasonable person of the same kind in the same circumstances.

ARTICLE 5.1.5

(Determination of kind of duty involved)

In determining the extent to which an obligation of a party involves a duty of best efforts in the performance of an activity or a duty to achieve a specific result, regard shall be had, among other factors, to

- (a) the way in which the obligation is expressed in the contract;
- (b) the contractual price and other terms of the contract;
- (c) the degree of risk normally involved in achieving the expected result;
- (d) the ability of the other party to influence the performance of the obligation.

ARTICLE 5.1.6

(Determination of quality of performance)

Where the quality of performance is neither fixed by, nor determinable from, the contract a party is bound to render a performance of a quality that is reasonable and not less than average in the circumstances.

ARTICLE 5.1.7

(Price determination)

(1) Where a contract does not fix or make provision for determining the price, the parties are considered, in the absence of any indication to the contrary, to have made reference to the price generally charged at the time of the conclusion of the contract for such performance in comparable circumstances in the trade concerned or, if no such price is available, to a reasonable price.

(2) Where the price is to be determined by one party and that determination is manifestly unreasonable, a reasonable price shall be substituted notwithstanding any contract term to the contrary.

(3) Where the price is to be fixed by one party or a third person, and that party or third person does not do so, the price shall be a reasonable price.

(4) Where the price is to be fixed by reference to factors which do not exist or have ceased to exist or to be accessible, the nearest equivalent factor shall be treated as a substitute.

ARTICLE 5.1.8

(Termination of a contract for an indefinite period)

A contract for an indefinite period may be terminated by either party by giving notice a reasonable time in advance. As to the effects of termination in general, and as to restitution, the provisions in Articles 7.3.5 and 7.3.7 apply.

ARTICLE 5.1.9

(Release by agreement)

(1) An obligee may release its right by agreement with the obligor.

(2) An offer to release a right gratuitously shall be deemed accepted if the obligor does not reject the offer without delay after having become aware of it.

SECTION 2: THIRD PARTY RIGHTS

ARTICLE 5.2.1

(Contracts in favour of third parties)

(1) The parties (the "promisor" and the "promisee") may confer by express or implied agreement a right on a third party (the "beneficiary").

(2) The existence and content of the beneficiary's right against the promisor are determined by the agreement of the parties and are subject to any conditions or other limitations under the agreement.

ARTICLE 5.2.2

(Third party identifiable)

The beneficiary must be identifiable with adequate certainty by the contract but need not be in existence at the time the contract is made.

ARTICLE 5.2.3

(Exclusion and limitation clauses)

The conferment of rights in the beneficiary includes the right to invoke a clause in the contract which excludes or limits the liability of the beneficiary.

ARTICLE 5.2.4

(Defences)

The promisor may assert against the beneficiary all defences which the promisor could assert against the promisee.

ARTICLE 5.2.5

(Revocation)

The parties may modify or revoke the rights conferred by the contract on the beneficiary until the beneficiary has accepted them or reasonably acted in reliance on them.

ARTICLE 5.2.6

(Renunciation)

The beneficiary may renounce a right conferred on it.

SECTION 3: CONDITIONS

ARTICLE 5.3.1

(Types of condition)

A contract or a contractual obligation may be made conditional upon the occurrence of a future uncertain event, so that the contract or the contractual obligation only takes effect if the event occurs (suspensive condition) or comes to an end if the event occurs (resolutive condition).

ARTICLE 5.3.2

(Effect of conditions)

Unless the parties otherwise agree:

(a) the relevant contract or contractual obligation takes effect upon fulfilment of a suspensive condition;

(b) the relevant contract or contractual obligation comes to an end upon fulfilment of a resolutive condition.

ARTICLE 5.3.3

(Interference with conditions)

(1) If fulfilment of a condition is prevented by a party, contrary to the duty of good faith and fair dealing or the duty of co-operation, that party may not rely on the non-fulfilment of the condition.

(2) If fulfilment of a condition is brought about by a party, contrary to the duty of good faith and fair dealing or the duty of co-operation, that party may not rely on the fulfilment of the condition.

ARTICLE 5.3.4

(Duty to preserve rights)

Pending fulfilment of a condition, a party may not, contrary to the duty to act in accordance with good faith and fair dealing, act so as to prejudice the other party's rights in case of fulfilment of the condition.

ARTICLE 5.3.5

(Restitution in case of fulfilment of a resolutive condition)

(1) On fulfilment of a resolutive condition, the rules on restitution set out in Articles 7.3.6 and 7.3.7 apply with appropriate adaptations.

(2) If the parties have agreed that the resolutive condition is to operate retroactively, the rules on restitution set out in Article 3.2.15 apply with appropriate adaptations.

CHAPTER 6—PERFORMANCE

SECTION 1: PERFORMANCE IN GENERAL

ARTICLE 6.1.1

(Time of performance)

A party must perform its obligations:

(a) if a time is fixed by or determinable from the contract, at that time;

(b) if a period of time is fixed by or determinable from the contract, at any time within that period unless circumstances indicate that the other party is to choose a time;

(c) in any other case, within a reasonable time after the conclusion of the contract.

(c) In any other case, within a reasonable time after the conclusion of the contract.

ARTICLE 6.1.2

(Performance at one time or in instalments)

In cases under Article 6.1.1(b) or (c), a party must perform its obligations at one time if that performance can be rendered at one time and the circumstances do not indicate otherwise.

ARTICLE 6.1.3

(Partial performance)

(1) The obligee may reject an offer to perform in part at the time performance is due, whether or not such offer is coupled with an assurance as to the balance of the performance, unless the obligee has no legitimate interest in so doing.

(2) Additional expenses caused to the obligee by partial performance are to be borne by the obligor without prejudice to any other remedy.

ARTICLE 6.1.4

(Order of performance)

(1) To the extent that the performances of the parties can be rendered simultaneously, the parties are bound to render them simultaneously unless the circumstances indicate otherwise.

(2) To the extent that the performance of only one party requires a period of time, that party is bound to render its performance first, unless the circumstances indicate otherwise.

ARTICLE 6.1.5

(Earlier performance)

(1) The obligee may reject an earlier performance unless it has no legitimate interest in so doing.

(2) Acceptance by a party of an earlier performance does not affect the time for the performance of its own obligations if that time has been fixed irrespective of the performance of the other party's obligations.

(3) Additional expenses caused to the obligee by earlier performance are to be borne by the obligor, without prejudice to any other remedy.

ARTICLE 6.1.6

(Place of performance)

(1) If the place of performance is neither fixed by, nor determinable from, the contract, a party is to perform:

UNIDROIT Principles 2016

(a) a monetary obligation, at the obligee's place of business;

(b) any other obligation, at its own place of business.

(2) A party must bear any increase in the expenses incidental to performance which is caused by a change in its place of business subsequent to the conclusion of the contract.

ARTICLE 6.1.7

(Payment by cheque or other instrument)

(1) Payment may be made in any form used in the ordinary course of business at the place for payment.

(2) However, an obligee who accepts, either by virtue of paragraph (1) or voluntarily, a cheque, any other order to pay or a promise to pay, is presumed to do so only on condition that it will be honoured.

ARTICLE 6.1.8

(Payment by funds transfer)

(1) Unless the obligee has indicated a particular account, payment may be made by a transfer to any of the financial institutions in which the obligee has made it known that it has an account.

(2) In case of payment by a transfer the obligation of the obligor is discharged when the transfer to the obligee's financial institution becomes effective.

ARTICLE 6.1.9

(Currency of payment)

(1) If a monetary obligation is expressed in a currency other than that of the place for payment, it may be paid by the obligor in the currency of the place for payment unless

(a) that currency is not freely convertible; or

(b) the parties have agreed that payment should be made only in the currency in which the monetary obligation is expressed.

(2) If it is impossible for the obligor to make payment in the currency in which the monetary obligation is expressed, the obligee may require payment in the currency of the place for payment, even in the case referred to in paragraph (1)(b).

(3) Payment in the currency of the place for payment is to be made according to the applicable rate of exchange prevailing there when payment is due.

(4) However, if the obligor has not paid at the time when payment is due, the obligee may require payment according to the applicable rate of exchange prevailing either when payment is due or at the time of actual payment.

ARTICLE 6.1.10

(Currency not expressed)

Where a monetary obligation is not expressed in a particular currency, payment must be made in the currency of the place where payment is to be made.

ARTICLE 6.1.11

(Costs of performance)

Each party shall bear the costs of performance of its obligations.

ARTICLE 6.1.12

(Imputation of payments)

(1) An obligor owing several monetary obligations to the same obligee may specify at the time of payment the debt to which it intends the payment to be applied.

However, the payment discharges first any expenses, then interest due and finally the principal.

(2) If the obligor makes no such specification, the obligee may, within a reasonable time after payment, declare to the obligor the obligation to which it imputes the payment, provided that the obligation is due and undisputed.

(3) In the absence of imputation under paragraphs (1) or (2), payment is imputed to that obligation which satisfies one of the following criteria in the order indicated:

- (a) an obligation which is due or which is the first to fall due;
- (b) the obligation for which the obligee has least security;
- (c) the obligation which is the most burdensome for the obligor;
- (d) the obligation which has arisen first.

If none of the preceding criteria applies, payment is imputed to all the obligations proportionally.

ARTICLE 6.1.13

(Imputation of non-monetary obligations)

Article 6.1.12 applies with appropriate adaptations to the imputation of performance of non-monetary obligations.

ARTICLE 6.1.14

(Application for public permission)

Where the law of a State requires a public permission affecting the validity of the contract or its performance and neither that law nor the circumstances indicate otherwise

(a) if only one party has its place of business in that State, that party shall take the measures necessary to obtain the permission;

(b) in any other case the party whose performance requires permission shall take the necessary measures.

ARTICLE 6.1.15

(Procedure in applying for permission)

(1) The party required to take the measures necessary to obtain the permission shall do so without undue delay and shall bear any expenses incurred.

(2) That party shall whenever appropriate give the other party notice of the grant or refusal of such permission without undue delay.

ARTICLE 6.1.16

(Permission neither granted nor refused)

(1) If, notwithstanding the fact that the party responsible has taken all measures required, permission is neither granted nor refused within an agreed period or, where no period has been agreed, within a reasonable time from the conclusion of the contract, either party is entitled to terminate the contract.

(2) Where the permission affects some terms only, paragraph (1) does not apply if, having regard to the circumstances, it is reasonable to uphold the remaining contract even if the permission is refused.

ARTICLE 6.1.17

(Permission refused)

(1) The refusal of a permission affecting the validity of the contract renders the contract void. If the refusal affects the validity of some terms only, only such terms are void if, having regard to the circumstances, it is reasonable to uphold the remaining contract.

(2) Where the refusal of a permission renders the performance of the contract impossible in whole or in part, the rules on non-performance apply.

SECTION 2: HARDSHIP

ARTICLE 6.2.1

(Contract to be observed)

Where the performance of a contract becomes more onerous for one of the parties, that party is nevertheless bound to perform its obligations subject to the following provisions on hardship.

ARTICLE 6.2.2

(Definition of hardship)

There is hardship where the occurrence of events fundamentally alters the equilibrium of the contract either because the cost of a party's performance has increased or because the value of the performance a party receives has diminished, and

(a) the events occur or become known to the disadvantaged party after the conclusion of the contract;

(b) the events could not reasonably have been taken into account by the disadvantaged party at the time of the conclusion of the contract;

(c) the events are beyond the control of the disadvantaged party; and

(d) the risk of the events was not assumed by the disadvantaged party.

ARTICLE 6.2.3

(Effects of hardship)

(1) In case of hardship the disadvantaged party is entitled to request renegotiations. The request shall be made without undue delay and shall indicate the grounds on which it is based.

(2) The request for renegotiation does not in itself entitle the disadvantaged party to withhold performance.

(3) Upon failure to reach agreement within a reasonable time either party may resort to the court.

(4) If the court finds hardship it may, if reasonable,

(a) terminate the contract at a date and on terms to be fixed, or

(b) adapt the contract with a view to restoring its equilibrium.

CHAPTER 7 — NON-PERFORMANCE

SECTION 1: NON-PERFORMANCE IN GENERAL

ARTICLE 7.1.1

(Non-performance defined)

Non-performance is failure by a party to perform any of its obligations under the contract, including defective performance or late performance.

ARTICLE 7.1.2

(Interference by the other party)

A party may not rely on the non-performance of the other party to the extent that such non-performance was caused by the first party's act or omission or by another event for which the first party bears the risk.

ARTICLE 7.1.3

(Withholding performance)

(1) Where the parties are to perform simultaneously, either party may withhold performance until the other party tenders its performance.

(2) Where the parties are to perform consecutively, the party that is to perform later may withhold its performance until the first party has performed.

ARTICLE 7.1.4

(Cure by non-performing party)

(1) The non-performing party may, at its own expense, cure any non-performance, provided that

(a) without undue delay, it gives notice indicating the proposed manner and timing of the cure;

(b) cure is appropriate in the circumstances;

(c) the aggrieved party has no legitimate interest in refusing cure; and

(d) cure is effected promptly.

(2) The right to cure is not precluded by notice of termination.

(3) Upon effective notice of cure, rights of the aggrieved party that are inconsistent with the non-performing party's performance are suspended until the time for cure has expired.

(4) The aggrieved party may withhold performance pending cure.

(5) Notwithstanding cure, the aggrieved party retains the right to claim damages for delay as well as for any harm caused or not prevented by the cure.

ARTICLE 7.1.5

(Additional period for performance)

(1) In a case of non-performance the aggrieved party may by notice to the other party allow an additional period of time for performance.

(2) During the additional period the aggrieved party may withhold performance of its own reciprocal obligations and may claim damages but may not resort to any other remedy. If it receives notice from the other party that the latter will not perform within that period, or if upon expiry of that period due performance has not been made, the aggrieved party may resort to any of the remedies that may be available under this Chapter.

(3) Where in a case of delay in performance which is not fundamental the aggrieved party has given notice allowing an additional period of time of reasonable length, it may terminate the contract at the end of that period. If the additional period allowed is not of reasonable length it shall be extended to a reasonable length. The aggrieved party may in its notice provide that if the other party fails to perform within the period allowed by the notice the contract shall automatically terminate.

(4) Paragraph (3) does not apply where the obligation which has not been performed is only a minor part of the contractual obligation of the non-performing party.

ARTICLE 7.1.6

(Exemption clauses)

A clause which limits or excludes one party's liability for non-performance or which permits one party to render performance substantially different from what the other party reasonably expected may not be invoked if it would be grossly unfair to do so, having regard to the purpose of the contract.

ARTICLE 7.1.7

(Force majeure)

(1) Non-performance by a party is excused if that party proves that the nonperformance was due to an impediment beyond its control and that it could not reasonably be expected to have taken the impediment into account at the time of the conclusion of the contract or to have avoided or overcome it or its consequences. (2) When the impediment is only temporary, the excuse shall have effect for such period as is reasonable having regard to the effect of the impediment on the performance of the contract.

(3) The party who fails to perform must give notice to the other party of the impediment and its effect on its ability to perform. If the notice is not received by the other party within a reasonable time after the party who fails to perform knew or ought to have known of the impediment, it is liable for damages resulting from such non-receipt.

(4) Nothing in this Article prevents a party from exercising a right to terminate the contract or to withhold performance or request interest on money due.

SECTION 2: RIGHT TO PERFORMANCE

ARTICLE 7.2.1

(Performance of monetary obligation)

Where a party who is obliged to pay money does not do so, the other party may require payment.

ARTICLE 7.2.2

(Performance of non-monetary obligation)

Where a party who owes an obligation other than one to pay money does not perform, the other party may require performance, unless

(a) performance is impossible in law or in fact;

(b) performance or, where relevant, enforcement is unreasonably burdensome or expensive;

(c) the party entitled to performance may reasonably obtain performance from another source;

(d) performance is of an exclusively personal character; or

(e) the party entitled to performance does not require performance within a reasonable time after it has, or ought to have, become aware of the non-performance.

ARTICLE 7.2.3

(Repair and replacement of defective performance)

The right to performance includes in appropriate cases the right to require repair, replacement, or other cure of defective performance. The provisions of Articles 7.2.1 and 7.2.2 apply accordingly.

ARTICLE 7.2.4

(Judicial penalty)

(1) Where the court orders a party to perform, it may also direct that this party pay a penalty if it does not comply with the order.

(2) The penalty shall be paid to the aggrieved party unless mandatory provisions of the law of the forum provide otherwise. Payment of the penalty to the aggrieved party does not exclude any claim for damages.

ARTICLE 7.2.5

(Change of remedy)

(1) An aggrieved party who has required performance of a non-monetary obligation and who has not received performance within a period fixed or otherwise within a reasonable period of time may invoke any other remedy.

(2) Where the decision of a court for performance of a non-monetary obligation cannot be enforced, the aggrieved party may invoke any other remedy.

SECTION 3: TERMINATION

ARTICLE 7.3.1

(*Right to terminate the contract*)

(1) A party may terminate the contract where the failure of the other party to perform an obligation under the contract amounts to a fundamental non-performance.

(2) In determining whether a failure to perform an obligation amounts to a fundamental non-performance regard shall be had, in particular, to whether

(a) the non-performance substantially deprives the aggrieved party of what it was entitled to expect under the contract unless the other party did not foresee and could not reasonably have foreseen such result;

(b) strict compliance with the obligation which has not been performed is of essence under the contract;

(c) the non-performance is intentional or reckless;

(d) the non-performance gives the aggrieved party reason to believe that it cannot rely on the other party's future performance;

(e) the non-performing party will suffer disproportionate loss as a result of the preparation or performance if the contract is terminated.

(3) In the case of delay the aggrieved party may also terminate the contract if the other party fails to perform before the time allowed it under Article 7.1.5 has expired.

ARTICLE 7.3.2

(Notice of termination)

(1) The right of a party to terminate the contract is exercised by notice to the other party.

(2) If performance has been offered late or otherwise does not conform to the contract the aggrieved party will lose its right to terminate the contract unless it gives notice to the other party within a reasonable time after it has or ought to have become aware of the offer or of the non-conforming performance.

ARTICLE 7.3.3

(Anticipatory non-performance)

Where prior to the date for performance by one of the parties it is clear that there will be a fundamental non-performance by that party, the other party may terminate the contract.

ARTICLE 7.3.4

(Adequate assurance of due performance)

A party who reasonably believes that there will be a fundamental non-performance by the other party may demand adequate assurance of due performance and may meanwhile withhold its own performance. Where this assurance is not provided within a reasonable time the party demanding it may terminate the contract.

ARTICLE 7.3.5

(Effects of termination in general)

(1) Termination of the contract releases both parties from their obligation to effect and to receive future performance.

(2) Termination does not preclude a claim for damages for non-performance.

(3) Termination does not affect any provision in the contract for the settlement of disputes or any other term of the contract which is to operate even after termination.

ARTICLE 7.3.6

(*Restitution with respect to contracts to be performed at one time*)

(1) On termination of a contract to be performed at one time either party may claim restitution of whatever it has supplied under the contract, provided that such party concurrently makes restitution of whatever it has received under the contract.

(2) If restitution in kind is not possible or appropriate, an allowance has to be made in money whenever reasonable.

(3) The recipient of the performance does not have to make an allowance in money if the impossibility to make restitution in kind is attributable to the other party.

(4) Compensation may be claimed for expenses reasonably required to preserve or maintain the performance received.

ARTICLE 7.3.7

(Restitution with respect to long-term contracts)

(1) On termination of a long-term contract restitution can only be claimed for the period after termination has taken effect, provided the contract is divisible.

(2) As far as restitution has to be made, the provisions of Article 7.3.6 apply.

SECTION 4: DAMAGES

ARTICLE 7.4.1

(Right to damages)

Any non-performance gives the aggrieved party a right to damages either exclusively or in conjunction with any other remedies except where the nonperformance is excused under these Principles.

ARTICLE 7.4.2

(Full compensation)

(1) The aggrieved party is entitled to full compensation for harm sustained as a result of the non-performance. Such harm includes both any loss which it suffered and any gain of which it was deprived, taking into account any gain to the aggrieved party resulting from its avoidance of cost or harm.

(2) Such harm may be non-pecuniary and includes, for instance, physical suffering or emotional distress.

ARTICLE 7.4.3

(Certainty of harm)

(1) Compensation is due only for harm, including future harm, that is established with a reasonable degree of certainty.

(2) Compensation may be due for the loss of a chance in proportion to the probability of its occurrence.

(3) Where the amount of damages cannot be established with a sufficient degree of certainty, the assessment is at the discretion of the court.

ARTICLE 7.4.4

(Foreseeability of harm)

The non-performing party is liable only for harm which it foresaw or could reasonably have foreseen at the time of the conclusion of the contract as being likely to result from its non-performance.

ARTICLE 7.4.5

(Proof of harm in case of replacement transaction)

Where the aggrieved party has terminated the contract and has made a replacement transaction within a reasonable time and in a reasonable manner it may recover the difference between the contract price and the price of the replacement transaction as well as damages for any further harm.

ARTICLE 7.4.6

(Proof of harm by current price)

(1) Where the aggrieved party has terminated the contract and has not made a replacement transaction but there is a current price for the performance contracted for, it may recover the difference between the contract price and the price current at the time the contract is terminated as well as damages for any further harm.

(2) Current price is the price generally charged for goods delivered or services rendered in comparable circumstances at the place where the contract should have been performed or, if there is no current price at that place, the current price at such other place that appears reasonable to take as a reference.

ARTICLE 7.4.7

(Harm due in part to aggrieved party)

Where the harm is due in part to an act or omission of the aggrieved party or to another event for which that party bears the risk, the amount of damages shall be reduced to the extent that these factors have contributed to the harm, having regard to the conduct of each of the parties.

ARTICLE 7.4.8

(Mitigation of harm)

(1) The non-performing party is not liable for harm suffered by the aggrieved party to the extent that the harm could have been reduced by the latter party's taking reasonable steps.

(2) The aggrieved party is entitled to recover any expenses reasonably incurred in attempting to reduce the harm.

ARTICLE 7.4.9

(Interest for failure to pay money)

(1) If a party does not pay a sum of money when it falls due the aggrieved party is entitled to interest upon that sum from the time when payment is due to the time of payment whether or not the non-payment is excused.

(2) The rate of interest shall be the average bank short-term lending rate to prime borrowers prevailing for the currency of payment at the place for payment, or where no such rate exists at that place, then the same rate in the State of the currency of payment. In the absence of such a rate at either place the rate of interest shall be the appropriate rate fixed by the law of the State of the currency of payment.

(3) The aggrieved party is entitled to additional damages if the non-payment caused it a greater harm.

ARTICLE 7.4.10

(Interest on damages)

Unless otherwise agreed, interest on damages for non-performance of non-monetary obligations accrues as from the time of non-performance.

ARTICLE 7.4.11

(Manner of monetary redress)

(1) Damages are to be paid in a lump sum. However, they may be payable in instalments where the nature of the harm makes this appropriate.

(2) Damages to be paid in instalments may be indexed.

ARTICLE 7.4.12

(Currency in which to assess damages)

Damages are to be assessed either in the currency in which the monetary obligation was expressed or in the currency in which the harm was suffered, whichever is more appropriate.

ARTICLE 7.4.13

(Agreed payment for non-performance)

(1) Where the contract provides that a party who does not perform is to pay a specified sum to the aggrieved party for such non-performance, the aggrieved party is entitled to that sum irrespective of its actual harm.

(2) However, notwithstanding any agreement to the contrary the specified sum may be reduced to a reasonable amount where it is grossly excessive in relation to the harm resulting from the non-performance and to the other circumstances.

CHAPTER 8—SET-OFF

ARTICLE 8.1

(Conditions of set-off)

(1) Where two parties owe each other money or other performances of the same kind, either of them ("the first party") may set off its obligation against that of its obligee ("the other party") if at the time of set-off,

(a) the first party is entitled to perform its obligation;

(b) the other party's obligation is ascertained as to its existence and amount and performance is due.

(2) If the obligations of both parties arise from the same contract, the first party may also set off its obligation against an obligation of the other party which is not ascertained as to its existence or to its amount.

ARTICLE 8.2

(Foreign currency set-off)

Where the obligations are to pay money in different currencies, the right of set-off may be exercised, provided that both currencies are freely convertible and the parties have not agreed that the first party shall pay only in a specified currency.

ARTICLE 8.3

(Set-off by notice)

The right of set-off is exercised by notice to the other party.

ARTICLE 8.4

(Content of notice)

(1) The notice must specify the obligations to which it relates.

(2) If the notice does not specify the obligation against which set-off is exercised, the other party may, within a reasonable time, declare to the first party the obligation to which set-off relates. If no such declaration is made, the set-off will relate to all the obligations proportionally.

ARTICLE 8.5

(Effect of set-off)

(1) Set-off discharges the obligations.

(2) If obligations differ in amount, set-off discharges the obligations up to the amount of the lesser obligation.

(3) Set-off takes effect as from the time of notice.

CHAPTER 9 — ASSIGNMENT OF RIGHTS, TRANSFER OF OBLIGATIONS, ASSIGNMENT OF CONTRACTS

SECTION 1: ASSIGNMENT OF RIGHTS

ARTICLE 9.1.1

(Definitions)

"Assignment of a right" means the transfer by agreement from one person (the "assignor") to another person (the "assignee"), including transfer by way of security, of the assignor's right to payment of a monetary sum or other performance from a third person ("the obligor").

ARTICLE 9.1.2

(Exclusions)

This Section does not apply to transfers made under the special rules governing the transfers:

(a) of instruments such as negotiable instruments, documents of title or financial instruments, or

(b) of rights in the course of transferring a business.

ARTICLE 9.1.3

(Assignability of non-monetary rights)

A right to non-monetary performance may be assigned only if the assignment does not render the obligation significantly more burdensome.

ARTICLE 9.1.4

(Partial assignment)

(1) A right to the payment of a monetary sum may be assigned partially.

(2) A right to other performance may be assigned partially only if it is divisible, and the assignment does not render the obligation significantly more burdensome.

ARTICLE 9.1.5

(Future rights)

A future right is deemed to be transferred at the time of the agreement, provided the right, when it comes into existence, can be identified as the right to which the assignment relates.

ARTICLE 9.1.6

(Rights assigned without individual specification)

A number of rights may be assigned without individual specification, provided such rights can be identified as rights to which the assignment relates at the time of the assignment or when they come into existence.

ARTICLE 9.1.7

(Agreement between assignor and assignee sufficient)

(1) A right is assigned by mere agreement between the assignor and the assignee, without notice to the obligor.

(2) The consent of the obligor is not required unless the obligation in the circumstances is of an essentially personal character.

ARTICLE 9.1.8

(Obligor's additional costs)

The obligor has a right to be compensated by the assignor or the assignee for any additional costs caused by the assignment.

ARTICLE 9.1.9

(Non-assignment clauses)

(1) The assignment of a right to the payment of a monetary sum is effective notwithstanding an agreement between the assignor and the obligor limiting or prohibiting such an assignment. However, the assignor may be liable to the obligor for breach of contract.

(2) The assignment of a right to other performance is ineffective if it is contrary to an agreement between the assignor and the obligor limiting or prohibiting the assignment. Nevertheless, the assignment is effective if the assignee, at the time of the assignment, neither knew nor ought to have known of the agreement. The assignor may then be liable to the obligor for breach of contract.

ARTICLE 9.1.10

(Notice to the obligor)

(1) Until the obligor receives a notice of the assignment from either the assignor or the assignee, it is discharged by paying the assignor.

(2) After the obligor receives such a notice, it is discharged only by paying the assignee.

ARTICLE 9.1.11

(Successive assignments)

If the same right has been assigned by the same assignor to two or more successive assignees, the obligor is discharged by paying according to the order in which the notices were received.

ARTICLE 9.1.12

(Adequate proof of assignment)

(1) If notice of the assignment is given by the assignee, the obligor may request the assignee to provide within a reasonable time adequate proof that the assignment has been made.

(2) Until adequate proof is provided, the obligor may withhold payment.

(3) Unless adequate proof is provided, notice is not effective.

(4) Adequate proof includes, but is not limited to, any writing emanating from the assignor and indicating that the assignment has taken place.

ARTICLE 9.1.13

(Defences and rights of set-off)

(1) The obligor may assert against the assignee all defences that the obligor could assert against the assignor.

(2) The obligor may exercise against the assignee any right of set-off available to the obligor against the assignor up to the time notice of assignment was received.

ARTICLE 9.1.14

(Rights related to the right assigned)

The assignment of a right transfers to the assignee:

(a) all the assignor's rights to payment or other performance under the contract in respect of the right assigned, and

(b) all rights securing performance of the right assigned.

ARTICLE 9.1.15

(Undertakings of the assignor)

The assignor undertakes towards the assignee, except as otherwise disclosed to the assignee, that:

(a) the assigned right exists at the time of the assignment, unless the right is a future right;

(b) the assignor is entitled to assign the right;

(c) the right has not been previously assigned to another assignee, and it is free from any right or claim from a third party;

(d) the obligor does not have any defences;

(e) neither the obligor nor the assignor has given notice of set-off concerning the assigned right and will not give any such notice;

(f) the assignor will reimburse the assignee for any payment received from the obligor before notice of the assignment was given.

SECTION 2: TRANSFER OF OBLIGATIONS

ARTICLE 9.2.1

(Modes of transfer)

An obligation to pay money or render other performance may be transferred from one person (the "original obligor") to another person (the "new obligor") either

(a) by an agreement between the original obligor and the new obligor subject to Article 9.2.3, or

(b) by an agreement between the obligee and the new obligor, by which the new obligor assumes the obligation.

ARTICLE 9.2.2

(Exclusion)

This Section does not apply to transfers of obligations made under the special rules governing transfers of obligations in the course of transferring a business.

ARTICLE 9.2.3

(Requirement of obligee's consent to transfer)

The transfer of an obligation by an agreement between the original obligor and the new obligor requires the consent of the obligee.

ARTICLE 9.2.4

(Advance consent of obligee)

(1) The obligee may give its consent in advance.

(2) If the obligee has given its consent in advance, the transfer of the obligation becomes effective when a notice of the transfer is given to the obligee or when the obligee acknowledges it.

ARTICLE 9.2.5

(Discharge of original obligor)

(1) The obligee may discharge the original obligor.

(2) The obligee may also retain the original obligor as an obligor in case the new obligor does not perform properly.

(3) Otherwise the original obligor and the new obligor are jointly and severally liable.

ARTICLE 9.2.6

(Third party performance)

(1) Without the obligee's consent, the obligor may contract with another person that this person will perform the obligation in place of the obligor, unless the obligation in the circumstances has an essentially personal character.

(2) The obligee retains its claim against the obligor.

ARTICLE 9.2.7

(Defences and rights of set-off)

(1) The new obligor may assert against the obligee all defences which the original obligor could assert against the obligee.

(2) The new obligor may not exercise against the obligee any right of set-off available to the original obligor against the obligee.

ARTICLE 9.2.8

(Rights related to the obligation transferred)

(1) The obligee may assert against the new obligor all its rights to payment or other performance under the contract in respect of the obligation transferred.

(2) If the original obligor is discharged under Article 9.2.5(1), a security granted by any person other than the new obligor for the performance of the obligation is discharged, unless that other person agrees that it should continue to be available to the obligee.

(3) Discharge of the original obligor also extends to any security of the original obligor given to the obligee for the performance of the obligation, unless the security is over an asset which is transferred as part of a transaction between the original obligor and the new obligor.

SECTION 3: ASSIGNMENT OF CONTRACTS

ARTICLE 9.3.1

(Definitions)

"Assignment of a contract" means the transfer by agreement from one person (the "assignor") to another person (the "assignee") of the assignor's rights and obligations arising out of a contract with another person (the "other party").

ARTICLE 9.3.2

(Exclusion)

This Section does not apply to the assignment of contracts made under the special rules governing transfers of contracts in the course of transferring a business.

ARTICLE 9.3.3

(Requirement of consent of the other party)

The assignment of a contract requires the consent of the other party.

ARTICLE 9.3.4

(Advance consent of the other party)

(1) The other party may give its consent in advance.

(2) If the other party has given its consent in advance, the assignment of the contract becomes effective when a notice of the assignment is given to the other party or when the other party acknowledges it.

ARTICLE 9.3.5

(Discharge of the assignor)

(1) The other party may discharge the assignor.

(2) The other party may also retain the assignor as an obligor in case the assignee does not perform properly.

(3) Otherwise the assignor and the assignee are jointly and severally liable.

ARTICLE 9.3.6

(Defences and rights of set-off)

(1) To the extent that the assignment of a contract involves an assignment of rights, Article 9.1.13 applies accordingly.

(2) To the extent that the assignment of a contract involves a transfer of obligations, Article 9.2.7 applies accordingly.

ARTICLE 9.3.7

(Rights transferred with the contract)

(1) To the extent that the assignment of a contract involves an assignment of rights, Article 9.1.14 applies accordingly.

(2) To the extent that the assignment of a contract involves a transfer of obligations, Article 9.2.8 applies accordingly.

CHAPTER 10 - LIMITATION PERIODS

ARTICLE 10.1

(Scope of the Chapter)

(1) The exercise of rights governed by the Principles is barred by the expiration of a period of time, referred to as "limitation period", according to the rules of this Chapter.

(2) This Chapter does not govern the time within which one party is required under the Principles, as a condition for the acquisition or exercise of its right, to give notice to the other party or to perform any act other than the institution of legal proceedings.

ARTICLE 10.2

(Limitation periods)

(1) The general limitation period is three years beginning on the day after the day the obligee knows or ought to know the facts as a result of which the obligee's right can be exercised.

(2) In any event, the maximum limitation period is ten years beginning on the day after the day the right can be exercised.

ARTICLE 10.3

(Modification of limitation periods by the parties)

- (1) The parties may modify the limitation periods.
- (2) However they may not
- (a) shorten the general limitation period to less than one year;
- (b) shorten the maximum limitation period to less than four years;
- (c) extend the maximum limitation period to more than fifteen years.

ARTICLE 10.4

(New limitation period by acknowledgement)

(1) Where the obligor before the expiration of the general limitation period acknowledges the right of the obligee, a new general limitation period begins on the day after the day of the acknowledgement.

(2) The maximum limitation period does not begin to run again, but may be exceeded by the beginning of a new general limitation period under Article 10.2(1).

ARTICLE 10.5

(Suspension by judicial proceedings)

(1) The running of the limitation period is suspended

(a) when the obligee performs any act, by commencing judicial proceedings or in judicial proceedings already instituted, that is recognised by the law of the court as asserting the obligee's right against the obligor;

(b) in the case of the obligor's insolvency when the obligee has asserted its rights in the insolvency proceedings; or

(c) in the case of proceedings for dissolution of the entity which is the obligor when the obligee has asserted its rights in the dissolution proceedings.

(2) Suspension lasts until a final decision has been issued or until the proceedings have been otherwise terminated.

ARTICLE 10.6

(Suspension by arbitral proceedings)

(1) The running of the limitation period is suspended when the obligee performs any act, by commencing arbitral proceedings or in arbitral proceedings already instituted, that is recognised by the law of the arbitral tribunal as asserting the obligee's right against the obligor. In the absence of regulations for arbitral proceedings or provisions determining the exact date of the commencement of arbitral proceedings, the proceedings are deemed to commence on the date on which a request that the right in dispute should be adjudicated reaches the obligor.

(2) Suspension lasts until a binding decision has been issued or until the proceedings have been otherwise terminated.

ARTICLE 10.7

(Alternative dispute resolution)

The provisions of Articles 10.5 and 10.6 apply with appropriate modifications to other proceedings whereby the parties request a third person to assist them in their attempt to reach an amicable settlement of their dispute.

ARTICLE 10.8

(Suspension in case of force majeure, death or incapacity)

(1) Where the obligee has been prevented by an impediment that is beyond its control and that it could neither avoid nor overcome, from causing a limitation period to cease to run under the preceding Articles, the general limitation period is suspended so as not to expire before one year after the relevant impediment has ceased to exist.

(2) Where the impediment consists of the incapacity or death of the obligee or obligor, suspension ceases when a representative for the incapacitated or deceased party or its estate has been appointed or a successor has inherited the respective party's position. The additional one-year period under paragraph (1) applies accordingly.

ARTICLE 10.9

(Effects of expiration of limitation period)

(1) The expiration of the limitation period does not extinguish the right.

(2) For the expiration of the limitation period to have effect, the obligor must assert it as a defence.

(3) A right may still be relied on as a defence even though the expiration of the limitation period for that right has been asserted.

ARTICLE 10.10

(Right of set-off)

The obligee may exercise the right of set-off until the obligor has asserted the expiration of the limitation period.

ARTICLE 10.11

(Restitution)

Where there has been performance in order to discharge an obligation, there is no right of restitution merely because the limitation period has expired.

CHAPTER 11 — PLURALITY OF OBLIGORS AND OF OBLIGEES

SECTION 1: PLURALITY OF OBLIGORS

ARTICLE 11.1.1

(Definitions)

When several obligors are bound by the same obligation towards an obligee:

(a) the obligations are joint and several when each obligor is bound for the whole obligation;

(b) the obligations are separate when each obligor is bound only for its share.

ARTICLE 11.1.2

(Presumption of joint and several obligations)

When several obligors are bound by the same obligation towards an obligee, they are presumed to be jointly and severally bound, unless the circumstances indicate otherwise.

ARTICLE 11.1.3

(Obligee's rights against joint and several obligors)

When obligors are jointly and severally bound, the obligee may require performance from any one of them, until full performance has been received.

ARTICLE 11.1.4

(Availability of defences and rights of set-off)

A joint and several obligor against whom a claim is made by the obligee may assert all the defences and rights of set-off that are personal to it or that are common to all the co-obligors, but may not assert defences or rights of set-off that are personal to one or several of the other co-obligors.

ARTICLE 11.1.5

(Effect of performance or set-off)

Performance or set-off by a joint and several obligor or set-off by the obligee against one joint and several obligor discharges the other obligors in relation to the obligee to the extent of the performance or set-off.

ARTICLE 11.1.6

(Effect of release or settlement)

(1) Release of one joint and several obligor, or settlement with one joint and several obligor, discharges all the other obligors for the share of the released or settling obligor, unless the circumstances indicate otherwise.

(2) When the other obligors are discharged for the share of the released obligor, they no longer have a contributory claim against the released obligor under Article 11.1.10.

ARTICLE 11.1.7

(Effect of expiration or suspension of limitation period)

(1) Expiration of the limitation period of the obligee's rights against one joint and several obligor does not affect:

(a) the obligations to the obligee of the other joint and several obligors; or

(b) the rights of recourse between the joint and several obligors under Article 11.1.10.

(2) If the obligee initiates proceedings under Articles 10.5, 10.6 or 10.7 against one joint and several obligor, the running of the limitation period is also suspended against the other joint and several obligors.

ARTICLE 11.1.8

(Effect of judgment)

(1) A decision by a court as to the liability to the obligee of one joint and several obligor does not affect:

(a) the obligations to the obligee of the other joint and several obligors; or

(b) the rights of recourse between the joint and several obligors under Article 11.1.10.

(2) However, the other joint and several obligors may rely on such a decision, except if it was based on grounds personal to the obligor concerned. In such a case, the rights of recourse between the joint and several obligors under Article 11.1.10 are affected accordingly.

ARTICLE 11.1.9

(Apportionment among joint and several obligors)

As among themselves, joint and several obligors are bound in equal shares, unless the circumstances indicate otherwise.

ARTICLE 11.1.10

(Extent of contributory claim)

A joint and several obligor who has performed more than its share may claim the excess from any of the other obligors to the extent of each obligor's unperformed share.

ARTICLE 11.1.11

(Rights of the obligee)

(1) A joint and several obligor to whom Article 11.1.10 applies may also exercise the rights of the obligee, including all rights securing their performance, to recover the excess from all or any of the other obligors to the extent of each obligor's unperformed share.

(2) An obligee who has not received full performance retains its rights against the co-obligors to the extent of the unperformed part, with precedence over co-obligors exercising contributory claims.

ARTICLE 11.1.12

(Defences in contributory claims)

A joint and several obligor against whom a claim is made by the co-obligor who has performed the obligation:

(a) may raise any common defences and rights of set-off that were available to be asserted by the co-obligor against the obligee ;

(b) may assert defences which are personal to itself;

(c) may not assert defences and rights of set-off which are personal to one or several of the other co-obligors.

ARTICLE 11.1.13

(Inability to recover)

If a joint and several obligor who has performed more than that obligor's share is unable, despite all reasonable efforts, to recover contribution from another joint and several obligor, the share of the others, including the one who has performed, is increased proportionally.

SECTION 2: PLURALITY OF OBLIGEES

ARTICLE 11.2.1

(Definitions)

When several obligees can claim performance of the same obligation from an obligor:

(a) the claims are separate when each obligee can only claim its share;

(b) the claims are joint and several when each obligee can claim the whole performance;

(c) the claims are joint when all obligees have to claim performance together.

ARTICLE 11.2.2

(Effects of joint and several claims)

Full performance of an obligation in favour of one of the joint and several obligees discharges the obligor towards the other obligees.

ARTICLE 11.2.3

(Availability of defences against joint and several obligees)

(1) The obligor may assert against any of the joint and several obligees all the defences and rights of set-off that are personal to its relationship to that obligee or that it can assert against all the co-obligees, but may not assert defences and rights of set-off that are personal to its relationship to one or several of the other co-obligees.

(2) The provisions of Articles 11.1.5, 11.1.6, 11.1.7 and 11.1.8 apply, with appropriate adaptations, to joint and several claims.

ARTICLE 11.2.4

(Allocation between joint and several obligees)

(1) As among themselves, joint and several obligees are entitled to equal shares, unless the circumstances indicate otherwise.

(2) An obligee who has received more than its share must transfer the excess to the other obligees to the extent of their respective shares.



PERMANENT COURT OF ARBITRATION ARBITRATION RULES 2012

<u>/ ភូមិ</u> #### %



PERMANENT COURT OF ARBITRATION ARBITRATION RULES 2012

Effective December 17, 2012

TABLE OF CONTENTS

Section I. Introductory rules	5
Scope of application	
Article 1	5
Article 2	5
Notice of arbitration	
Article 3	6
Response to the notice of arbitration	
Article 4	7
Representation and assistance	
Article 5	7
Appointing authority	
Article 6	
Section II. Composition of the arbitral tribunal	
Number of arbitrators	
Article 7	
Appointment of arbitrators	
Article 8	8
Article 9	
Article 10	
Disclosures by and challenge of arbitrators	······································
Article 11	10
Article 12	
Article 13	
Replacement of an arbitrator	10
Article 14	11
Repetition of hearings in the event of the replacement of an arbitrator	
Article 15	11
Exclusion of liability	
Article 16	
Section III. Arbitral proceedings	
General provisions Article 17	11
Place of arbitration	
Article 18	10
Language Article 19	10
Statement of claim	
Article 20	10
Statement of defence	
Article 21	12
Amendments to the claim or defence	
Article 22	12
Pleas as to the jurisdiction of the arbitral tribunal Article 23	10
Further written statements	
Article 24	1 /
Periods of time	
Article 25	1 /

Interim measures	
Article 26	14
Evidence	
Article 27	15
Hearings	
Article 28	15
Experts appointed by the arbitral tribunal	
Article 29	16
Default	
Article 30	16
Closure of hearings	
Article 31	17
Waiver of right to object	
Article 32	17
Section IV. The award	
Decisions	
Article 33	17
Form and effect of the award	
Article 34	17
Applicable law, <i>amiable compositeur</i>	
Article 35	18
Settlement or other grounds for termination	
Article 36	
Interpretation of the award	
Article 37	
Correction of the award	
Article 38	19
Additional award	
Article 39	
Definition of costs	
Article 40	
Article 41	
Allocation of costs	
Article 42	
Deposit of costs	
Article 43	
Annex	
Model arbitration clause for contracts	
Model arbitration clause for treaties and other agreements	
Possible waiver statement	
Model statements of impartiality and independence pursuant to article 11 of the Rules	

Introduction

These Rules are for use in arbitrating disputes involving at least one State, State-controlled entity, or intergovernmental organization. They add a new option for arbitration of disputes under the auspices of the Permanent Court of Arbitration (hereinafter the "PCA") without replacing the previously adopted PCA Rules, which remain valid and available. The Rules are optional and are based on the 2010 UNCITRAL Arbitration Rules with changes made in order to:

- (i) Reflect the public international law elements that may arise in disputes involving a State, Statecontrolled entity, and/or intergovernmental organization;
- (ii) Indicate the role of the Secretary-General and the International Bureau of the PCA; and
- (iii) Emphasize flexibility and party autonomy. For example:
 - (a) The Rules allow for arbitration of multiparty disputes involving a combination of States, State-controlled entities, intergovernmental organizations, and private parties;
 - (b) The Rules and the services of the Secretary-General and the International Bureau of the PCA are available for use by all States and their entities and enterprises, and are not restricted to disputes in which the State is a party to either the Hague Convention for the Pacific Settlement of International Disputes of 1899 or that of 1907;
 - (c) The Rules allow parties to choose an arbitral tribunal of one, three, or five persons; and
 - (d) The choice of arbitrators is not limited to persons who are listed as Members of the PCA.

Model clauses that parties may consider inserting in treaties, contracts, or other agreements to provide for arbitration of existing or future disputes are set forth in the annex to these Rules.

Section I. Introductory rules

Scope of application*

Article 1

- 1. Where a State, State-controlled entity, or intergovernmental organization has agreed with one or more States, State-controlled entities, intergovernmental organizations, or private parties that disputes between them in respect of a defined legal relationship, whether contractual, treaty-based, or otherwise, shall be referred to arbitration under the Permanent Court of Arbitration Arbitration Rules 2012 (hereinafter the "Rules"), then such disputes shall be settled in accordance with these Rules subject to such modification as the parties may agree.
- 2. Agreement by a State, State-controlled entity, or intergovernmental organization to arbitrate under these Rules with a party that is not a State, State-controlled entity, or intergovernmental organization constitutes a waiver of any right of immunity from jurisdiction in respect of the proceedings relating to the dispute in question to which such party might otherwise be entitled. A waiver of immunity relating to the execution of an arbitral award must be explicitly expressed.
- 3. The International Bureau of the Permanent Court of Arbitration at the Hague (hereinafter the "International Bureau") shall serve as registry for the proceedings and provide secretariat services.
- 4. The involvement of at least one State, State-controlled entity, or intergovernmental organization as a party to the dispute is not necessary for jurisdiction where all the parties have agreed to settle a dispute under these Rules. However, where the Secretary-General of the Permanent Court of Arbitration determines that no State, State-controlled entity, or intergovernmental organization is a party to the dispute, the Secretary-General may decide to limit the Permanent Court of Arbitration's role in the proceedings to the function of the Secretary-General as appointing authority, with the role of the International Bureau under these Rules to be assumed by the arbitral tribunal.

* Model arbitration clauses can be found in the annex to the Rules.

Notice and calculation of periods of time

- 1. A notice, including a notification, communication or proposal, may be transmitted by any means of communication that provides or allows for a record of its transmission.
- 2. If an address has been designated by a party specifically for this purpose or authorized by the arbitral tribunal, any notice shall be delivered to that party at that address, and if so delivered shall be deemed to have been received. Delivery by electronic means such as facsimile or e-mail may only be made to an address so designated or authorized.
- 3. In the absence of such designation or authorization, a notice is:
 - (a) Received if it is physically delivered to the addressee; or
 - (b) Deemed to have been received if it is delivered at the place of business, habitual residence or mailing address of the addressee.

- 4. If, after reasonable efforts, delivery cannot be effected in accordance with paragraphs 2 or 3, a notice is deemed to have been received if it is sent to the addressee's last-known place of business, habitual residence or mailing address by registered letter or any other means that provides a record of delivery or of attempted delivery.
- 5. A notice shall be deemed to have been received on the day it is delivered in accordance with paragraphs 2, 3 or 4, or attempted to be delivered in accordance with paragraph 4. A notice transmitted by electronic means is deemed to have been received on the day it is sent, except that a notice of arbitration so transmitted is only deemed to have been received on the day when it reaches the addressee's electronic address.
- 6. For the purpose of calculating a period of time under these Rules, such period shall begin to run on the day following the day when a notice is received. If the last day of such period is an official holiday or a non-business day at the residence or place of business of the addressee, the period is extended until the first business day which follows. Official holidays or non-business days occurring during the running of the period of time are included in calculating the period.

Notice of arbitration

- 1. The party or parties initiating recourse to arbitration (hereinafter the "claimant") shall communicate to the other party or parties (hereinafter the "respondent") and the International Bureau a notice of arbitration.
- 2. Arbitral proceedings shall be deemed to commence on the date on which the notice of arbitration is received by the respondent.
- 3. The notice of arbitration shall include the following:
 - (a) A demand that the dispute be referred to arbitration;
 - (b) The names and contact details of the parties;
 - (c) Identification of the arbitration agreement that is invoked;
 - (d) Identification of any rule, decision, agreement, contract, convention, treaty, constituent instrument of an organization or agency, or relationship out of, or in relation to which, the dispute arises;
 - (e) A brief description of the claim and an indication of the amount involved, if any;
 - (f) The relief or remedy sought;
 - (g) A proposal as to the number of arbitrators, language and place of arbitration, if the parties have not previously agreed thereon.
- 4. The notice of arbitration may also include:
 - (a) A proposal for the appointment of a sole arbitrator referred to in article 8, paragraph 1;
 - (b) Notification of the appointment of an arbitrator referred to in articles 9 or 10.
- 5. The constitution of the arbitral tribunal shall not be hindered by any controversy with respect to the sufficiency of the notice of arbitration, which shall be finally resolved by the arbitral tribunal.

Response to the notice of arbitration

Article 4

- 1. Within 30 days of the receipt of the notice of arbitration, or such other period as may be set by the International Bureau, the respondent shall communicate to the claimant and the International Bureau a response to the notice of arbitration, which shall include:
 - (a) The name and contact details of each respondent;
 - (b) A response to the information set forth in the notice of arbitration, pursuant to article 3, paragraphs 3 (c) to (g).
- 2. The response to the notice of arbitration may also include:
 - (a) Any plea that an arbitral tribunal to be constituted under these Rules lacks jurisdiction;
 - (b) A proposal for the appointment of a sole arbitrator referred to in article 8, paragraph 1;
 - (c) Notification of the appointment of an arbitrator referred to in articles 9 or 10;
 - (d) A brief description of counterclaims or claims for the purpose of a set-off, if any, including where relevant, an indication of the amounts involved, and the relief or remedy sought;
 - (e) A notice of arbitration in accordance with article 3 in case the respondent formulates a claim against a party to the arbitration agreement other than the claimant.
- 3. The constitution of the arbitral tribunal shall not be hindered by any controversy with respect to the respondent's failure to communicate a response to the notice of arbitration, or an incomplete or late response to the notice of arbitration, which shall be finally resolved by the arbitral tribunal.

Representation and assistance

- 1. In disputes involving only States and/or intergovernmental organizations, each party shall appoint an agent. Each party may also be assisted by persons of its choice.
- 2. In other disputes under these Rules, each party may be represented or assisted by persons chosen by it.
- 3. The names and addresses of agents, party representatives, and other persons assisting the parties must be communicated to all parties, to the International Bureau, and to the arbitral tribunal. Such communication must specify whether the appointment is being made for purposes of representation or assistance. Where a person is to act as an agent or representative of a party, the arbitral tribunal, on its own initiative or at the request of any party, may at any time require proof of authority granted to the agent or representative in such a form as the arbitral tribunal may determine.

Appointing authority

Article 6

- 1. The Secretary-General of the Permanent Court of Arbitration shall serve as appointing authority.
- 2. In exercising its functions under these Rules, the appointing authority may require from any party and the arbitrators the information it deems necessary and it shall give the parties and, where appropriate, the arbitrators, an opportunity to present their views in any manner it considers appropriate.
- 3. The appointing authority shall have regard to such considerations as are likely to secure the appointment of an independent and impartial arbitrator and shall take into account the advisability of appointing an arbitrator of a nationality other than the nationalities of the parties.

Section II. Composition of the arbitral tribunal

Number of arbitrators

Article 7

- 1. If the parties have not previously agreed on the number of arbitrators, and if within 30 days after the receipt by the respondent of the notice of arbitration the parties have not agreed on the number of arbitrators, three arbitrators shall be appointed.
- 2. Notwithstanding paragraph 1, if no other parties have responded to a party's proposal to appoint a sole arbitrator within the time limit provided for in paragraph 1 and the party or parties concerned have failed to appoint a second arbitrator in accordance with articles 9 or 10, the appointing authority may, at the request of a party, appoint a sole arbitrator pursuant to the procedure provided for in article 8, paragraph 2 if it determines that, in view of the circumstances of the case, this is more appropriate.

Appointment of arbitrators (articles 8 to 10)

- 1. If the parties have agreed that a sole arbitrator is to be appointed and if within 30 days after receipt by all other parties of a proposal of an individual who would serve as a sole arbitrator the parties have not reached agreement thereon, a sole arbitrator shall, at the request of a party, be appointed by the appointing authority.
- 2. The appointing authority shall appoint the sole arbitrator as promptly as possible. In making the appointment, the appointing authority shall use the following list-procedure, unless the parties agree that the list-procedure should not be used or unless the appointing authority determines in its discretion that the use of the list-procedure is not appropriate for the case:
 - (a) The appointing authority shall communicate to each of the parties an identical list containing at least three names;
 - (b) Within 15 days after the receipt of this list, or such other period as may be set by the International Bureau, each party may return the list to the appointing authority, without copying the other party, after having deleted the name or names to which it objects and numbered the remaining names on the list in the order of its preference;

- (c) After the expiration of the above period of time the appointing authority shall appoint the sole arbitrator from among the names approved on the lists returned to it and in accordance with the order of preference indicated by the parties;
- (d) If for any reason the appointment cannot be made according to this procedure, the appointing authority may exercise its discretion in appointing the sole arbitrator.

Article 9

- 1. If three arbitrators are to be appointed, each party shall appoint one arbitrator. The two arbitrators thus appointed shall choose the third arbitrator who will act as the presiding arbitrator of the arbitral tribunal. If five arbitrators are to be appointed, the two party-appointed arbitrators shall choose the remaining three arbitrators and designate one of those three as the presiding arbitrator of the tribunal.
- 2. If within 30 days after the receipt of a party's notification of the appointment of an arbitrator the other party has not notified the first party of the arbitrator it has appointed, the first party may request the appointing authority to appoint the second arbitrator.
- 3. If within 30 days after the appointment of the second arbitrator, or such other period as may be set by the International Bureau, the two arbitrators have not agreed on the choice of the remaining arbitrators and/or the presiding arbitrator, the remaining arbitrators and/or the presiding arbitrator shall be appointed by the appointing authority in the same way as a sole arbitrator would be appointed under article 8, paragraph 2.

- 1. For the purposes of article 9, paragraph 1, where three or five arbitrators are to be appointed and there are multiple parties as claimant or as respondent, unless the parties have agreed to another method of appointment of arbitrators, the multiple parties jointly, whether as claimant or as respondent, shall appoint an arbitrator.
- 2. If the parties have agreed that the arbitral tribunal is to be composed of a number of arbitrators other than one, three, or five, the arbitrators shall be appointed according to the method agreed upon by the parties.
- 3. In the event of any failure to constitute the arbitral tribunal under these Rules, the appointing authority shall, at the request of any party, constitute the arbitral tribunal and, in doing so, may revoke any appointment already made and appoint each of the arbitrators and designate one of them as the presiding arbitrator. The appointing authority may, if it deems it appropriate, reappoint previous appointees.
- 4. In appointing arbitrators pursuant to these Rules, the parties and the appointing authority are free to choose persons who are not Members of the Permanent Court of Arbitration.

Disclosures by and challenge of arbitrators** (articles 11 to 13)

Article 11

When a person is approached in connection with his or her possible appointment as an arbitrator, he or she shall disclose any circumstances likely to give rise to justifiable doubts as to his or her impartiality or independence. An arbitrator, from the time of his or her appointment and throughout the arbitral proceedings, shall without delay disclose any such circumstances to the parties and the other arbitrators unless they have already been informed by him or her of these circumstances.

** Model statements of independence pursuant to article 11 can be found in the annex to the Rules.

Article 12

- 1. Any arbitrator may be challenged if circumstances exist that give rise to justifiable doubts as to the arbitrator's impartiality or independence.
- 2. A party may challenge the arbitrator appointed by it only for reasons of which it becomes aware after the appointment has been made.
- 3. In the event that an arbitrator fails to act or in the event of the *de jure* or *de facto* impossibility of his or her performing his or her functions, the procedure in respect of the challenge of an arbitrator as provided in article 13 shall apply.
- 4. If an arbitrator on a tribunal of three, five, or more persons fails to participate in the arbitration, the other arbitrators shall, unless the parties agree otherwise, have the power in their sole discretion to continue the arbitration and to make any decision, ruling or award, notwithstanding the failure of one arbitrator to participate. In determining whether to continue the arbitration or to render any decision, ruling or award without the participation of an arbitrator, the other arbitrators shall take into account the stage of the arbitration, the reason, if any, expressed by the arbitrator for such non-participation, and such other matters as they consider appropriate in the circumstances of the case. In the event that the other arbitrators determine not to continue the arbitration without the non-participating arbitrator, the arbitrator shall be appointed pursuant to the provisions of articles 8 to 11, unless the parties agree on a different method of appointment.

- 1. A party that intends to challenge an arbitrator shall send notice of its challenge within 30 days after it has been notified of the appointment of the challenged arbitrator, or within 30 days after the circumstances mentioned in articles 11 and 12 became known to that party.
- 2. The notice of challenge shall be communicated to all other parties, to the arbitrator who is challenged, to the other arbitrators, and to the International Bureau. The notice of challenge shall state the reasons for the challenge.
- 3. When an arbitrator has been challenged by a party, all parties may agree to the challenge. The arbitrator may also, after the challenge, withdraw from his or her office. In neither case does this imply acceptance of the validity of the grounds for the challenge.
- 4. If, within 15 days from the date of the notice of challenge, all parties do not agree to the challenge or the challenged arbitrator does not withdraw, the party making the challenge may elect to pursue it. In that case, within 30 days from the date of the notice of challenge, it shall seek a decision on the challenge by the appointing authority.

5. In rendering a decision on the challenge, the appointing authority may indicate the reasons for the decision, unless the parties agree that no reasons shall be given.

Replacement of an arbitrator

Article 14

- 1. Subject to paragraph 2 of this article, in any event where an arbitrator has to be replaced during the course of the arbitral proceedings, a substitute arbitrator shall be appointed or chosen pursuant to the procedure provided for in articles 8 to 11 that was applicable to the appointment or choice of the arbitrator being replaced. This procedure shall apply even if during the process of appointing the arbitrator to be replaced, a party had failed to exercise its right to appoint or to participate in the appointment.
- 2. If, at the request of a party, the appointing authority determines that, in view of the exceptional circumstances of the case, it would be justified for a party to be deprived of its right to appoint a substitute arbitrator, the appointing authority may, after giving an opportunity to the parties and the remaining arbitrators to express their views, appoint the substitute arbitrator.

Repetition of hearings in the event of the replacement of an arbitrator

Article 15

If an arbitrator is replaced, the proceedings shall resume at the stage where the arbitrator who was replaced ceased to perform his or her functions, unless the arbitral tribunal decides otherwise.

Exclusion of liability

Article 16

The parties waive, to the fullest extent permitted under the applicable law, any claim against the arbitrators and any person appointed by the arbitral tribunal based on any act or omission in connection with the arbitration.

Section III. Arbitral proceedings

General provisions

- 1. Subject to these Rules, the arbitral tribunal may conduct the arbitration in such manner as it considers appropriate, provided that the parties are treated with equality and that at an appropriate stage of the proceedings each party is given a reasonable opportunity of presenting its case. The arbitral tribunal, in exercising its discretion, shall conduct the proceedings so as to avoid unnecessary delay and expense and to provide a fair and efficient process for resolving the parties' dispute.
- 2. As soon as practicable after its constitution and after inviting the parties to express their views, the arbitral tribunal shall establish the provisional timetable of the arbitration. The arbitral tribunal may, at any time, after inviting the parties to express their views, extend or abridge any period of time prescribed under these Rules or agreed by the parties.
- 3. If at an appropriate stage of the proceedings any party so requests, the arbitral tribunal shall hold hearings for the presentation of evidence by witnesses, including expert witnesses, or for oral argument. In the absence of such a request, the arbitral tribunal shall decide whether to hold such

hearings or whether the proceedings shall be conducted on the basis of documents and other materials.

- 4. All communications to the arbitral tribunal by one party shall be communicated by that party to all other parties and the International Bureau. Such communications shall be made at the same time, except as otherwise permitted by the arbitral tribunal if it may do so under applicable law.
- 5. The arbitral tribunal may, at the request of any party, allow one or more third persons to be joined in the arbitration as a party provided such person is a party to the arbitration agreement, unless the arbitral tribunal finds, after giving all parties, including the person or persons to be joined, the opportunity to be heard, that joinder should not be permitted because of prejudice to any of those parties. The arbitral tribunal may make a single award or several awards in respect of all parties so involved in the arbitration.

Place of arbitration

Article 18

- 1. If the parties have not previously agreed on the place of arbitration, the place of arbitration shall be determined by the arbitral tribunal having regard to the circumstances of the case. The award shall be deemed to have been made at the place of arbitration.
- 2. The arbitral tribunal may meet at any location it considers appropriate for deliberations. Unless otherwise agreed by the parties, the arbitral tribunal may also meet at any location it considers appropriate for any other purpose, including hearings.

Language

Article 19

- 1. Subject to an agreement by the parties, the arbitral tribunal shall, promptly after its appointment, determine the language or languages to be used in the proceedings. This determination shall apply to the statement of claim, the statement of defence, and any further written statements and, if oral hearings take place, to the language or languages to be used in such hearings.
- 2. The arbitral tribunal may order that any documents annexed to the statement of claim or statement of defence, and any supplementary documents or exhibits submitted in the course of the proceedings, delivered in their original language, shall be accompanied by a translation into the language or languages agreed upon by the parties or determined by the arbitral tribunal.

Statement of claim

- 1. The claimant shall communicate its statement of claim in writing to the respondent, to the International Bureau, and to each of the arbitrators within a period of time to be determined by the arbitral tribunal. The claimant may elect to treat its notice of arbitration referred to in article 3 as a statement of claim, provided that the notice of arbitration also complies with the requirements of paragraphs 2 to 4 of this article.
- 2. The statement of claim shall include the following particulars:
 - (a) The names and contact details of the parties;
 - (b) A statement of the facts supporting the claim;

- (c) The points at issue;
- (d) The relief or remedy sought;
- (e) The legal grounds or arguments supporting the claim.
- 3. A copy of any rule, decision, agreement, contract, convention, treaty, constituent instrument of an organization or agency, or relationship out of, or in relation to which, the dispute arises, and of the arbitration agreement shall be annexed to the statement of claim.
- 4. The statement of claim should, as far as possible, be accompanied by all documents and other evidence relied upon by the claimant, or contain references to them.

Statement of defence

Article 21

- 1. The respondent shall communicate its statement of defence in writing to the claimant, to the International Bureau, and to each of the arbitrators within a period of time to be determined by the arbitral tribunal. The respondent may elect to treat its response to the notice of arbitration referred to in article 4 as a statement of defence, provided that the response to the notice of arbitration also complies with the requirements of paragraph 2 of this article.
- 2. The statement of defence shall reply to the particulars (b) to (e) of the statement of claim (article 20, paragraph 2). The statement of defence should, as far as possible, be accompanied by all documents and other evidence relied upon by the respondent, or contain references to them.
- 3. In its statement of defence, or at a later stage in the arbitral proceedings if the arbitral tribunal decides that the delay was justified under the circumstances, the respondent may make a counterclaim or rely on a claim for the purpose of a set-off provided that the arbitral tribunal has jurisdiction over it.
- 4. The provisions of article 20, paragraphs 2 to 4 shall apply to a counterclaim, a claim under article 4, paragraph 2(e), and a claim relied on for the purpose of a set-off.

Amendments to the claim or defence

Article 22

During the course of the arbitral proceedings, a party may amend or supplement its claim or defence, including a counterclaim or a claim for the purpose of a set-off, unless the arbitral tribunal considers it inappropriate to allow such amendment or supplement having regard to the delay in making it or prejudice to other parties or any other circumstances. However, a claim or defence, including a counterclaim or a claim for the purpose of a set-off, may not be amended or supplemented in such a manner that the amended or supplemented claim or defence falls outside the jurisdiction of the arbitral tribunal.

Pleas as to the jurisdiction of the arbitral tribunal

Article 23

1. The arbitral tribunal shall have the power to rule on its own jurisdiction, including any objections with respect to the existence or validity of the arbitration agreement. For that purpose, an arbitration clause that forms part of a contract, treaty, or other agreement shall be treated as an

agreement independent of the other terms of the contract, treaty, or other agreement. A decision by the arbitral tribunal that the contract, treaty, or other agreement is null, void, or invalid shall not entail automatically the invalidity of the arbitration clause.

- 2. A plea that the arbitral tribunal does not have jurisdiction shall be raised no later than in the statement of defence or, with respect to a counterclaim or a claim for the purpose of a set-off, in the reply to the counterclaim or to the claim for the purpose of a set-off. A party is not precluded from raising such a plea by the fact that it has appointed, or participated in the appointment of, an arbitrator. A plea that the arbitral tribunal is exceeding the scope of its authority shall be raised as soon as the matter alleged to be beyond the scope of its authority is raised during the arbitral proceedings. The arbitral tribunal may, in either case, admit a later plea if it considers the delay justified.
- 3. The arbitral tribunal may rule on a plea referred to in paragraph 2 either as a preliminary question or in an award on the merits. The arbitral tribunal may continue the arbitral proceedings and make an award, notwithstanding any pending challenge to its jurisdiction before a competent authority.

Further written statements

Article 24

The arbitral tribunal shall decide which further written statements, in addition to the statement of claim and the statement of defence, shall be required from the parties or may be presented by them and shall fix the periods of time for communicating such statements.

Periods of time

Article 25

The periods of time fixed by the arbitral tribunal for the communication of written statements (including the statement of claim and statement of defence) should not exceed 45 days. However, the arbitral tribunal may extend the time limits if it concludes that an extension is justified.

Interim measures

- 1. The arbitral tribunal may, at the request of a party, grant interim measures.
- 2. An interim measure is any temporary measure by which, at any time prior to the issuance of the award by which the dispute is finally decided, the arbitral tribunal orders a party, for example and without limitation, to:
 - (a) Maintain or restore the status quo pending determination of the dispute;
 - (b) Take action that would prevent, or refrain from taking action that is likely to cause, (i) current or imminent harm or (ii) prejudice to the arbitral process itself;
 - (c) Provide a means of preserving assets out of which a subsequent award may be satisfied; or
 - (d) Preserve evidence that may be relevant and material to the resolution of the dispute.
- 3. The party requesting an interim measure under paragraphs 2 (a) to (c) shall satisfy the arbitral tribunal that:

- (a) Harm not adequately reparable by an award of damages is likely to result if the measure is not ordered, and such harm substantially outweighs the harm that is likely to result to the party against whom the measure is directed if the measure is granted; and
- (b) There is a reasonable possibility that the requesting party will succeed on the merits of the claim. The determination on this possibility shall not affect the discretion of the arbitral tribunal in making any subsequent determination.
- 4. With regard to a request for an interim measure under paragraph 2 (d), the requirements in paragraphs 3 (a) and (b) shall apply only to the extent the arbitral tribunal considers appropriate.
- 5. The arbitral tribunal may modify, suspend or terminate an interim measure it has granted, upon application of any party or, in exceptional circumstances and upon prior notice to the parties, on the arbitral tribunal's own initiative.
- 6. The arbitral tribunal may require the party requesting an interim measure to provide appropriate security in connection with the measure.
- 7. The arbitral tribunal may require any party promptly to disclose any material change in the circumstances on the basis of which the interim measure was requested or granted.
- 8. The party requesting an interim measure may be liable for any costs and damages caused by the measure to any party if the arbitral tribunal later determines that, in the circumstances then prevailing, the measure should not have been granted. The arbitral tribunal may award such costs and damages at any point during the proceedings.
- 9. A request for interim measures addressed by any party to a judicial authority shall not be deemed incompatible with the agreement to arbitrate, or as a waiver of that agreement.

Evidence

Article 27

- 1. Each party shall have the burden of proving the facts relied on to support its claim or defence.
- 2. Witnesses, including expert witnesses, who are presented by the parties to testify to the arbitral tribunal on any issue of fact or expertise may be any individual, notwithstanding that the individual is a party to the arbitration or in any way related to a party. Unless otherwise directed by the arbitral tribunal, statements by witnesses, including expert witnesses, may be presented in writing and signed by them.
- 3. At any time during the arbitral proceedings the arbitral tribunal may require the parties to produce documents, exhibits or other evidence within such a period of time as the arbitral tribunal shall determine. The arbitral tribunal may also, after consultation with the parties, perform a site visit.
- 4. The arbitral tribunal shall determine the admissibility, relevance, materiality and weight of the evidence offered.

Hearings

Article 28

1. In the event of an oral hearing, the arbitral tribunal shall give the parties adequate advance notice of the date, time and place thereof.

- 2. Witnesses, including expert witnesses, may be heard under the conditions and examined in the manner set by the arbitral tribunal.
- 3. Hearings shall be held *in camera* unless the parties agree otherwise. The arbitral tribunal may require the retirement of any witness or witnesses, including expert witnesses, during the testimony of such other witnesses, except that a witness, including an expert witness, who is a party to the arbitration shall not, in principle, be asked to retire.
- 4. The arbitral tribunal may direct that witnesses, including expert witnesses, be examined through means of telecommunication that do not require their physical presence at the hearing (such as videoconference).

Experts appointed by the arbitral tribunal

Article 29

- 1. After consultation with the parties, the arbitral tribunal may appoint one or more independent experts to report to it, in writing, on specific issues to be determined by the arbitral tribunal. A copy of the expert's terms of reference, established by the arbitral tribunal, shall be communicated to the parties.
- 2. The expert shall, in principle before accepting appointment, submit to the arbitral tribunal and to the parties a description of his or her qualifications and a statement of his or her impartiality and independence. Within the time ordered by the arbitral tribunal, the parties shall inform the arbitral tribunal whether they have any objections as to the expert's qualifications, impartiality or independence. The arbitral tribunal shall decide promptly whether to accept any such objections. After an expert's appointment, a party may object to the expert's qualifications, impartiality or independence only if the objection is for reasons of which the party becomes aware after the appointment has been made. The arbitral tribunal shall decide promptly what, if any, action to take.
- 3. The parties shall give the expert any relevant information or produce for his or her inspection any relevant documents or goods that he or she may require of them. Any dispute between a party and such expert as to the relevance of the required information or production shall be referred to the arbitral tribunal for decision.
- 4. Upon receipt of the expert's report, the arbitral tribunal shall communicate a copy of the report to the parties, which shall be given the opportunity to express, in writing, their opinion on the report. A party shall be entitled to examine any document on which the expert relied in his or her report.
- 5. If a party so requests or if the arbitral tribunal considers it necessary, the expert shall, after delivery of the report, participate in a hearing where the parties have the opportunity to put questions to him or her and to present expert witnesses in order to testify on the points at issue. The provisions of article 28 shall be applicable to such proceedings.

Default

Article 30

1. If, within the period of time fixed by these Rules or the arbitral tribunal, without showing sufficient cause:

- (a) The claimant has failed to communicate its statement of claim, the arbitral tribunal shall issue an order for the termination of the arbitral proceedings, unless there are remaining matters that may need to be decided and the arbitral tribunal considers it appropriate to do so;
- (b) The respondent has failed to communicate its response to the notice of arbitration or its statement of defence, the arbitral tribunal shall order that the proceedings continue, without treating such failure in itself as an admission of the claimant's allegations; the provisions of this subparagraph also apply to a claimant's failure to submit a defence to a counterclaim or to a claim for the purpose of a set-off.
- 2. If a party, duly notified under these Rules, fails to appear at a hearing, without showing sufficient cause for such failure, the arbitral tribunal may proceed with the arbitration.
- 3. If a party, duly invited by the arbitral tribunal to produce documents, exhibits or other evidence, fails to do so within the established period of time, without showing sufficient cause for such failure, the arbitral tribunal may make the award on the evidence before it.

Closure of proceedings

Article 31

- 1. When it is satisfied that the parties have had a reasonable opportunity to present their cases, the arbitral tribunal shall declare the proceedings closed.
- 2. The arbitral tribunal may, if it considers it necessary owing to exceptional circumstances, decide, on its own initiative or upon application of a party, to reopen the proceedings at any time before the award is made.

Waiver of right to object

Article 32

A failure by any party to object promptly to any non-compliance with these Rules or with any requirement of the arbitration agreement shall be deemed to be a waiver of the right of such party to make such an objection, unless such party can show that, under the circumstances, its failure to object was justified.

Section IV. The award

Decisions

Article 33

- 1. When there is more than one arbitrator, any award or other decision of the arbitral tribunal shall be made by a majority of the arbitrators.
- 2. In the case of questions of procedure, when there is no majority or when the arbitral tribunal so authorizes, the presiding arbitrator may decide alone, subject to revision, if any, by the arbitral tribunal.

Form and effect of the award

Article 34

1. The arbitral tribunal may make separate awards on different issues at different times.

- 2. All awards shall be made in writing and shall be final and binding on the parties. The parties shall carry out all awards without delay.
- 3. The arbitral tribunal shall state the reasons upon which the award is based, unless the parties have agreed that no reasons are to be given.
- 4. An award shall be signed by the arbitrators and it shall contain the date on which the award was made and indicate the place of arbitration. Where there is more than one arbitrator and any of them fails to sign, the award shall state the reason for the absence of the signature.
- 5. An award may be made public with the consent of all parties or where and to the extent disclosure is required of a party by legal duty, to protect or pursue a legal right or in relation to legal proceedings before a court or other competent authority.
- 6. Copies of the award signed by the arbitrators shall be communicated to the parties by the International Bureau.
- 7. In cases involving only States, the parties shall communicate to the International Bureau the laws, regulations, or other documents evidencing the execution of the award.

Applicable law, *amiable compositeur*

Article 35

- 1. The arbitral tribunal shall apply the rules of law designated by the parties as applicable to the substance of the dispute. Failing such designation by the parties, the arbitral tribunal shall:
 - (a) In cases involving only States, decide such disputes in accordance with international law by applying:
 - i. International conventions, whether general or particular, establishing rules expressly recognized by the contesting States;
 - ii. International custom, as evidence of a general practice accepted as law;
 - iii. The general principles of law recognized by civilized nations;
 - iv. Judicial and arbitral decisions and the teachings of the most highly qualified publicists of the various nations, as subsidiary means for the determination of rules of law.
 - (b) In cases involving only States and intergovernmental organizations, apply the rules of the organization concerned and the law applicable to any agreement or relationship between the parties, and, where appropriate, the general principles governing the law of intergovernmental organizations and the rules of general international law.
 - (c) In cases involving intergovernmental organizations and private parties, have regard both to the rules of the organization concerned and to the law applicable to the agreement or relationship out of or in relation to which the dispute arises, and, where appropriate, to the general principles governing the law of intergovernmental organizations and to the rules of general international law. In such cases, the arbitral tribunal shall decide in accordance with the terms of the agreement and shall take into account relevant trade usages.

- (d) In all other cases, apply the law which it determines to be appropriate. In such cases, the arbitral tribunal shall decide in accordance with the terms of the agreement and shall take into account relevant trade usages.
- 2. The arbitral tribunal shall decide as *amiable compositeur* or *ex aequo et bono* only if the parties have expressly authorized the arbitral tribunal to do so.

Settlement or other grounds for termination

Article 36

- 1. If, before the award is made, the parties agree on a settlement of the dispute, the arbitral tribunal shall either issue an order for the termination of the arbitral proceedings or, if requested by the parties and accepted by the arbitral tribunal, record the settlement in the form of an arbitral award on agreed terms. The arbitral tribunal is not obliged to give reasons for such an award.
- 2. If, before the award is made, the continuation of the arbitral proceedings becomes unnecessary or impossible for any reason not mentioned in paragraph 1, the arbitral tribunal shall inform the parties of its intention to issue an order for the termination of the proceedings. The arbitral tribunal shall have the power to issue such an order unless there are remaining matters that may need to be decided and the arbitral tribunal considers it appropriate to do so.
- 3. Copies of the order for termination of the arbitral proceedings or of the arbitral award on agreed terms, signed by the arbitrators, shall be communicated by the arbitral tribunal to the parties. Where an arbitral award on agreed terms is made, the provisions of article 34, paragraphs 2, 4 and 5 shall apply.

Interpretation of the award

Article 37

- 1. Within 30 days after the receipt of the award, a party, with notice to the other parties and the International Bureau, may request that the arbitral tribunal give an interpretation of the award.
- 2. The interpretation shall be given in writing within 45 days after the receipt of the request. The interpretation shall form part of the award and the provisions of article 34, paragraphs 2 to 6, shall apply.

Correction of the award

Article 38

- 1. Within 30 days after the receipt of the award, a party, with notice to the other parties and the International Bureau, may request the arbitral tribunal to correct in the award any error in computation, any clerical or typographical error, or any error or omission of a similar nature. If the arbitral tribunal considers that the request is justified, it shall make the correction within 45 days of receipt of the request.
- 2. The arbitral tribunal may within 30 days after the communication of the award make such corrections on its own initiative.
- 3. Such corrections shall be in writing and shall form part of the award. The provisions of article 34, paragraphs 2 to 6, shall apply.

Additional award

Article 39

- 1. Within 30 days after the receipt of the termination order or the award, a party, with notice to the other parties and the International Bureau, may request the arbitral tribunal to make an award or an additional award as to claims presented in the arbitral proceedings but not decided by the arbitral tribunal.
- 2. If the arbitral tribunal considers the request for an award or additional award to be justified, it shall render or complete its award within 60 days after the receipt of the request. The arbitral tribunal may extend, if necessary, the period of time within which it shall make the award.
- 3. When such an award or additional award is made, the provisions of article 34, paragraphs 2 to 6, shall apply.

Definition of costs

Article 40

- 1. The arbitral tribunal shall fix the costs of arbitration in the final award and, if it deems appropriate, in another decision.
- 2. The term "costs" includes only:
 - (a) The fees of the arbitral tribunal to be stated separately as to each arbitrator and to be fixed by the tribunal itself in accordance with article 41;
 - (b) The reasonable travel and other expenses incurred by the arbitrators;
 - (c) The reasonable costs of expert advice and of other assistance required by the arbitral tribunal;
 - (d) The reasonable travel and other expenses of witnesses to the extent such expenses are approved by the arbitral tribunal;
 - (e) The legal and other costs incurred by the parties in relation to the arbitration to the extent that the arbitral tribunal determines that the amount of such costs is reasonable;
 - (f) The fees and expenses of the International Bureau, including the fees and expenses of the appointing authority.
- 3. In relation to interpretation, correction or completion of any award under articles 37 to 39, the arbitral tribunal may charge the costs referred to in paragraphs 2 (b) to (f), but no additional fees.

Fees and expenses of arbitrators Article 41

- 1. The costs referred to in article 40, paragraphs 2 (a), (b) and (c) shall be reasonable in amount, taking into account the amount in dispute, the complexity of the subject matter, the time spent by the arbitrators and any experts appointed by the arbitral tribunal, and any other relevant circumstances of the case.
- 2. Promptly after its constitution, the arbitral tribunal shall inform the parties as to how it proposes to determine its fees and expenses, including any rates it intends to apply. Within 15 days of

receiving that proposal, any party may refer the proposal to the appointing authority for review. If the appointing authority finds that the proposal of the arbitral tribunal is inconsistent with paragraph 1, it shall make any necessary adjustments thereto, which shall be binding upon the arbitral tribunal.

- 3.
- (a) Before fixing the costs of arbitration pursuant to article 40, the arbitral tribunal shall submit its determination of the costs referred to in article 40, paragraphs 2 (a), (b) and (c), with an explanation of the manner in which the corresponding amounts have been calculated, to the appointing authority for review;
- (b) If the appointing authority finds that the arbitral tribunal's determination is inconsistent with the criteria in paragraph 1 or with the arbitral tribunal's proposal (and any adjustments thereto) under paragraph 2, it shall make any necessary adjustments to the arbitral tribunal's determination. Any such adjustments shall be binding upon the arbitral tribunal when it fixes the costs of arbitration pursuant to article 40.
- 4. Throughout the procedure under paragraphs 2 and 3 of this article, the arbitral tribunal shall proceed with the arbitration, in accordance with article 17, paragraph 1.

Allocation of costs

Article 42

- 1. The costs of arbitration shall in principle be borne by the unsuccessful party or parties. However, the arbitral tribunal may apportion each of such costs between the parties if it determines that apportionment is reasonable, taking into account the circumstances of the case.
- 2. The arbitral tribunal shall in the final award or, if it deems appropriate, in any other award, determine any amount that a party may have to pay to another party as a result of the decision on allocation of costs.

Deposit of costs

Article 43

- The International Bureau, following the commencement of the arbitration, may request the parties to deposit an equal amount as an advance for the costs referred to in article 40, paragraphs 2 (a), (b), (c), and (f). All amounts deposited by the parties pursuant to this paragraph and paragraph 2 of this article shall be directed to the International Bureau, and disbursed by it for such costs, including, *inter alia*, fees to the arbitrators, to the appointing authority, and to the International Bureau. The International Bureau shall ensure that any disbursements of arbitral tribunal fees and expenses made prior to the fixing of the costs of arbitration pursuant to article 40 are consistent with the criteria in article 41, paragraph 1 and with the arbitral tribunal's proposal (and any adjustments thereto) under article 41, paragraph 2.
- 2. During the course of the arbitral proceedings the International Bureau may request supplementary deposits from the parties.
- 3. Any deposit of security for costs ordered by the arbitral tribunal pursuant to article 26 shall be directed to the International Bureau and disbursed by it upon order from the arbitral tribunal.
- 4. If the requested deposits are not paid in full within 30 days after the receipt of the request or such other period as may be set by the International Bureau, the International Bureau shall so inform the parties in order that one or more of them may make the required payment. If such payment is

not made, the arbitral tribunal may order the suspension or termination of the arbitral proceedings.

5. After a termination order or final award has been made, the International Bureau shall render an accounting to the parties of the deposits received and return any unexpended balance to the parties.

Annex

Model arbitration clause for contracts

Any dispute, controversy or claim arising out of or relating to this contract, or the breach, termination or invalidity thereof, shall be settled by arbitration in accordance with the PCA Arbitration Rules 2012.

Model arbitration clause for treaties and other agreements

Any dispute, controversy or claim arising out of or in relation to this [agreement] [treaty], or the existence, interpretation, application, breach, termination, or invalidity thereof, shall be settled by arbitration in accordance with the PCA Arbitration Rules 2012.

Note — Parties should consider adding:

- (a) The number of arbitrators shall be ... (one, three, or five);
- (b) The place of arbitration shall be ... (town and country);
- (c) The language to be used in the arbitral proceedings shall be

Possible waiver statement

Note — If the parties wish to exclude recourse against the arbitral award that may be available under the applicable law, they may consider adding a provision to that effect as suggested below, considering, however, that the effectiveness and conditions of such an exclusion depend on the applicable law.

Waiver: The parties hereby waive their right to any form of recourse against an award to any court or other competent authority, insofar as such waiver can validly be made under the applicable law.

Model statements of impartiality and independence pursuant to article 11 of the Rules

No circumstances to disclose: I am impartial and independent of each of the parties and intend to remain so. To the best of my knowledge, there are no circumstances, past or present, likely to give rise to justifiable doubts as to my impartiality or independence. I shall promptly notify the parties and the other arbitrators of any such circumstances that may subsequently come to my attention during this arbitration.

Circumstances to disclose: I am impartial and independent of each of the parties and intend to remain so. Attached is a statement made pursuant to article 11 of the PCA Arbitration Rules 2012 of (a) my past and present professional, business and other relationships with the parties and (b) any other relevant circumstances. [Include statement] I confirm that those circumstances do not affect my independence and impartiality. I shall promptly notify the parties and the other arbitrators of any such further relationships or circumstances that may subsequently come to my attention during this arbitration.

Note — *Any party may consider requesting from the arbitrator the following addition to the statement of impartiality and independence:*

I confirm, on the basis of the information presently available to me, that I can devote the time necessary to conduct this arbitration diligently, efficiently and in accordance with the time limits in the Rules.



UNCITRAL MODEL LAW ON INTERNATIONAL COMMERCIAL ARBITRATION

/ / **क्रि**क || || || || || **UNCITRAL** UNITED NATIONS COMMISSION ON INTERNATIONAL TRADE LAW

UNCITRAL Model Law on International Commercial Arbitration

1985

With amendments as adopted in 2006



UNITED NATIONS

Further information may be obtained from:

UNCITRAL secretariat, Vienna International Centre, P.O. Box 500, 1400 Vienna, Austria

Telephone: (+43-1) 26060-4060 Internet: uncitral.un.org Telefax: (+43-1) 26060-5813 Email: uncitral@un.org

UNCITRAL Model Law on International Commercial Arbitration

1985

With amendments as adopted in 2006



UNITED NATIONS Vienna, 2008

NOTE

Symbols of United Nations documents are composed of capital letters combined with figures. Mention of such a symbol indicates a reference to a United Nations document.

UNITED NATIONS PUBLICATION Sales No. E.08.V.4 ISBN 978-92-1-133773-0

Contents

Page

Resolutions adopted by the General Assembly	vii
General Assembly Resolution 40/72 (11 December 1985)	vii
General Assembly Resolution 61/33 (4 December 2006)	iii

Part One

		DEL LAW ON INTERNATIONAL COMMERCIAL	1
Chapter I.	Gene	ral provisions	1
Article	1.	Scope of application	1
Article	2.	Definitions and rules of interpretation	
Article	2A.	International origin and general principles	3
Article	3.	Receipt of written communications	3
Article	4.	Waiver of right to object	3
Article	5.	Extent of court intervention	4
Article	6.	Court or other authority for certain functions of arbitration	
		assistance and supervision	4
Chapter II.	Arbi	tration agreement	4
Article	7.	<i>Option I</i> Definition and form of arbitration agreement <i>Option II</i> Definition of arbitration agreement	
Article	8.	Arbitration agreement and substantive claim before court	5
Article	9.	Arbitration agreement and interim measures by court	5
Chapter III.	Con	nposition of arbitral tribunal	6
Article	10.	Number of arbitrators	6
Article	11.	Appointment of arbitrators	6
Article	12.	Grounds for challenge	7
Article	13.	Challenge procedure	
Article	14.	Failure or impossibility to act	8
Article	15.	Appointment of substitute arbitrator	
Chapter IV.	Juri	sdiction of arbitral tribunal	8
		~	~

Article 16. Competence of arbitral tribunal to rule on its jurisdiction ... 8

Chapter IV A.	Interim measures and preliminary orders
Section 1. Article Article	17. Power of arbitral tribunal to order interim measures9
Section 2. Article	Preliminary orders
Article	
Section 3.	Provisions applicable to interim measures and preliminary orders
Article Article Article Article	17 D.Modification, suspension, termination
Section 4. Article Article	6
Section 5. Article	
Chapter V. Co	onduct of arbitral proceedings 14
Article 18. Article 19. Article 20. Article 21. Article 22. Article 23. Article 24. Article 25. Article 26. Article 27.	Equal treatment of parties14Determination of rules of procedure14Place of arbitration14Commencement of arbitral proceedings14Language15Statements of claim and defence15Hearings and written proceedings15Default of a party16Expert appointed by arbitral tribunal16Court assistance in taking evidence16
Chapter VI. Mo	aking of award and termination of proceedings 17
Article 28. Article 29. Article 30. Article 31. Article 32. Article 33.	Rules applicable to substance of dispute17Decision-making by panel of arbitrators.17Settlement17Form and contents of award18Termination of proceedings18Correction and interpretation of award; additional award18

Chapter VII. Red	course against award	19
	Application for setting aside as exclusive recourse against arbitral award	19
Chapter VIII. Re	cognition and enforcement of awards	20
	Recognition and enforcement Grounds for refusing recognition or enforcement	

Page

Part Two

EXPLANATORY NOTE BY THE UNCITRAL SECRETARIAT ON THE MODEL LAW ON INTERNATIONAL COMMERCIAL ARBITRATION..... 23

A.	Ba	ckground to the Model Law24
	1.	Inadequacy of domestic laws
	2.	Disparity between national laws
B.	Sal	ient features of the Model Law
	1.	Special procedural regime for international commercial
		arbitration
	2.	Arbitration agreement
	3.	Composition of arbitral tribunal
	4.	Jurisdiction of arbitral tribunal
	5.	Conduct of arbitral proceedings
	6.	Making of award and termination of proceedings
	7.	Recourse against award
	8.	Recognition and enforcement of awards

Part Three

Resolutions adopted by the General Assembly

40/72. Model Law on International Commercial Arbitration of the United Nations Commission on International Trade Law

The General Assembly,

Recognizing the value of arbitration as a method of settling disputes arising in international commercial relations,

Convinced that the establishment of a model law on arbitration that is acceptable to States with different legal, social and economic systems contributes to the development of harmonious international economic relations,

Noting that the Model Law on International Commercial Arbitration¹ was adopted by the United Nations Commission on International Trade Law at its eighteenth session, after due deliberation and extensive consultation with arbitral institutions and individual experts on international commercial arbitration,

Convinced that the Model Law, together with the Convention on the Recognition and Enforcement of Foreign Arbitral Awards² and the Arbitration Rules of the United Nations Commission on International Trade Law³ recommended by the General Assembly in its resolution 31/98 of 15 December 1976, significantly contributes to the establishment of a unified legal framework for the fair and efficient settlement of disputes arising in international commercial relations,

1. *Requests* the Secretary-General to transmit the text of the Model Law on International Commercial Arbitration of the United Nations Commission on International Trade Law, together with the *travaux préparatoires* from the eighteenth session of the Commission, to Governments and to arbitral institutions and other interested bodies, such as chambers of commerce;

2. *Recommends* that all States give due consideration to the Model Law on International Commercial Arbitration, in view of the desirability of uniformity of the law of arbitral procedures and the specific needs of international commercial arbitration practice.

> 112th plenary meeting 11 December 1985

¹Official Records of the General Assembly, Fortieth Session, Supplement No. 17 (A/40/17), annex I.

²United Nations, Treaty Series, vol. 330, No. 4739, p. 38.

³United Nations publication, Sales No. E.77.V.6.

[on the report of the Sixth Committee (A/61/453)]

61/33. Revised articles of the Model Law on International Commercial Arbitration of the United Nations Commission on International Trade Law, and the recommendation regarding the interpretation of article II, paragraph 2, and article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done at New York, 10 June 1958

The General Assembly,

Recognizing the value of arbitration as a method of settling disputes arising in the context of international commercial relations,

Recalling its resolution 40/72 of 11 December 1985 regarding the Model Law on International Commercial Arbitration,¹

Recognizing the need for provisions in the Model Law to conform to current practices in international trade and modern means of contracting with regard to the form of the arbitration agreement and the granting of interim measures,

Believing that revised articles of the Model Law on the form of the arbitration agreement and interim measures reflecting those current practices will significantly enhance the operation of the Model Law,

Noting that the preparation of the revised articles of the Model Law on the form of the arbitration agreement and interim measures was the subject of due deliberation and extensive consultations with Governments and interested circles and would contribute significantly to the establishment of a harmonized legal framework for a fair and efficient settlement of international commercial disputes,

Believing that, in connection with the modernization of articles of the Model Law, the promotion of a uniform interpretation and application of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done at New York, 10 June 1958,² is particularly timely,

1. *Expresses its appreciation* to the United Nations Commission on International Trade Law for formulating and adopting the revised articles of its Model Law on International Commercial Arbitration on the form of the arbitration agreement and interim measures, the text of which is contained in annex I to the report of the United Nations Commission on International Trade Law on the work of its thirtyninth session,³ and recommends that all States give favourable consideration to the enactment of the revised articles of the Model Law, or the revised Model Law on International Commercial Arbitration of the United Nations Commission on

¹Official Records of the General Assembly, Fortieth Session, Supplement No. 17 (A/40/17), annex I.

²United Nations, Treaty Series, vol. 330, No. 4739.

³Official Records of the General Assembly, Sixty-first Session, Supplement No. 17 (A/61/17).

International Trade Law, when they enact or revise their laws, in view of the desirability of uniformity of the law of arbitral procedures and the specific needs of international commercial arbitration practice;

2. Also expresses its appreciation to the United Nations Commission on International Trade Law for formulating and adopting the recommendation regarding the interpretation of article II, paragraph 2, and article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done at New York, 10 June 1958,² the text of which is contained in annex II to the report of the United Nations Commission on International Trade Law on the work of its thirty-ninth session;³

3. *Requests* the Secretary-General to make all efforts to ensure that the revised articles of the Model Law and the recommendation become generally known and available.

64th plenary meeting 4 December 2006

Part One

UNCITRAL Model Law on International Commercial Arbitration

(United Nations documents A/40/17, annex I and A/61/17, annex I)

(As adopted by the United Nations Commission on International Trade Law on 21 June 1985, and as amended by the United Nations Commission on International Trade Law on 7 July 2006)

CHAPTER I. GENERAL PROVISIONS

Article 1. Scope of application¹

(1) This Law applies to international commercial² arbitration, subject to any agreement in force between this State and any other State or States.

(2) The provisions of this Law, except articles 8, 9, 17 H, 17 I, 17 J, 35 and 36, apply only if the place of arbitration is in the territory of this State.

(Article 1(2) has been amended by the Commission at its thirty-ninth session, in 2006)

(3) An arbitration is international if:

(*a*) the parties to an arbitration agreement have, at the time of the conclusion of that agreement, their places of business in different States; or

 $^{{}^{1}\!}Article$ headings are for reference purposes only and are not to be used for purposes of interpretation.

²The term "commercial" should be given a wide interpretation so as to cover matters arising from all relationships of a commercial nature, whether contractual or not. Relationships of a commercial nature include, but are not limited to, the following transactions: any trade transaction for the supply or exchange of goods or services; distribution agreement; commercial representation or agency; factoring; leasing; construction of works; consulting; engineering; licensing; investment; financing; banking; insurance; exploitation agreement or concession; joint venture and other forms of industrial or business cooperation; carriage of goods or passengers by air, sea, rail or road.

(b) one of the following places is situated outside the State in which the parties have their places of business:

- (i) the place of arbitration if determined in, or pursuant to, the arbitration agreement;
- (ii) any place where a substantial part of the obligations of the commercial relationship is to be performed or the place with which the subject-matter of the dispute is most closely connected; or

(c) the parties have expressly agreed that the subject matter of the arbitration agreement relates to more than one country.

(4) For the purposes of paragraph (3) of this article:

(*a*) if a party has more than one place of business, the place of business is that which has the closest relationship to the arbitration agreement;

(b) if a party does not have a place of business, reference is to be made to his habitual residence.

(5) This Law shall not affect any other law of this State by virtue of which certain disputes may not be submitted to arbitration or may be submitted to arbitration only according to provisions other than those of this Law.

Article 2. Definitions and rules of interpretation

For the purposes of this Law:

(a) "arbitration" means any arbitration whether or not administered by a permanent arbitral institution;

(b) "arbitral tribunal" means a sole arbitrator or a panel of arbitrators;

(c) "court" means a body or organ of the judicial system of a State;

(d) where a provision of this Law, except article 28, leaves the parties free to determine a certain issue, such freedom includes the right of the parties to authorize a third party, including an institution, to make that determination;

(e) where a provision of this Law refers to the fact that the parties have agreed or that they may agree or in any other way refers to an agreement of the parties, such agreement includes any arbitration rules referred to in that agreement;

(f) where a provision of this Law, other than in articles 25(a) and 32(2) (a), refers to a claim, it also applies to a counter-claim, and where it refers to a defence, it also applies to a defence to such counter-claim.

Article 2 A. International origin and general principles (As adopted by the Commission at its thirty-ninth session, in 2006)

(1) In the interpretation of this Law, regard is to be had to its international origin and to the need to promote uniformity in its application and the observance of good faith.

(2) Questions concerning matters governed by this Law which are not expressly settled in it are to be settled in conformity with the general principles on which this Law is based.

Article 3. Receipt of written communications

(1) Unless otherwise agreed by the parties:

(a) any written communication is deemed to have been received if it is delivered to the addressee personally or if it is delivered at his place of business, habitual residence or mailing address; if none of these can be found after making a reasonable inquiry, a written communication is deemed to have been received if it is sent to the addressee's last-known place of business, habitual residence or mailing address by registered letter or any other means which provides a record of the attempt to deliver it;

(b) the communication is deemed to have been received on the day it is so delivered.

(2) The provisions of this article do not apply to communications in court proceedings.

Article 4. Waiver of right to object

A party who knows that any provision of this Law from which the parties may derogate or any requirement under the arbitration agreement has not been complied with and yet proceeds with the arbitration without stating his objection to such non-compliance without undue delay or, if a time-limit is provided therefor, within such period of time, shall be deemed to have waived his right to object.

Article 5. Extent of court intervention

In matters governed by this Law, no court shall intervene except where so provided in this Law.

Article 6. Court or other authority for certain functions of arbitration assistance and supervision

The functions referred to in articles 11(3), 11(4), 13(3), 14, 16(3) and 34(2) shall be performed by ... [Each State enacting this model law specifies the court, courts or, where referred to therein, other authority competent to perform these functions.]

CHAPTER II. ARBITRATION AGREEMENT

Option I

Article 7. Definition and form of arbitration agreement (As adopted by the Commission at its thirty-ninth session, in 2006)

(1) "Arbitration agreement" is an agreement by the parties to submit to arbitration all or certain disputes which have arisen or which may arise between them in respect of a defined legal relationship, whether contractual or not. An arbitration agreement may be in the form of an arbitration clause in a contract or in the form of a separate agreement.

(2) The arbitration agreement shall be in writing.

(3) An arbitration agreement is in writing if its content is recorded in any form, whether or not the arbitration agreement or contract has been concluded orally, by conduct, or by other means.

(4) The requirement that an arbitration agreement be in writing is met by an electronic communication if the information contained therein is accessible so as to be useable for subsequent reference; "electronic communication" means any communication that the parties make by means of data messages; "data message" means information generated, sent, received or stored by electronic, magnetic, optical or similar means, including, but not limited to, electronic data interchange (EDI), electronic mail, telegram, telex or telecopy.

(5) Furthermore, an arbitration agreement is in writing if it is contained in an exchange of statements of claim and defence in which the existence of an agreement is alleged by one party and not denied by the other.

(6) The reference in a contract to any document containing an arbitration clause constitutes an arbitration agreement in writing, provided that the reference is such as to make that clause part of the contract.

Option II

Article 7. Definition of arbitration agreement

(As adopted by the Commission at its thirty-ninth session, in 2006)

"Arbitration agreement" is an agreement by the parties to submit to arbitration all or certain disputes which have arisen or which may arise between them in respect of a defined legal relationship, whether contractual or not.

Article 8. Arbitration agreement and substantive claim before court

(1) A court before which an action is brought in a matter which is the subject of an arbitration agreement shall, if a party so requests not later than when submitting his first statement on the substance of the dispute, refer the parties to arbitration unless it finds that the agreement is null and void, inoperative or incapable of being performed.

(2) Where an action referred to in paragraph (1) of this article has been brought, arbitral proceedings may nevertheless be commenced or continued, and an award may be made, while the issue is pending before the court.

Article 9. Arbitration agreement and interim measures by court

It is not incompatible with an arbitration agreement for a party to request, before or during arbitral proceedings, from a court an interim measure of protection and for a court to grant such measure.

CHAPTER III. COMPOSITION OF ARBITRAL TRIBUNAL

Article 10. Number of arbitrators

(1) The parties are free to determine the number of arbitrators.

(2) Failing such determination, the number of arbitrators shall be three.

Article 11. Appointment of arbitrators

(1) No person shall be precluded by reason of his nationality from acting as an arbitrator, unless otherwise agreed by the parties.

(2) The parties are free to agree on a procedure of appointing the arbitrator or arbitrators, subject to the provisions of paragraphs (4) and (5) of this article.

(3) Failing such agreement,

(a) in an arbitration with three arbitrators, each party shall appoint one arbitrator, and the two arbitrators thus appointed shall appoint the third arbitrator; if a party fails to appoint the arbitrator within thirty days of receipt of a request to do so from the other party, or if the two arbitrators fail to agree on the third arbitrator within thirty days of their appointment, the appointment shall be made, upon request of a party, by the court or other authority specified in article 6;

(b) in an arbitration with a sole arbitrator, if the parties are unable to agree on the arbitrator, he shall be appointed, upon request of a party, by the court or other authority specified in article 6.

(4) Where, under an appointment procedure agreed upon by the parties,

(a) a party fails to act as required under such procedure, or

(b) the parties, or two arbitrators, are unable to reach an agreement expected of them under such procedure, or

(c) a third party, including an institution, fails to perform any function entrusted to it under such procedure,

any party may request the court or other authority specified in article 6 to take the necessary measure, unless the agreement on the appointment procedure provides other means for securing the appointment.

(5) A decision on a matter entrusted by paragraph (3) or (4) of this article to the court or other authority specified in article 6 shall be subject to no

appeal. The court or other authority, in appointing an arbitrator, shall have due regard to any qualifications required of the arbitrator by the agreement of the parties and to such considerations as are likely to secure the appointment of an independent and impartial arbitrator and, in the case of a sole or third arbitrator, shall take into account as well the advisability of appointing an arbitrator of a nationality other than those of the parties.

Article 12. Grounds for challenge

(1) When a person is approached in connection with his possible appointment as an arbitrator, he shall disclose any circumstances likely to give rise to justifiable doubts as to his impartiality or independence. An arbitrator, from the time of his appointment and throughout the arbitral proceedings, shall without delay disclose any such circumstances to the parties unless they have already been informed of them by him.

(2) An arbitrator may be challenged only if circumstances exist that give rise to justifiable doubts as to his impartiality or independence, or if he does not possess qualifications agreed to by the parties. A party may challenge an arbitrator appointed by him, or in whose appointment he has participated, only for reasons of which he becomes aware after the appointment has been made.

Article 13. Challenge procedure

(1) The parties are free to agree on a procedure for challenging an arbitrator, subject to the provisions of paragraph (3) of this article.

(2) Failing such agreement, a party who intends to challenge an arbitrator shall, within fifteen days after becoming aware of the constitution of the arbitral tribunal or after becoming aware of any circumstance referred to in article 12(2), send a written statement of the reasons for the challenge to the arbitral tribunal. Unless the challenged arbitrator withdraws from his office or the other party agrees to the challenge, the arbitral tribunal shall decide on the challenge.

(3) If a challenge under any procedure agreed upon by the parties or under the procedure of paragraph (2) of this article is not successful, the challenging party may request, within thirty days after having received notice of the decision rejecting the challenge, the court or other authority specified in article 6 to decide on the challenge, which decision shall be subject to no appeal; while such a request is pending, the arbitral tribunal, including the challenged arbitrator, may continue the arbitral proceedings and make an award.

Article 14. Failure or impossibility to act

(1) If an arbitrator becomes *de jure* or *de facto* unable to perform his functions or for other reasons fails to act without undue delay, his mandate terminates if he withdraws from his office or if the parties agree on the termination. Otherwise, if a controversy remains concerning any of these grounds, any party may request the court or other authority specified in article 6 to decide on the termination of the mandate, which decision shall be subject to no appeal.

(2) If, under this article or article 13(2), an arbitrator withdraws from his office or a party agrees to the termination of the mandate of an arbitrator, this does not imply acceptance of the validity of any ground referred to in this article or article 12(2).

Article 15. Appointment of substitute arbitrator

Where the mandate of an arbitrator terminates under article 13 or 14 or because of his withdrawal from office for any other reason or because of the revocation of his mandate by agreement of the parties or in any other case of termination of his mandate, a substitute arbitrator shall be appointed according to the rules that were applicable to the appointment of the arbitrator being replaced.

CHAPTER IV. JURISDICTION OF ARBITRAL TRIBUNAL

Article 16. Competence of arbitral tribunal to rule on its jurisdiction

(1) The arbitral tribunal may rule on its own jurisdiction, including any objections with respect to the existence or validity of the arbitration agreement. For that purpose, an arbitration clause which forms part of a contract shall be treated as an agreement independent of the other terms of the contract. A decision by the arbitral tribunal that the contract is null and void shall not entail *ipso jure* the invalidity of the arbitration clause.

(2) A plea that the arbitral tribunal does not have jurisdiction shall be raised not later than the submission of the statement of defence. A party is not precluded from raising such a plea by the fact that he has appointed, or participated in the appointment of, an arbitrator. A plea that the arbitral tribunal is exceeding the scope of its authority shall be raised as soon as the matter alleged to be beyond the scope of its authority is raised during the arbitral proceedings. The arbitral tribunal may, in either case, admit a later plea if it considers the delay justified.

(3) The arbitral tribunal may rule on a plea referred to in paragraph (2) of this article either as a preliminary question or in an award on the merits. If the arbitral tribunal rules as a preliminary question that it has jurisdiction, any party may request, within thirty days after having received notice of that ruling, the court specified in article 6 to decide the matter, which decision shall be subject to no appeal; while such a request is pending, the arbitral tribunal may continue the arbitral proceedings and make an award.

CHAPTER IV A. INTERIM MEASURES AND PRELIMINARY ORDERS

(As adopted by the Commission at its thirty-ninth session, in 2006)

Section 1. Interim measures

Article 17. Power of arbitral tribunal to order interim measures

(1) Unless otherwise agreed by the parties, the arbitral tribunal may, at the request of a party, grant interim measures.

(2) An interim measure is any temporary measure, whether in the form of an award or in another form, by which, at any time prior to the issuance of the award by which the dispute is finally decided, the arbitral tribunal orders a party to:

(a) Maintain or restore the status quo pending determination of the dispute;

(b) Take action that would prevent, or refrain from taking action that is likely to cause, current or imminent harm or prejudice to the arbitral process itself;

(c) Provide a means of preserving assets out of which a subsequent award may be satisfied; or

(d) Preserve evidence that may be relevant and material to the resolution of the dispute.

Article 17 A. Conditions for granting interim measures

(1) The party requesting an interim measure under article 17(2)(a), (b) and (c) shall satisfy the arbitral tribunal that:

(a) Harm not adequately reparable by an award of damages is likely to result if the measure is not ordered, and such harm substantially outweighs the harm that is likely to result to the party against whom the measure is directed if the measure is granted; and

(b) There is a reasonable possibility that the requesting party will succeed on the merits of the claim. The determination on this possibility shall not affect the discretion of the arbitral tribunal in making any subsequent determination.

(2) With regard to a request for an interim measure under article 17(2)(d), the requirements in paragraphs (1)(a) and (b) of this article shall apply only to the extent the arbitral tribunal considers appropriate.

Section 2. Preliminary orders

Article 17 B. Applications for preliminary orders and conditions for granting preliminary orders

(1) Unless otherwise agreed by the parties, a party may, without notice to any other party, make a request for an interim measure together with an application for a preliminary order directing a party not to frustrate the purpose of the interim measure requested.

(2) The arbitral tribunal may grant a preliminary order provided it considers that prior disclosure of the request for the interim measure to the party against whom it is directed risks frustrating the purpose of the measure.

(3) The conditions defined under article 17A apply to any preliminary order, provided that the harm to be assessed under article 17A(1)(a), is the harm likely to result from the order being granted or not.

Article 17 C. Specific regime for preliminary orders

(1) Immediately after the arbitral tribunal has made a determination in respect of an application for a preliminary order, the arbitral tribunal shall give notice to all parties of the request for the interim measure, the application for

the preliminary order, the preliminary order, if any, and all other communications, including by indicating the content of any oral communication, between any party and the arbitral tribunal in relation thereto.

(2) At the same time, the arbitral tribunal shall give an opportunity to any party against whom a preliminary order is directed to present its case at the earliest practicable time.

(3) The arbitral tribunal shall decide promptly on any objection to the preliminary order.

(4) A preliminary order shall expire after twenty days from the date on which it was issued by the arbitral tribunal. However, the arbitral tribunal may issue an interim measure adopting or modifying the preliminary order, after the party against whom the preliminary order is directed has been given notice and an opportunity to present its case.

(5) A preliminary order shall be binding on the parties but shall not be subject to enforcement by a court. Such a preliminary order does not constitute an award.

Section 3. Provisions applicable to interim measures and preliminary orders

Article 17 D. Modification, suspension, termination

The arbitral tribunal may modify, suspend or terminate an interim measure or a preliminary order it has granted, upon application of any party or, in exceptional circumstances and upon prior notice to the parties, on the arbitral tribunal's own initiative.

Article 17 E. Provision of security

(1) The arbitral tribunal may require the party requesting an interim measure to provide appropriate security in connection with the measure.

(2) The arbitral tribunal shall require the party applying for a preliminary order to provide security in connection with the order unless the arbitral tribunal considers it inappropriate or unnecessary to do so.

Article 17 F. Disclosure

(1) The arbitral tribunal may require any party promptly to disclose any material change in the circumstances on the basis of which the measure was requested or granted.

(2) The party applying for a preliminary order shall disclose to the arbitral tribunal all circumstances that are likely to be relevant to the arbitral tribunal's determination whether to grant or maintain the order, and such obligation shall continue until the party against whom the order has been requested has had an opportunity to present its case. Thereafter, paragraph (1) of this article shall apply.

Article 17 G. Costs and damages

The party requesting an interim measure or applying for a preliminary order shall be liable for any costs and damages caused by the measure or the order to any party if the arbitral tribunal later determines that, in the circumstances, the measure or the order should not have been granted. The arbitral tribunal may award such costs and damages at any point during the proceedings.

Section 4. Recognition and enforcement of interim measures

Article 17 H. Recognition and enforcement

(1) An interim measure issued by an arbitral tribunal shall be recognized as binding and, unless otherwise provided by the arbitral tribunal, enforced upon application to the competent court, irrespective of the country in which it was issued, subject to the provisions of article 17 I.

(2) The party who is seeking or has obtained recognition or enforcement of an interim measure shall promptly inform the court of any termination, suspension or modification of that interim measure.

(3) The court of the State where recognition or enforcement is sought may, if it considers it proper, order the requesting party to provide appropriate security if the arbitral tribunal has not already made a determination with respect to security or where such a decision is necessary to protect the rights of third parties.

Article 17 I. Grounds for refusing recognition or enforcement³

(1) Recognition or enforcement of an interim measure may be refused only:

(a) At the request of the party against whom it is invoked if the court is satisfied that:

- (i) Such refusal is warranted on the grounds set forth in article 36(1)(a)(i), (ii), (iii) or (iv); or
- (ii) The arbitral tribunal's decision with respect to the provision of security in connection with the interim measure issued by the arbitral tribunal has not been complied with; or
- (iii) The interim measure has been terminated or suspended by the arbitral tribunal or, where so empowered, by the court of the State in which the arbitration takes place or under the law of which that interim measure was granted; or
- (b) If the court finds that:
 - (i) The interim measure is incompatible with the powers conferred upon the court unless the court decides to reformulate the interim measure to the extent necessary to adapt it to its own powers and procedures for the purposes of enforcing that interim measure and without modifying its substance; or
 - (ii) Any of the grounds set forth in article 36(1)(b)(i) or (ii), apply to the recognition and enforcement of the interim measure.

(2) Any determination made by the court on any ground in paragraph (1) of this article shall be effective only for the purposes of the application to recognize and enforce the interim measure. The court where recognition or enforcement is sought shall not, in making that determination, undertake a review of the substance of the interim measure.

Section 5. Court-ordered interim measures

Article 17 J. Court-ordered interim measures

A court shall have the same power of issuing an interim measure in relation to arbitration proceedings, irrespective of whether their place is in

³The conditions set forth in article 17 I are intended to limit the number of circumstances in which the court may refuse to enforce an interim measure. It would not be contrary to the level of harmonization sought to be achieved by these model provisions if a State were to adopt fewer circumstances in which enforcement may be refused.

the territory of this State, as it has in relation to proceedings in courts. The court shall exercise such power in accordance with its own procedures in consideration of the specific features of international arbitration.

CHAPTER V. CONDUCT OF ARBITRAL PROCEEDINGS

Article 18. Equal treatment of parties

The parties shall be treated with equality and each party shall be given a full opportunity of presenting his case.

Article 19. Determination of rules of procedure

(1) Subject to the provisions of this Law, the parties are free to agree on the procedure to be followed by the arbitral tribunal in conducting the proceedings.

(2) Failing such agreement, the arbitral tribunal may, subject to the provisions of this Law, conduct the arbitration in such manner as it considers appropriate. The power conferred upon the arbitral tribunal includes the power to determine the admissibility, relevance, materiality and weight of any evidence.

Article 20. Place of arbitration

(1) The parties are free to agree on the place of arbitration. Failing such agreement, the place of arbitration shall be determined by the arbitral tribunal having regard to the circumstances of the case, including the convenience of the parties.

(2) Notwithstanding the provisions of paragraph (1) of this article, the arbitral tribunal may, unless otherwise agreed by the parties, meet at any place it considers appropriate for consultation among its members, for hearing witnesses, experts or the parties, or for inspection of goods, other property or documents.

Article 21. Commencement of arbitral proceedings

Unless otherwise agreed by the parties, the arbitral proceedings in respect of a particular dispute commence on the date on which a request for that dispute to be referred to arbitration is received by the respondent.

Article 22. Language

(1) The parties are free to agree on the language or languages to be used in the arbitral proceedings. Failing such agreement, the arbitral tribunal shall determine the language or languages to be used in the proceedings. This agreement or determination, unless otherwise specified therein, shall apply to any written statement by a party, any hearing and any award, decision or other communication by the arbitral tribunal.

(2) The arbitral tribunal may order that any documentary evidence shall be accompanied by a translation into the language or languages agreed upon by the parties or determined by the arbitral tribunal.

Article 23. Statements of claim and defence

(1) Within the period of time agreed by the parties or determined by the arbitral tribunal, the claimant shall state the facts supporting his claim, the points at issue and the relief or remedy sought, and the respondent shall state his defence in respect of these particulars, unless the parties have otherwise agreed as to the required elements of such statements. The parties may submit with their statements all documents they consider to be relevant or may add a reference to the documents or other evidence they will submit.

(2) Unless otherwise agreed by the parties, either party may amend or supplement his claim or defence during the course of the arbitral proceedings, unless the arbitral tribunal considers it inappropriate to allow such amendment having regard to the delay in making it.

Article 24. Hearings and written proceedings

(1) Subject to any contrary agreement by the parties, the arbitral tribunal shall decide whether to hold oral hearings for the presentation of evidence or for oral argument, or whether the proceedings shall be conducted on the basis of documents and other materials. However, unless the parties have agreed that no hearings shall be held, the arbitral tribunal shall hold such hearings at an appropriate stage of the proceedings, if so requested by a party.

(2) The parties shall be given sufficient advance notice of any hearing and of any meeting of the arbitral tribunal for the purposes of inspection of goods, other property or documents.

(3) All statements, documents or other information supplied to the arbitral tribunal by one party shall be communicated to the other party. Also any expert report or evidentiary document on which the arbitral tribunal may rely in making its decision shall be communicated to the parties.

Article 25. Default of a party

Unless otherwise agreed by the parties, if, without showing sufficient cause,

(a) the claimant fails to communicate his statement of claim in accordance with article 23(1), the arbitral tribunal shall terminate the proceedings;

(b) the respondent fails to communicate his statement of defence in accordance with article 23(1), the arbitral tribunal shall continue the proceedings without treating such failure in itself as an admission of the claimant's allegations;

(c) any party fails to appear at a hearing or to produce documentary evidence, the arbitral tribunal may continue the proceedings and make the award on the evidence before it.

Article 26. Expert appointed by arbitral tribunal

(1) Unless otherwise agreed by the parties, the arbitral tribunal

(a) may appoint one or more experts to report to it on specific issues to be determined by the arbitral tribunal;

(b) may require a party to give the expert any relevant information or to produce, or to provide access to, any relevant documents, goods or other property for his inspection.

(2) Unless otherwise agreed by the parties, if a party so requests or if the arbitral tribunal considers it necessary, the expert shall, after delivery of his written or oral report, participate in a hearing where the parties have the opportunity to put questions to him and to present expert witnesses in order to testify on the points at issue.

Article 27. Court assistance in taking evidence

The arbitral tribunal or a party with the approval of the arbitral tribunal may request from a competent court of this State assistance in taking evidence. The court may execute the request within its competence and according to its rules on taking evidence.

CHAPTER VI. MAKING OF AWARD AND TERMINATION OF PROCEEDINGS

Article 28. Rules applicable to substance of dispute

(1) The arbitral tribunal shall decide the dispute in accordance with such rules of law as are chosen by the parties as applicable to the substance of the dispute. Any designation of the law or legal system of a given State shall be construed, unless otherwise expressed, as directly referring to the substantive law of that State and not to its conflict of laws rules.

(2) Failing any designation by the parties, the arbitral tribunal shall apply the law determined by the conflict of laws rules which it considers applicable.

(3) The arbitral tribunal shall decide *ex aequo et bono* or as *amiable compositeur* only if the parties have expressly authorized it to do so.

(4) In all cases, the arbitral tribunal shall decide in accordance with the terms of the contract and shall take into account the usages of the trade applicable to the transaction.

Article 29. Decision-making by panel of arbitrators

In arbitral proceedings with more than one arbitrator, any decision of the arbitral tribunal shall be made, unless otherwise agreed by the parties, by a majority of all its members. However, questions of procedure may be decided by a presiding arbitrator, if so authorized by the parties or all members of the arbitral tribunal.

Article 30. Settlement

(1) If, during arbitral proceedings, the parties settle the dispute, the arbitral tribunal shall terminate the proceedings and, if requested by the parties and not objected to by the arbitral tribunal, record the settlement in the form of an arbitral award on agreed terms.

(2) An award on agreed terms shall be made in accordance with the provisions of article 31 and shall state that it is an award. Such an award has the same status and effect as any other award on the merits of the case.

Article 31. Form and contents of award

(1) The award shall be made in writing and shall be signed by the arbitrator or arbitrators. In arbitral proceedings with more than one arbitrator, the signatures of the majority of all members of the arbitral tribunal shall suffice, provided that the reason for any omitted signature is stated.

(2) The award shall state the reasons upon which it is based, unless the parties have agreed that no reasons are to be given or the award is an award on agreed terms under article 30.

(3) The award shall state its date and the place of arbitration as determined in accordance with article 20(1). The award shall be deemed to have been made at that place.

(4) After the award is made, a copy signed by the arbitrators in accordance with paragraph (1) of this article shall be delivered to each party.

Article 32. Termination of proceedings

(1) The arbitral proceedings are terminated by the final award or by an order of the arbitral tribunal in accordance with paragraph (2) of this article.

(2) The arbitral tribunal shall issue an order for the termination of the arbitral proceedings when:

(a) the claimant withdraws his claim, unless the respondent objects thereto and the arbitral tribunal recognizes a legitimate interest on his part in obtaining a final settlement of the dispute;

(b) the parties agree on the termination of the proceedings;

(c) the arbitral tribunal finds that the continuation of the proceedings has for any other reason become unnecessary or impossible.

(3) The mandate of the arbitral tribunal terminates with the termination of the arbitral proceedings, subject to the provisions of articles 33 and 34(4).

Article 33. Correction and interpretation of award; additional award

(1) Within thirty days of receipt of the award, unless another period of time has been agreed upon by the parties:

(a) a party, with notice to the other party, may request the arbitral

tribunal to correct in the award any errors in computation, any clerical or typographical errors or any errors of similar nature;

(b) if so agreed by the parties, a party, with notice to the other party, may request the arbitral tribunal to give an interpretation of a specific point or part of the award.

If the arbitral tribunal considers the request to be justified, it shall make the correction or give the interpretation within thirty days of receipt of the request. The interpretation shall form part of the award.

(2) The arbitral tribunal may correct any error of the type referred to in paragraph (1)(a) of this article on its own initiative within thirty days of the date of the award.

(3) Unless otherwise agreed by the parties, a party, with notice to the other party, may request, within thirty days of receipt of the award, the arbitral tribunal to make an additional award as to claims presented in the arbitral proceedings but omitted from the award. If the arbitral tribunal considers the request to be justified, it shall make the additional award within sixty days.

(4) The arbitral tribunal may extend, if necessary, the period of time within which it shall make a correction, interpretation or an additional award under paragraph (1) or (3) of this article.

(5) The provisions of article 31 shall apply to a correction or interpretation of the award or to an additional award.

CHAPTER VII. RECOURSE AGAINST AWARD

Article 34. Application for setting aside as exclusive recourse against arbitral award

(1) Recourse to a court against an arbitral award may be made only by an application for setting aside in accordance with paragraphs (2) and (3) of this article.

(2) An arbitral award may be set aside by the court specified in article 6 only if:

- (a) the party making the application furnishes proof that:
 - (i) a party to the arbitration agreement referred to in article 7 was under some incapacity; or the said agreement is not

valid under the law to which the parties have subjected it or, failing any indication thereon, under the law of this State; or

- (ii) the party making the application was not given proper notice of the appointment of an arbitrator or of the arbitral proceedings or was otherwise unable to present his case; or
- (iii) the award deals with a dispute not contemplated by or not falling within the terms of the submission to arbitration, or contains decisions on matters beyond the scope of the submission to arbitration, provided that, if the decisions on matters submitted to arbitration can be separated from those not so submitted, only that part of the award which contains decisions on matters not submitted to arbitration may be set aside; or
- (iv) the composition of the arbitral tribunal or the arbitral procedure was not in accordance with the agreement of the parties, unless such agreement was in conflict with a provision of this Law from which the parties cannot derogate, or, failing such agreement, was not in accordance with this Law; or
- (*b*) the court finds that:
 - (i) the subject-matter of the dispute is not capable of settlement by arbitration under the law of this State; or
 - (ii) the award is in conflict with the public policy of this State.

(3) An application for setting aside may not be made after three months have elapsed from the date on which the party making that application had received the award or, if a request had been made under article 33, from the date on which that request had been disposed of by the arbitral tribunal.

(4) The court, when asked to set aside an award, may, where appropriate and so requested by a party, suspend the setting aside proceedings for a period of time determined by it in order to give the arbitral tribunal an opportunity to resume the arbitral proceedings or to take such other action as in the arbitral tribunal's opinion will eliminate the grounds for setting aside.

CHAPTER VIII. RECOGNITION AND ENFORCEMENT OF AWARDS

Article 35. Recognition and enforcement

(1) An arbitral award, irrespective of the country in which it was made, shall be recognized as binding and, upon application in writing to the

competent court, shall be enforced subject to the provisions of this article and of article 36.

(2) The party relying on an award or applying for its enforcement shall supply the original award or a copy thereof. If the award is not made in an official language of this State, the court may request the party to supply a translation thereof into such language.⁴

(Article 35(2) has been amended by the Commission at its thirty-ninth session, in 2006)

Article 36. Grounds for refusing recognition or enforcement

(1) Recognition or enforcement of an arbitral award, irrespective of the country in which it was made, may be refused only:

(*a*) at the request of the party against whom it is invoked, if that party furnishes to the competent court where recognition or enforcement is sought proof that:

- a party to the arbitration agreement referred to in article 7 was under some incapacity; or the said agreement is not valid under the law to which the parties have subjected it or, failing any indication thereon, under the law of the country where the award was made; or
- (ii) the party against whom the award is invoked was not given proper notice of the appointment of an arbitrator or of the arbitral proceedings or was otherwise unable to present his case; or
- (iii) the award deals with a dispute not contemplated by or not falling within the terms of the submission to arbitration, or it contains decisions on matters beyond the scope of the submission to arbitration, provided that, if the decisions on matters submitted to arbitration can be separated from those not so submitted, that part of the award which contains decisions on matters submitted to arbitration may be recognized and enforced; or
- (iv) the composition of the arbitral tribunal or the arbitral procedure was not in accordance with the agreement of the parties or, failing such agreement, was not in accordance with the law of the country where the arbitration took place; or

⁴The conditions set forth in this paragraph are intended to set maximum standards. It would, thus, not be contrary to the harmonization to be achieved by the model law if a State retained even less onerous conditions.

- (v) the award has not yet become binding on the parties or has been set aside or suspended by a court of the country in which, or under the law of which, that award was made; or
- (b) if the court finds that:
 - (i) the subject-matter of the dispute is not capable of settlement by arbitration under the law of this State; or
 - (ii) the recognition or enforcement of the award would be contrary to the public policy of this State.

(2) If an application for setting aside or suspension of an award has been made to a court referred to in paragraph (1)(a)(v) of this article, the court where recognition or enforcement is sought may, if it considers it proper, adjourn its decision and may also, on the application of the party claiming recognition or enforcement of the award, order the other party to provide appropriate security.

Part Two

Explanatory Note by the UNCITRAL secretariat on the 1985 Model Law on International Commercial Arbitration as amended in 2006¹

1. The UNCITRAL Model Law on International Commercial Arbitration ("the Model Law") was adopted by the United Nations Commission on International Trade Law (UNCITRAL) on 21 June 1985, at the end of the eighteenth session of the Commission. The General Assembly, in its resolution 40/72 of 11 December 1985, recommended "that all States give due consideration to the Model Law on International Commercial Arbitration, in view of the desirability of uniformity of the law of arbitral procedures and the specific needs of international commercial arbitration practice". The Model Law was amended by UNCITRAL on 7 July 2006, at the thirty-ninth session of the Commission (see below, paragraphs 4, 19, 20, 27, 29 and 53). The General Assembly, in its resolution 61/33 of 4 December 2006, recommended "that all States give favourable consideration to the enactment of the revised articles of the UNCITRAL Model Law on International Commercial Arbitration, or the revised UNCITRAL Model Law on International Commercial Arbitration, when they enact or revise their laws (...)".

2. The Model Law constitutes a sound basis for the desired harmonization and improvement of national laws. It covers all stages of the arbitral process from the arbitration agreement to the recognition and enforcement of the arbitral award and reflects a worldwide consensus on the principles and important issues of international arbitration practice. It is acceptable to States of all regions and the different legal or economic systems of the world. Since its adoption by UNCITRAL, the Model Law has come to represent the accepted international legislative standard for a modern arbitration law and a significant number of jurisdictions have enacted arbitration legislation based on the Model Law.

3. The form of a model law was chosen as the vehicle for harmonization and modernization in view of the flexibility it gives to States in preparing new arbitration laws. Notwithstanding that flexibility, and in order to increase the likelihood of achieving a satisfactory degree of harmonization, States are encouraged to make

¹This note was prepared by the secretariat of the United Nations Commission on International Trade Law (UNCITRAL) for informational purposes only; it is not an official commentary on the Model Law. A commentary prepared by the Secretariat on an early draft of the Model Law appears in document A/CN.9/264 (reproduced in UNCITRAL Yearbook, vol. XVI — 1985, United Nations publication, Sales No. E.87.V.4).

as few changes as possible when incorporating the Model Law into their legal systems. Efforts to minimize variation from the text adopted by UNCITRAL are also expected to increase the visibility of harmonization, thus enhancing the confidence of foreign parties, as the primary users of international arbitration, in the reliability of arbitration law in the enacting State.

4. The revision of the Model Law adopted in 2006 includes article 2 A, which is designed to facilitate interpretation by reference to internationally accepted principles and is aimed at promoting a uniform understanding of the Model Law. Other substantive amendments to the Model Law relate to the form of the arbitration agreement and to interim measures. The original 1985 version of the provision on the form of the arbitration agreement (article 7) was modelled on the language used in article II (2) of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards (New York, 1958) ("the New York Convention"). The revision of article 7 is intended to address evolving practice in international trade and technological developments. The extensive revision of article 17 on interim measures was considered necessary in light of the fact that such measures are increasingly relied upon in the practice of international commercial arbitration. The revision also includes an enforcement regime for such measures in recognition of the fact that the effectiveness of arbitration frequently depends upon the possibility of enforcing interim measures. The new provisions are contained in a new chapter of the Model Law on interim measures and preliminary orders (chapter IV A).

A. Background to the Model Law

5. The Model Law was developed to address considerable disparities in national laws on arbitration. The need for improvement and harmonization was based on findings that national laws were often particularly inappropriate for international cases.

1. Inadequacy of domestic laws

6. Recurrent inadequacies to be found in outdated national laws include provisions that equate the arbitral process with court litigation and fragmentary provisions that fail to address all relevant substantive law issues. Even most of those laws that appear to be up-to-date and comprehensive were drafted with domestic arbitration primarily, if not exclusively, in mind. While this approach is understandable in view of the fact that even today the bulk of cases governed by arbitration law would be of a purely domestic nature, the unfortunate consequence is that traditional local concepts are imposed on international cases and the needs of modern practice are often not met.

7. The expectations of the parties as expressed in a chosen set of arbitration rules or a "one-off" arbitration agreement may be frustrated, especially by mandatory provisions of applicable law. Unexpected and undesired restrictions found in national laws may prevent the parties, for example, from submitting future disputes to arbitration, from selecting the arbitrator freely, or from having the arbitral proceedings conducted according to agreed rules of procedure and with no more court involvement than appropriate. Frustration may also ensue from non-mandatory provisions that may impose undesired requirements on unwary parties who may not think about the need to provide otherwise when drafting the arbitration agreement. Even the absence of any legislative provision may cause difficulties simply by leaving unanswered some of the many procedural issues relevant in arbitration and not always settled in the arbitration agreement. The Model Law is intended to reduce the risk of such possible frustration, difficulties or surprise.

2. Disparity between national laws

8. Problems stemming from inadequate arbitration laws or from the absence of specific legislation governing arbitration are aggravated by the fact that national laws differ widely. Such differences are a frequent source of concern in international arbitration, where at least one of the parties is, and often both parties are, confronted with foreign and unfamiliar provisions and procedures. Obtaining a full and precise account of the law applicable to the arbitration is, in such circumstances often expensive, impractical or impossible.

9. Uncertainty about the local law with the inherent risk of frustration may adversely affect the functioning of the arbitral process and also impact on the selection of the place of arbitration. Due to such uncertainty, a party may hesitate or refuse to agree to a place, which for practical reasons would otherwise be appropriate. The range of places of arbitration acceptable to parties is thus widened and the smooth functioning of the arbitral proceedings is enhanced where States adopt the Model Law, which is easily recognizable, meets the specific needs of international commercial arbitration and provides an international standard based on solutions acceptable to parties from different legal systems.

B. Salient features of the Model Law

1. Special procedural regime for international commercial arbitration

10. The principles and solutions adopted in the Model Law aim at reducing or eliminating the above-mentioned concerns and difficulties. As a response to the inadequacies and disparities of national laws, the Model Law presents a special legal regime tailored to international commercial arbitration, without affecting any relevant treaty in force in the State adopting the Model Law. While the Model Law was designed with international commercial arbitration in mind, it offers a set of basic rules that are not, in and of themselves, unsuitable to any other type of arbitration. States may thus consider extending their enactment of the Model Law to cover also domestic disputes, as a number of enacting States already have.

(a) Substantive and territorial scope of application

11. Article 1 defines the scope of application of the Model Law by reference to the notion of "international commercial arbitration". The Model Law defines an arbitration as international if "the parties to an arbitration agreement have, at the time of the conclusion of that agreement, their places of business in different States" (article 1 (3)). The vast majority of situations commonly regarded as international will meet this criterion. In addition, article 1 (3) broadens the notion of internationality so that the Model Law also covers cases where the place of arbitration, the place of contract performance, or the place of the subject-matter of the dispute is situated outside the State where the parties have their place of business, or cases where the parties have expressly agreed that the subject-matter of the arbitration agreement relates to more than one country. Article 1 thus recognizes extensively the freedom of the parties to submit a dispute to the legal regime established pursuant to the Model Law.

12. In respect of the term "commercial", the Model Law provides no strict definition. The footnote to article 1 (1) calls for "a wide interpretation" and offers an illustrative and open-ended list of relationships that might be described as commercial in nature, "whether contractual or not". The purpose of the footnote is to circumvent any technical difficulty that may arise, for example, in determining which transactions should be governed by a specific body of "commercial law" that may exist in some legal systems.

13. Another aspect of applicability is the territorial scope of application. The principle embodied in article 1 (2) is that the Model Law as enacted in a given State applies only if the place of arbitration is in the territory of that State. However, article 1 (2) also contains important exceptions to that principle, to the effect that certain articles apply, irrespective of whether the place of arbitration is in the enacting State or elsewhere (or, as the case may be, even before the place of arbitration is determined). These articles are the following: articles 8 (1) and 9, which deal with the recognition of arbitration agreements, including their compatibility with interim measures ordered by a court, article 17 J on court-ordered interim measures, articles 17 H and 17 I on the recognition and enforcement of interim measures ordered by an arbitral tribunal, and articles 35 and 36 on the recognition and enforcement of arbitration awards.

14. The territorial criterion governing most of the provisions of the Model Law was adopted for the sake of certainty and in view of the following facts. In most legal systems, the place of arbitration is the exclusive criterion for determining the applicability of national law and, where the national law allows parties to choose the procedural law of a State other than that where the arbitration takes place, experience shows that parties rarely make use of that possibility. Incidentally, enactment of the Model Law reduces any need for the parties to choose a "foreign" law, since the Model Law grants the parties wide freedom in shaping the rules of the arbitral proceedings. In addition to designating the law governing the arbitral proceedure, the territorial criterion is of considerable practical importance in respect of articles 11, 13, 14, 16, 27 and 34, which entrust State courts at the place of

arbitration with functions of supervision and assistance to arbitration. It should be noted that the territorial criterion legally triggered by the parties' choice regarding the place of arbitration does not limit the arbitral tribunal's ability to meet at any place it considers appropriate for the conduct of the proceedings, as provided by article 20 (2).

(b) Delimitation of court assistance and supervision

15. Recent amendments to arbitration laws reveal a trend in favour of limiting and clearly defining court involvement in international commercial arbitration. This is justified in view of the fact that the parties to an arbitration agreement make a conscious decision to exclude court jurisdiction and prefer the finality and expediency of the arbitral process.

16. In this spirit, the Model Law envisages court involvement in the following instances. A first group comprises issues of appointment, challenge and termination of the mandate of an arbitrator (articles 11, 13 and 14), jurisdiction of the arbitral tribunal (article 16) and setting aside of the arbitral award (article 34). These instances are listed in article 6 as functions that should be entrusted, for the sake of centralization, specialization and efficiency, to a specially designated court or, with respect to articles 11, 13 and 14, possibly to another authority (for example, an arbitral institution or a chamber of commerce). A second group comprises issues of court assistance in taking evidence (article 27), recognition of the arbitration agreement, including its compatibility with court-ordered interim measures (articles 8 and 9), court-ordered interim measures (articles 17 J), and recognition and enforcement of interim measures (articles 35 and 36).

17. Beyond the instances in these two groups, "no court shall intervene, in matters governed by this Law". Article 5 thus guarantees that all instances of possible court intervention are found in the piece of legislation enacting the Model Law, except for matters not regulated by it (for example, consolidation of arbitral proceedings, contractual relationship between arbitrators and parties or arbitral institutions, or fixing of costs and fees, including deposits). Protecting the arbitral process from unpredictable or disruptive court interference is essential to parties who choose arbitration (in particular foreign parties).

2. Arbitration agreement

18. Chapter II of the Model Law deals with the arbitration agreement, including its recognition by courts.

(a) Definition and form of arbitration agreement

19. The original 1985 version of the provision on the definition and form of arbitration agreement (article 7) closely followed article II (2) of the New York

Convention, which requires that an arbitration agreement be in writing. If the parties have agreed to arbitrate, but they entered into the arbitration agreement in a manner that does not meet the form requirement, any party may have grounds to object to the jurisdiction of the arbitral tribunal. It was pointed out by practitioners that, in a number of situations, the drafting of a written document was impossible or impractical. In such cases, where the willingness of the parties to arbitrate was not in question, the validity of the arbitration agreement should be recognized. For that reason, article 7 was amended in 2006 to better conform to international contract practices. In amending article 7, the Commission adopted two options, which reflect two different approaches on the question of definition and form of arbitration agreement. The first approach follows the detailed structure of the original 1985 text. It confirms the validity and effect of a commitment by the parties to submit to arbitration an existing dispute ("compromis") or a future dispute ("clause compromissoire"). It follows the New York Convention in requiring the written form of the arbitration agreement but recognizes a record of the "contents" of the agreement "in any form" as equivalent to traditional "writing". The agreement to arbitrate may be entered into in any form (e.g. including orally) as long as the content of the agreement is recorded. This new rule is significant in that it no longer requires signatures of the parties or an exchange of messages between the parties. It modernizes the language referring to the use of electronic commerce by adopting wording inspired from the 1996 UNCITRAL Model Law on Electronic Commerce and the 2005 United Nations Convention on the Use of Electronic Communications in International Contracts. It covers the situation of "an exchange of statements of claim and defence in which the existence of an agreement is alleged by one party and not denied by another". It also states that "the reference in a contract to any document" (for example, general conditions) "containing an arbitration clause constitutes an arbitration agreement in writing provided that the reference is such as to make that clause part of the contract". It thus clarifies that applicable contract law remains available to determine the level of consent necessary for a party to become bound by an arbitration agreement allegedly made "by reference". The second approach defines the arbitration agreement in a manner that omits any form requirement. No preference was expressed by the Commission in favour of either option I or II, both of which are offered for enacting States to consider, depending on their particular needs, and by reference to the legal context in which the Model Law is enacted, including the general contract law of the enacting State. Both options are intended to preserve the enforceability of arbitration agreements under the New York Convention.

20. In that respect, the Commission also adopted, at its thirty-ninth session in 2006, a "Recommendation regarding the interpretation of article II, paragraph 2, and article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958" (A/61/17, Annex 2).² The General Assembly, in its resolution 61/33 of 4 December 2006 noted that "in connection with the modernization of articles of the Model Law, the promotion of a uniform interpretation and application of the Convention on the Recognition and

²Reproduced in Part Three hereafter.

Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958, is particularly timely". The Recommendation was drafted in recognition of the widening use of electronic commerce and enactments of domestic legislation as well as case law, which are more favourable than the New York Convention in respect of the form requirement governing arbitration agreements, arbitration proceedings, and the enforcement of arbitral awards. The Recommendation encourages States to apply article II (2) of the New York Convention "recognizing that the circumstances described therein are not exhaustive". In addition, the Recommendation encourages States to adopt the revised article 7 of the Model Law. Both options of the revised article 7 establish a more favourable regime for the recognition and enforcement of arbitral awards than that provided under the New York Convention. By virtue of the "more favourable law provision" contained in article VII (1) of the New York Convention, the Recommendation clarifies that "any interested party" should be allowed "to avail itself of rights it may have, under the law or treaties of the country where an arbitration agreement is sought to be relied upon, to seek recognition of the validity of such an arbitration agreement".

(b) Arbitration agreement and the courts

21. Articles 8 and 9 deal with two important aspects of the complex relationship between the arbitration agreement and the resort to courts. Modelled on article II (3) of the New York Convention, article 8 (1) of the Model Law places any court under an obligation to refer the parties to arbitration if the court is seized with a claim on the same subject-matter unless it finds that the arbitration agreement is null and void, inoperative or incapable of being performed. The referral is dependent on a request, which a party may make not later than when submitting its first statement on the substance of the dispute. This provision, where adopted by a State enacting the Model Law, is by its nature binding only on the courts of that State. However, since article 8 is not limited in scope to agreements providing for arbitration to take place in the enacting State, it promotes the universal recognition and effect of international commercial arbitration agreements.

22. Article 9 expresses the principle that any interim measures of protection that may be obtained from courts under their procedural law (for example, pre-award attachments) are compatible with an arbitration agreement. That provision is ultimately addressed to the courts of any State, insofar as it establishes the compatibility between interim measures possibly issued by any court and an arbitration agreement, irrespective of the place of arbitration. Wherever a request for interim measures may be made to a court, it may not be relied upon, under the Model Law, as a waiver or an objection against the existence or effect of the arbitration agreement.

3. Composition of arbitral tribunal

23. Chapter III contains a number of detailed provisions on appointment, challenge, termination of mandate and replacement of an arbitrator. The chapter illustrates the

general approach taken by the Model Law in eliminating difficulties that arise from inappropriate or fragmentary laws or rules. First, the approach recognizes the freedom of the parties to determine, by reference to an existing set of arbitration rules or by an ad hoc agreement, the procedure to be followed, subject to the fundamental requirements of fairness and justice. Secondly, where the parties have not exercised their freedom to lay down the rules of procedure or they have failed to cover a particular issue, the Model Law ensures, by providing a set of suppletive rules, that the arbitration may commence and proceed effectively until the dispute is resolved.

24. Where under any procedure, agreed upon by the parties or based upon the suppletive rules of the Model Law, difficulties arise in the process of appointment, challenge or termination of the mandate of an arbitrator, articles 11, 13 and 14 provide for assistance by courts or other competent authorities designated by the enacting State. In view of the urgency of matters relating to the composition of the arbitral tribunal or its ability to function, and in order to reduce the risk and effect of any dilatory tactics, short time-periods are set and decisions rendered by courts or other authorities on such matters are not appealable.

4. Jurisdiction of arbitral tribunal

(a) Competence to rule on own jurisdiction

25. Article 16 (1) adopts the two important (not yet generally recognized) principles of "*Kompetenz-Kompetenz*" and of separability or autonomy of the arbitration clause. "*Kompetenz-Kompetenz*" means that the arbitral tribunal may independently rule on the question of whether it has jurisdiction, including any objections with respect to the existence or validity of the arbitration agreement, without having to resort to a court. Separability means that an arbitration clause shall be treated as an agreement independent of the other terms of the contract. As a consequence, a decision by the arbitral tribunal that the contract is null and void shall not entail *ipso jure* the invalidity of the arbitration clause. Detailed provisions in paragraph (2) require that any objections relating to the arbitrators' jurisdiction be made at the earliest possible time.

26. The competence of the arbitral tribunal to rule on its own jurisdiction (i.e. on the foundation, content and extent of its mandate and power) is, of course, subject to court control. Where the arbitral tribunal rules as a preliminary question that it has jurisdiction, article 16 (3) allows for immediate court control in order to avoid waste of time and money. However, three procedural safeguards are added to reduce the risk and effect of dilatory tactics: short time-period for resort to court (30 days), court decision not appealable, and discretion of the arbitral tribunal to continue the proceedings and make an award while the matter is pending before the court. In those cases where the arbitral tribunal decides to combine its decision on jurisdiction with an award on the merits, judicial review on the question of jurisdiction is available in setting aside proceedings under article 34 or in enforcement proceedings under article 36.

(b) Power to order interim measures and preliminary orders

27. Chapter IV A on interim measures and preliminary orders was adopted by the Commission in 2006. It replaces article 17 of the original 1985 version of the Model Law. Section 1 provides a generic definition of interim measures and sets out the conditions for granting such measures. An important innovation of the revision lies in the establishment (in section 4) of a regime for the recognition and enforcement of interim measures, which was modelled, as appropriate, on the regime for the recognition and enforcement of arbitral awards under articles 35 and 36 of the Model Law.

Section 2 of chapter IV A deals with the application for, and conditions for 28. the granting of, preliminary orders. Preliminary orders provide a means for preserving the status quo until the arbitral tribunal issues an interim measure adopting or modifying the preliminary order. Article 17 B (1) provides that "a party may, without notice to any other party, make a request for an interim measure together with an application for a preliminary order directing a party not to frustrate the purpose of the interim measure requested". Article 17 B (2) permits an arbitral tribunal to grant a preliminary order if "it considers that prior disclosure of the request for the interim measure to the party against whom it is directed risks frustrating the purpose of the measure". Article 17 C contains carefully drafted safeguards for the party against whom the preliminary order is directed, such as prompt notification of the application for the preliminary order and of the preliminary order itself (if any), and an opportunity for that party to present its case "at the earliest practicable time". In any event, a preliminary order has a maximum duration of twenty days and, while binding on the parties, is not subject to court enforcement and does not constitute an award. The term "preliminary order" is used to emphasize its limited nature.

29. Section 3 sets out rules applicable to both preliminary orders and interim measures.

30. Section 5 includes article 17 J on interim measures ordered by courts in support of arbitration, and provides that "a court shall have the same power of issuing an interim measure in relation to arbitration proceedings irrespective of whether their place is in the territory of the enacting State, as it has in relation to proceedings in courts". That article has been added in 2006 to put it beyond any doubt that the existence of an arbitration agreement does not infringe on the powers of the competent court to issue interim measures and that the party to such an arbitration agreement is free to approach the court with a request to order interim measures.

5. Conduct of arbitral proceedings

31. Chapter V provides the legal framework for a fair and effective conduct of the arbitral proceedings. Article 18, which sets out fundamental requirements of procedural justice, and article 19 on the rights and powers to determine the rules of procedure, express principles that are central to the Model Law.

(a) Fundamental procedural rights of a party

32. Article 18 embodies the principles that the parties shall be treated with equality and given a full opportunity of presenting their case. A number of provisions illustrate those principles. For example, article 24 (1) provides that, unless the parties have agreed that no oral hearings be held for the presentation of evidence or for oral argument, the arbitral tribunal shall hold such hearings at an appropriate stage of the proceedings, if so requested by a party. It should be noted that article 24 (1) deals only with the general entitlement of a party to oral hearings (as an alternative to proceedings conducted on the basis of documents and other materials) and not with the procedural aspects, such as the length, number or timing of hearings.

33. Another illustration of those principles relates to evidence by an expert appointed by the arbitral tribunal. Article 26 (2) requires the expert, after delivering his or her written or oral report, to participate in a hearing where the parties may put questions to the expert and present expert witnesses to testify on the points at issue, if such a hearing is requested by a party or deemed necessary by the arbitral tribunal. As another provision aimed at ensuring fairness, objectivity and impartiality, article 24 (3) provides that all statements, documents and other information supplied to the arbitral tribunal by one party shall be communicated to the other party, and that any expert report or evidentiary document on which the arbitral tribunal may rely in making its decision shall be communicated to the parties. In order to enable the parties to be present at any hearing and at any meeting of the arbitral tribunal for inspection purposes, they shall be given sufficient notice in advance (article 24 (2)).

(b) Determination of rules of procedure

34. Article 19 guarantees the parties' freedom to agree on the procedure to be followed by the arbitral tribunal in conducting the proceedings, subject to a few mandatory provisions on procedure, and empowers the arbitral tribunal, failing agreement by the parties, to conduct the arbitration in such a manner as it considers appropriate. The power conferred upon the arbitral tribunal includes the power to determine the admissibility, relevance, materiality and weight of any evidence.

35. Autonomy of the parties in determining the rules of procedure is of special importance in international cases since it allows the parties to select or tailor the rules according to their specific wishes and needs, unimpeded by traditional and possibly conflicting domestic concepts, thus obviating the earlier mentioned risk of frustration or surprise (see above, paras. 7 and 9). The supplementary discretion of the arbitral tribunal is equally important in that it allows the tribunal to tailor the conduct of the proceedings to the specific features of the case without being hindered by any restraint that may stem from traditional local law, including any domestic rule on evidence. Moreover, it provides grounds for displaying initiative in solving any procedural question not regulated in the arbitration agreement or the Model Law.

36. In addition to the general provisions of article 19, other provisions in the Model Law recognize party autonomy and, failing agreement, empower the arbitral tribunal to decide on certain matters. Examples of particular practical importance in international cases are article 20 on the place of arbitration and article 22 on the language to be used in the proceedings.

(c) Default of a party

37. The arbitral proceedings may be continued in the absence of a party, provided that due notice has been given. This applies, in particular, to the failure of the respondent to communicate its statement of defence (article 25 (b)). The arbitral tribunal may also continue the proceedings where a party fails to appear at a hearing or to produce documentary evidence without showing sufficient cause for the failure (article 25 (c)). However, if the claimant fails to submit its statement of claim, the arbitral tribunal is obliged to terminate the proceedings (article 25 (a)).

38. Provisions that empower the arbitral tribunal to carry out its task even if one of the parties does not participate are of considerable practical importance. As experience shows, it is not uncommon for one of the parties to have little interest in cooperating or expediting matters. Such provisions therefore provide international commercial arbitration its necessary effectiveness, within the limits of fundamental requirements of procedural justice.

6. Making of award and termination of proceedings

(a) Rules applicable to substance of dispute

39. Article 28 deals with the determination of the rules of law governing the substance of the dispute. Under paragraph (1), the arbitral tribunal decides the dispute in accordance with the rules of law chosen by the parties. This provision is significant in two respects. It grants the parties the freedom to choose the applicable substantive law, which is important where the national law does not clearly or fully recognize that right. In addition, by referring to the choice of "rules of law" instead of "law", the Model Law broadens the range of options available to the parties as regards the designation of the law applicable to the substance of the dispute. For example, parties may agree on rules of law that have been elaborated by an international forum but have not yet been incorporated into any national legal system. Parties could also choose directly an instrument such as the United Nations Convention on Contracts for the International Sale of Goods as the body of substantive law governing the arbitration, without having to refer to the national law of any State party to that Convention. The power of the arbitral tribunal, on the other hand, follows more traditional lines. When the parties have not chosen the applicable law, the arbitral tribunal shall apply the law (i.e., the national law) determined by the conflict-of-laws rules that it considers applicable.

40. Article 28 (3) recognizes that the parties may authorize the arbitral tribunal to decide the dispute *ex aequo et bono* or as *amiables compositeur*. This type of arbitration (where the arbitral tribunal may decide the dispute on the basis of principles it believes to be just, without having to refer to any particular body of law) is currently not known or used in all legal systems. The Model Law does not intend to regulate this area. It simply calls the attention of the parties on the need to provide clarification in the arbitration agreement and specifically to empower the arbitrat tribunal. However, paragraph (4) makes it clear that in all cases where the dispute relates to a contract (including arbitration *ex aequo et bono*) the arbitrat tribunal must decide in accordance with the terms of the contract and shall take into account the usages of the trade applicable to the transaction.

(b) Making of award and other decisions

41. In its rules on the making of the award (articles 29-31), the Model Law focuses on the situation where the arbitral tribunal consists of more than one arbitrator. In such a situation, any award and other decision shall be made by a majority of the arbitrators, except on questions of procedure, which may be left to a presiding arbitrator. The majority principle applies also to the signing of the award, provided that the reason for any omitted signature is stated.

42. Article 31 (3) provides that the award shall state the place of arbitration and shall be deemed to have been made at that place. The effect of the deeming provision is to emphasize that the final making of the award constitutes a legal act, which in practice does not necessarily coincide with one factual event. For the same reason that the arbitral proceedings need not be carried out at the place designated as the legal "place of arbitration", the making of the award may be completed through deliberations held at various places, by telephone or correspondence. In addition, the award does not have to be signed by the arbitrators physically gathering at the same place.

43. The arbitral award must be in writing and state its date. It must also state the reasons on which it is based, unless the parties have agreed otherwise or the award is "on agreed terms" (i.e., an award that records the terms of an amicable settlement by the parties). It may be added that the Model Law neither requires nor prohibits "dissenting opinions".

7. Recourse against award

44. The disparity found in national laws as regards the types of recourse against an arbitral award available to the parties presents a major difficulty in harmonizing international arbitration legislation. Some outdated laws on arbitration, by establishing parallel regimes for recourse against arbitral awards or against court decisions, provide various types of recourse, various (and often long) time periods for exercising the recourse, and extensive lists of grounds on which recourse may be based. That situation (of considerable concern to those involved in international commercial arbitration) is greatly improved by the Model Law, which provides uniform grounds upon which (and clear time periods within which) recourse against an arbitral award may be made.

(a) Application for setting aside as exclusive recourse

45. The first measure of improvement is to allow only one type of recourse, to the exclusion of any other recourse regulated in any procedural law of the State in question. Article 34 (1) provides that the sole recourse against an arbitral award is by application for setting aside, which must be made within three months of receipt of the award (article 34 (3)). In regulating "recourse" (i.e., the means through which a party may actively "attack" the award), article 34 does not preclude a party from seeking court control by way of defence in enforcement proceedings (articles 35 and 36). Article 34 is limited to action before a court (i.e., an organ of the judicial system of a State). However, a party is not precluded from appealing to an arbitral tribunal of second instance if the parties have agreed on such a possibility (as is common in certain commodity trades).

(b) Grounds for setting aside

46. As a further measure of improvement, the Model Law lists exhaustively the grounds on which an award may be set aside. This list essentially mirrors that contained in article 36 (1), which is taken from article V of the New York Convention. The grounds provided in article 34 (2) are set out in two categories. Grounds which are to be proven by one party are as follows: lack of capacity of the parties to conclude an arbitration agreement; lack of a valid arbitration agreement; lack of notice of appointment of an arbitrator or of the arbitral proceedings or inability of a party to present its case; the award deals with matters not covered by the submission to arbitration; the composition of the arbitral tribunal or the conduct of arbitral proceedings are contrary to the effective agreement of the parties or, failing such agreement, to the Model Law. Grounds that a court may consider of its own initiative are as follows: non-arbitrability of the subject-matter of the dispute or violation of public policy (which is to be understood as serious departures from fundamental notions of procedural justice).

47. The approach under which the grounds for setting aside an award under the Model Law parallel the grounds for refusing recognition and enforcement of the award under article V of the New York Convention is reminiscent of the approach taken in the European Convention on International Commercial Arbitration (Geneva, 1961). Under article IX of the latter Convention, the decision of a foreign court to set aside an award for a reason other than the ones listed in article V of the New York Convention does not constitute a ground for refusing enforcement. The Model Law takes this philosophy one step further by directly limiting the reasons for setting aside.

48. Although the grounds for setting aside as set out in article 34 (2) are almost identical to those for refusing recognition or enforcement as set out in article 36 (1), a practical difference should be noted. An application for setting aside under article 34 (2) may only be made to a court in the State where the award was rendered whereas an application for enforcement might be made in a court in any State. For that reason, the grounds relating to public policy and non-arbitrability may vary in substance with the law applied by the court (in the State of setting aside or in the State of enforcement).

8. Recognition and enforcement of awards

49. The eighth and last chapter of the Model Law deals with the recognition and enforcement of awards. Its provisions reflect the significant policy decision that the same rules should apply to arbitral awards whether made in the country of enforcement or abroad, and that those rules should follow closely the New York Convention.

(a) Towards uniform treatment of all awards irrespective of country of origin

50. By treating awards rendered in international commercial arbitration in a uniform manner irrespective of where they were made, the Model Law distinguishes between "international" and "non-international" awards instead of relying on the traditional distinction between "foreign" and "domestic" awards. This new line is based on substantive grounds rather than territorial borders, which are inappropriate in view of the limited importance of the place of arbitration in international cases. The place of arbitration is often chosen for reasons of convenience of the parties and the dispute may have little or no connection with the State where the arbitration legally takes place. Consequently, the recognition and enforcement of "international" awards, whether "foreign" or "domestic", should be governed by the same provisions.

51. By modelling the recognition and enforcement rules on the relevant provisions of the New York Convention, the Model Law supplements, without conflicting with, the regime of recognition and enforcement created by that successful Convention.

(b) Procedural conditions of recognition and enforcement

52. Under article 35 (1) any arbitral award, irrespective of the country in which it was made, shall be recognized as binding and enforceable, subject to the provisions of article 35 (2) and of article 36 (the latter of which sets forth the grounds on which recognition or enforcement may be refused). Based on the above consideration of the limited importance of the place of arbitration in international cases and the desire of overcoming territorial restrictions, reciprocity is not included as a condition for recognition and enforcement.

53. The Model Law does not lay down procedural details of recognition and enforcement, which are left to national procedural laws and practices. The Model Law merely sets certain conditions for obtaining enforcement under article 35 (2). It was amended in 2006 to liberalize formal requirements and reflect the amendment made to article 7 on the form of the arbitration agreement. Presentation of a copy of the arbitration agreement is no longer required under article 35 (2).

(c) Grounds for refusing recognition or enforcement

54. Although the grounds on which recognition or enforcement may be refused under the Model Law are identical to those listed in article V of the New York Convention, the grounds listed in the Model Law are relevant not only to foreign awards but to all awards rendered in the sphere of application of the piece of legislation enacting the Model Law. Generally, it was deemed desirable to adopt, for the sake of harmony, the same approach and wording as this important Convention. However, the first ground on the list as contained in the New York Convention (which provides that recognition and enforcement may be refused if "the parties to the arbitration agreement were, under the law applicable to them, under some incapacity") was modified since it was viewed as containing an incomplete and potentially misleading conflict-of-laws rule.

Further information on the Model Law may be obtained from:

UNCITRAL secretariat Vienna International Centre P.O. Box 500 1400 Vienna Austria

Telephone: (+43-1) 26060-4060 Telefax: (+43-1) 26060-5813 Internet: uncitral.un.org Email: uncitral@un.org

Part Three

Recommendation regarding the interpretation of article II, paragraph 2, and article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958, adopted by the United Nations Commission on International Trade Law on 7 July 2006 at its thirty-ninth session

The United Nations Commission on International Trade Law,

Recalling General Assembly resolution 2205 (XXI) of 17 December 1966, which established the United Nations Commission on International Trade Law with the object of promoting the progressive harmonization and unification of the law of international trade by, inter alia, promoting ways and means of ensuring a uniform interpretation and application of international conventions and uniform laws in the field of the law of international trade,

Conscious of the fact that the different legal, social and economic systems of the world, together with different levels of development, are represented in the Commission,

Recalling successive resolutions of the General Assembly reaffirming the mandate of the Commission as the core legal body within the United Nations system in the field of international trade law to coordinate legal activities in this field,

Convinced that the wide adoption of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York on 10 June 1958,¹ has been a significant achievement in the promotion of the rule of law, particularly in the field of international trade,

Recalling that the Conference of Plenipotentiaries which prepared and opened the Convention for signature adopted a resolution, which states, inter alia, that the Conference "considers that greater uniformity of national laws on arbitration would further the effectiveness of arbitration in the settlement of private law disputes",

Bearing in mind differing interpretations of the form requirements under the Convention that result in part from differences of expression as between the five equally authentic texts of the Convention,

¹United Nations, Treaty Series, vol. 330, No. 4739.

Taking into account article VII, paragraph 1, of the Convention, a purpose of which is to enable the enforcement of foreign arbitral awards to the greatest extent, in particular by recognizing the right of any interested party to avail itself of law or treaties of the country where the award is sought to be relied upon, including where such law or treaties offer a regime more favourable than the Convention,

Considering the wide use of electronic commerce,

Taking into account international legal instruments, such as the 1985 UNCITRAL Model Law on International Commercial Arbitration,² as subsequently revised, particularly with respect to article 7,³ the UNCITRAL Model Law on Electronic Commerce,⁴ the UNCITRAL Model Law on Electronic Signatures⁵ and the United Nations Convention on the Use of Electronic Communications in International Contracts,⁶

Taking into account also enactments of domestic legislation, as well as case law, more favourable than the Convention in respect of form requirement governing arbitration agreements, arbitration proceedings and the enforcement of arbitral awards,

Considering that, in interpreting the Convention, regard is to be had to the need to promote recognition and enforcement of arbitral awards,

1. *Recommends* that article II, paragraph 2, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958, be applied recognizing that the circumstances described therein are not exhaustive;

2. *Recommends also* that article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958, should be applied to allow any interested party to avail itself of rights it may have, under the law or treaties of the country where an arbitration agreement is sought to be relied upon, to seek recognition of the validity of such an arbitration agreement.

²Official Records of the General Assembly, Fortieth Session, Supplement No. 17 (A/40/17), annex I, and United Nations publication, Sales No. E.95.V.18.

³Ibid., Sixty-first Session, Supplement No. 17 (A/61/17), annex I.

⁴Ibid., *Fifty-first Session, Supplement No. 17* (A/51/17), annex I, and United Nations publication, Sales No. E.99.V.4, which contains also an additional article 5 bis, adopted in 1998, and the accompanying Guide to Enactment.

⁵Ibid., *Fifty-sixth Session, Supplement No. 17* and corrigendum (A/56/17 and Corr.3), annex II, and United Nations publication, Sales No. E.02.V.8, which contains also the accompanying Guide to Enactment.

⁶General Assembly resolution 60/21, annex.



CONVENTION ON THE RECOGNITION AND ENFORCEMENT OF FOREIGN ARBITRAL AWARDS

//#n

Convention on the Recognition and Enforcement of Foreign Arbitral Awards (New York, 1958)





The United Nations Commission on International Trade Law (UNCITRAL) is a subsidiary body of the General Assembly. It plays an important role in improving the legal framework for international trade by preparing international legislative texts for use by States in modernizing the law of international trade and non-legislative texts for use by commercial parties in negotiating transactions. UNCITRAL legislative texts address international sale of goods; international commercial dispute resolution, including both arbitration and conciliation; electronic commerce; insolvency, including cross-border insolvency; international transport of goods; international payments; procurement and infrastructure development; and security interests. Non-legislative texts include rules for conduct of arbitration and conciliation proceedings; notes on organizing and conducting arbitral proceedings; and legal guides on industrial construction contracts and countertrade.

Further information may be obtained from:

UNCITRAL secretariat, Vienna International Centre, P.O. Box 500, 1400 Vienna, Austria

Telephone: (+43-1) 26060-4060 Internet: www.uncitral.org Telefax: (+43-1) 26060-5813 E-mail: uncitral@uncitral.org

Convention on the Recognition and Enforcement of Foreign Arbitral Awards (New York, 1958)



UNITED NATIONS New York, 2015

NOTE

Symbols of United Nations documents are composed of capital letters combined with figures. Mention of such a symbol indicates a reference to a United Nations document.

The publication reproduced here is a revised version in which part three of the original publication of 2009 has been removed.

Material in this publication may be freely quoted or reprinted, but acknowledgement is requested, together with a copy of the publication containing the quotation or reprint.

Contents

		F	Page
Intro	oductio	0 n	. 1
Part	one.	United Nations Conference on International Commercial Arbitration, New York, 20 May–10 June 1958	. 5
		pts from the Final Act of the United Nations Conference on ational Commercial Arbitration	. 5
		ention on the Recognition and Enforcement of Foreign Arbitral ls	. 8
Part	two.	paragraph 2, and article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral	
		Awards	
	Genera	al Assembly resolution 61/33 of 4 December 2006	15
	Recommendation regarding the interpretation of article II, paragraph 2, and article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958, adopted by the United Nations Commission on International Trade Law on 7 July 2006 at its thirty-ninth session		1.5

Introduction

Objectives

Recognizing the growing importance of international arbitration as a means of settling international commercial disputes, the Convention on the Recognition and Enforcement of Foreign Arbitral Awards (the Convention) seeks to provide common legislative standards for the recognition of arbitration agreements and court recognition and enforcement of foreign and nondomestic arbitral awards. The term "non-domestic" appears to embrace awards which, although made in the state of enforcement, are treated as "foreign" under its law because of some foreign element in the proceedings, e.g. another State's procedural laws are applied.

The Convention's principal aim is that foreign and non-domestic arbitral awards will not be discriminated against and it obliges Parties to ensure such awards are recognized and generally capable of enforcement in their jurisdiction in the same way as domestic awards. An ancillary aim of the Convention is to require courts of Parties to give full effect to arbitration agreements by requiring courts to deny the parties access to court in contravention of their agreement to refer the matter to an arbitral tribunal.

Key provisions

The Convention applies to awards made in any State other than the State in which recognition and enforcement is sought. It also applies to awards "not considered as domestic awards". When consenting to be bound by the Convention, a State may declare that it will apply the Convention (a) in respect to awards made only in the territory of another Party and (b) only to legal relationships that are considered "commercial" under its domestic law.

The Convention contains provisions on arbitration agreements. This aspect was covered in recognition of the fact that an award could be refused enforcement on the grounds that the agreement upon which it was based might not be recognized. Article II (1) provides that Parties shall recognize written arbitration agreements. In that respect, UNCITRAL adopted, at its thirty-ninth session in 2006, a Recommendation that seeks to provide guidance to Parties on the interpretation of the requirement in article II (2) that an arbitration agreement be in writing and to encourage application of article VII (1) to allow any interested party to avail itself of rights it may have, under the law or treaties of the country where an arbitration agreement is sought to be relied upon, to seek recognition of the validity of such an arbitration agreement.

The central obligation imposed upon Parties is to recognize all arbitral awards within the scheme as binding and enforce them, if requested to do so, under the lex fori. Each Party may determine the procedural mechanisms that may be followed where the Convention does not prescribe any requirement.

The Convention defines five grounds upon which recognition and enforcement may be refused at the request of the party against whom it is invoked. The grounds include incapacity of the parties, invalidity of the arbitration agreement, due process, scope of the arbitration agreement, jurisdiction of the arbitral tribunal, setting aside or suspension of an award in the country in which, or under the law of which, that award was made. The Convention defines two additional grounds upon which the court may, on its own motion, refuse recognition and enforcement of an award. Those grounds relate to arbitrability and public policy.

The Convention seeks to encourage recognition and enforcement of awards in the greatest number of cases as possible. That purpose is achieved through article VII (1) of the Convention by removing conditions for recognition and enforcement in national laws that are more stringent than the conditions in the Convention, while allowing the continued application of any national provisions that give special or more favourable rights to a party seeking to enforce an award. That article recognizes the right of any interested party to avail itself of law or treaties of the country where the award is sought to be relied upon, including where such law or treaties offer a regime more favourable than the Convention.

Entry into force

The Convention entered into force on 7 June 1959 (article XII).

How to become a party

The Convention is closed for signature. It is subject to ratification, and is open to accession by any Member State of the United Nations, any other State which is a member of any specialized agency of the United Nations, or is a Party to the Statute of the International Court of Justice (articles VIII and IX).

Optional and/or mandatory declarations and notifications

When signing, ratifying or acceding to the Convention, or notifying a territorial extension under article X, any State may on the basis of reciprocity declare that it will apply the Convention to the recognition and enforcement of awards made only in the territory of another Party to the Convention. It may also declare that it will apply the Convention only to differences arising out of legal relationships, whether contractual or not, which are considered as commercial under the national law of the State making such declaration (article I).

Denunciation/Withdrawal

Any Party may denounce this Convention by a written notification to the Secretary-General of the United Nations. Denunciation shall take effect one year after the date of the receipt of the notification by the Secretary-General (article XIII).

Part one

UNITED NATIONS CONFERENCE ON INTERNATIONAL COMMERCIAL ARBITRATION, NEW YORK, 20 MAY–10 JUNE 1958

*Excerpts from the Final Act of the United Nations Conference on International Commercial Arbitration*¹

"1. The Economic and Social Council of the United Nations, by resolution 604 (XXI) adopted on 3 May 1956, decided to convene a Conference of Plenipotentiaries for the purpose of concluding a convention on the recognition and enforcement of foreign arbitral awards, and to consider other possible measures for increasing the effectiveness of arbitration in the settlement of private law disputes.

[...]

"12. The Economic and Social Council, by its resolution convening the Conference, requested it to conclude a convention on the basis of the draft convention prepared by the Committee on the Enforcement of International Arbitral Awards, taking into account the comments and suggestions made by Governments and non-governmental organizations, as well as the discussion at the twenty-first session of the Council.

"13. On the basis of the deliberations, as recorded in the reports of the working parties and in the records of the plenary meetings, the Conference prepared and opened for signature the Convention on the Recognition and Enforcement of Foreign Arbitral Awards which is annexed to this Final Act.

[...]

"16. In addition the Conference adopted, on the basis of proposals made by the Committee on Other Measures as recorded in its report, the following resolution:

¹The full text of the Final Act of the United Nations Conference on International Commercial Arbitration (E/CONF.26/8Rev.1) is available at http://www.uncitral.org

"The Conference,

"Believing that, in addition to the convention on the recognition and enforcement of foreign arbitral awards just concluded, which would contribute to increasing the effectiveness of arbitration in the settlement of private law disputes, additional measures should be taken in this field,

"Having considered the able survey and analysis of possible measures for increasing the effectiveness of arbitration in the settlement of private law disputes prepared by the Secretary-General (document E/CONF.26/6),

"Having given particular attention to the suggestions made therein for possible ways in which interested governmental and other organizations may make practical contributions to the more effective use of arbitration,

"Expresses the following views with respect to the principal matters dealt with in the note of the Secretary-General:

"1. It considers that wider diffusion of information on arbitration laws, practices and facilities contributes materially to progress in commercial arbitration; recognizes that work has already been done in this field by interested organizations,² and expresses the wish that such organizations, so far as they have not concluded them, continue their activities in this regard, with particular attention to coordinating their respective efforts;

"2. It recognizes the desirability of encouraging where necessary the establishment of new arbitration facilities and the improvement of existing facilities, particularly in some geographic regions and branches of trade; and believes that useful work may be done in this field by appropriate governmental and other organizations, which may be active in arbitration matters, due regard being given to the need to avoid duplication of effort and to concentrate upon those measures of greatest practical benefit to the regions and branches of trade concerned;

"3. It recognizes the value of technical assistance in the development of effective arbitral legislation and institutions; and suggests that interested Governments and other organizations endeavour to furnish such assistance, within the means available, to those seeking it;

"4. It recognizes that regional study groups, seminars or working parties may in appropriate circumstances have productive results; believes that consideration should be given to the advisability of the convening of

²For example, the Economic Commission for Europe and the Inter-American Council of Jurists.

such meetings by the appropriate regional commissions of the United Nations and other bodies, but regards it as important that any such action be taken with careful regard to avoiding duplication and assuring economy of effort and of resources;

"5. It considers that greater uniformity of national laws on arbitration would further the effectiveness of arbitration in the settlement of private law disputes, notes the work already done in this field by various existing organizations,³ and suggests that by way of supplementing the efforts of these bodies appropriate attention be given to defining suitable subject matter for model arbitration statutes and other appropriate measures for encouraging the development of such legislation;

"Expresses the wish that the United Nations, through its appropriate organs, take such steps as it deems feasible to encourage further study of measures for increasing the effectiveness of arbitration in the settlement of private law disputes through the facilities of existing regional bodies and non-governmental organizations and through such other institutions as may be established in the future;

"Suggests that any such steps be taken in a manner that will assure proper coordination of effort, avoidance of duplication and due observance of budgetary considerations;

"*Requests* that the Secretary-General submit this resolution to the appropriate organs of the United Nations."

³For example, the International Institute for the Unification of Private Law and the Inter-American Council of Jurists.

CONVENTION ON THE RECOGNITION AND ENFORCEMENT OF FOREIGN ARBITRAL AWARDS

Article I

1. This Convention shall apply to the recognition and enforcement of arbitral awards made in the territory of a State other than the State where the recognition and enforcement of such awards are sought, and arising out of differences between persons, whether physical or legal. It shall also apply to arbitral awards not considered as domestic awards in the State where their recognition and enforcement are sought.

2. The term "arbitral awards" shall include not only awards made by arbitrators appointed for each case but also those made by permanent arbitral bodies to which the parties have submitted.

3. When signing, ratifying or acceding to this Convention, or notifying extension under article X hereof, any State may on the basis of reciprocity declare that it will apply the Convention to the recognition and enforcement of awards made only in the territory of another Contracting State. It may also declare that it will apply the Convention only to differences arising out of legal relationships, whether contractual or not, which are considered as commercial under the national law of the State making such declaration.

Article II

1. Each Contracting State shall recognize an agreement in writing under which the parties undertake to submit to arbitration all or any differences which have arisen or which may arise between them in respect of a defined legal relationship, whether contractual or not, concerning a subject matter capable of settlement by arbitration.

2. The term "agreement in writing" shall include an arbitral clause in a contract or an arbitration agreement, signed by the parties or contained in an exchange of letters or telegrams.

3. The court of a Contracting State, when seized of an action in a matter in respect of which the parties have made an agreement within the meaning of this article, shall, at the request of one of the parties, refer the parties to arbitration, unless it finds that the said agreement is null and void, inoperative or incapable of being performed.

Article III

Each Contracting State shall recognize arbitral awards as binding and enforce them in accordance with the rules of procedure of the territory where the award is relied upon, under the conditions laid down in the following articles. There shall not be imposed substantially more onerous conditions or higher fees or charges on the recognition or enforcement of arbitral awards to which this Convention applies than are imposed on the recognition or enforcement of domestic arbitral awards.

Article IV

1. To obtain the recognition and enforcement mentioned in the preceding article, the party applying for recognition and enforcement shall, at the time of the application, supply:

(a) The duly authenticated original award or a duly certified copy thereof;

(b) The original agreement referred to in article II or a duly certified copy thereof.

2. If the said award or agreement is not made in an official language of the country in which the award is relied upon, the party applying for recognition and enforcement of the award shall produce a translation of these documents into such language. The translation shall be certified by an official or sworn translator or by a diplomatic or consular agent.

Article V

1. Recognition and enforcement of the award may be refused, at the request of the party against whom it is invoked, only if that party furnishes to the competent authority where the recognition and enforcement is sought, proof that:

(*a*) The parties to the agreement referred to in article II were, under the law applicable to them, under some incapacity, or the said agreement is not valid under the law to which the parties have subjected it or, failing any indication thereon, under the law of the country where the award was made; or

(b) The party against whom the award is invoked was not given proper notice of the appointment of the arbitrator or of the arbitration proceedings or was otherwise unable to present his case; or (c) The award deals with a difference not contemplated by or not falling within the terms of the submission to arbitration, or it contains decisions on matters beyond the scope of the submission to arbitration, provided that, if the decisions on matters submitted to arbitration can be separated from those not so submitted, that part of the award which contains decisions on matters submitted to arbitration may be recognized and enforced; or

(d) The composition of the arbitral authority or the arbitral procedure was not in accordance with the agreement of the parties, or, failing such agreement, was not in accordance with the law of the country where the arbitration took place; or

(e) The award has not yet become binding on the parties, or has been set aside or suspended by a competent authority of the country in which, or under the law of which, that award was made.

2. Recognition and enforcement of an arbitral award may also be refused if the competent authority in the country where recognition and enforcement is sought finds that:

(a) The subject matter of the difference is not capable of settlement by arbitration under the law of that country; or

(b) The recognition or enforcement of the award would be contrary to the public policy of that country.

Article VI

If an application for the setting aside or suspension of the award has been made to a competent authority referred to in article V (1) (e), the authority before which the award is sought to be relied upon may, if it considers it proper, adjourn the decision on the enforcement of the award and may also, on the application of the party claiming enforcement of the award, order the other party to give suitable security.

Article VII

1. The provisions of the present Convention shall not affect the validity of multilateral or bilateral agreements concerning the recognition and enforcement of arbitral awards entered into by the Contracting States nor deprive any interested party of any right he may have to avail himself of an arbitral award in the manner and to the extent allowed by the law or the treaties of the country where such award is sought to be relied upon. 2. The Geneva Protocol on Arbitration Clauses of 1923 and the Geneva Convention on the Execution of Foreign Arbitral Awards of 1927 shall cease to have effect between Contracting States on their becoming bound and to the extent that they become bound, by this Convention.

Article VIII

1. This Convention shall be open until 31 December 1958 for signature on behalf of any Member of the United Nations and also on behalf of any other State which is or hereafter becomes a member of any specialized agency of the United Nations, or which is or hereafter becomes a party to the Statute of the International Court of Justice, or any other State to which an invitation has been addressed by the General Assembly of the United Nations.

2. This Convention shall be ratified and the instrument of ratification shall be deposited with the Secretary-General of the United Nations.

Article IX

1. This Convention shall be open for accession to all States referred to in article VIII.

2. Accession shall be effected by the deposit of an instrument of accession with the Secretary-General of the United Nations.

Article X

1. Any State may, at the time of signature, ratification or accession, declare that this Convention shall extend to all or any of the territories for the international relations of which it is responsible. Such a declaration shall take effect when the Convention enters into force for the State concerned.

2. At any time thereafter any such extension shall be made by notification addressed to the Secretary-General of the United Nations and shall take effect as from the ninetieth day after the day of receipt by the Secretary-General of the United Nations of this notification, or as from the date of entry into force of the Convention for the State concerned, whichever is the later.

3. With respect to those territories to which this Convention is not extended at the time of signature, ratification or accession, each State concerned shall consider the possibility of taking the necessary steps in order to extend the application of this Convention to such territories, subject, where necessary for constitutional reasons, to the consent of the Governments of such territories.

Article XI

In the case of a federal or non-unitary State, the following provisions shall apply:

(*a*) With respect to those articles of this Convention that come within the legislative jurisdiction of the federal authority, the obligations of the federal Government shall to this extent be the same as those of Contracting States which are not federal States;

(b) With respect to those articles of this Convention that come within the legislative jurisdiction of constituent states or provinces which are not, under the constitutional system of the federation, bound to take legislative action, the federal Government shall bring such articles with a favourable recommendation to the notice of the appropriate authorities of constituent states or provinces at the earliest possible moment;

(c) A federal State Party to this Convention shall, at the request of any other Contracting State transmitted through the Secretary-General of the United Nations, supply a statement of the law and practice of the federation and its constituent units in regard to any particular provision of this Convention, showing the extent to which effect has been given to that provision by legislative or other action.

Article XII

1. This Convention shall come into force on the ninetieth day following the date of deposit of the third instrument of ratification or accession.

2. For each State ratifying or acceding to this Convention after the deposit of the third instrument of ratification or accession, this Convention shall enter into force on the ninetieth day after deposit by such State of its instrument of ratification or accession.

Article XIII

1. Any Contracting State may denounce this Convention by a written notification to the Secretary-General of the United Nations. Denunciation

shall take effect one year after the date of receipt of the notification by the Secretary-General.

2. Any State which has made a declaration or notification under article X may, at any time thereafter, by notification to the Secretary-General of the United Nations, declare that this Convention shall cease to extend to the territory concerned one year after the date of the receipt of the notification by the Secretary-General.

3. This Convention shall continue to be applicable to arbitral awards in respect of which recognition or enforcement proceedings have been instituted before the denunciation takes effect.

Article XIV

A Contracting State shall not be entitled to avail itself of the present Convention against other Contracting States except to the extent that it is itself bound to apply the Convention.

Article XV

The Secretary-General of the United Nations shall notify the States contemplated in article VIII of the following:

(a) Signatures and ratifications in accordance with article VIII;

(b) Accessions in accordance with article IX;

(c) Declarations and notifications under articles I, X and XI;

(*d*) The date upon which this Convention enters into force in accordance with article XII;

(e) Denunciations and notifications in accordance with article XIII.

Article XVI

1. This Convention, of which the Chinese, English, French, Russian and Spanish texts shall be equally authentic, shall be deposited in the archives of the United Nations.

2. The Secretary-General of the United Nations shall transmit a certified copy of this Convention to the States contemplated in article VIII.

Part two

RECOMMENDATION REGARDING THE INTERPRETATION OF ARTICLE II, PARAGRAPH 2, AND ARTICLE VII, PARAGRAPH 1, OF THE CONVENTION ON THE RECOGNITION AND ENFORCEMENT OF FOREIGN ARBITRAL AWARDS

General Assembly resolution 61/33 of 4 December 2006

The General Assembly,

Recognizing the value of arbitration as a method of settling disputes arising in the context of international commercial relations,

Recalling its resolution 40/72 of 11 December 1985 regarding the Model Law on International Commercial Arbitration,¹

Recognizing the need for provisions in the Model Law to conform to current practices in international trade and modern means of contracting with regard to the form of the arbitration agreement and the granting of interim measures,

Believing that revised articles of the Model Law on the form of the arbitration agreement and interim measures reflecting those current practices will significantly enhance the operation of the Model Law,

Noting that the preparation of the revised articles of the Model Law on the form of the arbitration agreement and interim measures was the subject of due deliberation and extensive consultations with Governments and interested circles and would contribute significantly to the establishment of a harmonized legal framework for a fair and efficient settlement of international commercial disputes,

¹Official Records of the General Assembly, Fortieth Session, Supplement No. 17 (A/40/17), annex I.

Believing that, in connection with the modernization of articles of the Model Law, the promotion of a uniform interpretation and application of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done at New York, 10 June 1958,² is particularly timely,

1. *Expresses its appreciation* to the United Nations Commission on International Trade Law for formulating and adopting the revised articles of its Model Law on International Commercial Arbitration on the form of the arbitration agreement and interim measures, the text of which is contained in annex I to the report of the United Nations Commission on International Trade Law on the work of its thirty-ninth session,³ and recommends that all States give favourable consideration to the enactment of the revised articles of the Model Law, or the revised Model Law on International Trade Law, when they enact or revise their laws, in view of the desirability of uniformity of the law of arbitral procedures and the specific needs of international commercial arbitration practice;

2. Also expresses its appreciation to the United Nations Commission on International Trade Law for formulating and adopting the recommendation regarding the interpretation of article II, paragraph 2, and article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done at New York, 10 June 1958,² the text of which is contained in annex II to the report of the United Nations Commission on International Trade Law on the work of its thirty-ninth session;³

3. *Requests* the Secretary-General to make all efforts to ensure that the revised articles of the Model Law and the recommendation become generally known and available.

64th plenary meeting 4 December 2006

²United Nations, *Treaty Series*, vol. 330, No. 4739.

³Official Records of the General Assembly, Sixty-first Session, Supplement No. 17 (A/61/17).

RECOMMENDATION REGARDING THE INTERPRETATION OF ARTICLE II, PARAGRAPH 2, AND ARTICLE VII, PARAGRAPH 1, OF THE CONVENTION ON THE RECOGNITION AND ENFORCEMENT OF FOREIGN ARBITRAL AWARDS, DONE IN NEW YORK, 10 JUNE 1958, ADOPTED BY THE UNITED NATIONS COMMISSION ON INTERNATIONAL TRADE LAW ON 7 JULY 2006 AT ITS THIRTY-NINTH SESSION

The United Nations Commission on International Trade Law,

Recalling General Assembly resolution 2205 (XXI) of 17 December 1966, which established the United Nations Commission on International Trade Law with the object of promoting the progressive harmonization and unification of the law of international trade by, inter alia, promoting ways and means of ensuring a uniform interpretation and application of international conventions and uniform laws in the field of the law of international trade,

Conscious of the fact that the different legal, social and economic systems of the world, together with different levels of development, are represented in the Commission,

Recalling successive resolutions of the General Assembly reaffirming the mandate of the Commission as the core legal body within the United Nations system in the field of international trade law to coordinate legal activities in this field,

Convinced that the wide adoption of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York on 10 June 1958,⁴ has been a significant achievement in the promotion of the rule of law, particularly in the field of international trade,

Recalling that the Conference of Plenipotentiaries which prepared and opened the Convention for signature adopted a resolution, which states, inter alia, that the Conference "considers that greater uniformity of national laws on arbitration would further the effectiveness of arbitration in the settlement of private law disputes",

Bearing in mind differing interpretations of the form requirements under the Convention that result in part from differences of expression as between the five equally authentic texts of the Convention,

Taking into account article VII, paragraph 1, of the Convention, a purpose of which is to enable the enforcement of foreign arbitral awards to

⁴United Nations, *Treaty Series*, vol. 330, No. 4739.

the greatest extent, in particular by recognizing the right of any interested party to avail itself of law or treaties of the country where the award is sought to be relied upon, including where such law or treaties offer a regime more favourable than the Convention,

Considering the wide use of electronic commerce,

Taking into account international legal instruments, such as the 1985 UNCITRAL Model Law on International Commercial Arbitration,⁵ as subsequently revised, particularly with respect to article 7,⁶ the UNCITRAL Model Law on Electronic Commerce,⁷ the UNCITRAL Model Law on Electronic Signatures⁸ and the United Nations Convention on the Use of Electronic Communications in International Contracts,⁹

Taking into account also enactments of domestic legislation, as well as case law, more favourable than the Convention in respect of form requirement governing arbitration agreements, arbitration proceedings and the enforcement of arbitral awards,

Considering that, in interpreting the Convention, regard is to be had to the need to promote recognition and enforcement of arbitral awards,

1. *Recommends* that article II, paragraph 2, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958, be applied recognizing that the circumstances described therein are not exhaustive;

2. *Recommends also* that article VII, paragraph 1, of the Convention on the Recognition and Enforcement of Foreign Arbitral Awards, done in New York, 10 June 1958, should be applied to allow any interested party to avail itself of rights it may have, under the law or treaties of the country where an arbitration agreement is sought to be relied upon, to seek recognition of the validity of such an arbitration agreement.

⁵Official Records of the General Assembly, Fortieth Session, Supplement No. 17 (A/40/17), annex I, and United Nations publication, Sales No. E.95.V.18.

⁶Ibid., Sixty-first Session, Supplement No. 17 (A/61/17), annex I.

⁷Ibid., *Fifty-first Session, Supplement No. 17* (A/51/17), annex I, and United Nations publication, Sales No. E.99.V.4, which contains also an additional article 5 bis, adopted in 1998, and the accompanying Guide to Enactment.

⁸Ibid., *Fifty-sixth Session, Supplement No. 17* and corrigendum (A/56/17 and Corr.3), annex II, and United Nations publication, Sales No. E.02.V.8, which contains also the accompanying Guide to Enactment.

⁹General Assembly resolution 60/21, annex.



UNITED NATIONS CONVENTION AGAINST CORRUPTION

//#**फ** 111 <u>111 11 12</u>

United Nations Convention against Corruption

Preamble

The States Parties to this Convention,

Concerned about the seriousness of problems and threats posed by corruption to the stability and security of societies, undermining the institutions and values of democracy, ethical values and justice and jeopardizing sustainable development and the rule of law,

Concerned also about the links between corruption and other forms of crime, in particular organized crime and economic crime, including money-laundering,

Concerned further about cases of corruption that involve vast quantities of assets, which may constitute a substantial proportion of the resources of States, and that threaten the political stability and sustainable development of those States,

Convinced that corruption is no longer a local matter but a transnational phenomenon that affects all societies and economies, making international cooperation to prevent and control it essential,

Convinced also that a comprehensive and multidisciplinary approach is required to prevent and combat corruption effectively,

Convinced further that the availability of technical assistance can play an important role in enhancing the ability of States, including by strengthening capacity and by institution-building, to prevent and combat corruption effectively,

Convinced that the illicit acquisition of personal wealth can be particularly damaging to democratic institutions, national economies and the rule of law,

Determined to prevent, detect and deter in a more effective manner international transfers of illicitly acquired assets and to strengthen international cooperation in asset recovery,

Acknowledging the fundamental principles of due process of law in criminal proceedings and in civil or administrative proceedings to adjudicate property rights,

Bearing in mind that the prevention and eradication of corruption is a responsibility of all States and that they must cooperate with one another, with the support and involvement of individuals and groups outside the public sector, such as civil society, non-governmental organizations and community-based organizations, if their efforts in this area are to be effective,

Bearing also in mind the principles of proper management of public affairs and public property, fairness, responsibility and equality before the law and the need to safeguard integrity and to foster a culture of rejection of corruption,

Commending the work of the Commission on Crime Prevention and Criminal Justice and the United Nations Office on Drugs and Crime in preventing and combating corruption,

1

Recalling the work carried out by other international and regional organizations in this field, including the activities of the African Union, the Council of Europe, the Customs Cooperation Council (also known as the World Customs Organization), the European Union, the League of Arab States, the Organisation for Economic Cooperation and Development and the Organization of American States,

Taking note with appreciation of multilateral instruments to prevent and combat corruption, including, inter alia, the Inter-American Convention against Corruption, adopted by the Organization of American States on 29 March 1996,¹ the Convention on the Fight against Corruption involving Officials of the European Communities or Officials of Member States of the European Union, adopted by the Council of the European Union on 26 May 1997,² the Convention on Combating Bribery of Foreign Public Officials in International Business Transactions, adopted by the Organisation for Economic Cooperation and Development on 21 November 1997,³ the Criminal Law Convention on Corruption, adopted by the Committee of Ministers of the Council of Europe on 27 January 1999,⁴ the Civil Law Convention on Corruption, adopted by the African Union Convention on Preventing and Combating Corruption, adopted by the Heads of State and Government of the African Union on 12 July 2003,

Welcoming the entry into force on 29 September 2003 of the United Nations Convention against Transnational Organized Crime,⁶

Have agreed as follows:

Chapter I General provisions

Article 1 Statement of purpose

The purposes of this Convention are:

(a) To promote and strengthen measures to prevent and combat corruption more efficiently and effectively;

(b) To promote, facilitate and support international cooperation and technical assistance in the prevention of and fight against corruption, including in asset recovery;

(c) To promote integrity, accountability and proper management of public affairs and public property.

¹ See E/1996/99.

² Official Journal of the European Communities, C 195, 25 June 1997.

³ See Corruption and Integrity Improvement Initiatives in Developing Countries (United Nations publication, Sales No. E.98.III.B.18).

⁴ Council of Europe, European Treaty Series, No. 173.

⁵ Ibid., No. 174.

⁶ General Assembly resolution 55/25, annex I.

Article 2

Use of terms

For the purposes of this Convention:

(a) "Public official" shall mean: (i) any person holding a legislative, executive, administrative or judicial office of a State Party, whether appointed or elected, whether permanent or temporary, whether paid or unpaid, irrespective of that person's seniority; (ii) any other person who performs a public function, including for a public agency or public enterprise, or provides a public service, as defined in the domestic law of the State Party and as applied in the pertinent area of law of that State Party; (iii) any other person defined as a "public official" in the domestic law of a State Party. However, for the purpose of some specific measures contained in chapter II of this Convention, "public official" may mean any person who performs a public function or provides a public service as defined in the domestic law of the State Party and as applied in the pertinent area of some specific measures contained in chapter II of this Convention, "public official" may mean any person who performs a public function or provides a public service as defined in the domestic law of the State Party and as applied in the pertinent area of law of that State Party and as applied in the pertinent area of law of that State Party and as applied in the pertinent area of law of that State Party and as applied in the pertinent area of law of that State Party and as applied in the pertinent area of law of that State Party and as applied in the pertinent area of law of that State Party;

(b) "Foreign public official" shall mean any person holding a legislative, executive, administrative or judicial office of a foreign country, whether appointed or elected; and any person exercising a public function for a foreign country, including for a public agency or public enterprise;

(c) "Official of a public international organization" shall mean an international civil servant or any person who is authorized by such an organization to act on behalf of that organization;

(d) "Property" shall mean assets of every kind, whether corporeal or incorporeal, movable or immovable, tangible or intangible, and legal documents or instruments evidencing title to or interest in such assets;

(e) "Proceeds of crime" shall mean any property derived from or obtained, directly or indirectly, through the commission of an offence;

(f) "Freezing" or "seizure" shall mean temporarily prohibiting the transfer, conversion, disposition or movement of property or temporarily assuming custody or control of property on the basis of an order issued by a court or other competent authority;

(g) "Confiscation", which includes forfeiture where applicable, shall mean the permanent deprivation of property by order of a court or other competent authority;

(h) "Predicate offence" shall mean any offence as a result of which proceeds have been generated that may become the subject of an offence as defined in article 23 of this Convention;

(i) "Controlled delivery" shall mean the technique of allowing illicit or suspect consignments to pass out of, through or into the territory of one or more States, with the knowledge and under the supervision of their competent authorities, with a view to the investigation of an offence and the identification of persons involved in the commission of the offence.

Article 3

Scope of application

1. This Convention shall apply, in accordance with its terms, to the prevention, investigation and prosecution of corruption and to the freezing, seizure, confiscation and return of the proceeds of offences established in accordance with this Convention.

2. For the purposes of implementing this Convention, it shall not be necessary, except as otherwise stated herein, for the offences set forth in it to result in damage or harm to state property.

Article 4

Protection of sovereignty

1. States Parties shall carry out their obligations under this Convention in a manner consistent with the principles of sovereign equality and territorial integrity of States and that of non-intervention in the domestic affairs of other States.

2. Nothing in this Convention shall entitle a State Party to undertake in the territory of another State the exercise of jurisdiction and performance of functions that are reserved exclusively for the authorities of that other State by its domestic law.

Chapter II Preventive measures

Article 5

Preventive anti-corruption policies and practices

1. Each State Party shall, in accordance with the fundamental principles of its legal system, develop and implement or maintain effective, coordinated anticorruption policies that promote the participation of society and reflect the principles of the rule of law, proper management of public affairs and public property, integrity, transparency and accountability.

2. Each State Party shall endeavour to establish and promote effective practices aimed at the prevention of corruption.

3. Each State Party shall endeavour to periodically evaluate relevant legal instruments and administrative measures with a view to determining their adequacy to prevent and fight corruption.

4. States Parties shall, as appropriate and in accordance with the fundamental principles of their legal system, collaborate with each other and with relevant international and regional organizations in promoting and developing the measures referred to in this article. That collaboration may include participation in international programmes and projects aimed at the prevention of corruption.

Article 6 Preventive anti-corruption body or bodies

1. Each State Party shall, in accordance with the fundamental principles of its legal system, ensure the existence of a body or bodies, as appropriate, that prevent corruption by such means as:

(a) Implementing the policies referred to in article 5 of this Convention and, where appropriate, overseeing and coordinating the implementation of those policies;

(b) Increasing and disseminating knowledge about the prevention of corruption.

2. Each State Party shall grant the body or bodies referred to in paragraph 1 of this article the necessary independence, in accordance with the fundamental principles of its legal system, to enable the body or bodies to carry out its or their functions effectively and free from any undue influence. The necessary material resources and specialized staff, as well as the training that such staff may require to carry out their functions, should be provided.

3. Each State Party shall inform the Secretary-General of the United Nations of the name and address of the authority or authorities that may assist other States Parties in developing and implementing specific measures for the prevention of corruption.

Article 7

Public sector

1. Each State Party shall, where appropriate and in accordance with the fundamental principles of its legal system, endeavour to adopt, maintain and strengthen systems for the recruitment, hiring, retention, promotion and retirement of civil servants and, where appropriate, other non-elected public officials:

(a) That are based on principles of efficiency, transparency and objective criteria such as merit, equity and aptitude;

(b) That include adequate procedures for the selection and training of individuals for public positions considered especially vulnerable to corruption and the rotation, where appropriate, of such individuals to other positions;

(c) That promote adequate remuneration and equitable pay scales, taking into account the level of economic development of the State Party;

(d) That promote education and training programmes to enable them to meet the requirements for the correct, honourable and proper performance of public functions and that provide them with specialized and appropriate training to enhance their awareness of the risks of corruption inherent in the performance of their functions. Such programmes may make reference to codes or standards of conduct in applicable areas.

2. Each State Party shall also consider adopting appropriate legislative and administrative measures, consistent with the objectives of this Convention and in accordance with the fundamental principles of its domestic law, to prescribe criteria concerning candidature for and election to public office.

3. Each State Party shall also consider taking appropriate legislative and administrative measures, consistent with the objectives of this Convention and in accordance with the fundamental principles of its domestic law, to enhance transparency in the funding of candidatures for elected public office and, where applicable, the funding of political parties.

4. Each State Party shall, in accordance with the fundamental principles of its domestic law, endeavour to adopt, maintain and strengthen systems that promote transparency and prevent conflicts of interest.

Article 8

Codes of conduct for public officials

1. In order to fight corruption, each State Party shall promote, inter alia, integrity, honesty and responsibility among its public officials, in accordance with the fundamental principles of its legal system.

2. In particular, each State Party shall endeavour to apply, within its own institutional and legal systems, codes or standards of conduct for the correct, honourable and proper performance of public functions.

3. For the purposes of implementing the provisions of this article, each State Party shall, where appropriate and in accordance with the fundamental principles of its legal system, take note of the relevant initiatives of regional, interregional and multilateral organizations, such as the International Code of Conduct for Public Officials contained in the annex to General Assembly resolution 51/59 of 12 December 1996.

4. Each State Party shall also consider, in accordance with the fundamental principles of its domestic law, establishing measures and systems to facilitate the reporting by public officials of acts of corruption to appropriate authorities, when such acts come to their notice in the performance of their functions.

5. Each State Party shall endeavour, where appropriate and in accordance with the fundamental principles of its domestic law, to establish measures and systems requiring public officials to make declarations to appropriate authorities regarding, inter alia, their outside activities, employment, investments, assets and substantial gifts or benefits from which a conflict of interest may result with respect to their functions as public officials.

6. Each State Party shall consider taking, in accordance with the fundamental principles of its domestic law, disciplinary or other measures against public officials who violate the codes or standards established in accordance with this article.

Article 9

Public procurement and management of public finances

1. Each State Party shall, in accordance with the fundamental principles of its legal system, take the necessary steps to establish appropriate systems of procurement, based on transparency, competition and objective criteria in decision-making, that are effective, inter alia, in preventing corruption. Such systems, which may take into account appropriate threshold values in their application, shall address, inter alia: (a) The public distribution of information relating to procurement procedures and contracts, including information on invitations to tender and relevant or pertinent information on the award of contracts, allowing potential tenderers sufficient time to prepare and submit their tenders;

(b) The establishment, in advance, of conditions for participation, including selection and award criteria and tendering rules, and their publication;

(c) The use of objective and predetermined criteria for public procurement decisions, in order to facilitate the subsequent verification of the correct application of the rules or procedures;

(d) An effective system of domestic review, including an effective system of appeal, to ensure legal recourse and remedies in the event that the rules or procedures established pursuant to this paragraph are not followed;

(e) Where appropriate, measures to regulate matters regarding personnel responsible for procurement, such as declaration of interest in particular public procurements, screening procedures and training requirements.

2. Each State Party shall, in accordance with the fundamental principles of its legal system, take appropriate measures to promote transparency and accountability in the management of public finances. Such measures shall encompass, inter alia:

- (a) Procedures for the adoption of the national budget;
- (b) Timely reporting on revenue and expenditure;
- (c) A system of accounting and auditing standards and related oversight;
- (d) Effective and efficient systems of risk management and internal control; and

(e) Where appropriate, corrective action in the case of failure to comply with the requirements established in this paragraph.

3. Each State Party shall take such civil and administrative measures as may be necessary, in accordance with the fundamental principles of its domestic law, to preserve the integrity of accounting books, records, financial statements or other documents related to public expenditure and revenue and to prevent the falsification of such documents.

Article 10

Public reporting

Taking into account the need to combat corruption, each State Party shall, in accordance with the fundamental principles of its domestic law, take such measures as may be necessary to enhance transparency in its public administration, including with regard to its organization, functioning and decision-making processes, where appropriate. Such measures may include, inter alia:

(a) Adopting procedures or regulations allowing members of the general public to obtain, where appropriate, information on the organization, functioning and decision-making processes of its public administration and, with due regard for the protection of privacy and personal data, on decisions and legal acts that concern members of the public;

(b) Simplifying administrative procedures, where appropriate, in order to facilitate public access to the competent decision-making authorities; and

(c) Publishing information, which may include periodic reports on the risks of corruption in its public administration.

Article 11

Measures relating to the judiciary and prosecution services

1. Bearing in mind the independence of the judiciary and its crucial role in combating corruption, each State Party shall, in accordance with the fundamental principles of its legal system and without prejudice to judicial independence, take measures to strengthen integrity and to prevent opportunities for corruption among members of the judiciary. Such measures may include rules with respect to the conduct of members of the judiciary.

2. Measures to the same effect as those taken pursuant to paragraph 1 of this article may be introduced and applied within the prosecution service in those States Parties where it does not form part of the judiciary but enjoys independence similar to that of the judicial service.

Article 12

Private sector

1. Each State Party shall take measures, in accordance with the fundamental principles of its domestic law, to prevent corruption involving the private sector, enhance accounting and auditing standards in the private sector and, where appropriate, provide effective, proportionate and dissuasive civil, administrative or criminal penalties for failure to comply with such measures.

2. Measures to achieve these ends may include, inter alia:

(a) Promoting cooperation between law enforcement agencies and relevant private entities;

(b) Promoting the development of standards and procedures designed to safeguard the integrity of relevant private entities, including codes of conduct for the correct, honourable and proper performance of the activities of business and all relevant professions and the prevention of conflicts of interest, and for the promotion of the use of good commercial practices among businesses and in the contractual relations of businesses with the State;

(c) Promoting transparency among private entities, including, where appropriate, measures regarding the identity of legal and natural persons involved in the establishment and management of corporate entities;

(d) Preventing the misuse of procedures regulating private entities, including procedures regarding subsidies and licences granted by public authorities for commercial activities;

(e) Preventing conflicts of interest by imposing restrictions, as appropriate and for a reasonable period of time, on the professional activities of former public officials or on the employment of public officials by the private sector after their resignation or retirement, where such activities or employment relate directly to the functions held or supervised by those public officials during their tenure; (f) Ensuring that private enterprises, taking into account their structure and size, have sufficient internal auditing controls to assist in preventing and detecting acts of corruption and that the accounts and required financial statements of such private enterprises are subject to appropriate auditing and certification procedures.

3. In order to prevent corruption, each State Party shall take such measures as may be necessary, in accordance with its domestic laws and regulations regarding the maintenance of books and records, financial statement disclosures and accounting and auditing standards, to prohibit the following acts carried out for the purpose of committing any of the offences established in accordance with this Convention:

- (a) The establishment of off-the-books accounts;
- (b) The making of off-the-books or inadequately identified transactions;
- (c) The recording of non-existent expenditure;
- (d) The entry of liabilities with incorrect identification of their objects;
- (e) The use of false documents; and

(f) The intentional destruction of bookkeeping documents earlier than foreseen by the law.

4. Each State Party shall disallow the tax deductibility of expenses that constitute bribes, the latter being one of the constituent elements of the offences established in accordance with articles 15 and 16 of this Convention and, where appropriate, other expenses incurred in furtherance of corrupt conduct.

Article 13

Participation of society

1. Each State Party shall take appropriate measures, within its means and in accordance with fundamental principles of its domestic law, to promote the active participation of individuals and groups outside the public sector, such as civil society, non-governmental organizations and community-based organizations, in the prevention of and the fight against corruption and to raise public awareness regarding the existence, causes and gravity of and the threat posed by corruption. This participation should be strengthened by such measures as:

(a) Enhancing the transparency of and promoting the contribution of the public to decision-making processes;

(b) Ensuring that the public has effective access to information;

(c) Undertaking public information activities that contribute to non-tolerance of corruption, as well as public education programmes, including school and university curricula;

(d) Respecting, promoting and protecting the freedom to seek, receive, publish and disseminate information concerning corruption. That freedom may be subject to certain restrictions, but these shall only be such as are provided for by law and are necessary:

(i) For respect of the rights or reputations of others;

(ii) For the protection of national security or *ordre public* or of public health or morals.

2. Each State Party shall take appropriate measures to ensure that the relevant anti-corruption bodies referred to in this Convention are known to the public and shall provide access to such bodies, where appropriate, for the reporting, including anonymously, of any incidents that may be considered to constitute an offence established in accordance with this Convention.

Article 14 Measures to prevent money-laundering

1. Each State Party shall:

(a) Institute a comprehensive domestic regulatory and supervisory regime for banks and non-bank financial institutions, including natural or legal persons that provide formal or informal services for the transmission of money or value and, where appropriate, other bodies particularly susceptible to money-laundering, within its competence, in order to deter and detect all forms of money-laundering, which regime shall emphasize requirements for customer and, where appropriate, beneficial owner identification, record-keeping and the reporting of suspicious transactions;

(b) Without prejudice to article 46 of this Convention, ensure that administrative, regulatory, law enforcement and other authorities dedicated to combating money-laundering (including, where appropriate under domestic law, judicial authorities) have the ability to cooperate and exchange information at the national and international levels within the conditions prescribed by its domestic law and, to that end, shall consider the establishment of a financial intelligence unit to serve as a national centre for the collection, analysis and dissemination of information regarding potential money-laundering.

2. States Parties shall consider implementing feasible measures to detect and monitor the movement of cash and appropriate negotiable instruments across their borders, subject to safeguards to ensure proper use of information and without impeding in any way the movement of legitimate capital. Such measures may include a requirement that individuals and businesses report the cross-border transfer of substantial quantities of cash and appropriate negotiable instruments.

3. States Parties shall consider implementing appropriate and feasible measures to require financial institutions, including money remitters:

(a) To include on forms for the electronic transfer of funds and related messages accurate and meaningful information on the originator;

(b) To maintain such information throughout the payment chain; and

(c) To apply enhanced scrutiny to transfers of funds that do not contain complete information on the originator.

4. In establishing a domestic regulatory and supervisory regime under the terms of this article, and without prejudice to any other article of this Convention, States Parties are called upon to use as a guideline the relevant initiatives of regional, interregional and multilateral organizations against money-laundering.

5. States Parties shall endeavour to develop and promote global, regional, subregional and bilateral cooperation among judicial, law enforcement and financial regulatory authorities in order to combat money-laundering.

Chapter III Criminalization and law enforcement

Article 15 Bribery of national public officials

Each State Party shall adopt such legislative and other measures as may be necessary to establish as criminal offences, when committed intentionally:

(a) The promise, offering or giving, to a public official, directly or indirectly, of an undue advantage, for the official himself or herself or another person or entity, in order that the official act or refrain from acting in the exercise of his or her official duties;

(b) The solicitation or acceptance by a public official, directly or indirectly, of an undue advantage, for the official himself or herself or another person or entity, in order that the official act or refrain from acting in the exercise of his or her official duties.

Article 16 Bribery of foreign public officials and officials of public international organizations

1. Each State Party shall adopt such legislative and other measures as may be necessary to establish as a criminal offence, when committed intentionally, the promise, offering or giving to a foreign public official or an official of a public international organization, directly or indirectly, of an undue advantage, for the official himself or herself or another person or entity, in order that the official act or refrain from acting in the exercise of his or her official duties, in order to obtain or retain business or other undue advantage in relation to the conduct of international business.

2. Each State Party shall consider adopting such legislative and other measures as may be necessary to establish as a criminal offence, when committed intentionally, the solicitation or acceptance by a foreign public official or an official of a public international organization, directly or indirectly, of an undue advantage, for the official himself or herself or another person or entity, in order that the official act or refrain from acting in the exercise of his or her official duties.

Article 17

Embezzlement, misappropriation or other diversion of property by a public official

Each State Party shall adopt such legislative and other measures as may be necessary to establish as criminal offences, when committed intentionally, the embezzlement, misappropriation or other diversion by a public official for his or her benefit or for the benefit of another person or entity, of any property, public or private funds or securities or any other thing of value entrusted to the public official by virtue of his or her position.

Article 18

Trading in influence

Each State Party shall consider adopting such legislative and other measures as may be necessary to establish as criminal offences, when committed intentionally:

(a) The promise, offering or giving to a public official or any other person, directly or indirectly, of an undue advantage in order that the public official or the person abuse his or her real or supposed influence with a view to obtaining from an administration or public authority of the State Party an undue advantage for the original instigator of the act or for any other person;

(b) The solicitation or acceptance by a public official or any other person, directly or indirectly, of an undue advantage for himself or herself or for another person in order that the public official or the person abuse his or her real or supposed influence with a view to obtaining from an administration or public authority of the State Party an undue advantage.

Article 19

Abuse of functions

Each State Party shall consider adopting such legislative and other measures as may be necessary to establish as a criminal offence, when committed intentionally, the abuse of functions or position, that is, the performance of or failure to perform an act, in violation of laws, by a public official in the discharge of his or her functions, for the purpose of obtaining an undue advantage for himself or herself or for another person or entity.

Article 20

Illicit enrichment

Subject to its constitution and the fundamental principles of its legal system, each State Party shall consider adopting such legislative and other measures as may be necessary to establish as a criminal offence, when committed intentionally, illicit enrichment, that is, a significant increase in the assets of a public official that he or she cannot reasonably explain in relation to his or her lawful income.

Article 21

Bribery in the private sector

Each State Party shall consider adopting such legislative and other measures as may be necessary to establish as criminal offences, when committed intentionally in the course of economic, financial or commercial activities:

(a) The promise, offering or giving, directly or indirectly, of an undue advantage to any person who directs or works, in any capacity, for a private sector entity, for the person himself or herself or for another person, in order that he or she, in breach of his or her duties, act or refrain from acting;

(b) The solicitation or acceptance, directly or indirectly, of an undue advantage by any person who directs or works, in any capacity, for a private sector

entity, for the person himself or herself or for another person, in order that he or she, in breach of his or her duties, act or refrain from acting.

Article 22

Embezzlement of property in the private sector

Each State Party shall consider adopting such legislative and other measures as may be necessary to establish as a criminal offence, when committed intentionally in the course of economic, financial or commercial activities, embezzlement by a person who directs or works, in any capacity, in a private sector entity of any property, private funds or securities or any other thing of value entrusted to him or her by virtue of his or her position.

Article 23

Laundering of proceeds of crime

1. Each State Party shall adopt, in accordance with fundamental principles of its domestic law, such legislative and other measures as may be necessary to establish as criminal offences, when committed intentionally:

(a) (i) The conversion or transfer of property, knowing that such property is the proceeds of crime, for the purpose of concealing or disguising the illicit origin of the property or of helping any person who is involved in the commission of the predicate offence to evade the legal consequences of his or her action;

(ii) The concealment or disguise of the true nature, source, location, disposition, movement or ownership of or rights with respect to property, knowing that such property is the proceeds of crime;

(b) Subject to the basic concepts of its legal system:

(i) The acquisition, possession or use of property, knowing, at the time of receipt, that such property is the proceeds of crime;

(ii) Participation in, association with or conspiracy to commit, attempts to commit and aiding, abetting, facilitating and counselling the commission of any of the offences established in accordance with this article.

2. For purposes of implementing or applying paragraph 1 of this article:

(a) Each State Party shall seek to apply paragraph 1 of this article to the widest range of predicate offences;

(b) Each State Party shall include as predicate offences at a minimum a comprehensive range of criminal offences established in accordance with this Convention;

(c) For the purposes of subparagraph (b) above, predicate offences shall include offences committed both within and outside the jurisdiction of the State Party in question. However, offences committed outside the jurisdiction of a State Party shall constitute predicate offences only when the relevant conduct is a criminal offence under the domestic law of the State where it is committed and would be a criminal offence under the domestic law of the State Party implementing or applying this article had it been committed there; (d) Each State Party shall furnish copies of its laws that give effect to this article and of any subsequent changes to such laws or a description thereof to the Secretary-General of the United Nations;

(e) If required by fundamental principles of the domestic law of a State Party, it may be provided that the offences set forth in paragraph 1 of this article do not apply to the persons who committed the predicate offence.

Article 24

Concealment

Without prejudice to the provisions of article 23 of this Convention, each State Party shall consider adopting such legislative and other measures as may be necessary to establish as a criminal offence, when committed intentionally after the commission of any of the offences established in accordance with this Convention without having participated in such offences, the concealment or continued retention of property when the person involved knows that such property is the result of any of the offences established in accordance with this Convention.

Article 25

Obstruction of justice

Each State Party shall adopt such legislative and other measures as may be necessary to establish as criminal offences, when committed intentionally:

(a) The use of physical force, threats or intimidation or the promise, offering or giving of an undue advantage to induce false testimony or to interfere in the giving of testimony or the production of evidence in a proceeding in relation to the commission of offences established in accordance with this Convention;

(b) The use of physical force, threats or intimidation to interfere with the exercise of official duties by a justice or law enforcement official in relation to the commission of offences established in accordance with this Convention. Nothing in this subparagraph shall prejudice the right of States Parties to have legislation that protects other categories of public official.

Article 26 Liability of legal persons

1. Each State Party shall adopt such measures as may be necessary, consistent with its legal principles, to establish the liability of legal persons for participation in the offences established in accordance with this Convention.

2. Subject to the legal principles of the State Party, the liability of legal persons may be criminal, civil or administrative.

3. Such liability shall be without prejudice to the criminal liability of the natural persons who have committed the offences.

4. Each State Party shall, in particular, ensure that legal persons held liable in accordance with this article are subject to effective, proportionate and dissuasive criminal or non-criminal sanctions, including monetary sanctions.

Article 27 Participation and attempt

1. Each State Party shall adopt such legislative and other measures as may be necessary to establish as a criminal offence, in accordance with its domestic law, participation in any capacity such as an accomplice, assistant or instigator in an offence established in accordance with this Convention.

2. Each State Party may adopt such legislative and other measures as may be necessary to establish as a criminal offence, in accordance with its domestic law, any attempt to commit an offence established in accordance with this Convention.

3. Each State Party may adopt such legislative and other measures as may be necessary to establish as a criminal offence, in accordance with its domestic law, the preparation for an offence established in accordance with this Convention.

Article 28 Knowledge, intent and purpose as elements of an offence

Knowledge, intent or purpose required as an element of an offence established in accordance with this Convention may be inferred from objective factual circumstances.

Article 29

Statute of limitations

Each State Party shall, where appropriate, establish under its domestic law a long statute of limitations period in which to commence proceedings for any offence established in accordance with this Convention and establish a longer statute of limitations period or provide for the suspension of the statute of limitations where the alleged offender has evaded the administration of justice.

Article 30

Prosecution, adjudication and sanctions

1. Each State Party shall make the commission of an offence established in accordance with this Convention liable to sanctions that take into account the gravity of that offence.

2. Each State Party shall take such measures as may be necessary to establish or maintain, in accordance with its legal system and constitutional principles, an appropriate balance between any immunities or jurisdictional privileges accorded to its public officials for the performance of their functions and the possibility, when necessary, of effectively investigating, prosecuting and adjudicating offences established in accordance with this Convention.

3. Each State Party shall endeavour to ensure that any discretionary legal powers under its domestic law relating to the prosecution of persons for offences established in accordance with this Convention are exercised to maximize the effectiveness of law enforcement measures in respect of those offences and with due regard to the need to deter the commission of such offences.

4. In the case of offences established in accordance with this Convention, each State Party shall take appropriate measures, in accordance with its domestic

law and with due regard to the rights of the defence, to seek to ensure that conditions imposed in connection with decisions on release pending trial or appeal take into consideration the need to ensure the presence of the defendant at subsequent criminal proceedings.

5. Each State Party shall take into account the gravity of the offences concerned when considering the eventuality of early release or parole of persons convicted of such offences.

6. Each State Party, to the extent consistent with the fundamental principles of its legal system, shall consider establishing procedures through which a public official accused of an offence established in accordance with this Convention may, where appropriate, be removed, suspended or reassigned by the appropriate authority, bearing in mind respect for the principle of the presumption of innocence.

7. Where warranted by the gravity of the offence, each State Party, to the extent consistent with the fundamental principles of its legal system, shall consider establishing procedures for the disqualification, by court order or any other appropriate means, for a period of time determined by its domestic law, of persons convicted of offences established in accordance with this Convention from:

(a) Holding public office; and

(b) Holding office in an enterprise owned in whole or in part by the State.

8. Paragraph 1 of this article shall be without prejudice to the exercise of disciplinary powers by the competent authorities against civil servants.

9. Nothing contained in this Convention shall affect the principle that the description of the offences established in accordance with this Convention and of the applicable legal defences or other legal principles controlling the lawfulness of conduct is reserved to the domestic law of a State Party and that such offences shall be prosecuted and punished in accordance with that law.

10. States Parties shall endeavour to promote the reintegration into society of persons convicted of offences established in accordance with this Convention.

Article 31

Freezing, seizure and confiscation

1. Each State Party shall take, to the greatest extent possible within its domestic legal system, such measures as may be necessary to enable confiscation of:

(a) Proceeds of crime derived from offences established in accordance with this Convention or property the value of which corresponds to that of such proceeds;

(b) Property, equipment or other instrumentalities used in or destined for use in offences established in accordance with this Convention.

2. Each State Party shall take such measures as may be necessary to enable the identification, tracing, freezing or seizure of any item referred to in paragraph 1 of this article for the purpose of eventual confiscation.

3. Each State Party shall adopt, in accordance with its domestic law, such legislative and other measures as may be necessary to regulate the administration by

the competent authorities of frozen, seized or confiscated property covered in paragraphs 1 and 2 of this article.

4. If such proceeds of crime have been transformed or converted, in part or in full, into other property, such property shall be liable to the measures referred to in this article instead of the proceeds.

5. If such proceeds of crime have been intermingled with property acquired from legitimate sources, such property shall, without prejudice to any powers relating to freezing or seizure, be liable to confiscation up to the assessed value of the intermingled proceeds.

6. Income or other benefits derived from such proceeds of crime, from property into which such proceeds of crime have been transformed or converted or from property with which such proceeds of crime have been intermingled shall also be liable to the measures referred to in this article, in the same manner and to the same extent as proceeds of crime.

7. For the purpose of this article and article 55 of this Convention, each State Party shall empower its courts or other competent authorities to order that bank, financial or commercial records be made available or seized. A State Party shall not decline to act under the provisions of this paragraph on the ground of bank secrecy.

8. States Parties may consider the possibility of requiring that an offender demonstrate the lawful origin of such alleged proceeds of crime or other property liable to confiscation, to the extent that such a requirement is consistent with the fundamental principles of their domestic law and with the nature of judicial and other proceedings.

9. The provisions of this article shall not be so construed as to prejudice the rights of bona fide third parties.

10. Nothing contained in this article shall affect the principle that the measures to which it refers shall be defined and implemented in accordance with and subject to the provisions of the domestic law of a State Party.

Article 32

Protection of witnesses, experts and victims

1. Each State Party shall take appropriate measures in accordance with its domestic legal system and within its means to provide effective protection from potential retaliation or intimidation for witnesses and experts who give testimony concerning offences established in accordance with this Convention and, as appropriate, for their relatives and other persons close to them.

2. The measures envisaged in paragraph 1 of this article may include, inter alia, without prejudice to the rights of the defendant, including the right to due process:

(a) Establishing procedures for the physical protection of such persons, such as, to the extent necessary and feasible, relocating them and permitting, where appropriate, non-disclosure or limitations on the disclosure of information concerning the identity and whereabouts of such persons; (b) Providing evidentiary rules to permit witnesses and experts to give testimony in a manner that ensures the safety of such persons, such as permitting testimony to be given through the use of communications technology such as video or other adequate means.

3. States Parties shall consider entering into agreements or arrangements with other States for the relocation of persons referred to in paragraph 1 of this article.

4. The provisions of this article shall also apply to victims insofar as they are witnesses.

5. Each State Party shall, subject to its domestic law, enable the views and concerns of victims to be presented and considered at appropriate stages of criminal proceedings against offenders in a manner not prejudicial to the rights of the defence.

Article 33

Protection of reporting persons

Each State Party shall consider incorporating into its domestic legal system appropriate measures to provide protection against any unjustified treatment for any person who reports in good faith and on reasonable grounds to the competent authorities any facts concerning offences established in accordance with this Convention.

Article 34

Consequences of acts of corruption

With due regard to the rights of third parties acquired in good faith, each State Party shall take measures, in accordance with the fundamental principles of its domestic law, to address consequences of corruption. In this context, States Parties may consider corruption a relevant factor in legal proceedings to annul or rescind a contract, withdraw a concession or other similar instrument or take any other remedial action.

Article 35 Compensation for damage

Each State Party shall take such measures as may be necessary, in accordance with principles of its domestic law, to ensure that entities or persons who have suffered damage as a result of an act of corruption have the right to initiate legal proceedings against those responsible for that damage in order to obtain compensation.

Article 36

Specialized authorities

Each State Party shall, in accordance with the fundamental principles of its legal system, ensure the existence of a body or bodies or persons specialized in combating corruption through law enforcement. Such body or bodies or persons shall be granted the necessary independence, in accordance with the fundamental principles of the legal system of the State Party, to be able to carry out their

functions effectively and without any undue influence. Such persons or staff of such body or bodies should have the appropriate training and resources to carry out their tasks.

Article 37

Cooperation with law enforcement authorities

1. Each State Party shall take appropriate measures to encourage persons who participate or who have participated in the commission of an offence established in accordance with this Convention to supply information useful to competent authorities for investigative and evidentiary purposes and to provide factual, specific help to competent authorities that may contribute to depriving offenders of the proceeds of crime and to recovering such proceeds.

2. Each State Party shall consider providing for the possibility, in appropriate cases, of mitigating punishment of an accused person who provides substantial cooperation in the investigation or prosecution of an offence established in accordance with this Convention.

3. Each State Party shall consider providing for the possibility, in accordance with fundamental principles of its domestic law, of granting immunity from prosecution to a person who provides substantial cooperation in the investigation or prosecution of an offence established in accordance with this Convention.

4. Protection of such persons shall be, mutatis mutandis, as provided for in article 32 of this Convention.

5. Where a person referred to in paragraph 1 of this article located in one State Party can provide substantial cooperation to the competent authorities of another State Party, the States Parties concerned may consider entering into agreements or arrangements, in accordance with their domestic law, concerning the potential provision by the other State Party of the treatment set forth in paragraphs 2 and 3 of this article.

Article 38

Cooperation between national authorities

Each State Party shall take such measures as may be necessary to encourage, in accordance with its domestic law, cooperation between, on the one hand, its public authorities, as well as its public officials, and, on the other hand, its authorities responsible for investigating and prosecuting criminal offences. Such cooperation may include:

(a) Informing the latter authorities, on their own initiative, where there are reasonable grounds to believe that any of the offences established in accordance with articles 15, 21 and 23 of this Convention has been committed; or

(b) Providing, upon request, to the latter authorities all necessary information.

Article 39

Cooperation between national authorities and the private sector

1. Each State Party shall take such measures as may be necessary to encourage, in accordance with its domestic law, cooperation between national investigating and prosecuting authorities and entities of the private sector, in particular financial institutions, relating to matters involving the commission of offences established in accordance with this Convention.

2. Each State Party shall consider encouraging its nationals and other persons with a habitual residence in its territory to report to the national investigating and prosecuting authorities the commission of an offence established in accordance with this Convention.

Article 40 Bank secrecv

Each State Party shall ensure that, in the case of domestic criminal investigations of offences established in accordance with this Convention, there are appropriate mechanisms available within its domestic legal system to overcome obstacles that may arise out of the application of bank secrecy laws.

Article 41 Criminal record

Each State Party may adopt such legislative or other measures as may be necessary to take into consideration, under such terms as and for the purpose that it deems appropriate, any previous conviction in another State of an alleged offender for the purpose of using such information in criminal proceedings relating to an offence established in accordance with this Convention.

Article 42

Jurisdiction

1. Each State Party shall adopt such measures as may be necessary to establish its jurisdiction over the offences established in accordance with this Convention when:

(a) The offence is committed in the territory of that State Party; or

(b) The offence is committed on board a vessel that is flying the flag of that State Party or an aircraft that is registered under the laws of that State Party at the time that the offence is committed.

2. Subject to article 4 of this Convention, a State Party may also establish its jurisdiction over any such offence when:

(a) The offence is committed against a national of that State Party; or

(b) The offence is committed by a national of that State Party or a stateless person who has his or her habitual residence in its territory; or

(c) The offence is one of those established in accordance with article 23, paragraph 1 (b) (ii), of this Convention and is committed outside its territory with a

view to the commission of an offence established in accordance with article 23, paragraph 1 (a) (i) or (ii) or (b) (i), of this Convention within its territory; or

(d) The offence is committed against the State Party.

3. For the purposes of article 44 of this Convention, each State Party shall take such measures as may be necessary to establish its jurisdiction over the offences established in accordance with this Convention when the alleged offender is present in its territory and it does not extradite such person solely on the ground that he or she is one of its nationals.

4. Each State Party may also take such measures as may be necessary to establish its jurisdiction over the offences established in accordance with this Convention when the alleged offender is present in its territory and it does not extradite him or her.

5. If a State Party exercising its jurisdiction under paragraph 1 or 2 of this article has been notified, or has otherwise learned, that any other States Parties are conducting an investigation, prosecution or judicial proceeding in respect of the same conduct, the competent authorities of those States Parties shall, as appropriate, consult one another with a view to coordinating their actions.

6. Without prejudice to norms of general international law, this Convention shall not exclude the exercise of any criminal jurisdiction established by a State Party in accordance with its domestic law.

Chapter IV International cooperation

Article 43

International cooperation

1. States Parties shall cooperate in criminal matters in accordance with articles 44 to 50 of this Convention. Where appropriate and consistent with their domestic legal system, States Parties shall consider assisting each other in investigations of and proceedings in civil and administrative matters relating to corruption.

2. In matters of international cooperation, whenever dual criminality is considered a requirement, it shall be deemed fulfilled irrespective of whether the laws of the requested State Party place the offence within the same category of offence or denominate the offence by the same terminology as the requesting State Party, if the conduct underlying the offence for which assistance is sought is a criminal offence under the laws of both States Parties.

Article 44 Extradition

1. This article shall apply to the offences established in accordance with this Convention where the person who is the subject of the request for extradition is present in the territory of the requested State Party, provided that the offence for which extradition is sought is punishable under the domestic law of both the requesting State Party and the requested State Party.

2. Notwithstanding the provisions of paragraph 1 of this article, a State Party whose law so permits may grant the extradition of a person for any of the offences covered by this Convention that are not punishable under its own domestic law.

3. If the request for extradition includes several separate offences, at least one of which is extraditable under this article and some of which are not extraditable by reason of their period of imprisonment but are related to offences established in accordance with this Convention, the requested State Party may apply this article also in respect of those offences.

4. Each of the offences to which this article applies shall be deemed to be included as an extraditable offence in any extradition treaty existing between States Parties. States Parties undertake to include such offences as extraditable offences in every extradition treaty to be concluded between them. A State Party whose law so permits, in case it uses this Convention as the basis for extradition, shall not consider any of the offences established in accordance with this Convention to be a political offence.

5. If a State Party that makes extradition conditional on the existence of a treaty receives a request for extradition from another State Party with which it has no extradition treaty, it may consider this Convention the legal basis for extradition in respect of any offence to which this article applies.

6. A State Party that makes extradition conditional on the existence of a treaty shall:

(a) At the time of deposit of its instrument of ratification, acceptance or approval of or accession to this Convention, inform the Secretary-General of the United Nations whether it will take this Convention as the legal basis for cooperation on extradition with other States Parties to this Convention; and

(b) If it does not take this Convention as the legal basis for cooperation on extradition, seek, where appropriate, to conclude treaties on extradition with other States Parties to this Convention in order to implement this article.

7. States Parties that do not make extradition conditional on the existence of a treaty shall recognize offences to which this article applies as extraditable offences between themselves.

8. Extradition shall be subject to the conditions provided for by the domestic law of the requested State Party or by applicable extradition treaties, including, inter alia, conditions in relation to the minimum penalty requirement for extradition and the grounds upon which the requested State Party may refuse extradition.

9. States Parties shall, subject to their domestic law, endeavour to expedite extradition procedures and to simplify evidentiary requirements relating thereto in respect of any offence to which this article applies.

10. Subject to the provisions of its domestic law and its extradition treaties, the requested State Party may, upon being satisfied that the circumstances so warrant and are urgent and at the request of the requesting State Party, take a person whose extradition is sought and who is present in its territory into custody or take other appropriate measures to ensure his or her presence at extradition proceedings.

11. A State Party in whose territory an alleged offender is found, if it does not extradite such person in respect of an offence to which this article applies solely on the ground that he or she is one of its nationals, shall, at the request of the State Party seeking extradition, be obliged to submit the case without undue delay to its competent authorities for the purpose of prosecution. Those authorities shall take their decision and conduct their proceedings in the same manner as in the case of any other offence of a grave nature under the domestic law of that State Party. The States Parties concerned shall cooperate with each other, in particular on procedural and evidentiary aspects, to ensure the efficiency of such prosecution.

12. Whenever a State Party is permitted under its domestic law to extradite or otherwise surrender one of its nationals only upon the condition that the person will be returned to that State Party to serve the sentence imposed as a result of the trial or proceedings for which the extradition or surrender of the person was sought and that State Party and the State Party seeking the extradition of the person agree with this option and other terms that they may deem appropriate, such conditional extradition or surrender shall be sufficient to discharge the obligation set forth in paragraph 11 of this article.

13. If extradition, sought for purposes of enforcing a sentence, is refused because the person sought is a national of the requested State Party, the requested State Party shall, if its domestic law so permits and in conformity with the requirements of such law, upon application of the requesting State Party, consider the enforcement of the sentence imposed under the domestic law of the requesting State Party or the remainder thereof.

14. Any person regarding whom proceedings are being carried out in connection with any of the offences to which this article applies shall be guaranteed fair treatment at all stages of the proceedings, including enjoyment of all the rights and guarantees provided by the domestic law of the State Party in the territory of which that person is present.

15. Nothing in this Convention shall be interpreted as imposing an obligation to extradite if the requested State Party has substantial grounds for believing that the request has been made for the purpose of prosecuting or punishing a person on account of that person's sex, race, religion, nationality, ethnic origin or political opinions or that compliance with the request would cause prejudice to that person's position for any one of these reasons.

16. States Parties may not refuse a request for extradition on the sole ground that the offence is also considered to involve fiscal matters.

17. Before refusing extradition, the requested State Party shall, where appropriate, consult with the requesting State Party to provide it with ample opportunity to present its opinions and to provide information relevant to its allegation.

18. States Parties shall seek to conclude bilateral and multilateral agreements or arrangements to carry out or to enhance the effectiveness of extradition.

Article 45 Transfer of sentenced persons

States Parties may consider entering into bilateral or multilateral agreements or arrangements on the transfer to their territory of persons sentenced to imprisonment or other forms of deprivation of liberty for offences established in accordance with this Convention in order that they may complete their sentences there.

Article 46 Mutual legal assistance

1. States Parties shall afford one another the widest measure of mutual legal assistance in investigations, prosecutions and judicial proceedings in relation to the offences covered by this Convention.

2. Mutual legal assistance shall be afforded to the fullest extent possible under relevant laws, treaties, agreements and arrangements of the requested State Party with respect to investigations, prosecutions and judicial proceedings in relation to the offences for which a legal person may be held liable in accordance with article 26 of this Convention in the requesting State Party.

3. Mutual legal assistance to be afforded in accordance with this article may be requested for any of the following purposes:

- (a) Taking evidence or statements from persons;
- (b) Effecting service of judicial documents;
- (c) Executing searches and seizures, and freezing;
- (d) Examining objects and sites;
- (e) Providing information, evidentiary items and expert evaluations;

(f) Providing originals or certified copies of relevant documents and records, including government, bank, financial, corporate or business records;

(g) Identifying or tracing proceeds of crime, property, instrumentalities or other things for evidentiary purposes;

(h) Facilitating the voluntary appearance of persons in the requesting State Party;

(i) Any other type of assistance that is not contrary to the domestic law of the requested State Party;

(j) Identifying, freezing and tracing proceeds of crime in accordance with the provisions of chapter V of this Convention;

(k) The recovery of assets, in accordance with the provisions of chapter V of this Convention.

4. Without prejudice to domestic law, the competent authorities of a State Party may, without prior request, transmit information relating to criminal matters to a competent authority in another State Party where they believe that such information could assist the authority in undertaking or successfully concluding inquiries and criminal proceedings or could result in a request formulated by the latter State Party pursuant to this Convention.

5. The transmission of information pursuant to paragraph 4 of this article shall be without prejudice to inquiries and criminal proceedings in the State of the competent authorities providing the information. The competent authorities receiving the information shall comply with a request that said information remain confidential, even temporarily, or with restrictions on its use. However, this shall not prevent the receiving State Party from disclosing in its proceedings information that is exculpatory to an accused person. In such a case, the receiving State Party shall notify the transmitting State Party prior to the disclosure and, if so requested, consult with the transmitting State Party. If, in an exceptional case, advance notice is not possible, the receiving State Party shall inform the transmitting State Party of the disclosure without delay.

6. The provisions of this article shall not affect the obligations under any other treaty, bilateral or multilateral, that governs or will govern, in whole or in part, mutual legal assistance.

7. Paragraphs 9 to 29 of this article shall apply to requests made pursuant to this article if the States Parties in question are not bound by a treaty of mutual legal assistance. If those States Parties are bound by such a treaty, the corresponding provisions of that treaty shall apply unless the States Parties agree to apply paragraphs 9 to 29 of this article in lieu thereof. States Parties are strongly encouraged to apply those paragraphs if they facilitate cooperation.

8. States Parties shall not decline to render mutual legal assistance pursuant to this article on the ground of bank secrecy.

9. (a) A requested State Party, in responding to a request for assistance pursuant to this article in the absence of dual criminality, shall take into account the purposes of this Convention, as set forth in article 1;

(b) States Parties may decline to render assistance pursuant to this article on the ground of absence of dual criminality. However, a requested State Party shall, where consistent with the basic concepts of its legal system, render assistance that does not involve coercive action. Such assistance may be refused when requests involve matters of a *de minimis* nature or matters for which the cooperation or assistance sought is available under other provisions of this Convention;

(c) Each State Party may consider adopting such measures as may be necessary to enable it to provide a wider scope of assistance pursuant to this article in the absence of dual criminality.

10. A person who is being detained or is serving a sentence in the territory of one State Party whose presence in another State Party is requested for purposes of identification, testimony or otherwise providing assistance in obtaining evidence for investigations, prosecutions or judicial proceedings in relation to offences covered by this Convention may be transferred if the following conditions are met:

(a) The person freely gives his or her informed consent;

(b) The competent authorities of both States Parties agree, subject to such conditions as those States Parties may deem appropriate.

11. For the purposes of paragraph 10 of this article:

(a) The State Party to which the person is transferred shall have the authority and obligation to keep the person transferred in custody, unless otherwise requested or authorized by the State Party from which the person was transferred;

(b) The State Party to which the person is transferred shall without delay implement its obligation to return the person to the custody of the State Party from which the person was transferred as agreed beforehand, or as otherwise agreed, by the competent authorities of both States Parties;

(c) The State Party to which the person is transferred shall not require the State Party from which the person was transferred to initiate extradition proceedings for the return of the person;

(d) The person transferred shall receive credit for service of the sentence being served in the State from which he or she was transferred for time spent in the custody of the State Party to which he or she was transferred.

12. Unless the State Party from which a person is to be transferred in accordance with paragraphs 10 and 11 of this article so agrees, that person, whatever his or her nationality, shall not be prosecuted, detained, punished or subjected to any other restriction of his or her personal liberty in the territory of the State to which that person is transferred in respect of acts, omissions or convictions prior to his or her departure from the territory of the State from which he or she was transferred.

Each State Party shall designate a central authority that shall have the 13. responsibility and power to receive requests for mutual legal assistance and either to execute them or to transmit them to the competent authorities for execution. Where a State Party has a special region or territory with a separate system of mutual legal assistance, it may designate a distinct central authority that shall have the same function for that region or territory. Central authorities shall ensure the speedy and proper execution or transmission of the requests received. Where the central authority transmits the request to a competent authority for execution, it shall encourage the speedy and proper execution of the request by the competent authority. The Secretary-General of the United Nations shall be notified of the central authority designated for this purpose at the time each State Party deposits its instrument of ratification, acceptance or approval of or accession to this Convention. Requests for mutual legal assistance and any communication related thereto shall be transmitted to the central authorities designated by the States Parties. This requirement shall be without prejudice to the right of a State Party to require that such requests and communications be addressed to it through diplomatic channels and, in urgent circumstances, where the States Parties agree, through the International Criminal Police Organization, if possible.

14. Requests shall be made in writing or, where possible, by any means capable of producing a written record, in a language acceptable to the requested State Party, under conditions allowing that State Party to establish authenticity. The Secretary-General of the United Nations shall be notified of the language or languages acceptable to each State Party at the time it deposits its instrument of ratification, acceptance or approval of or accession to this Convention. In urgent

circumstances and where agreed by the States Parties, requests may be made orally but shall be confirmed in writing forthwith.

- 15. A request for mutual legal assistance shall contain:
- (a) The identity of the authority making the request;

(b) The subject matter and nature of the investigation, prosecution or judicial proceeding to which the request relates and the name and functions of the authority conducting the investigation, prosecution or judicial proceeding;

(c) A summary of the relevant facts, except in relation to requests for the purpose of service of judicial documents;

(d) A description of the assistance sought and details of any particular procedure that the requesting State Party wishes to be followed;

(e) Where possible, the identity, location and nationality of any person concerned; and

(f) The purpose for which the evidence, information or action is sought.

16. The requested State Party may request additional information when it appears necessary for the execution of the request in accordance with its domestic law or when it can facilitate such execution.

17. A request shall be executed in accordance with the domestic law of the requested State Party and, to the extent not contrary to the domestic law of the requested State Party and where possible, in accordance with the procedures specified in the request.

18. Wherever possible and consistent with fundamental principles of domestic law, when an individual is in the territory of a State Party and has to be heard as a witness or expert by the judicial authorities of another State Party, the first State Party may, at the request of the other, permit the hearing to take place by video conference if it is not possible or desirable for the individual in question to appear in person in the territory of the requesting State Party. States Parties may agree that the hearing shall be conducted by a judicial authority of the requesting State Party.

19. The requesting State Party shall not transmit or use information or evidence furnished by the requested State Party for investigations, prosecutions or judicial proceedings other than those stated in the request without the prior consent of the requested State Party. Nothing in this paragraph shall prevent the requesting State Party from disclosing in its proceedings information or evidence that is exculpatory to an accused person. In the latter case, the requesting State Party shall notify the requested State Party prior to the disclosure and, if so requested, consult with the requested State Party. If, in an exceptional case, advance notice is not possible, the requesting State Party shall inform the requested State Party of the disclosure without delay.

20. The requesting State Party may require that the requested State Party keep confidential the fact and substance of the request, except to the extent necessary to execute the request. If the requested State Party cannot comply with the requirement of confidentiality, it shall promptly inform the requesting State Party.

21. Mutual legal assistance may be refused:

(a) If the request is not made in conformity with the provisions of this article;

(b) If the requested State Party considers that execution of the request is likely to prejudice its sovereignty, security, *ordre public* or other essential interests;

(c) If the authorities of the requested State Party would be prohibited by its domestic law from carrying out the action requested with regard to any similar offence, had it been subject to investigation, prosecution or judicial proceedings under their own jurisdiction;

(d) If it would be contrary to the legal system of the requested State Party relating to mutual legal assistance for the request to be granted.

22. States Parties may not refuse a request for mutual legal assistance on the sole ground that the offence is also considered to involve fiscal matters.

23. Reasons shall be given for any refusal of mutual legal assistance.

24. The requested State Party shall execute the request for mutual legal assistance as soon as possible and shall take as full account as possible of any deadlines suggested by the requesting State Party and for which reasons are given, preferably in the request. The requesting State Party may make reasonable requests for information on the status and progress of measures taken by the requested State Party to satisfy its request. The requested State Party shall respond to reasonable requests by the requesting State Party on the status, and progress in its handling, of the request. The requesting State Party shall promptly inform the requested State Party when the assistance sought is no longer required.

25. Mutual legal assistance may be postponed by the requested State Party on the ground that it interferes with an ongoing investigation, prosecution or judicial proceeding.

26. Before refusing a request pursuant to paragraph 21 of this article or postponing its execution pursuant to paragraph 25 of this article, the requested State Party shall consult with the requesting State Party to consider whether assistance may be granted subject to such terms and conditions as it deems necessary. If the requesting State Party accepts assistance subject to those conditions, it shall comply with the conditions.

27. Without prejudice to the application of paragraph 12 of this article, a witness, expert or other person who, at the request of the requesting State Party, consents to give evidence in a proceeding or to assist in an investigation, prosecution or judicial proceeding in the territory of the requesting State Party shall not be prosecuted, detained, punished or subjected to any other restriction of his or her personal liberty in that territory in respect of acts, omissions or convictions prior to his or her departure from the territory of the requested State Party. Such safe conduct shall cease when the witness, expert or other person having had, for a period of fifteen consecutive days or for any period agreed upon by the States Parties from the date on which he or she has been officially informed that his or her presence is no longer required by the judicial authorities, an opportunity of leaving, has nevertheless remained voluntarily in the territory of the requesting State Party or, having left it, has returned of his or her own free will.

28. The ordinary costs of executing a request shall be borne by the requested State Party, unless otherwise agreed by the States Parties concerned. If expenses of a substantial or extraordinary nature are or will be required to fulfil the request, the States Parties shall consult to determine the terms and conditions under which the request will be executed, as well as the manner in which the costs shall be borne.

29. The requested State Party:

(a) Shall provide to the requesting State Party copies of government records, documents or information in its possession that under its domestic law are available to the general public;

(b) May, at its discretion, provide to the requesting State Party in whole, in part or subject to such conditions as it deems appropriate, copies of any government records, documents or information in its possession that under its domestic law are not available to the general public.

30. States Parties shall consider, as may be necessary, the possibility of concluding bilateral or multilateral agreements or arrangements that would serve the purposes of, give practical effect to or enhance the provisions of this article.

Article 47 Transfer of criminal proceedings

States Parties shall consider the possibility of transferring to one another proceedings for the prosecution of an offence established in accordance with this Convention in cases where such transfer is considered to be in the interests of the proper administration of justice, in particular in cases where several jurisdictions are involved, with a view to concentrating the prosecution.

Article 48

Law enforcement cooperation

1. States Parties shall cooperate closely with one another, consistent with their respective domestic legal and administrative systems, to enhance the effectiveness of law enforcement action to combat the offences covered by this Convention. States Parties shall, in particular, take effective measures:

(a) To enhance and, where necessary, to establish channels of communication between their competent authorities, agencies and services in order to facilitate the secure and rapid exchange of information concerning all aspects of the offences covered by this Convention, including, if the States Parties concerned deem it appropriate, links with other criminal activities;

(b) To cooperate with other States Parties in conducting inquiries with respect to offences covered by this Convention concerning:

(i) The identity, whereabouts and activities of persons suspected of involvement in such offences or the location of other persons concerned;

(ii) The movement of proceeds of crime or property derived from the commission of such offences;

(iii) The movement of property, equipment or other instrumentalities used or intended for use in the commission of such offences;

(c) To provide, where appropriate, necessary items or quantities of substances for analytical or investigative purposes;

(d) To exchange, where appropriate, information with other States Parties concerning specific means and methods used to commit offences covered by this Convention, including the use of false identities, forged, altered or false documents and other means of concealing activities;

(e) To facilitate effective coordination between their competent authorities, agencies and services and to promote the exchange of personnel and other experts, including, subject to bilateral agreements or arrangements between the States Parties concerned, the posting of liaison officers;

(f) To exchange information and coordinate administrative and other measures taken as appropriate for the purpose of early identification of the offences covered by this Convention.

2. With a view to giving effect to this Convention, States Parties shall consider entering into bilateral or multilateral agreements or arrangements on direct cooperation between their law enforcement agencies and, where such agreements or arrangements already exist, amending them. In the absence of such agreements or arrangements between the States Parties concerned, the States Parties may consider this Convention to be the basis for mutual law enforcement cooperation in respect of the offences covered by this Convention. Whenever appropriate, States Parties shall make full use of agreements or arrangements, including international or regional organizations, to enhance the cooperation between their law enforcement agencies.

3. States Parties shall endeavour to cooperate within their means to respond to offences covered by this Convention committed through the use of modern technology.

Article 49

Joint investigations

States Parties shall consider concluding bilateral or multilateral agreements or arrangements whereby, in relation to matters that are the subject of investigations, prosecutions or judicial proceedings in one or more States, the competent authorities concerned may establish joint investigative bodies. In the absence of such agreements or arrangements, joint investigations may be undertaken by agreement on a case-by-case basis. The States Parties involved shall ensure that the sovereignty of the State Party in whose territory such investigation is to take place is fully respected.

Article 50

Special investigative techniques

1. In order to combat corruption effectively, each State Party shall, to the extent permitted by the basic principles of its domestic legal system and in accordance with the conditions prescribed by its domestic law, take such measures as may be necessary, within its means, to allow for the appropriate use by its competent authorities of controlled delivery and, where it deems appropriate, other special investigative techniques, such as electronic or other forms of surveillance

and undercover operations, within its territory, and to allow for the admissibility in court of evidence derived therefrom.

2. For the purpose of investigating the offences covered by this Convention, States Parties are encouraged to conclude, when necessary, appropriate bilateral or multilateral agreements or arrangements for using such special investigative techniques in the context of cooperation at the international level. Such agreements or arrangements shall be concluded and implemented in full compliance with the principle of sovereign equality of States and shall be carried out strictly in accordance with the terms of those agreements or arrangements.

3. In the absence of an agreement or arrangement as set forth in paragraph 2 of this article, decisions to use such special investigative techniques at the international level shall be made on a case-by-case basis and may, when necessary, take into consideration financial arrangements and understandings with respect to the exercise of jurisdiction by the States Parties concerned.

4. Decisions to use controlled delivery at the international level may, with the consent of the States Parties concerned, include methods such as intercepting and allowing the goods or funds to continue intact or be removed or replaced in whole or in part.

Chapter V Asset recovery

Article 51 General provision

The return of assets pursuant to this chapter is a fundamental principle of this Convention, and States Parties shall afford one another the widest measure of cooperation and assistance in this regard.

Article 52

Prevention and detection of transfers of proceeds of crime

1. Without prejudice to article 14 of this Convention, each State Party shall take such measures as may be necessary, in accordance with its domestic law, to require financial institutions within its jurisdiction to verify the identity of customers, to take reasonable steps to determine the identity of beneficial owners of

funds deposited into high-value accounts and to conduct enhanced scrutiny of accounts sought or maintained by or on behalf of individuals who are, or have been, entrusted with prominent public functions and their family members and close associates. Such enhanced scrutiny shall be reasonably designed to detect suspicious transactions for the purpose of reporting to competent authorities and should not be so construed as to discourage or prohibit financial institutions from doing business with any legitimate customer.

2. In order to facilitate implementation of the measures provided for in paragraph 1 of this article, each State Party, in accordance with its domestic law and inspired by relevant initiatives of regional, interregional and multilateral organizations against money-laundering, shall:

(a) Issue advisories regarding the types of natural or legal person to whose accounts financial institutions within its jurisdiction will be expected to apply enhanced scrutiny, the types of accounts and transactions to which to pay particular attention and appropriate account-opening, maintenance and record-keeping measures to take concerning such accounts; and

(b) Where appropriate, notify financial institutions within its jurisdiction, at the request of another State Party or on its own initiative, of the identity of particular natural or legal persons to whose accounts such institutions will be expected to apply enhanced scrutiny, in addition to those whom the financial institutions may otherwise identify.

3. In the context of paragraph 2 (a) of this article, each State Party shall implement measures to ensure that its financial institutions maintain adequate records, over an appropriate period of time, of accounts and transactions involving the persons mentioned in paragraph 1 of this article, which should, as a minimum, contain information relating to the identity of the customer as well as, as far as possible, of the beneficial owner.

4. With the aim of preventing and detecting transfers of proceeds of offences established in accordance with this Convention, each State Party shall implement appropriate and effective measures to prevent, with the help of its regulatory and oversight bodies, the establishment of banks that have no physical presence and that are not affiliated with a regulated financial group. Moreover, States Parties may consider requiring their financial institutions to refuse to enter into or continue a correspondent banking relationship with such institutions and to guard against establishing relations with foreign financial institutions that permit their accounts to be used by banks that have no physical presence and that are not affiliated with a regulated financial group.

5. Each State Party shall consider establishing, in accordance with its domestic law, effective financial disclosure systems for appropriate public officials and shall provide for appropriate sanctions for non-compliance. Each State Party shall also consider taking such measures as may be necessary to permit its competent authorities to share that information with the competent authorities in other States Parties when necessary to investigate, claim and recover proceeds of offences established in accordance with this Convention.

6. Each State Party shall consider taking such measures as may be necessary, in accordance with its domestic law, to require appropriate public officials having an interest in or signature or other authority over a financial account in a foreign country to report that relationship to appropriate authorities and to maintain appropriate records related to such accounts. Such measures shall also provide for appropriate sanctions for non-compliance.

Article 53

Measures for direct recovery of property

Each State Party shall, in accordance with its domestic law:

(a) Take such measures as may be necessary to permit another State Party to initiate civil action in its courts to establish title to or ownership of property

acquired through the commission of an offence established in accordance with this Convention;

(b) Take such measures as may be necessary to permit its courts to order those who have committed offences established in accordance with this Convention to pay compensation or damages to another State Party that has been harmed by such offences; and

(c) Take such measures as may be necessary to permit its courts or competent authorities, when having to decide on confiscation, to recognize another State Party's claim as a legitimate owner of property acquired through the commission of an offence established in accordance with this Convention.

Article 54 Mechanisms for recovery of property through international cooperation in confiscation

1. Each State Party, in order to provide mutual legal assistance pursuant to article 55 of this Convention with respect to property acquired through or involved in the commission of an offence established in accordance with this Convention, shall, in accordance with its domestic law:

(a) Take such measures as may be necessary to permit its competent authorities to give effect to an order of confiscation issued by a court of another State Party;

(b) Take such measures as may be necessary to permit its competent authorities, where they have jurisdiction, to order the confiscation of such property of foreign origin by adjudication of an offence of money-laundering or such other offence as may be within its jurisdiction or by other procedures authorized under its domestic law; and

(c) Consider taking such measures as may be necessary to allow confiscation of such property without a criminal conviction in cases in which the offender cannot be prosecuted by reason of death, flight or absence or in other appropriate cases.

2. Each State Party, in order to provide mutual legal assistance upon a request made pursuant to paragraph 2 of article 55 of this Convention, shall, in accordance with its domestic law:

(a) Take such measures as may be necessary to permit its competent authorities to freeze or seize property upon a freezing or seizure order issued by a court or competent authority of a requesting State Party that provides a reasonable basis for the requested State Party to believe that there are sufficient grounds for taking such actions and that the property would eventually be subject to an order of confiscation for purposes of paragraph 1 (a) of this article;

(b) Take such measures as may be necessary to permit its competent authorities to freeze or seize property upon a request that provides a reasonable basis for the requested State Party to believe that there are sufficient grounds for taking such actions and that the property would eventually be subject to an order of confiscation for purposes of paragraph 1 (a) of this article; and (c) Consider taking additional measures to permit its competent authorities to preserve property for confiscation, such as on the basis of a foreign arrest or criminal charge related to the acquisition of such property.

Article 55

International cooperation for purposes of confiscation

1. A State Party that has received a request from another State Party having jurisdiction over an offence established in accordance with this Convention for confiscation of proceeds of crime, property, equipment or other instrumentalities referred to in article 31, paragraph 1, of this Convention situated in its territory shall, to the greatest extent possible within its domestic legal system:

(a) Submit the request to its competent authorities for the purpose of obtaining an order of confiscation and, if such an order is granted, give effect to it; or

(b) Submit to its competent authorities, with a view to giving effect to it to the extent requested, an order of confiscation issued by a court in the territory of the requesting State Party in accordance with articles 31, paragraph 1, and 54, paragraph 1 (a), of this Convention insofar as it relates to proceeds of crime, property, equipment or other instrumentalities referred to in article 31, paragraph 1, situated in the territory of the requested State Party.

2. Following a request made by another State Party having jurisdiction over an offence established in accordance with this Convention, the requested State Party shall take measures to identify, trace and freeze or seize proceeds of crime, property, equipment or other instrumentalities referred to in article 31, paragraph 1, of this Convention for the purpose of eventual confiscation to be ordered either by the requesting State Party or, pursuant to a request under paragraph 1 of this article, by the requested State Party.

3. The provisions of article 46 of this Convention are applicable, mutatis mutandis, to this article. In addition to the information specified in article 46, paragraph 15, requests made pursuant to this article shall contain:

(a) In the case of a request pertaining to paragraph 1 (a) of this article, a description of the property to be confiscated, including, to the extent possible, the location and, where relevant, the estimated value of the property and a statement of the facts relied upon by the requesting State Party sufficient to enable the requested State Party to seek the order under its domestic law;

(b) In the case of a request pertaining to paragraph 1 (b) of this article, a legally admissible copy of an order of confiscation upon which the request is based issued by the requesting State Party, a statement of the facts and information as to the extent to which execution of the order is requested, a statement specifying the measures taken by the requesting State Party to provide adequate notification to bona fide third parties and to ensure due process and a statement that the confiscation order is final;

(c) In the case of a request pertaining to paragraph 2 of this article, a statement of the facts relied upon by the requesting State Party and a description of

the actions requested and, where available, a legally admissible copy of an order on which the request is based.

4. The decisions or actions provided for in paragraphs 1 and 2 of this article shall be taken by the requested State Party in accordance with and subject to the provisions of its domestic law and its procedural rules or any bilateral or multilateral agreement or arrangement to which it may be bound in relation to the requesting State Party.

5. Each State Party shall furnish copies of its laws and regulations that give effect to this article and of any subsequent changes to such laws and regulations or a description thereof to the Secretary-General of the United Nations.

6. If a State Party elects to make the taking of the measures referred to in paragraphs 1 and 2 of this article conditional on the existence of a relevant treaty, that State Party shall consider this Convention the necessary and sufficient treaty basis.

7. Cooperation under this article may also be refused or provisional measures lifted if the requested State Party does not receive sufficient and timely evidence or if the property is of a *de minimis* value.

8. Before lifting any provisional measure taken pursuant to this article, the requested State Party shall, wherever possible, give the requesting State Party an opportunity to present its reasons in favour of continuing the measure.

9. The provisions of this article shall not be construed as prejudicing the rights of bona fide third parties.

Article 56

Special cooperation

Without prejudice to its domestic law, each State Party shall endeavour to take measures to permit it to forward, without prejudice to its own investigations, prosecutions or judicial proceedings, information on proceeds of offences established in accordance with this Convention to another State Party without prior request, when it considers that the disclosure of such information might assist the receiving State Party in initiating or carrying out investigations, prosecutions or judicial proceedings or might lead to a request by that State Party under this chapter of the Convention.

Article 57

Return and disposal of assets

1. Property confiscated by a State Party pursuant to article 31 or 55 of this Convention shall be disposed of, including by return to its prior legitimate owners, pursuant to paragraph 3 of this article, by that State Party in accordance with the provisions of this Convention and its domestic law.

2. Each State Party shall adopt such legislative and other measures, in accordance with the fundamental principles of its domestic law, as may be necessary to enable its competent authorities to return confiscated property, when acting on the request made by another State Party, in accordance with this Convention, taking into account the rights of bona fide third parties.

3. In accordance with articles 46 and 55 of this Convention and paragraphs 1 and 2 of this article, the requested State Party shall:

(a) In the case of embezzlement of public funds or of laundering of embezzled public funds as referred to in articles 17 and 23 of this Convention, when confiscation was executed in accordance with article 55 and on the basis of a final judgement in the requesting State Party, a requirement that can be waived by the requested State Party, return the confiscated property to the requesting State Party;

(b) In the case of proceeds of any other offence covered by this Convention, when the confiscation was executed in accordance with article 55 of this Convention and on the basis of a final judgement in the requesting State Party, a requirement that can be waived by the requested State Party, return the confiscated property to the requesting State Party, when the requesting State Party reasonably establishes its prior ownership of such confiscated property to the requested State Party or when the requested State Party recognizes damage to the requesting State Party as a basis for returning the confiscated property;

(c) In all other cases, give priority consideration to returning confiscated property to the requesting State Party, returning such property to its prior legitimate owners or compensating the victims of the crime.

4. Where appropriate, unless States Parties decide otherwise, the requested State Party may deduct reasonable expenses incurred in investigations, prosecutions or judicial proceedings leading to the return or disposition of confiscated property pursuant to this article.

5. Where appropriate, States Parties may also give special consideration to concluding agreements or mutually acceptable arrangements, on a case-by-case basis, for the final disposal of confiscated property.

Article 58

Financial intelligence unit

States Parties shall cooperate with one another for the purpose of preventing and combating the transfer of proceeds of offences established in accordance with this Convention and of promoting ways and means of recovering such proceeds and, to that end, shall consider establishing a financial intelligence unit to be responsible for receiving, analysing and disseminating to the competent authorities reports of suspicious financial transactions.

Article 59

Bilateral and multilateral agreements and arrangements

States Parties shall consider concluding bilateral or multilateral agreements or arrangements to enhance the effectiveness of international cooperation undertaken pursuant to this chapter of the Convention.

Chapter VI Technical assistance and information exchange

Article 60 Training and technical assistance

1. Each State Party shall, to the extent necessary, initiate, develop or improve specific training programmes for its personnel responsible for preventing and combating corruption. Such training programmes could deal, inter alia, with the following areas:

(a) Effective measures to prevent, detect, investigate, punish and control corruption, including the use of evidence-gathering and investigative methods;

(b) Building capacity in the development and planning of strategic anticorruption policy;

(c) Training competent authorities in the preparation of requests for mutual legal assistance that meet the requirements of this Convention;

(d) Evaluation and strengthening of institutions, public service management and the management of public finances, including public procurement, and the private sector;

(e) Preventing and combating the transfer of proceeds of offences established in accordance with this Convention and recovering such proceeds;

(f) Detecting and freezing of the transfer of proceeds of offences established in accordance with this Convention;

(g) Surveillance of the movement of proceeds of offences established in accordance with this Convention and of the methods used to transfer, conceal or disguise such proceeds;

(h) Appropriate and efficient legal and administrative mechanisms and methods for facilitating the return of proceeds of offences established in accordance with this Convention;

(i) Methods used in protecting victims and witnesses who cooperate with judicial authorities; and

(j) Training in national and international regulations and in languages.

2. States Parties shall, according to their capacity, consider affording one another the widest measure of technical assistance, especially for the benefit of developing countries, in their respective plans and programmes to combat corruption, including material support and training in the areas referred to in

paragraph 1 of this article, and training and assistance and the mutual exchange of relevant experience and specialized knowledge, which will facilitate international cooperation between States Parties in the areas of extradition and mutual legal assistance.

3. States Parties shall strengthen, to the extent necessary, efforts to maximize operational and training activities in international and regional organizations and in the framework of relevant bilateral and multilateral agreements or arrangements.

4. States Parties shall consider assisting one another, upon request, in conducting evaluations, studies and research relating to the types, causes, effects

and costs of corruption in their respective countries, with a view to developing, with the participation of competent authorities and society, strategies and action plans to combat corruption.

5. In order to facilitate the recovery of proceeds of offences established in accordance with this Convention, States Parties may cooperate in providing each other with the names of experts who could assist in achieving that objective.

6. States Parties shall consider using subregional, regional and international conferences and seminars to promote cooperation and technical assistance and to stimulate discussion on problems of mutual concern, including the special problems and needs of developing countries and countries with economies in transition.

7. States Parties shall consider establishing voluntary mechanisms with a view to contributing financially to the efforts of developing countries and countries with economies in transition to apply this Convention through technical assistance programmes and projects.

8. Each State Party shall consider making voluntary contributions to the United Nations Office on Drugs and Crime for the purpose of fostering, through the Office, programmes and projects in developing countries with a view to implementing this Convention.

Article 61

Collection, exchange and analysis of information on corruption

1. Each State Party shall consider analysing, in consultation with experts, trends in corruption in its territory, as well as the circumstances in which corruption offences are committed.

2. States Parties shall consider developing and sharing with each other and through international and regional organizations statistics, analytical expertise concerning corruption and information with a view to developing, insofar as possible, common definitions, standards and methodologies, as well as information on best practices to prevent and combat corruption.

3. Each State Party shall consider monitoring its policies and actual measures to combat corruption and making assessments of their effectiveness and efficiency.

Article 62

Other measures: implementation of the Convention through economic development and technical assistance

1. States Parties shall take measures conducive to the optimal implementation of this Convention to the extent possible, through international cooperation, taking into account the negative effects of corruption on society in general, in particular on sustainable development.

2. States Parties shall make concrete efforts to the extent possible and in coordination with each other, as well as with international and regional organizations:

(a) To enhance their cooperation at various levels with developing countries, with a view to strengthening the capacity of the latter to prevent and combat corruption;

(b) To enhance financial and material assistance to support the efforts of developing countries to prevent and fight corruption effectively and to help them implement this Convention successfully;

(c) To provide technical assistance to developing countries and countries with economies in transition to assist them in meeting their needs for the implementation of this Convention. To that end, States Parties shall endeavour to make adequate and regular voluntary contributions to an account specifically designated for that purpose in a United Nations funding mechanism. States Parties may also give special consideration, in accordance with their domestic law and the provisions of this Convention, to contributing to that account a percentage of the money or of the corresponding value of proceeds of crime or property confiscated in accordance with the provisions of this Convention;

(d) To encourage and persuade other States and financial institutions as appropriate to join them in efforts in accordance with this article, in particular by providing more training programmes and modern equipment to developing countries in order to assist them in achieving the objectives of this Convention.

3. To the extent possible, these measures shall be without prejudice to existing foreign assistance commitments or to other financial cooperation arrangements at the bilateral, regional or international level.

4. States Parties may conclude bilateral or multilateral agreements or arrangements on material and logistical assistance, taking into consideration the financial arrangements necessary for the means of international cooperation provided for by this Convention to be effective and for the prevention, detection and control of corruption.

Chapter VII Mechanisms for implementation

Article 63

Conference of the States Parties to the Convention

1. A Conference of the States Parties to the Convention is hereby established to improve the capacity of and cooperation between States Parties to achieve the objectives set forth in this Convention and to promote and review its implementation.

2. The Secretary-General of the United Nations shall convene the Conference of the States Parties not later than one year following the entry into force of this Convention. Thereafter, regular meetings of the Conference of the States Parties shall be held in accordance with the rules of procedure adopted by the Conference.

3. The Conference of the States Parties shall adopt rules of procedure and rules governing the functioning of the activities set forth in this article, including

rules concerning the admission and participation of observers, and the payment of expenses incurred in carrying out those activities.

4. The Conference of the States Parties shall agree upon activities, procedures and methods of work to achieve the objectives set forth in paragraph 1 of this article, including:

(a) Facilitating activities by States Parties under articles 60 and 62 and chapters II to V of this Convention, including by encouraging the mobilization of voluntary contributions;

(b) Facilitating the exchange of information among States Parties on patterns and trends in corruption and on successful practices for preventing and combating it and for the return of proceeds of crime, through, inter alia, the publication of relevant information as mentioned in this article;

(c) Cooperating with relevant international and regional organizations and mechanisms and non-governmental organizations;

(d) Making appropriate use of relevant information produced by other international and regional mechanisms for combating and preventing corruption in order to avoid unnecessary duplication of work;

(e) Reviewing periodically the implementation of this Convention by its States Parties;

(f) Making recommendations to improve this Convention and its implementation;

(g) Taking note of the technical assistance requirements of States Parties with regard to the implementation of this Convention and recommending any action it may deem necessary in that respect.

5. For the purpose of paragraph 4 of this article, the Conference of the States Parties shall acquire the necessary knowledge of the measures taken by States

Parties in implementing this Convention and the difficulties encountered by them in doing so through information provided by them and through such supplemental review mechanisms as may be established by the Conference of the States Parties.

6. Each State Party shall provide the Conference of the States Parties with information on its programmes, plans and practices, as well as on legislative and administrative measures to implement this Convention, as required by the Conference of the States Parties. The Conference of the States Parties shall examine the most effective way of receiving and acting upon information, including, inter alia, information received from States Parties and from competent international organizations. Inputs received from relevant non-governmental organizations duly accredited in accordance with procedures to be decided upon by the Conference of the States Parties may also be considered.

7. Pursuant to paragraphs 4 to 6 of this article, the Conference of the States Parties shall establish, if it deems it necessary, any appropriate mechanism or body to assist in the effective implementation of the Convention.

> Article 64 Secretariat

1. The Secretary-General of the United Nations shall provide the necessary secretariat services to the Conference of the States Parties to the Convention.

2. The secretariat shall:

(a) Assist the Conference of the States Parties in carrying out the activities set forth in article 63 of this Convention and make arrangements and provide the necessary services for the sessions of the Conference of the States Parties;

(b) Upon request, assist States Parties in providing information to the Conference of the States Parties as envisaged in article 63, paragraphs 5 and 6, of this Convention; and

(c) Ensure the necessary coordination with the secretariats of relevant international and regional organizations.

Chapter VIII Final provisions

Article 65

Implementation of the Convention

1. Each State Party shall take the necessary measures, including legislative and administrative measures, in accordance with fundamental principles of its domestic law, to ensure the implementation of its obligations under this Convention.

2. Each State Party may adopt more strict or severe measures than those provided for by this Convention for preventing and combating corruption.

Article 66

Settlement of disputes

1. States Parties shall endeavour to settle disputes concerning the interpretation or application of this Convention through negotiation.

2. Any dispute between two or more States Parties concerning the interpretation or application of this Convention that cannot be settled through negotiation within a reasonable time shall, at the request of one of those States Parties, be submitted to arbitration. If, six months after the date of the request for arbitration, those States Parties are unable to agree on the organization of the arbitration, any one of those States Parties may refer the dispute to the International Court of Justice by request in accordance with the Statute of the Court.

3. Each State Party may, at the time of signature, ratification, acceptance or approval of or accession to this Convention, declare that it does not consider itself bound by paragraph 2 of this article. The other States Parties shall not be bound by paragraph 2 of this article with respect to any State Party that has made such a reservation.

4. Any State Party that has made a reservation in accordance with paragraph 3 of this article may at any time withdraw that reservation by notification to the Secretary-General of the United Nations.

Article 67

Signature, ratification, acceptance, approval and accession

1. This Convention shall be open to all States for signature from 9 to 11 December 2003 in Merida, Mexico, and thereafter at United Nations Headquarters in New York until 9 December 2005.

2. This Convention shall also be open for signature by regional economic integration organizations provided that at least one member State of such organization has signed this Convention in accordance with paragraph 1 of this article.

3. This Convention is subject to ratification, acceptance or approval. Instruments of ratification, acceptance or approval shall be deposited with the Secretary-General of the United Nations. A regional economic integration organization may deposit its instrument of ratification, acceptance or approval if at least one of its member States has done likewise. In that instrument of ratification, acceptance or approval, such organization shall declare the extent of its competence with respect to the matters governed by this Convention. Such organization shall also inform the depositary of any relevant modification in the extent of its competence.

4. This Convention is open for accession by any State or any regional economic integration organization of which at least one member State is a Party to this Convention. Instruments of accession shall be deposited with the Secretary-General of the United Nations. At the time of its accession, a regional economic integration organization shall declare the extent of its competence with respect to matters governed by this Convention. Such organization shall also inform the depositary of any relevant modification in the extent of its competence.

Article 68

Entry into force

1. This Convention shall enter into force on the ninetieth day after the date of deposit of the thirtieth instrument of ratification, acceptance, approval or accession. For the purpose of this paragraph, any instrument deposited by a regional economic integration organization shall not be counted as additional to those deposited by member States of such organization.

2. For each State or regional economic integration organization ratifying, accepting, approving or acceding to this Convention after the deposit of the thirtieth instrument of such action, this Convention shall enter into force on the thirtieth day after the date of deposit by such State or organization of the relevant instrument or on the date this Convention enters into force pursuant to paragraph 1 of this article, whichever is later.

Article 69

Amendment

1. After the expiry of five years from the entry into force of this Convention, a State Party may propose an amendment and transmit it to the Secretary-General of the United Nations, who shall thereupon communicate the proposed amendment to the States Parties and to the Conference of the States Parties to the Convention for the purpose of considering and deciding on the proposal. The Conference of the States Parties shall make every effort to achieve consensus on each amendment. If all efforts at consensus have been exhausted and no agreement has been reached, the amendment shall, as a last resort, require for its adoption a two-thirds majority vote of the States Parties present and voting at the meeting of the Conference of the States Parties.

2. Regional economic integration organizations, in matters within their competence, shall exercise their right to vote under this article with a number of votes equal to the number of their member States that are Parties to this Convention. Such organizations shall not exercise their right to vote if their member States exercise theirs and vice versa.

3. An amendment adopted in accordance with paragraph 1 of this article is subject to ratification, acceptance or approval by States Parties.

4. An amendment adopted in accordance with paragraph 1 of this article shall enter into force in respect of a State Party ninety days after the date of the deposit with the Secretary-General of the United Nations of an instrument of ratification, acceptance or approval of such amendment.

5. When an amendment enters into force, it shall be binding on those States Parties which have expressed their consent to be bound by it. Other States Parties shall still be bound by the provisions of this Convention and any earlier amendments that they have ratified, accepted or approved.

Article 70

Denunciation

1. A State Party may denounce this Convention by written notification to the Secretary-General of the United Nations. Such denunciation shall become effective one year after the date of receipt of the notification by the Secretary-General.

2. A regional economic integration organization shall cease to be a Party to this Convention when all of its member States have denounced it.

Article 71

Depositary and languages

1. The Secretary-General of the United Nations is designated depositary of this Convention.

2. The original of this Convention, of which the Arabic, Chinese, English, French, Russian and Spanish texts are equally authentic, shall be deposited with the Secretary-General of the United Nations.

IN WITNESS WHEREOF, the undersigned plenipotentiaries, being duly authorized thereto by their respective Governments, have signed this Convention.



UNCITRAL ARBITRATION RULES (2021) UNCITRAL EXPEDITED ARBITRATION RULES UNCITRAL RULES ON TRANSPARENCY IN TREATY-BASED INVESTOR-STATE ARBITRATION



UNCITRAL Arbitration Rules (2021)

UNCITRAL Expedited Arbitration Rules

UNCITRAL Rules on Transparency in Treaty-based Investor-State Arbitration



Further information may be obtained from:
UNCITRAL secretariat, Vienna International Centre
P.O. Box 500, 1400 Vienna, AustriaTelephone: (+43-1) 26060-4060Telefax: (+43-1) 26060-5813Internet: uncitral.un.orgE-mail: uncitral@un.org

UNCITRAL Arbitration Rules

(with article 1, paragraph 4, as adopted in 2013 and article 1, paragraph 5, as adopted in 2021)

UNCITRAL Expedited Arbitration Rules

UNCITRAL

Rules on Transparency in Treaty-based Investor-State Arbitration



UNITED NATIONS Vienna, 2021

 $\ensuremath{\mathbb{O}}$ United Nations: United Nations Commission on International Trade Law. 2021. All rights reserved worldwide.

This publication has not been formally edited.

Publishing production: English, Publishing and Library Section. United Nations Office at Vienna.

Contents

	Page
General Assembly Resolution 76/108	1
UNCITRAL Arbitration Rules (with article 1, paragraph 4, as adopted in 2013 and article 1,	
paragraph 5, as adopted in 2021)	3
Section I. Introductory rules	3
Scope of application (article 1)	3
Notice and calculation of periods of time (article 2)	4
Notice of arbitration (article 3)	5
Response to the notice of arbitration (article 4)	6
Representation and assistance (article 5)	6
Designating and appointing authorities (article 6)	7
Section II. Composition of the arbitral tribunal	9
Number of arbitrators (article 7)	9
Appointment of arbitrators (articles 8 to 10)	9
Disclosures by and challenge of arbitrators (articles 11 to 13)	11
Replacement of an arbitrator (article 14)	12
Repetition of hearings in the event of the	12
replacement of an arbitrator (article 15)	13
Exclusion of liability (article 16)	13
Section III. Arbitral proceedings	14
General provisions (article 17)	14
Place of arbitration (article 18)	15
Language (article 19)	15
Statement of claim (article 20)	15
Statement of defence (article 21)	16
Amendments to the claim or defence (article 22)	17
Pleas as to the jurisdiction of the arbitral tribunal	
(article 23)	17
Further written statements (article 24)	18
Periods of time (article 25)	18
Interim measures (article 26)	18

Evidence (article 27)	19
Hearings (article 28)	20
Experts appointed by the arbitral tribunal (article 29)	20
Default (article 30)	21
Closure of hearings (article 31)	22
Waiver of right to object (article 32)	22
Section IV. The award	23
Decisions (article 33)	23
Form and effect of the award (article 34)	23
Applicable law, amiable compositeur (article 35)	24
Settlement or other grounds for termination (article 36)	24
Interpretation of the award (article 37)	25
Correction of the award (article 38)	25
Additional award (article 39)	25
Definition of costs (article 40)	26
Fees and expenses of arbitrators (article 41)	26
Allocation of costs (article 42)	28
Deposit of costs (article 43)	28
Annex	30
Model arbitration clause for contracts	30
Possible waiver statement	30
Model statements of independence pursuant to	
article 11 of the Rules	30
Appendix – UNCITRAL Expedited Arbitration Rules	32
Scope of application (articles 1 and 2)	32
Conduct of the parties and the arbitral	
tribunal (article 3)	32
Notice of arbitration and statement	
of claim (article 4)	33
Response to the notice of arbitration and statement of defence (article 5)	33
Designating and appointing authorities (article 6)	33 34
Number of arbitrators (article 7)	34
Appointment of a sole arbitrator (article 8)	34
Consultation with the parties (article 9)	34 34
• • •	54
Discretion of the arbitral tribunal with regard to periods of time (article 10)	35
Hearings (article 11)	35
	55

Counterclaims or claims for the purpose	
of set-off (article 12)	35
Amendments and supplements to a claim	25
or defence (article 13)	35 36
Further written statements (article 14)	30 36
Evidence (article 15)	
Period of time for making the award (article 16) Annex to the UNCITRAL Expedited	36
Arbitration Rules	38
Model arbitration clause for contracts	38
Model statement	38
UNCITRAL Rules on Transparency in Treaty-based	
Investor-State Arbitration	39
Article 1. Scope of application	39
Applicability of the Rules.	39
Application of the Rules.	39
Discretion and authority of the arbitral tribunal	40
Applicable instrument in case of conflict	40
Application in non-UNCITRAL arbitrations	41
Article 2. Publication of information at the	
commencement of arbitral proceedings	41
Article 3. Publication of documents	41
Article 4. Submission by a third person	42
Article 5. Submission by a non-disputing Party to	
the treaty	43
Article 6. Hearings	44
Article 7. Exceptions to transparency	44
Confidential or protected information	44
Integrity of the arbitral process	46
Article 8. Repository of published information	46
Explanatory Note to the	
UNCITRAL Expedited Arbitration Rules	47
A. Scope of application	47
B. General provision on expedited arbitration	51
C. Notice of arbitration, response thereto, statements of claim and defence	52
D. Designating and appointing authorities	

E.	Number of arbitrators	57
F.	Appointment of the arbitrator	57
G.	Consultation with the parties	59
H.	Time frames and the discretion of the arbitral tribunal	61
I.	Hearings	61
J.	Counterclaims and claims for the purpose of set-off	63
K.	Amendments and supplements to the claim or defence	63
L.	Further written statements	63
M.	Evidence	64
N.	Period of time for making of the award	64
О.	Model arbitration clause for expedited arbitration.	68
P.	The Expedited Rules and the Transparency Rules	69
Q.	Time frames in the Expedited Rules	70

Resolution adopted by the General Assembly on 9 December 2021

76/108. Expedited Arbitration Rules of the United Nations Commission on International Trade Law

The General Assembly,

Recalling its resolution 2205 (XXI) of 17 December 1966, by which it established the United Nations Commission on International Trade Law with a mandate to further the progressive harmonization and unification of the law of international trade and in that respect to bear in mind the interests of all peoples, in particular those of developing countries, in the extensive development of international trade,

Recalling also its resolution 31/98 of 15 December 1976 recommending the use of the Arbitration Rules of the United Nations Commission on International Trade Law¹ and its resolution 65/22 of 6 December 2010 recommending the use of the Arbitration Rules as revised in 2010,²

Mindful of the value of arbitration as a method of settling disputes that may arise in the context of international commercial relations,

Noting the value of expedited arbitration as a streamlined and simplified procedure for settling disputes that arise in the context of international commercial relations within a shortened time frame, and its increased use in international and domestic commercial practice for parties to reach a final resolution of the dispute in a cost- and time-effective manner,

¹ Official Records of the General Assembly, Thirty-first Session, Supplement No. 17 (A/31/17), chap. V, sect. C.

²Ibid., Sixty-fifth Session, Supplement No. 17 (A/65/17), annex I.

Aware of the need to balance the efficiency of the arbitral proceedings and the rights of the disputing parties to due process and fair treatment,

Noting that the preparation of the UNCITRAL Expedited Arbitration Rules and the accompanying explanatory note benefited greatly from consultations with Governments and interested intergovernmental and international non-governmental organizations,

Noting also that the Expedited Arbitration Rules were adopted by the United Nations Commission on International Trade Law at its fifty-fourth session, after due deliberations,³

1. *Expresses its appreciation* to the United Nations Commission on International Trade Law for having formulated and adopted the Expedited Arbitration Rules, the text of which is contained in annex IV to the report of the United Nations Commission on International Trade Law on the work of its fiftyfourth session⁴ and which came into effect on 19 September 2021;

2. *Recommends* the use of the UNCITRAL Expedited Arbitration Rules in the settlement of disputes arising in the context of international commercial relations;

3. *Requests* the Secretary-General to make all efforts to ensure that the UNCITRAL Expedited Arbitration Rules become generally known and available.

³ Ibid., Seventy-sixth Session, Supplement No. 17 (A/76/17), chap. VII. ⁴ Ibid., annex IV.

UNCITRAL Arbitration Rules (with article 1, paragraph 4, as adopted in 2013 and article 1, paragraph 5, as adopted in 2021)

Section I. Introductory rules

Scope of application*

Article 1

1. Where parties have agreed that disputes between them in respect of a defined legal relationship, whether contractual or not, shall be referred to arbitration under the UNCITRAL Arbitration Rules, then such disputes shall be settled in accordance with these Rules subject to such modification as the parties may agree.

2. The parties to an arbitration agreement concluded after 15 August 2010 shall be presumed to have referred to the Rules in effect on the date of commencement of the arbitration, unless the parties have agreed to apply a particular version of the Rules. That presumption does not apply where the arbitration agreement has been concluded by accepting after 15 August 2010 an offer made before that date.

3. These Rules shall govern the arbitration except that where any of these Rules is in conflict with a provision of the law applicable to the arbitration from which the parties cannot derogate, that provision shall prevail.

4. For investor-State arbitration initiated pursuant to a treaty providing for the protection of investments or investors, these Rules include the UNCITRAL Rules on Transparency in Treaty-based Investor-State Arbitration ("Rules on Transparency"), subject to article 1 of the Rules on Transparency.

5. The Expedited Arbitration Rules in the appendix shall apply to the arbitration where the parties so agree.

 $^{^{\}ast}A$ model arbitration clause for contracts can be found in the annex to the Rules.

Notice and calculation of periods of time

Article 2

1. A notice, including a notification, communication or proposal, may be transmitted by any means of communication that provides or allows for a record of its transmission.

2. If an address has been designated by a party specifically for this purpose or authorized by the arbitral tribunal, any notice shall be delivered to that party at that address, and if so delivered shall be deemed to have been received. Delivery by electronic means such as facsimile or e-mail may only be made to an address so designated or authorized.

3. In the absence of such designation or authorization, a notice is:

(a) Received if it is physically delivered to the addressee; or

(b) Deemed to have been received if it is delivered at the place of business, habitual residence or mailing address of the addressee.

4. If, after reasonable efforts, delivery cannot be effected in accordance with paragraphs 2 or 3, a notice is deemed to have been received if it is sent to the addressee's last-known place of business, habitual residence or mailing address by registered letter or any other means that provides a record of delivery or of attempted delivery.

5. A notice shall be deemed to have been received on the day it is delivered in accordance with paragraphs 2, 3 or 4, or attempted to be delivered in accordance with paragraph 4. A notice transmitted by electronic means is deemed to have been received on the day it is sent, except that a notice of arbitration so transmitted is only deemed to have been received on the day when it reaches the addressee's electronic address.

6. For the purpose of calculating a period of time under these Rules, such period shall begin to run on the day following the day when a notice is received. If the last day of such period is an official holiday or a non-business day at the residence or place of business of the addressee, the period is extended until the first business day which follows. Official holidays or nonbusiness days occurring during the running of the period of time are included in calculating the period.

Notice of arbitration

Article 3

1. The party or parties initiating recourse to arbitration (hereinafter called the "claimant") shall communicate to the other party or parties (hereinafter called the "respondent") a notice of arbitration.

2. Arbitral proceedings shall be deemed to commence on the date on which the notice of arbitration is received by the respondent.

3. The notice of arbitration shall include the following:

(a) A demand that the dispute be referred to arbitration;

(b) The names and contact details of the parties;

(c) Identification of the arbitration agreement that is invoked;

(d) Identification of any contract or other legal instrument out of or in relation to which the dispute arises or, in the absence of such contract or instrument, a brief description of the relevant relationship;

(e) A brief description of the claim and an indication of the amount involved, if any;

(f) The relief or remedy sought;

(g) A proposal as to the number of arbitrators, language and place of arbitration, if the parties have not previously agreed thereon.

4. The notice of arbitration may also include:

(a) A proposal for the designation of an appointing authority referred to in article 6, paragraph 1;

(b) A proposal for the appointment of a sole arbitrator referred to in article 8, paragraph 1;

(c) Notification of the appointment of an arbitrator referred to in article 9 or 10.

5. The constitution of the arbitral tribunal shall not be hindered by any controversy with respect to the sufficiency of the notice of arbitration, which shall be finally resolved by the arbitral tribunal.

Response to the notice of arbitration

Article 4

1. Within 30 days of the receipt of the notice of arbitration, the respondent shall communicate to the claimant a response to the notice of arbitration, which shall include:

(a) The name and contact details of each respondent;

(b) A response to the information set forth in the notice of arbitration, pursuant to article 3, paragraphs 3 (c) to (g).

2. The response to the notice of arbitration may also include:

(a) Any plea that an arbitral tribunal to be constituted under these Rules lacks jurisdiction;

(b) A proposal for the designation of an appointing authority referred to in article 6, paragraph 1;

(c) A proposal for the appointment of a sole arbitrator referred to in article 8, paragraph 1;

(*d*) Notification of the appointment of an arbitrator referred to in article 9 or 10;

(e) A brief description of counterclaims or claims for the purpose of a set-off, if any, including where relevant, an indication of the amounts involved, and the relief or remedy sought;

(f) A notice of arbitration in accordance with article 3 in case the respondent formulates a claim against a party to the arbitration agreement other than the claimant.

3. The constitution of the arbitral tribunal shall not be hindered by any controversy with respect to the respondent's failure to communicate a response to the notice of arbitration, or an incomplete or late response to the notice of arbitration, which shall be finally resolved by the arbitral tribunal.

Representation and assistance

Article 5

Each party may be represented or assisted by persons chosen by it. The names and addresses of such persons must be communicated to all parties and to the arbitral tribunal. Such communication must specify whether the appointment is being made for purposes of representation or assistance. Where a person is to act as a representative of a party, the arbitral tribunal, on its own initiative or at the request of any party, may at any time require proof of authority granted to the representative in such a form as the arbitral tribunal may determine.

Designating and appointing authorities

Article 6

1. Unless the parties have already agreed on the choice of an appointing authority, a party may at any time propose the name or names of one or more institutions or persons, including the Secretary-General of the Permanent Court of Arbitration at The Hague (hereinafter called the "PCA"), one of whom would serve as appointing authority.

2. If all parties have not agreed on the choice of an appointing authority within 30 days after a proposal made in accordance with paragraph 1 has been received by all other parties, any party may request the Secretary-General of the PCA to designate the appointing authority.

3. Where these Rules provide for a period of time within which a party must refer a matter to an appointing authority and no appointing authority has been agreed on or designated, the period is suspended from the date on which a party initiates the procedure for agreeing on or designating an appointing authority until the date of such agreement or designation.

4. Except as referred to in article 41, paragraph 4, if the appointing authority refuses to act, or if it fails to appoint an arbitrator within 30 days after it receives a party's request to do so, fails to act within any other period provided by these Rules, or fails to decide on a challenge to an arbitrator within a reasonable time after receiving a party's request to do so, any party may request the Secretary-General of the PCA to designate a substitute appointing authority.

5. In exercising their functions under these Rules, the appointing authority and the Secretary-General of the PCA may require from any party and the arbitrators the information they deem necessary and they shall give the parties and, where appropriate, the arbitrators, an opportunity to present their views in any manner they consider appropriate. All such communications to and from the appointing authority and the Secretary-General of the PCA shall also be provided by the sender to all other parties.

6. When the appointing authority is requested to appoint an arbitrator pursuant to articles 8, 9, 10 or 14, the party making the request shall send to the appointing authority copies of the notice of arbitration and, if it exists, any response to the notice of arbitration.

7. The appointing authority shall have regard to such considerations as are likely to secure the appointment of an independent and impartial arbitrator and shall take into account the advisability of appointing an arbitrator of a nationality other than the nationalities of the parties.

Section II. Composition of the arbitral tribunal

Number of arbitrators

Article 7

1. If the parties have not previously agreed on the number of arbitrators, and if within 30 days after the receipt by the respondent of the notice of arbitration the parties have not agreed that there shall be only one arbitrator, three arbitrators shall be appointed.

2. Notwithstanding paragraph 1, if no other parties have responded to a party's proposal to appoint a sole arbitrator within the time limit provided for in paragraph 1 and the party or parties concerned have failed to appoint a second arbitrator in accordance with article 9 or 10, the appointing authority may, at the request of a party, appoint a sole arbitrator pursuant to the procedure provided for in article 8, paragraph 2, if it determines that, in view of the circumstances of the case, this is more appropriate.

Appointment of arbitrators (articles 8 to 10)

Article 8

1. If the parties have agreed that a sole arbitrator is to be appointed and if within 30 days after receipt by all other parties of a proposal for the appointment of a sole arbitrator the parties have not reached agreement thereon, a sole arbitrator shall, at the request of a party, be appointed by the appointing authority.

2. The appointing authority shall appoint the sole arbitrator as promptly as possible. In making the appointment, the appointing authority shall use the following list-procedure, unless the parties agree that the list-procedure should not be used or unless the appointing authority determines in its discretion that the use of the list-procedure is not appropriate for the case:

(a) The appointing authority shall communicate to each of the parties an identical list containing at least three names;

(b) Within 15 days after the receipt of this list, each party may return the list to the appointing authority after having deleted the name or names to which it objects and numbered the remaining names on the list in the order of its preference;

(c) After the expiration of the above period of time the appointing authority shall appoint the sole arbitrator from among the names approved on the lists returned to it and in accordance with the order of preference indicated by the parties;

(*d*) If for any reason the appointment cannot be made according to this procedure, the appointing authority may exercise its discretion in appointing the sole arbitrator.

Article 9

1. If three arbitrators are to be appointed, each party shall appoint one arbitrator. The two arbitrators thus appointed shall choose the third arbitrator who will act as the presiding arbitrator of the arbitral tribunal.

2. If within 30 days after the receipt of a party's notification of the appointment of an arbitrator the other party has not notified the first party of the arbitrator it has appointed, the first party may request the appointing authority to appoint the second arbitrator.

3. If within 30 days after the appointment of the second arbitrator the two arbitrators have not agreed on the choice of the presiding arbitrator, the presiding arbitrator shall be appointed by the appointing authority in the same way as a sole arbitrator would be appointed under article 8.

Article 10

1. For the purposes of article 9, paragraph 1, where three arbitrators are to be appointed and there are multiple parties as claimant or as respondent, unless the parties have agreed to another method of appointment of arbitrators, the multiple parties jointly, whether as claimant or as respondent, shall appoint an arbitrator.

2. If the parties have agreed that the arbitral tribunal is to be composed of a number of arbitrators other than one or three, the arbitrators shall be appointed according to the method agreed upon by the parties. 3. In the event of any failure to constitute the arbitral tribunal under these Rules, the appointing authority shall, at the request of any party, constitute the arbitral tribunal and, in doing so, may revoke any appointment already made and appoint or reappoint each of the arbitrators and designate one of them as the presiding arbitrator.

Disclosures by and challenge of arbitrators** (articles 11 to 13)

Article 11

When a person is approached in connection with his or her possible appointment as an arbitrator, he or she shall disclose any circumstances likely to give rise to justifiable doubts as to his or her impartiality or independence. An arbitrator, from the time of his or her appointment and throughout the arbitral proceedings, shall without delay disclose any such circumstances to the parties and the other arbitrators unless they have already been informed by him or her of these circumstances.

Article 12

1. Any arbitrator may be challenged if circumstances exist that give rise to justifiable doubts as to the arbitrator's impartiality or independence.

2. A party may challenge the arbitrator appointed by it only for reasons of which it becomes aware after the appointment has been made.

3. In the event that an arbitrator fails to act or in the event of the de jure or de facto impossibility of his or her performing his or her functions, the procedure in respect of the challenge of an arbitrator as provided in article 13 shall apply.

Article 13

1. A party that intends to challenge an arbitrator shall send notice of its challenge within 15 days after it has been notified

^{**}Model statements of independence pursuant to article 11 can be found in the annex to the Rules.

of the appointment of the challenged arbitrator, or within 15 days after the circumstances mentioned in articles 11 and 12 became known to that party.

2. The notice of challenge shall be communicated to all other parties, to the arbitrator who is challenged and to the other arbitrators. The notice of challenge shall state the reasons for the challenge.

3. When an arbitrator has been challenged by a party, all parties may agree to the challenge. The arbitrator may also, after the challenge, withdraw from his or her office. In neither case does this imply acceptance of the validity of the grounds for the challenge.

4. If, within 15 days from the date of the notice of challenge, all parties do not agree to the challenge or the challenged arbitrator does not withdraw, the party making the challenge may elect to pursue it. In that case, within 30 days from the date of the notice of challenge, it shall seek a decision on the challenge by the appointing authority.

Replacement of an arbitrator

Article 14

1. Subject to paragraph 2, in any event where an arbitrator has to be replaced during the course of the arbitral proceedings, a substitute arbitrator shall be appointed or chosen pursuant to the procedure provided for in articles 8 to 11 that was applicable to the appointment or choice of the arbitrator being replaced. This procedure shall apply even if during the process of appointing the arbitrator to be replaced, a party had failed to exercise its right to appoint or to participate in the appointment.

2. If, at the request of a party, the appointing authority determines that, in view of the exceptional circumstances of the case, it would be justified for a party to be deprived of its right to appoint a substitute arbitrator, the appointing authority may, after giving an opportunity to the parties and the remaining arbitrators to express their views: (a) appoint the substitute arbitrator; or (b) after the closure of the hearings, authorize the other arbitrators to proceed with the arbitration and make any decision or award.

Repetition of hearings in the event of the replacement of an arbitrator

Article 15

If an arbitrator is replaced, the proceedings shall resume at the stage where the arbitrator who was replaced ceased to perform his or her functions, unless the arbitral tribunal decides otherwise.

Exclusion of liability

Article 16

Save for intentional wrongdoing, the parties waive, to the fullest extent permitted under the applicable law, any claim against the arbitrators, the appointing authority and any person appointed by the arbitral tribunal based on any act or omission in connection with the arbitration.

Section III. Arbitral proceedings

General provisions

Article 17

1. Subject to these Rules, the arbitral tribunal may conduct the arbitration in such manner as it considers appropriate, provided that the parties are treated with equality and that at an appropriate stage of the proceedings each party is given a reasonable opportunity of presenting its case. The arbitral tribunal, in exercising its discretion, shall conduct the proceedings so as to avoid unnecessary delay and expense and to provide a fair and efficient process for resolving the parties' dispute.

2. As soon as practicable after its constitution and after inviting the parties to express their views, the arbitral tribunal shall establish the provisional timetable of the arbitration. The arbitral tribunal may, at any time, after inviting the parties to express their views, extend or abridge any period of time prescribed under these Rules or agreed by the parties.

3. If at an appropriate stage of the proceedings any party so requests, the arbitral tribunal shall hold hearings for the presentation of evidence by witnesses, including expert witnesses, or for oral argument. In the absence of such a request, the arbitral tribunal shall decide whether to hold such hearings or whether the proceedings shall be conducted on the basis of documents and other materials.

4. All communications to the arbitral tribunal by one party shall be communicated by that party to all other parties. Such communications shall be made at the same time, except as otherwise permitted by the arbitral tribunal if it may do so under applicable law.

5. The arbitral tribunal may, at the request of any party, allow one or more third persons to be joined in the arbitration as a party provided such person is a party to the arbitration agreement, unless the arbitral tribunal finds, after giving all parties, including the person or persons to be joined, the opportunity to be heard, that joinder should not be permitted because of prejudice to any of those parties. The arbitral tribunal may make a single award or several awards in respect of all parties so involved in the arbitration.

Place of arbitration

Article 18

1. If the parties have not previously agreed on the place of arbitration, the place of arbitration shall be determined by the arbitral tribunal having regard to the circumstances of the case. The award shall be deemed to have been made at the place of arbitration.

2. The arbitral tribunal may meet at any location it considers appropriate for deliberations. Unless otherwise agreed by the parties, the arbitral tribunal may also meet at any location it considers appropriate for any other purpose, including hearings.

Language

Article 19

1. Subject to an agreement by the parties, the arbitral tribunal shall, promptly after its appointment, determine the language or languages to be used in the proceedings. This determination shall apply to the statement of claim, the statement of defence, and any further written statements and, if oral hearings take place, to the language or languages to be used in such hearings.

2. The arbitral tribunal may order that any documents annexed to the statement of claim or statement of defence, and any supplementary documents or exhibits submitted in the course of the proceedings, delivered in their original language, shall be accompanied by a translation into the language or languages agreed upon by the parties or determined by the arbitral tribunal.

Statement of claim

Article 20

1. The claimant shall communicate its statement of claim in writing to the respondent and to each of the arbitrators within

a period of time to be determined by the arbitral tribunal. The claimant may elect to treat its notice of arbitration referred to in article 3 as a statement of claim, provided that the notice of arbitration also complies with the requirements of paragraphs 2 to 4 of this article.

2. The statement of claim shall include the following particulars:

- (a) The names and contact details of the parties;
- (b) A statement of the facts supporting the claim;
- (c) The points at issue;
- (d) The relief or remedy sought;
- (e) The legal grounds or arguments supporting the claim.

3. A copy of any contract or other legal instrument out of or in relation to which the dispute arises and of the arbitration agreement shall be annexed to the statement of claim.

4. The statement of claim should, as far as possible, be accompanied by all documents and other evidence relied upon by the claimant, or contain references to them.

Statement of defence

Article 21

1. The respondent shall communicate its statement of defence in writing to the claimant and to each of the arbitrators within a period of time to be determined by the arbitral tribunal. The respondent may elect to treat its response to the notice of arbitration referred to in article 4 as a statement of defence, provided that the response to the notice of arbitration also complies with the requirements of paragraph 2 of this article.

2. The statement of defence shall reply to the particulars (b) to (e) of the statement of claim (art. 20, para. 2). The statement of defence should, as far as possible, be accompanied by all documents and other evidence relied upon by the respondent, or contain references to them.

3. In its statement of defence, or at a later stage in the arbitral proceedings if the arbitral tribunal decides that the delay was justified under the circumstances, the respondent may make a counterclaim or rely on a claim for the purpose of a set-off provided that the arbitral tribunal has jurisdiction over it.

4. The provisions of article 20, paragraphs 2 to 4, shall apply to a counterclaim, a claim under article 4, paragraph 2 (f), and a claim relied on for the purpose of a set-off.

Amendments to the claim or defence

Article 22

During the course of the arbitral proceedings, a party may amend or supplement its claim or defence, including a counterclaim or a claim for the purpose of a set-off, unless the arbitral tribunal considers it inappropriate to allow such amendment or supplement having regard to the delay in making it or prejudice to other parties or any other circumstances. However, a claim or defence, including a counterclaim or a claim for the purpose of a set-off, may not be amended or supplemented in such a manner that the amended or supplemented claim or defence falls outside the jurisdiction of the arbitral tribunal.

Pleas as to the jurisdiction of the arbitral tribunal

Article 23

1. The arbitral tribunal shall have the power to rule on its own jurisdiction, including any objections with respect to the existence or validity of the arbitration agreement. For that purpose, an arbitration clause that forms part of a contract shall be treated as an agreement independent of the other terms of the contract. A decision by the arbitral tribunal that the contract is null shall not entail automatically the invalidity of the arbitration clause.

2. A plea that the arbitral tribunal does not have jurisdiction shall be raised no later than in the statement of defence or, with respect to a counterclaim or a claim for the purpose of a set-off, in the reply to the counterclaim or to the claim for the purpose of a set-off. A party is not precluded from raising such a plea by the fact that it has appointed, or participated in the appointment of, an arbitrator. A plea that the arbitral tribunal is exceeding the scope of its authority shall be raised as soon as the matter alleged to be beyond the scope of its authority is raised during the arbitral proceedings. The arbitral tribunal may, in either case, admit a later plea if it considers the delay justified. 3. The arbitral tribunal may rule on a plea referred to in paragraph 2 either as a preliminary question or in an award on the merits. The arbitral tribunal may continue the arbitral proceedings and make an award, notwithstanding any pending challenge to its jurisdiction before a court.

Further written statements

Article 24

The arbitral tribunal shall decide which further written statements, in addition to the statement of claim and the statement of defence, shall be required from the parties or may be presented by them and shall fix the periods of time for communicating such statements.

Periods of time

Article 25

The periods of time fixed by the arbitral tribunal for the communication of written statements (including the statement of claim and statement of defence) should not exceed 45 days. However, the arbitral tribunal may extend the time limits if it concludes that an extension is justified.

Interim measures

Article 26

1. The arbitral tribunal may, at the request of a party, grant interim measures.

2. An interim measure is any temporary measure by which, at any time prior to the issuance of the award by which the dispute is finally decided, the arbitral tribunal orders a party, for example and without limitation, to:

(a) Maintain or restore the status quo pending determination of the dispute;

(b) Take action that would prevent, or refrain from taking action that is likely to cause, (i) current or imminent harm or (ii) prejudice to the arbitral process itself;

(c) Provide a means of preserving assets out of which a subsequent award may be satisfied; or

(*d*) Preserve evidence that may be relevant and material to the resolution of the dispute.

3. The party requesting an interim measure under paragraphs 2 (a) to (c) shall satisfy the arbitral tribunal that:

(a) Harm not adequately reparable by an award of damages is likely to result if the measure is not ordered, and such harm substantially outweighs the harm that is likely to result to the party against whom the measure is directed if the measure is granted; and

(b) There is a reasonable possibility that the requesting party will succeed on the merits of the claim. The determination on this possibility shall not affect the discretion of the arbitral tribunal in making any subsequent determination.

4. With regard to a request for an interim measure under paragraph 2 (d), the requirements in paragraphs 3 (a) and (b) shall apply only to the extent the arbitral tribunal considers appropriate.

5. The arbitral tribunal may modify, suspend or terminate an interim measure it has granted, upon application of any party or, in exceptional circumstances and upon prior notice to the parties, on the arbitral tribunal's own initiative.

6. The arbitral tribunal may require the party requesting an interim measure to provide appropriate security in connection with the measure.

7. The arbitral tribunal may require any party promptly to disclose any material change in the circumstances on the basis of which the interim measure was requested or granted.

8. The party requesting an interim measure may be liable for any costs and damages caused by the measure to any party if the arbitral tribunal later determines that, in the circumstances then prevailing, the measure should not have been granted. The arbitral tribunal may award such costs and damages at any point during the proceedings.

9. A request for interim measures addressed by any party to a judicial authority shall not be deemed incompatible with the agreement to arbitrate, or as a waiver of that agreement.

Evidence

Article 27

1. Each party shall have the burden of proving the facts relied on to support its claim or defence. 2. Witnesses, including expert witnesses, who are presented by the parties to testify to the arbitral tribunal on any issue of fact or expertise may be any individual, notwithstanding that the individual is a party to the arbitration or in any way related to a party. Unless otherwise directed by the arbitral tribunal, statements by witnesses, including expert witnesses, may be presented in writing and signed by them.

3. At any time during the arbitral proceedings the arbitral tribunal may require the parties to produce documents, exhibits or other evidence within such a period of time as the arbitral tribunal shall determine.

4. The arbitral tribunal shall determine the admissibility, relevance, materiality and weight of the evidence offered.

Hearings

Article 28

1. In the event of an oral hearing, the arbitral tribunal shall give the parties adequate advance notice of the date, time and place thereof.

2. Witnesses, including expert witnesses, may be heard under the conditions and examined in the manner set by the arbitral tribunal.

3. Hearings shall be held in camera unless the parties agree otherwise. The arbitral tribunal may require the retirement of any witness or witnesses, including expert witnesses, during the testimony of such other witnesses, except that a witness, including an expert witness, who is a party to the arbitration shall not, in principle, be asked to retire.

4. The arbitral tribunal may direct that witnesses, including expert witnesses, be examined through means of telecommunication that do not require their physical presence at the hearing (such as videoconference).

Experts appointed by the arbitral tribunal

Article 29

1. After consultation with the parties, the arbitral tribunal may appoint one or more independent experts to report to it,

in writing, on specific issues to be determined by the arbitral tribunal. A copy of the expert's terms of reference, established by the arbitral tribunal, shall be communicated to the parties.

2. The expert shall, in principle before accepting appointment, submit to the arbitral tribunal and to the parties a description of his or her qualifications and a statement of his or her impartiality and independence. Within the time ordered by the arbitral tribunal, the parties shall inform the arbitral tribunal whether they have any objections as to the expert's qualifications, impartiality or independence. The arbitral tribunal shall decide promptly whether to accept any such objections. After an expert's appointment, a party may object to the expert's qualifications, impartiality or independence only if the objection is for reasons of which the party becomes aware after the appointment has been made. The arbitral tribunal shall decide promptly what, if any, action to take.

3. The parties shall give the expert any relevant information or produce for his or her inspection any relevant documents or goods that he or she may require of them. Any dispute between a party and such expert as to the relevance of the required information or production shall be referred to the arbitral tribunal for decision.

4. Upon receipt of the expert's report, the arbitral tribunal shall communicate a copy of the report to the parties, which shall be given the opportunity to express, in writing, their opinion on the report. A party shall be entitled to examine any document on which the expert has relied in his or her report.

5. At the request of any party, the expert, after delivery of the report, may be heard at a hearing where the parties shall have the opportunity to be present and to interrogate the expert. At this hearing, any party may present expert witnesses in order to testify on the points at issue. The provisions of article 28 shall be applicable to such proceedings.

Default

Article 30

1. If, within the period of time fixed by these Rules or the arbitral tribunal, without showing sufficient cause:

(a) The claimant has failed to communicate its statement of claim, the arbitral tribunal shall issue an order for the termination of the arbitral proceedings, unless there are remaining matters that may need to be decided and the arbitral tribunal considers it appropriate to do so;

(b) The respondent has failed to communicate its response to the notice of arbitration or its statement of defence, the arbitral tribunal shall order that the proceedings continue, without treating such failure in itself as an admission of the claimant's allegations; the provisions of this subparagraph also apply to a claimant's failure to submit a defence to a counterclaim or to a claim for the purpose of a set-off.

2. If a party, duly notified under these Rules, fails to appear at a hearing, without showing sufficient cause for such failure, the arbitral tribunal may proceed with the arbitration.

3. If a party, duly invited by the arbitral tribunal to produce documents, exhibits or other evidence, fails to do so within the established period of time, without showing sufficient cause for such failure, the arbitral tribunal may make the award on the evidence before it.

Closure of hearings

Article 31

1. The arbitral tribunal may inquire of the parties if they have any further proof to offer or witnesses to be heard or submissions to make and, if there are none, it may declare the hearings closed.

2. The arbitral tribunal may, if it considers it necessary owing to exceptional circumstances, decide, on its own initiative or upon application of a party, to reopen the hearings at any time before the award is made.

Waiver of right to object

Article 32

A failure by any party to object promptly to any non-compliance with these Rules or with any requirement of the arbitration agreement shall be deemed to be a waiver of the right of such party to make such an objection, unless such party can show that, under the circumstances, its failure to object was justified.

Section IV. The award

Decisions

Article 33

1. When there is more than one arbitrator, any award or other decision of the arbitral tribunal shall be made by a majority of the arbitrators.

2. In the case of questions of procedure, when there is no majority or when the arbitral tribunal so authorizes, the presiding arbitrator may decide alone, subject to revision, if any, by the arbitral tribunal.

Form and effect of the award

Article 34

1. The arbitral tribunal may make separate awards on different issues at different times.

2. All awards shall be made in writing and shall be final and binding on the parties. The parties shall carry out all awards without delay.

3. The arbitral tribunal shall state the reasons upon which the award is based, unless the parties have agreed that no reasons are to be given.

4. An award shall be signed by the arbitrators and it shall contain the date on which the award was made and indicate the place of arbitration. Where there is more than one arbitrator and any of them fails to sign, the award shall state the reason for the absence of the signature.

5. An award may be made public with the consent of all parties or where and to the extent disclosure is required of a party by legal duty, to protect or pursue a legal right or in relation to legal proceedings before a court or other competent authority.

6. Copies of the award signed by the arbitrators shall be communicated to the parties by the arbitral tribunal.

Applicable law, amiable compositeur

Article 35

1. The arbitral tribunal shall apply the rules of law designated by the parties as applicable to the substance of the dispute. Failing such designation by the parties, the arbitral tribunal shall apply the law which it determines to be appropriate.

2. The arbitral tribunal shall decide as amiable compositeur or ex aequo et bono only if the parties have expressly authorized the arbitral tribunal to do so.

3. In all cases, the arbitral tribunal shall decide in accordance with the terms of the contract, if any, and shall take into account any usage of trade applicable to the transaction.

Settlement or other grounds for termination

Article 36

1. If, before the award is made, the parties agree on a settlement of the dispute, the arbitral tribunal shall either issue an order for the termination of the arbitral proceedings or, if requested by the parties and accepted by the arbitral tribunal, record the settlement in the form of an arbitral award on agreed terms. The arbitral tribunal is not obliged to give reasons for such an award.

2. If, before the award is made, the continuation of the arbitral proceedings becomes unnecessary or impossible for any reason not mentioned in paragraph 1, the arbitral tribunal shall inform the parties of its intention to issue an order for the termination of the proceedings. The arbitral tribunal shall have the power to issue such an order unless there are remaining matters that may need to be decided and the arbitral tribunal considers it appropriate to do so.

3. Copies of the order for termination of the arbitral proceedings or of the arbitral award on agreed terms, signed by the arbitrators, shall be communicated by the arbitral tribunal to the parties. Where an arbitral award on agreed terms is made, the provisions of article 34, paragraphs 2, 4 and 5, shall apply.

Interpretation of the award

Article 37

1. Within 30 days after the receipt of the award, a party, with notice to the other parties, may request that the arbitral tribunal give an interpretation of the award.

2. The interpretation shall be given in writing within 45 days after the receipt of the request. The interpretation shall form part of the award and the provisions of article 34, paragraphs 2 to 6, shall apply.

Correction of the award

Article 38

1. Within 30 days after the receipt of the award, a party, with notice to the other parties, may request the arbitral tribunal to correct in the award any error in computation, any clerical or typographical error, or any error or omission of a similar nature. If the arbitral tribunal considers that the request is justified, it shall make the correction within 45 days of receipt of the request.

2. The arbitral tribunal may within 30 days after the communication of the award make such corrections on its own initiative.

3. Such corrections shall be in writing and shall form part of the award. The provisions of article 34, paragraphs 2 to 6, shall apply.

Additional award

Article 39

1. Within 30 days after the receipt of the termination order or the award, a party, with notice to the other parties, may request the arbitral tribunal to make an award or an additional award as to claims presented in the arbitral proceedings but not decided by the arbitral tribunal.

2. If the arbitral tribunal considers the request for an award or additional award to be justified, it shall render or complete its award within 60 days after the receipt of the request. The arbitral tribunal may extend, if necessary, the period of time within which it shall make the award.

3. When such an award or additional award is made, the provisions of article 34, paragraphs 2 to 6, shall apply.

Definition of costs

Article 40

1. The arbitral tribunal shall fix the costs of arbitration in the final award and, if it deems appropriate, in another decision.

2. The term "costs" includes only:

(*a*) The fees of the arbitral tribunal to be stated separately as to each arbitrator and to be fixed by the tribunal itself in accordance with article 41;

(b) The reasonable travel and other expenses incurred by the arbitrators;

(c) The reasonable costs of expert advice and of other assistance required by the arbitral tribunal;

(*d*) The reasonable travel and other expenses of witnesses to the extent such expenses are approved by the arbitral tribunal;

(e) The legal and other costs incurred by the parties in relation to the arbitration to the extent that the arbitral tribunal determines that the amount of such costs is reasonable;

(f) Any fees and expenses of the appointing authority as well as the fees and expenses of the Secretary-General of the PCA.

3. In relation to interpretation, correction or completion of any award under articles 37 to 39, the arbitral tribunal may charge the costs referred to in paragraphs 2 (b) to (f), but no additional fees.

Fees and expenses of arbitrators

Article 41

1. The fees and expenses of the arbitrators shall be reasonable in amount, taking into account the amount in dispute, the complexity of the subject matter, the time spent by the arbitrators and any other relevant circumstances of the case. 2. If there is an appointing authority and it applies or has stated that it will apply a schedule or particular method for determining the fees for arbitrators in international cases, the arbitral tribunal in fixing its fees shall take that schedule or method into account to the extent that it considers appropriate in the circumstances of the case.

3. Promptly after its constitution, the arbitral tribunal shall inform the parties as to how it proposes to determine its fees and expenses, including any rates it intends to apply. Within 15 days of receiving that proposal, any party may refer the proposal to the appointing authority for review. If, within 45 days of receipt of such a referral, the appointing authority finds that the proposal of the arbitral tribunal is inconsistent with paragraph 1, it shall make any necessary adjustments thereto, which shall be binding upon the arbitral tribunal.

4. (a) When informing the parties of the arbitrators' fees and expenses that have been fixed pursuant to article 40, paragraphs 2 (a) and (b), the arbitral tribunal shall also explain the manner in which the corresponding amounts have been calculated;

(b) Within 15 days of receiving the arbitral tribunal's determination of fees and expenses, any party may refer for review such determination to the appointing authority. If no appointing authority has been agreed upon or designated, or if the appointing authority fails to act within the time specified in these Rules, then the review shall be made by the Secretary-General of the PCA;

(c) If the appointing authority or the Secretary-General of the PCA finds that the arbitral tribunal's determination is inconsistent with the arbitral tribunal's proposal (and any adjustment thereto) under paragraph 3 or is otherwise manifestly excessive, it shall, within 45 days of receiving such a referral, make any adjustments to the arbitral tribunal's determination that are necessary to satisfy the criteria in paragraph 1. Any such adjustments shall be binding upon the arbitral tribunal;

(*d*) Any such adjustments shall either be included by the arbitral tribunal in its award or, if the award has already been issued, be implemented in a correction to the award, to which the procedure of article 38, paragraph 3, shall apply.

5. Throughout the procedure under paragraphs 3 and 4, the arbitral tribunal shall proceed with the arbitration, in accordance with article 17, paragraph 1.

6. A referral under paragraph 4 shall not affect any determination in the award other than the arbitral tribunal's fees and expenses; nor shall it delay the recognition and enforcement of all parts of the award other than those relating to the determination of the arbitral tribunal's fees and expenses.

Allocation of costs

Article 42

1. The costs of the arbitration shall in principle be borne by the unsuccessful party or parties. However, the arbitral tribunal may apportion each of such costs between the parties if it determines that apportionment is reasonable, taking into account the circumstances of the case.

2. The arbitral tribunal shall in the final award or, if it deems appropriate, in any other award, determine any amount that a party may have to pay to another party as a result of the decision on allocation of costs.

Deposit of costs

Article 43

1. The arbitral tribunal, on its establishment, may request the parties to deposit an equal amount as an advance for the costs referred to in article 40, paragraphs 2 (a) to (c).

2. During the course of the arbitral proceedings the arbitral tribunal may request supplementary deposits from the parties.

3. If an appointing authority has been agreed upon or designated, and when a party so requests and the appointing authority consents to perform the function, the arbitral tribunal shall fix the amounts of any deposits or supplementary deposits only after consultation with the appointing authority, which may make any comments to the arbitral tribunal that it deems appropriate concerning the amount of such deposits and supplementary deposits.

4. If the required deposits are not paid in full within 30 days after the receipt of the request, the arbitral tribunal shall so inform the parties in order that one or more of them may make the required payment. If such payment is not made, the arbitral tribunal may order the suspension or termination of the arbitral proceedings.

5. After a termination order or final award has been made, the arbitral tribunal shall render an accounting to the parties of the deposits received and return any unexpended balance to the parties.

Annex

Model arbitration clause for contracts

Any dispute, controversy or claim arising out of or relating to this contract, or the breach, termination or invalidity thereof, shall be settled by arbitration in accordance with the UNCITRAL Arbitration Rules.

Note. Parties should consider adding:

- (*a*) The appointing authority shall be ... [name of institution or person];
- (b) The number of arbitrators shall be [one or three];

(c) The place of arbitration shall be . . . [town and country];

(d) The language to be used in the arbitral proceedings shall be \ldots .

Possible waiver statement

Note. If the parties wish to exclude recourse against the arbitral award that may be available under the applicable law, they may consider adding a provision to that effect as suggested below, considering, however, that the effectiveness and conditions of such an exclusion depend on the applicable law.

Waiver

The parties hereby waive their right to any form of recourse against an award to any court or other competent authority, insofar as such waiver can validly be made under the applicable law.

Model statements of independence pursuant to article 11 of the Rules

No circumstances to disclose

I am impartial and independent of each of the parties and

intend to remain so. To the best of my knowledge, there are no circumstances, past or present, likely to give rise to justifiable doubts as to my impartiality or independence. I shall promptly notify the parties and the other arbitrators of any such circumstances that may subsequently come to my attention during this arbitration.

Circumstances to disclose

I am impartial and independent of each of the parties and intend to remain so. Attached is a statement made pursuant to article 11 of the UNCITRAL Arbitration Rules of (a) my past and present professional, business and other relationships with the parties and (b) any other relevant circumstances. [Include statement.] I confirm that those circumstances do not affect my independence and impartiality. I shall promptly notify the parties and the other arbitrators of any such further relationships or circumstances that may subsequently come to my attention during this arbitration.

Note. Any party may consider requesting from the arbitrator the following addition to the statement of independence:

I confirm, on the basis of the information presently available to me, that I can devote the time necessary to conduct this arbitration diligently, efficiently and in accordance with the time limits in the Rules.

Appendix – UNCITRAL Expedited Arbitration Rules

Scope of application

Article 1

Where parties have agreed that disputes between them in respect of a defined legal relationship, whether contractual or not, shall be referred to arbitration under the UNCITRAL Expedited Arbitration Rules ("Expedited Rules"), then such disputes shall be settled in accordance with the UNCITRAL Arbitration Rules as modified by these Expedited Rules and subject to such modification as the parties may agree.^{***}

Article 2

1. At any time during the proceedings, the parties may agree that the Expedited Rules shall no longer apply to the arbitration.

2. At the request of a party, the arbitral tribunal may, in exceptional circumstances and after inviting the parties to express their views, determine that the Expedited Rules shall no longer apply to the arbitration. The arbitral tribunal shall state the reasons upon which that determination is based.

3. When the Expedited Rules no longer apply to the arbitration pursuant to paragraph 1 or 2, the arbitral tribunal shall remain in place and conduct the arbitration in accordance with the UNCITRAL Arbitration Rules.

Conduct of the parties and the arbitral tribunal

Article 3

1. The parties shall act expeditiously throughout the proceedings.

2. The arbitral tribunal shall conduct the proceedings

^{***}Unless otherwise agreed by the parties, the following articles in the UNCITRAL Arbitration Rules do not apply to expedited arbitration: article 3(4)(a) and (b); article 6(2); article 7; article 8(1); first sentence of article 20(1); first sentence of article 21(1); article 21(3); article 22; and second sentence of article 27(2).

expeditiously taking into account the fact that the parties agreed to refer their dispute to expedited arbitration and the time frames in the Expedited Rules.

3. The arbitral tribunal may, after inviting the parties to express their views and taking into account the circumstances of the case, utilize any technological means as it considers appropriate to conduct the proceedings, including to communicate with the parties and to hold consultations and hearings remotely.

Notice of arbitration and statement of claim

Article 4

1. A notice of arbitration shall also include:

(a) A proposal for the designation of an appointing authority, unless the parties have previously agreed thereon; and

(b) A proposal for the appointment of an arbitrator.

2. When communicating its notice of arbitration to the respondent, the claimant shall also communicate its statement of claim.

3. The claimant shall communicate the notice of arbitration and the statement of claim to the arbitral tribunal as soon as it is constituted.

Response to the notice of arbitration and statement of defence

Article 5

1. Within 15 days of the receipt of the notice of arbitration, the respondent shall communicate to the claimant a response to the notice of arbitration, which shall also include responses to the information set forth in the notice of arbitration pursuant to article 4(1)(a) and (b) of the Expedited Rules.

2. The respondent shall communicate its statement of defence to the claimant and the arbitral tribunal within 15 days of the constitution of the arbitral tribunal.

Designating and appointing authorities

Article 6

1. If all parties have not agreed on the choice of an appointing authority 15 days after a proposal for the designation of an appointing authority has been received by all other parties, any party may request the Secretary-General of the Permanent Court of Arbitration (hereinafter called the "PCA") to designate the appointing authority or to serve as appointing authority.

2. When making the request under article 6(4) of the UNCI-TRAL Arbitration Rules, a party may request the Secretary-General of the PCA to serve as appointing authority.

3. If requested to serve as appointing authority in accordance with paragraph 1 or 2, the Secretary-General of the PCA will serve as appointing authority unless it determines that in view of the circumstances of the case, it is more appropriate to designate an appointing authority.

Number of arbitrators

Article 7

Unless otherwise agreed by the parties, there shall be one arbitrator.

Appointment of a sole arbitrator

Article 8

1. A sole arbitrator shall be appointed jointly by the parties.

2. If the parties have not reached agreement on the appointment of a sole arbitrator 15 days after a proposal has been received by all other parties, a sole arbitrator shall, at the request of a party, be appointed by the appointing authority in accordance with article 8(2) of the UNCITRAL Arbitration Rules.

Consultation with the parties

Article 9

Promptly after and within 15 days of its constitution, the arbitral tribunal shall consult the parties, through a case management conference or otherwise, on the manner in which it will conduct the arbitration.

Discretion of the arbitral tribunal with regard to periods of time

Article 10

Subject to article 16 of the Expedited Rules, the arbitral tribunal may at any time, after inviting the parties to express their views, extend or abridge any period of time prescribed under the UNCITRAL Arbitration Rules and the Expedited Rules or agreed by the parties.

Hearings

Article 11

The arbitral tribunal may, after inviting the parties to express their views and in the absence of a request to hold hearings, decide that hearings shall not be held.

Counterclaims or claims for the purpose of set-off

Article 12

1. A counterclaim or a claim for the purpose of a set-off shall be made no later than in the statement of defence provided that the arbitral tribunal has jurisdiction over it.

2. The respondent may not make a counterclaim or rely on a claim for the purpose of a set-off at a later stage in the arbitral proceedings, unless the arbitral tribunal considers it appropriate to allow such claim having regard to the delay in making it or prejudice to other parties or any other circumstances.

Amendments and supplements to a claim or defence

Article 13

During the course of the arbitral proceedings, a party may not amend or supplement its claim or defence, including a counterclaim or a claim for the purpose of a set-off, unless the arbitral tribunal considers it appropriate to allow such amendment or supplement having regard to when it is requested or prejudice to other parties or any other circumstances. However, a claim or defence, including a counterclaim or a claim for the purpose of a set-off, may not be amended or supplemented in such a manner that the amended or supplemented claim or defence falls outside the jurisdiction of the arbitral tribunal.

Further written statements

Article 14

The arbitral tribunal may, after inviting the parties to express their views, decide whether any further written statement shall be required from the parties or may be presented by them.

Evidence

Article 15

1. The arbitral tribunal may decide which documents, exhibits or other evidence the parties should produce. The arbitral tribunal may reject any request, unless made by all parties, to establish a procedure whereby each party can request another party to produce documents.

2. Unless otherwise directed by the arbitral tribunal, statements by witnesses, including expert witnesses, shall be presented in writing and signed by them.

3. The arbitral tribunal may decide which witnesses, including expert witnesses, shall testify to the arbitral tribunal if hearings are held.

Period of time for making the award

Article 16

1. The award shall be made within six months from the date of the constitution of the arbitral tribunal unless otherwise agreed by the parties.

2. The arbitral tribunal may, in exceptional circumstances and after inviting the parties to express their views, extend the period of time established in accordance with paragraph 1. The extended period of time shall not exceed a total of nine months from the date of the constitution of the arbitral tribunal.

3. If the arbitral tribunal concludes that it is at risk of not rendering an award within nine months from the date of the constitution of the arbitral tribunal, it shall propose a final extended time limit, state the reasons for the proposal, and invite the parties to express their views within a fixed period of time. The extension shall be adopted only if all parties express their agreement to the proposal within the fixed period of time. 4. If there is no agreement to the extension in paragraph 3, any party may make a request that the Expedited Rules no longer apply to the arbitration. After inviting the parties to express their views, the arbitral tribunal may determine to continue to conduct the arbitration in accordance with the UNCITRAL Arbitration Rules

Annex to the UNCITRAL Expedited Arbitration Rules

Model arbitration clause for contracts

Any dispute, controversy or claim arising out of or relating to this contract, or the breach, termination or invalidity thereof, shall be settled by arbitration in accordance with the UNCITRAL Expedited Arbitration Rules.

Note: Parties should consider adding:

(*a*) The appointing authority shall be . . . [name of institution or person];

(b) The place of arbitration shall be . . . [town and country];

(c) The language to be used in the arbitral proceedings shall be ...;

Model statement

Note. Parties should consider requesting from the arbitrator the following addition to the statement of independence pursuant to article 11 of the UNCITRAL Arbitration Rules:

I confirm, on the basis of the information presently available to me, that I can devote the time necessary to conduct this arbitration diligently, efficiently, expeditiously and in accordance with the time limits in the UNCITRAL Arbitration Rules and the UNCITRAL Expedited Arbitration Rules.

UNCITRAL Rules on Transparency in Treaty-based Investor-State Arbitration

Article 1. Scope of application

Applicability of the Rules

1. The UNCITRAL Rules on Transparency in Treaty-based Investor-State Arbitration ("Rules on Transparency") shall apply to investor-State arbitration initiated under the UNCITRAL Arbitration Rules pursuant to a treaty providing for the protection of investments or investors ("treaty")* concluded on or after 1 April 2014 unless the Parties to the treaty** have agreed otherwise.

2. In investor-State arbitrations initiated under the UNCITRAL Arbitration Rules pursuant to a treaty concluded before 1 April 2014, these Rules shall apply only when:

(a) The parties to an arbitration (the "disputing parties") agree to their application in respect of that arbitration; or

(b) The Parties to the treaty or, in the case of a multilateral treaty, the State of the claimant and the respondent State, have agreed after 1 April 2014 to their application.

Application of the Rules

3. In any arbitration in which the Rules on Transparency apply pursuant to a treaty or to an agreement by the Parties to that treaty:

(*a*) The disputing parties may not derogate from these Rules, by agreement or otherwise, unless permitted to do so by the treaty;

^{*}For the purposes of the Rules on Transparency, a "treaty" shall be understood broadly as encompassing any bilateral or multilateral treaty that contains provisions on the protection of investments or investors and a right for investors to resort to arbitration against Parties to the treaty, including any treaty commonly referred to as a free trade agreement, economic integration agreement, trade and investment framework or cooperation agreement, or bilateral investment treaty.

^{**}For the purposes of the Rules on Transparency, any reference to a "Party to the treaty" or a "State" includes, for example, a regional economic integration organization where it is a Party to the treaty.

(b) The arbitral tribunal shall have the power, besides its discretionary authority under certain provisions of these Rules, to adapt the requirements of any specific provision of these Rules to the particular circumstances of the case, after consultation with the disputing parties, if such adaptation is necessary to conduct the arbitration in a practical manner and is consistent with the transparency objective of these Rules.

Discretion and authority of the arbitral tribunal

4. Where the Rules on Transparency provide for the arbitral tribunal to exercise discretion, the arbitral tribunal in exercising such discretion shall take into account:

(a) The public interest in transparency in treaty-based investor-State arbitration and in the particular arbitral proceedings; and

(b) The disputing parties' interest in a fair and efficient resolution of their dispute.

5. These Rules shall not affect any authority that the arbitral tribunal may otherwise have under the UNCITRAL Arbitration Rules to conduct the arbitration in such a manner as to promote transparency, for example by accepting submissions from third persons.

6. In the presence of any conduct, measure or other action having the effect of wholly undermining the transparency objectives of these Rules, the arbitral tribunal shall ensure that those objectives prevail.

Applicable instrument in case of conflict

7. Where the Rules on Transparency apply, they shall supplement any applicable arbitration rules. Where there is a conflict between the Rules on Transparency and the applicable arbitration rules, the Rules on Transparency shall prevail. Notwithstanding any provision in these Rules, where there is a conflict between the Rules on Transparency and the treaty, the provisions of the treaty shall prevail.

8. Where any of these Rules is in conflict with a provision of the law applicable to the arbitration from which the disputing parties cannot derogate, that provision shall prevail.

Application in non-UNCITRAL arbitrations

9. These Rules are available for use in investor-State arbitrations initiated under rules other than the UNCITRAL Arbitration Rules or in ad hoc proceedings.

Article 2. Publication of information at the commencement of arbitral proceedings

Once the notice of arbitration has been received by the respondent, each of the disputing parties shall promptly communicate a copy of the notice of arbitration to the repository referred to under article 8. Upon receipt of the notice of arbitration from the respondent, or upon receipt of the notice of arbitration and a record of its transmission to the respondent, the repository shall promptly make available to the public information regarding the name of the disputing parties, the economic sector involved and the treaty under which the claim is being made.

Article 3. Publication of documents

1. Subject to article 7, the following documents shall be made available to the public: the notice of arbitration, the response to the notice of arbitration, the statement of claim, the statement of defence and any further written statements or written submissions by any disputing party; a table listing all exhibits to the aforesaid documents and to expert reports and witness statements, if such table has been prepared for the proceedings, but not the exhibits themselves; any written submissions by the non-disputing Party (or Parties) to the treaty and by third persons, transcripts of hearings, where available; and orders, decisions and awards of the arbitral tribunal.

2. Subject to article 7, expert reports and witness statements, exclusive of the exhibits thereto, shall be made available to the public, upon request by any person to the arbitral tribunal.

3. Subject to article 7, the arbitral tribunal may decide, on its own initiative or upon request from any person, and after consultation with the disputing parties, whether and how to make available exhibits and any other documents provided to, or issued by, the arbitral tribunal not falling within paragraphs 1 or 2 above. This may include, for example, making such documents available at a specified site. 4. The documents to be made available to the public pursuant to paragraphs 1 and 2 shall be communicated by the arbitral tribunal to the repository referred to under article 8 as soon as possible, subject to any relevant arrangements or time limits for the protection of confidential or protected information prescribed under article 7. The documents to be made available pursuant to paragraph 3 may be communicated by the arbitral tribunal to the repository referred to under article 8 as they become available and, if applicable, in a redacted form in accordance with article 7. The repository shall make all documents available in a timely manner, in the form and in the language in which it receives them.

5. A person granted access to documents under paragraph 3 shall bear any administrative costs of making those documents available to that person, such as the costs of photocopying or shipping documents to that person, but not the costs of making those documents available to the public through the repository.

Article 4. Submission by a third person

1. After consultation with the disputing parties, the arbitral tribunal may allow a person that is not a disputing party, and not a non-disputing Party to the treaty ("third person(s)"), to file a written submission with the arbitral tribunal regarding a matter within the scope of the dispute.

2. A third person wishing to make a submission shall apply to the arbitral tribunal, and shall, in a concise written statement, which is in a language of the arbitration and complies with any page limits set by the arbitral tribunal:

(a) Describe the third person, including, where relevant, its membership and legal status (e.g., trade association or other non-governmental organization), its general objectives, the nature of its activities and any parent organization (including any organization that directly or indirectly controls the third person);

(b) Disclose any connection, direct or indirect, which the third person has with any disputing party;

(c) Provide information on any government, person or organization that has provided to the third person (i) any financial or other assistance in preparing the submission; or (ii) substantial assistance in either of the two years preceding the application by the third person under this article (e.g. funding around 20 per cent of its overall operations annually); (*d*) Describe the nature of the interest that the third person has in the arbitration; and

(e) Identify the specific issues of fact or law in the arbitration that the third person wishes to address in its written submission.

3. In determining whether to allow such a submission, the arbitral tribunal shall take into consideration, among other factors it determines to be relevant:

(a) Whether the third person has a significant interest in the arbitral proceedings; and

(b) The extent to which the submission would assist the arbitral tribunal in the determination of a factual or legal issue related to the arbitral proceedings by bringing a perspective, particular knowledge or insight that is different from that of the disputing parties.

4. The submission filed by the third person shall:

(a) Be dated and signed by the person filing the submission on behalf of the third person;

(b) Be concise, and in no case longer than as authorized by the arbitral tribunal;

(c) Set out a precise statement of the third person's position on issues; and

(d) Address only matters within the scope of the dispute.

5. The arbitral tribunal shall ensure that any submission does not disrupt or unduly burden the arbitral proceedings, or unfairly prejudice any disputing party.

6. The arbitral tribunal shall ensure that the disputing parties are given a reasonable opportunity to present their observations on any submission by the third person.

Article 5. Submission by a non-disputing Party to the treaty

1. The arbitral tribunal shall, subject to paragraph 4, allow, or, after consultation with the disputing parties, may invite, submissions on issues of treaty interpretation from a non-disputing Party to the treaty.

2. The arbitral tribunal, after consultation with the disputing parties, may allow submissions on further matters within the

scope of the dispute from a non-disputing Party to the treaty. In determining whether to allow such submissions, the arbitral tribunal shall take into consideration, among other factors it determines to be relevant, the factors referred to in article 4, paragraph 3, and, for greater certainty, the need to avoid submissions which would support the claim of the investor in a manner tantamount to diplomatic protection.

3. The arbitral tribunal shall not draw any inference from the absence of any submission or response to any invitation pursuant to paragraphs 1 or 2.

4. The arbitral tribunal shall ensure that any submission does not disrupt or unduly burden the arbitral proceedings, or unfairly prejudice any disputing party.

5. The arbitral tribunal shall ensure that the disputing parties are given a reasonable opportunity to present their observations on any submission by a non-disputing Party to the treaty.

Article 6. Hearings

1. Subject to article 6, paragraphs 2 and 3, hearings for the presentation of evidence or for oral argument ("hearings") shall be public.

2. Where there is a need to protect confidential information or the integrity of the arbitral process pursuant to article 7, the arbitral tribunal shall make arrangements to hold in private that part of the hearing requiring such protection.

3. The arbitral tribunal shall make logistical arrangements to facilitate the public access to hearings (including where appropriate by organizing attendance through video links or such other means as it deems appropriate). However, the arbitral tribunal may, after consultation with the disputing parties, decide to hold all or part of the hearings in private where this becomes necessary for logistical reasons, such as when the circumstances render any original arrangement for public access to a hearing infeasible.

Article 7. Exceptions to transparency

Confidential or protected information

1. Confidential or protected information, as defined in paragraph 2 and as identified pursuant to the arrangements referred to in paragraphs 3 and 4, shall not be made available to the public pursuant to articles 2 to 6.

2. Confidential or protected information consists of:

(a) Confidential business information;

(b) Information that is protected against being made available to the public under the treaty;

(c) Information that is protected against being made available to the public, in the case of the information of the respondent State, under the law of the respondent State, and in the case of other information, under any law or rules determined by the arbitral tribunal to be applicable to the disclosure of such information; or

(d) Information the disclosure of which would impede law enforcement.

3. The arbitral tribunal, after consultation with the disputing parties, shall make arrangements to prevent any confidential or protected information from being made available to the public, including by putting in place, as appropriate:

(*a*) Time limits in which a disputing party, non-disputing Party to the treaty or third person shall give notice that it seeks protection for such information in documents;

(b) Procedures for the prompt designation and redaction of the particular confidential or protected information in such documents; and

(c) Procedures for holding hearings in private to the extent required by article 6, paragraph 2.

Any determination as to whether information is confidential or protected shall be made by the arbitral tribunal after consultation with the disputing parties.

4. Where the arbitral tribunal determines that information should not be redacted from a document, or that a document should not be prevented from being made available to the public, any disputing party, non-disputing Party to the treaty or third person that voluntarily introduced the document into the record shall be permitted to withdraw all or part of the document from the record of the arbitral proceedings.

5. Nothing in these Rules requires a respondent State to make available to the public information the disclosure of which it considers to be contrary to its essential security interests.

Integrity of the arbitral process

6. Information shall not be made available to the public pursuant to articles 2 to 6 where the information, if made available to the public, would jeopardize the integrity of the arbitral process as determined pursuant to paragraph 7.

7. The arbitral tribunal may, on its own initiative or upon the application of a disputing party, after consultation with the disputing parties where practicable, take appropriate measures to restrain or delay the publication of information where such publication would jeopardize the integrity of the arbitral process because it could hamper the collection or production of evidence, lead to the intimidation of witnesses, lawyers acting for disputing parties or members of the arbitral tribunal, or in comparably exceptional circumstances.

Article 8. Repository of published information

The repository of published information under the Rules on Transparency shall be the Secretary-General of the United Nations or an institution named by UNCITRAL.

Explanatory Note to the UNCITRAL Expedited Arbitration Rules

1. Expedited arbitration is a streamlined and simplified procedure with a shortened time frame, which makes it possible for the parties to reach a final resolution of the dispute in a cost- and time-effective manner. The UNCITRAL Expedited Arbitration Rules (hereinafter the "Expedited Rules") provide a set of rules which parties may agree for expedited arbitration. The Expedited Rules balance on the one hand, the efficiency of the arbitral proceedings and on the other, the need to preserve due process and fair treatment.

2. Article 1(5) of the UNCITRAL Arbitration Rules (hereinafter the "UARs") incorporates the Expedited Rules, which are presented as an appendix to the UARs. The phrase "where the parties so agree" in that paragraph emphasizes the need for the parties' express consent for the Expedited Rules to apply to the arbitration.

3. In the following, any reference to "article(s)" is to that in the Expedited Rules, unless otherwise expressly indicated.

A. Scope of application

Article 1

4. Article 1 provides that express consent of the parties is required for the application of the Expedited Rules.

5. Parties are free to agree on the application of the Expedited Rules at any time even after the dispute has arisen (see model arbitration clause in the annex to the Expedited Rules). For example, parties that had concluded an arbitration agreement or had initiated arbitration under the UARs before the effective date of the Expedited Rules (19 September 2021) can subsequently agree to refer their dispute to arbitration under the Expedited Rules. Likewise, a party may propose to the other party or parties that the Expedited Rules shall apply to the arbitration.

6. However, parties should be mindful of the consequences when changing from non-expedited to expedited arbitration. For

example, a notice of arbitration communicated in accordance with article 3 of the UARs might not meet the requirements in article 4 of the Expedited Rules, whereby the claimant has to communicate proposals for the designation of an appointing authority and for the appointment of a sole arbitrator. Therefore, it would be prudent for the parties to agree on how such requirements could be met, should they agree to refer their dispute to arbitration under the Expedited Rules after the proceedings had begun. Similarly, if a three-member arbitral tribunal was constituted in accordance with the UARs, the parties may wish to consider appointing a sole arbitrator in accordance with article 8. If the constitution of the arbitral tribunal is changed, the parties may also need consider the status of statements and evidence submitted to the former tribunal.

7. Article 1 provides that the UARs generally apply to expedited arbitration, unless and as modified by the Expedited Rules. The phrase "as modified by these Expedited Rules" means that rules in the UARs and the Expedited Rules need to be read in conjunction for the proper conduct of the proceedings. The rules in the UARs are either supplemented or replaced by those in the Expedited Rules. For the avoidance of doubt, the footnote to article 1 provides a list of articles in the UARs that would not apply in the context of expedited arbitration. However, parties retain the flexibility to tailor the rules to their proceedings.

8. As the Expedited Rules are presented as an appendix to the UARs, reference to "the Rules" or "these Rules" in the UARs (see articles 1(2), 1(3), 1(4), 2(6), 4(2), 6(3), 6(4), 6(5), 10(3), 17(1), 17(2), 30(1), 30(2), 32 and 41(4)(b) of the UARs) include the Expedited Rules in the context of expedited arbitration.

9. In relation to article 1(2) of the UARs, parties to an arbitration agreement concluded before the entry into force of the Expedited Rules will not be presumed to have referred their dispute to the Expedited Rules, even though the Expedited Rules are presented as an appendix to the UARs. This is because the Expedited Rules only apply when so expressly agreed by the parties. On the other hand, if a subsequent version of the Expedited Rules were to be prepared, article 1(2) of the UARs would apply, which means that the Expedited Rules in effect on the date of commencement of the expedited arbitration would apply, unless the parties have agreed on the current or any other version of the Expedited Rules.

Article 2

10. Even when the parties had initially agreed to refer their dispute to arbitration under the Expedited Rules, the circumstances may be such that the Expedited Rules are not appropriate to resolve the particular dispute. Article 2 addresses such circumstances, with paragraph 1 allowing parties to agree to withdraw from expedited arbitration.

11. In accordance with paragraph 2, a party that had agreed to refer the dispute to arbitration under the Expedited Rules may subsequently request withdrawal from expedited arbitration, when the circumstances evolved in a manner that would make expedited arbitration no longer suitable for resolving the dispute (see also para. 91 below). While there is no time limit within which a party can request withdrawal, the arbitral tribunal should consider at which stage of the proceedings the request is being made.

12. The phrase "in exceptional circumstances" means that the party requesting withdrawal should provide convincing and justified reasons for the request and that the arbitral tribunal should uphold the request only in certain circumstances. It introduces a high threshold for allowing a unilateral withdrawal from expedited arbitration.

13. When making the determination, the arbitral tribunal should consider whether the Expedited Rules are no longer appropriate for the resolution of the dispute. It may wish to take into account, among others, the following:

- The urgency of resolving the dispute;

- The stage of the proceedings at which the request is made;

The complexity of the dispute (for example, the anticipated volume of documentary evidence and the number of witnesses);

- The anticipated amount in dispute (the sum of claims made in the notice of arbitration, any counterclaim made in the response thereto as well as any amendment or supplement);

- The terms of the parties' agreement to expedited arbitration and whether the current circumstances were foreseeable at the time of the agreement; and

- The consequences of the determination on the proceedings.

14. The above is a non-exhaustive list of elements that can be taken into account and it would not be necessary for the arbitral tribunal to consider all the elements therein.

15. When making the determination, the arbitral tribunal, in accordance with article 17(1) of the UARs, may decide that the Expedited Rules in their entirety would no longer apply or that certain articles would no longer apply to the arbitration. When deciding that certain articles of the Expedited Rules would no longer apply, the arbitral tribunal should make clear to the parties how the arbitration would be conducted and on the basis of which articles.

16. If the arbitral tribunal is not yet constituted, the determination would need to be made after it is constituted. However, if the parties are not able to reach an agreement on the arbitrator or if there is a disagreement between the parties on (i) whether the Expedited Rules apply or (ii) whether the criteria in the arbitration agreement triggering the application of the Expedited Rules are met, the appointing authority may be involved in constituting the arbitral tribunal in accordance with article 10(3) of the UARs. The appointing authority will make a prima facie decision on whether the arbitration would be conducted under the Expedited Rules. However, the ultimate determination on the application of the Expedited Rules would be left to the arbitral tribunal.

17. When the Expedited Rules no longer apply to the arbitration pursuant to paragraph 1 or 2, the arbitral tribunal shall conduct the arbitration in accordance with the UARs. However, this does not mean that the arbitral tribunal, if already constituted, would have to be re-constituted in accordance with the UARs. Instead, the arbitral tribunal shall remain in place in accordance with paragraph 3. There may, however, be instances where the parties agree to replace an arbitrator or reconstitute the arbitral tribunal. There may also be instances where the arbitrator resigns, for example, if the arbitrator appointed under the Expedited Rules believes that his or her schedule of future commitments does not allow for the conduct of non-expedited arbitration.

18. Unless the arbitral tribunal decides otherwise, the nonexpedited proceeding should resume at the stage where the expedited proceeding was when the parties agreed to withdraw or the arbitral tribunal made the determination. Decisions made during the expedited proceeding should remain applicable to the non-expedited proceedings, unless the arbitral tribunal decides to depart from its earlier decisions or from a decision made by the previous tribunal.

B. General provision on expedited arbitration

19. Considering that a fair and efficient resolution of the dispute is a common goal of arbitration under both the UARs and the Expedited Rules, article 3 highlights the expeditious nature of the proceedings under the Expedited Rules and emphasizes the obligation of the parties and the arbitral tribunal to act expeditiously.

20. Paragraph 1 is a reminder to parties that when referring their dispute to arbitration under the Expedited Rules, they are agreeing to cooperate in ensuring the efficiency of the proceeding as well as for a swift resolution of the dispute, particularly in an ad hoc setting where there is no administering institution to further expedite the process.

21. Paragraph 2 should be read along with article 17(1) of the UARs. Therefore, arbitral tribunals in expedited arbitration have the same duty to conduct the proceedings so as to avoid unnecessary delay and expense and to provide a fair and efficient process to resolve the dispute. The arbitral tribunal should also comply with any due process requirements.

22. When conducting arbitration under the Expedited Rules, arbitral tribunals should be mindful of the objectives of the Expedited Rules, of the parties' intentions and expectations when they chose the Expedited Rules and of the time frames therein, particularly those in article 16 with regard to the rendering of the award. The annex to the Expedited Rules includes a model statement which parties could request the arbitrator to add to the statement of independence. The model statement highlights that the arbitrator would conduct the arbitration expeditiously and in accordance with the time frames in the UARs and the Expedited Rules.

23. Designating and appointing authorities as well as arbitral institutions administering arbitration under the Expedited Rules should also be mindful of the objectives of the Expedited Rules as well as any applicable time frames (see para. 58 below).

24. Paragraph 3 emphasizes the discretion provided to the arbitral tribunal to make use of a wide range of technological means to conduct the proceeding, including when communicating with the parties and when holding consultations and hearings. It also mentions that consultations and hearings can be held without the physical presence of the participants and in different locations. The inclusion of such a rule in the Expedited Rules does not imply that the use of technological means is

available to arbitral tribunals only in expedited arbitration. The rule aims to assist the arbitral tribunal in streamlining the proceedings and avoiding unnecessary delay and expense, both of which are in line with the objectives of expedited arbitration. The arbitral tribunal should be mindful that the use of technological means is subject to the rules in the UARs to provide for a fair proceeding and to give each party a reasonable opportunity to present its case. Thus, the arbitral tribunal should also be mindful of any due process requirements. In that light, the arbitral tribunal should give the parties an opportunity to express their views on the use of such technological means and consider the overall circumstances of the case, including whether such technological means are at the disposal of the parties.

C. Notice of arbitration, response thereto, statements of claim and defence

Article 4

25. Article 4 addresses the initiation of recourse to arbitration by the claimant and modifies articles 3(4) and 20(1) of the UARs.

26. Two elements, which are optional under article 3(4) of the UARs, are required in the notice of arbitration under the Expedited Rules. This is to facilitate the speedy constitution of the arbitral tribunal in expedited arbitration. In accordance with paragraph 1, the claimant is required to propose an appointing authority (unless the parties have previously agreed thereon) and the appointment of the arbitrator. It is important for the claimant to include such information in its notice of arbitration because the 15-day time frames in articles 6 and 8 begin with the receipt by the respondent of the respective proposals.

27. A proposal for the appointment of the arbitrator does not mean that a party needs to put forward the name of the arbitrator; rather, a party may suggest a list of suitable candidates or qualifications, or a mechanism to be used by the parties for agreeing on the arbitrator. This would also cater for cases where the parties agreed to more than one arbitrator in expedited arbitration.

28. To further expedite the process, paragraph 2 requires the claimant to communicate its statement of claim along with its notice of arbitration. This modifies the rule in article 20(1) of the UARs, which provides that the statement of claim should be communicated within a period of time to be determined by the arbitral tribunal.

29. In summary, when initiating recourse to expedited arbitration, the claimant needs to include the following in its notice of arbitration and the statement of claim:

- A demand that the dispute be referred to arbitration (UARs art. 3(3)(a));

- The names and contact details of the parties (UARs arts. 3(3)(b) and 20(2)(a));

- Identification of the arbitration agreement that is invoked (UARs art. 3(3)(c)) and a copy thereof (UARs art. 20(3));

- Identification of any contract or other legal instrument out of or in relation to which the dispute arises (UARs art. 3(3) (*d*)) and a copy thereof (UARs art. 20(3)) – in the absence of such contract or instrument, a brief description of the relevant relationship (UARs art. 3(3)(*d*));

- A brief description of the claim and an indication of the amount involved, if any (UARs art. 3(3)(e));

- The relief or remedy sought (UARs arts. 3(3)(f) and 20(2)(d));

- A proposal as to the language and place of arbitration, if the parties have not previously agreed thereon (UARs art. 3(3)(g));

- A proposal for the designation of an appointing authority, unless the parties have previously agreed thereon (Expedited Rules art. 4(1)(a));

- A proposal for the appointment of an arbitrator (Expedited Rules art. 4(1)(b));

- A statement of the facts supporting the claim (UARs art. 20(2)(b));

- The points at issue (UARs art. 20(2)(c));

- The legal grounds or arguments supporting the claim (UARs art. 20(2)(e)); and

 As far as possible, all documents and other evidence relied upon by the claimant, or references to them (UARs art. 20(4)).

30. In light of article 7 which provides a default rule of a sole arbitrator, the claimant would not need to propose the number of arbitrators in its notice of arbitration, unless it wishes to suggest the constitution of an arbitral tribunal of more than one arbitrator.

31. With respect to the last item on the above list, the objective is to require the presentation of the complete case for the sake of efficiency. It does not, however, mean that all evidence has to be communicated at this stage, which may be burdensome and counterproductive. This is highlighted by the words "as far as possible" and the claimant may decide to make reference to the evidence to be relied upon. For example, witness statements need not be submitted at this stage. The claimant could instead identify in its statement of claim: (i) any witness whose testimony it would rely on; (ii) the subject matter of the testimony; and (iii) any subject matter for which the claimant intends to submit expert reports. It would be preferable to determine which evidence is to be submitted during the consultation between the arbitral tribunal and the parties (see para. 62 below).

32. The claimant may elect to treat its notice of arbitration as its statement of claim, as long as the notice of arbitration complies with the requirement of the statement of claim (see second sentence of article 20(1) of the UARs). In that case, the claimant would be communicating a single document combining its notice of arbitration and statement of claim.

33. Paragraph 3 requires the claimant to communicate its notice of arbitration and statement of claim to the arbitral tribunal as soon as it is constituted. In the case that the arbitral tribunal consists of more than one arbitrator, the claimant would, in practice, communicate its notice of arbitration and statement of claim to each of the arbitrators upon his or her appointment.

Article 5

34. Article 5 addresses the actions required by the respondent upon receipt of a notice of arbitration and a statement of claim from the claimant. It envisages a two-stage reply with a shorter time frame for the response to the notice of arbitration (hereinafter the "response") and a longer one for the statement of defence. This is to facilitate the speedy constitution of the arbitral tribunal and to provide sufficient time for the respondent to prepare its case.

35. The respondent is required to communicate a response within 15 days of the receipt of the notice. Article 5(1) thus modifies article 4(1) of the UARs, which provides for a 30-day time frame. A shorter time frame is imposed on the response, as it addresses procedural issues, in particular those relating to the constitution of the arbitral tribunal.

36. The response shall respond to the information set forth in the notice of arbitration. As article 4(1) of the Expedited Rules requires a claimant to include in its notice of arbitration proposals on an appointing authority and on the appointment of the arbitrator, the respondent is required to include a response to those proposals. If the respondent disagrees with the proposals, the respondent is free to make its own proposals in accordance with article 4(2)(b) and (c) of the UARs.

37. In summary, the respondent would need to provide, within 15 days of the receipt of the notice of arbitration, the following in the response:

- The name and contact details of each respondent (UARs art. 4(1)(a));

- A response to the information set forth in the notice of arbitration, pursuant to article 3(3)(c) to (g) of the UARs (UARs art. 4(1)(b)); and

- A response to the information set forth in the notice of arbitration, pursuant to article 4(1)(a) and (b) of the Expedited Rules (Expedited Rules art. 5(1)).

38. To provide the respondent with sufficient time to prepare its statement of defence and to ensure equality of the process, the respondent has 15 days from the constitution of the arbitral tribunal to communicate its statement of defence. Article 5(2)introduces a 15-day time frame in contrast to article 21(1) of the UARs, which provides that the statement of defence shall be communicated within a period of time determined by the arbitral tribunal. If the respondent requests for additional time, the arbitral tribunal may extend the 15-day time frame in accordance with article 10.

39. The respondent may elect to treat its response to the notice of arbitration as its statement of defence, as long as the response complies with the requirement of article 21(2) of the UARs (see second sentence of article 21(1) of the UARs).

D. Designating and appointing authorities

40. The appointing authority has a significant role in expediting the proceedings, especially with regard to the constitution of the arbitral tribunal. Therefore, it is important that the parties agree on the choice of an appointing authority (see model arbitration clause, paragraph (a)). When the parties have not agreed on that choice, article 6 of the Expedited Rules provides a mechanism for the Secretary-General of the Permanent Court of Arbitration (PCA) to designate an appointing authority or to serve as one, both of which would lead to an earlier engagement of the appointing authority.

41. Article 6(1) simplifies the process provided for in article 6(2) of the UARs by allowing a party to request the Secretary-General of the PCA to serve as the appointing authority. It provides a streamlined and flexible process, while providing a level of discretion to the Secretary-General of the PCA.

42. The process is accelerated by allowing any party to engage with the Secretary-General of the PCA any time after 15 days have lapsed from the receipt by all parties of a proposal on an appointing authority. In practice, this means that a claimant that has included in its notice of arbitration a proposal for an appointing authority in accordance with article 4(1) is able to make the request to the Secretary-General of the PCA immediately upon the lapse of the 15-day time frame in article 5(1).

43. It should, however, be noted that article 5(1) provides the respondent 15 days to respond to the notice of arbitration, which should also include a response to the proposal for an appointing authority. Therefore, it would be prudent for the claimant to consider such response before engaging the Secretary-General of the PCA. In any case, the Secretary-General of the PCA in exercising its functions under article 6(1) would be required to give the parties an opportunity to present their views, including any proposals on the appointing authority.

44. Similar to article 6(1), article 6(2) modifies article 6(4) of the UARs and allows a party to request the Secretary-General of the PCA to designate a substitute appointing authority or to serve as one, where the appointing authority refuses or fails to act. However, this would not be possible when the Secretary-General of the PCA is already serving as the appointing authority.

45. Paragraph 3 provides a level of discretion to the Secretary-General of the PCA to address practical questions that could arise, for example, (i) when a party has previously rejected or rejects a proposal for the Secretary-General of the PCA to serve as appointing authority; (ii) when a party requests the Secretary-General of the PCA to serve as appointing authority and the other party requests it to serve as designating authority; and (iii) when a party requests the Secretary-General of the PCA to either designate an appointing authority or to serve as an appointing authority.

46. Paragraphs 1, 3, 5, 6 and 7 of article 6 of the UARs continue to apply to expedited arbitration.

E. Number of arbitrators

47. Article 7 provides that an arbitral tribunal composed of a single arbitrator is the default rule in expedited arbitration. As such, article 7(1) of the UARs is replaced by article 7 of the Expedited Rules. Parties, however, can agree on more than one arbitrator, in light of the particulars of the dispute and if collective decision-making is preferred. However, they should be mindful that proceedings involving an arbitral tribunal composed of more than one arbitrator may be less expeditious (see para. 59 below).

48. When the parties have referred their dispute to arbitration under the Expedited Rules and there is no separate agreement on the number of arbitrators, the appointing authority should not have any role in determining that number and should appoint a sole arbitrator in accordance with articles 7 and 8. While the appointing authority may make a prima facie decision on whether the arbitration is to be conducted under the Expedited Rules, the ultimate determination on the application of the Expedited Rules would be left to the arbitral tribunal (see para. 16 above).

49. Article 7(2) of the UARs continues to apply in expedited arbitration when the parties agreed to constitute the arbitral tribunal with more than one arbitrator.

F. Appointment of the arbitrator

50. Article 8 addresses how a sole arbitrator is to be appointed in expedited arbitration. If the parties agreed on more than one arbitrator, articles 9 and 10 of the UARs apply.

51. Paragraph 1 encourages the parties to reach an agreement on the sole arbitrator.

52. Paragraph 2 provides a mechanism in the absence of an agreement by the parties on a sole arbitrator. Any party may request the engagement of the appointing authority 15 days after a proposal for the appointment of a sole arbitrator has been received by all other parties. This is shorter than the 30-day time frame in article 8(1) of the UARs. The involvement of the appointing authority can only be triggered by a request by one of the parties.

53. Considering that the claimant is required to include a proposal for the appointment of a sole arbitrator in the notice of arbitration (see article 4(1) and para. 27 above), if there is no agreement within 15 days after the respondent's receipt of the notice of arbitration, the claimant would be able to make a request to the appointing authority, if previously agreed by the parties. If a proposal is not included in the notice, the 15-day time frame would commence when the proposal is made.

54. It should, however, be noted that article 5(1) provides the respondent 15 days to respond to the notice of arbitration, which should also include a response to the claimant's proposal for the appointment of a sole arbitrator. Therefore, it would be prudent for the claimant to consider the response before engaging with the appointing authority. If the respondent foresees that an agreement cannot be reached, it could also engage with the appointing authority at the same time it communicates the response to the notice of arbitration.

55. If there is no agreement by the parties on the appointing authority and the sole arbitrator 15 days after the receipt of the notice by the respondent, any party may request the Secretary-General of the PCA to designate the appointing authority or to serve as appointing authority in accordance with article 6(1). In the latter case, a party can also request the appointment of a sole arbitrator in accordance with article 8(2), which would likely facilitate a speedy constitution of the arbitral tribunal.

56. Article 8(2) of the UARs, which mentions a list-procedure for the appointment of a sole arbitrator, also applies to expedited arbitration.

57. In exercising the functions under the Expedited Rules, the appointing authority and the Secretary-General of the PCA should be mindful of article 6(5) of the UARs, which requires them to give the parties and the arbitrators (if appointed and

when appropriate) an opportunity to present their views. Any proposal made and comments thereon by the parties on the appointment of a sole arbitrator should thus be taken into account.

58. When appointing an arbitrator for expedited arbitration, the appointing authority shall make an effort to secure not only an independent and impartial arbitrator in accordance with article 6(7) of the UARs but also an arbitrator who is available and ready to conduct the arbitration expeditiously in accordance with article 3(2) of the Expedited Rules. The appointing authority may wish to require the prospective arbitrator to make a statement as provided in the annex to the Expedited Rules.

59. The time frames in article 9 of the UARs on the constitution of a three-member arbitral tribunal apply to expedited arbitration. However, parties may wish to reduce the time frames therein to expedite the constitution of a three-member arbitral tribunal.

G. Consultation with the parties

60. Consultation between the arbitral tribunal and the parties at an early stage of the proceedings is particularly key to an efficient and fair organization of expedited arbitration. The terms "consult" and "consultation" are used in article 9 to highlight the interactive nature of the engagement between the arbitral tribunal and the parties when discussing how the arbitration would be conducted. In general, the phrase "after inviting the parties to express their views" is used throughout the UARs as well as in articles 2, 3, 10, 11, 14 and 16 of the Expedited Rules to refer to a situation where the arbitral tribunal is required to give the parties an opportunity to express their support, concerns or objections before the arbitral tribunal takes a decision on a certain matter.

61. Article 9 requires the arbitral tribunal to consult the parties on how to organize the proceedings. It thus conveys the expectation that the arbitral tribunal will engage actively with the parties rather than to simply invite them to express their views. A case management conference is one way of conducting such consultation and can be an important procedural tool, particularly in expedited arbitration, as it permits an arbitral tribunal to give parties a timely indication as to the organization of the proceedings and the manner in which it intends to proceed. 62. A number of issues could be discussed during consultations so as to create a basis for a common understanding of the proceedings, for example: (i) a list of points at issue including those that need to be addressed with priority; (ii) the need for further written statements and evidence; (iii) whether and how to conduct further consultations as well as hearings, including whether they would be in person or using technological means, including remotely; (iv) other procedural issues as well as the timetable. Similarly, if the parties indicate that they intend to present witnesses, whether statements by witnesses shall be in writing and the time for the presentation of the witness statements could be discussed during consultations.

Article 9 introduces a short time frame within which the 63. arbitral tribunal should consult the parties as it is useful for this to be done at the very early stages of the proceedings. The arbitral tribunal should conduct the consultation with the parties promptly after and within 15 days of its constitution. In certain cases, the respondent might not yet have communicated its statement of defence as it is to be communicated within 15 days of the constitution of the arbitral tribunal (see article 5(2)). Nonetheless, it would be useful for the arbitral tribunal to consult the parties at an early stage based on the notice of arbitration, response thereto as well as the statement of claim. Upon receipt of the statement of defence from the respondent. further consultations may be required, particularly if an agreement on a provisional timetable has been deferred pending the arbitral tribunal's review of the statement of defence or if the agreed timetable requires an update following such review.

64. Consultations may be conducted through a meeting in person, in writing, by telephone or videoconference or other means of communication as provided for in article 3(3). Considering that sufficient flexibility is provided to the arbitral tribunal, it should not be burdensome to meet the 15-day time frame in article 9.

65. In accordance with article 17(2) of the UARs, the arbitral tribunal should establish the provisional timetable. In so doing, the arbitral tribunal should be mindful of the time frames in the Expedited Rules, in particular those in article 16. Similarly, following the consultations, the arbitral tribunal should communicate to the parties the outcome of the consultations to ensure that the parties are fully aware of the time frames and avoid delays.

H. Time frames and the discretion of the arbitral tribunal

66. Article 10 addresses the discretion of the arbitral tribunal with regard to time frames in expedited arbitration. It should be read along with the second sentence of article 17(2) of the UARs.

67. Article 10 clarifies that the arbitral tribunal may extend or abridge any period of time prescribed under the UARs and the Expedited Rules or agreed by the parties. Even after a time frame has been fixed in accordance with article 10, flexibility is provided to adjust the time period when the adjustment is justified. However, this discretion is subject to article 16, which provides a specific rule with regard to the time frames for rendering the award and their extensions (see paras. 84–92 below).

68. Article 10 clarifies and reinforces the discretionary power of the arbitral tribunal to adapt the proceedings to the circumstances of the case, further limiting the risk of challenges at the enforcement stage. In other words, it provides the arbitral tribunal with a robust mandate to act decisively without fearing that its award could be set aside for a breach of due process.

69. While shorter time frames constitute one of the key characteristics of expedited arbitration, arbitral tribunals should preserve the flexible nature of the proceedings and comply with due process requirements.

70. With regard to the consequences of non-compliance by the parties with the time frames, article 30 of the UARs on default applies to expedited arbitration. With regard to late submissions, considering that flexibility is provided to the arbitral tribunal in setting and modifying time frames, the arbitral tribunal can reject or disregard such submissions, while such discretion should be exercised with care.

I. Hearings

71. Article 11 emphasizes the discretionary power of the arbitral tribunal to not hold hearings in expedited arbitration in the absence of a request by any party. It should be read together with article 17(3) of the UARs, which provides that: (i) the arbitral tribunal shall hold hearings if any party so requests at an appropriate stage of the proceedings; and (ii) in the absence of such a request, the arbitral tribunal shall decide whether to hold hearings. Parties themselves may agree to hold hearings, in which case that agreement is binding on the arbitral tribunal.

72. A hearing may cause delays particularly if the scheduling of the parties and the arbitral tribunal need to be coordinated. A hearing may be useful, however, when witness testimony and expert opinions are critical for the arbitral tribunal's decision-making. Moreover, a direct exchange between the parties and the arbitral tribunal at a hearing (whether in person or remotely) may facilitate a better understanding of the case and make the proceedings more efficient.

73. Considering the short time frame of six months for rendering the award in expedited arbitration, the arbitral tribunal may wish to decide at an early stage of the proceedings whether to hold hearings. A request to hold a hearing at a later stage may delay the proceedings and may have a negative impact on the arbitral tribunal complying with that time frame.

74. As parties have a right to request the holding of a hearing, article 11 requires the arbitral tribunal to invite the parties to express their views on whether hearings are to be held. This may also be done during the consultation with the parties (see para. 62 above). If a party so requests at that stage, the arbitral tribunal will need to hold a hearing in accordance with article 17(3) of the UARs. In the absence of such a request prior to and during the consultation, the arbitral tribunal may go ahead and decide to not hold a hearing.

75. This means that the proceedings shall be conducted on the basis of documents and other materials. A request by a party to hold a hearing after a decision by the arbitral tribunal to not hold one can be denied as the request might no longer be considered as being made at "an appropriate stage of the proceedings" (see article 17(3) of the UARs). Insofar, article 11 could have the effect of limiting the time frame during which a request for holding a hearing can be made.

76. As provided for in article 3(3) of the Expedited Rules and article 28(4) of the UARs, the arbitral tribunal may utilize any technological means to hold hearings without the physical presence of the parties or witnesses, including remotely. The remaining paragraphs of article 28 of the UARs also apply to the conduct of hearings in expedited arbitration. The arbitral tribunal has a broad discretion on how to conduct the hearings in a streamlined manner. Efforts should be made to limit the duration of the hearings, the number of witnesses as well as cross-examination and at the same time, to maintain due process.

J. Counterclaims and claims for the purpose of set-off

77. Article 12 preserves the right of the parties to make counterclaims and claims for the purpose of set-off (hereinafter referred to as "counterclaims"), but introduces certain qualifications, which can be lifted by the arbitral tribunal. This is to ensure that counterclaims do not result in delays in expedited arbitration.

78. Article 12 replaces article 21(3) of the UARs and introduces a higher threshold for counterclaims. Paragraph 1 requires the respondent to make any counterclaim at the latest in its statement of defence. A counterclaim can be made at a later stage of the proceedings, but only when the arbitral tribunal considers it appropriate under the circumstances.

K. Amendments and supplements to a claim or defence

79. Article 13 replaces article 22 of the UARs. It introduces a higher threshold for parties to make amendments and supplements to a claim or defence, including a counterclaim or a claim for the purposes of set-off (hereinafter referred to as "amendments") in the context of expedited arbitration. Nonetheless, it provides flexibility in its application to different circumstances. Accordingly, a party is not allowed to make amendments unless the arbitral tribunal considers it appropriate to allow such amendments. When determining whether to allow amendments, the arbitral tribunal should take into account at which stage of the proceedings such a request for the amendment is made, prejudice to other parties in allowing the amendment and any other circumstances.

80. Counterclaims and amendments might result in the expedited arbitration no longer being appropriate for resolving the dispute. In such a circumstance, parties may agree that that the Expedited Rules shall no longer apply to the arbitration or a party may request the arbitral tribunal to determine that the Expedited Rules shall no longer apply in accordance with article 2 (see paras. 10–14 above).

L. Further written statements

81. Article 14 emphasises the discretionary power of the arbitral tribunal under article 24 of the UARs to limit further written

statements. It clarifies that the arbitral tribunal may decide that the statement of claim and the statement of defence are sufficient and that no further written statements are required from the parties. It should, however, not be interpreted that arbitral tribunals do not have such discretion under article 24 of the UARs.

M. Evidence

82. Article 15 clarifies the discretionary power of the arbitral tribunal with regard to taking of evidence in expedited arbitration. Article 27(3) of the UARs provides that the arbitral tribunal may require the parties to produce documents and other evidence during the proceedings. The first sentence of article 15(1) clarifies that the arbitral tribunal may decide which documents or other evidence are to be produced by the parties. The second sentence reaffirms the discretionary power of the arbitral tribunal to not provide for a procedure where a party requests another party to produce documents (often referred to as the "document production" phase). The inclusion of article 15(1) in the Expedited Rules should, however, not be interpreted as meaning that arbitral tribunals do not have such discretion under article 27(3) of the UARs.

83. Article 15(2) provides that in expedited arbitration, statements by witnesses shall be presented in written form and signed by them. Paragraph 2 thus replaces the second sentence of article 27(2) of the UARs. While the rules for meeting the requirements of "in writing" and "signature" through electronic communication vary depending on the jurisdiction, it should be noted that article 9(2) and (3) of the United Nations Convention on the Use of Electronic Communications in International Contracts provides a functional equivalence rule.

N. Period of time for making the award

84. Article 16 provides the time frame for making the award, which refers to the final award. Paragraph 1 provides for a sixmonth time frame for making the award and a mechanism for extending that time frame in certain circumstances. The sixmonth time frame for rendering the award commences with the constitution of the arbitral tribunal. Parties are free to agree on a time frame different from that in paragraph 1, which may be shorter or longer depending on their needs.

85. The general discretion provided to arbitral tribunals under article 10 to extend or abridge any period of time prescribed

under the Expedited Rules and those agreed by the parties is subject to article 16. The first sentence of article 16(2) specifically authorizes the arbitral tribunal to extend the time frame for rendering the award established pursuant to paragraph 1, but only in exceptional circumstances and after inviting the parties to express their views. It would be up to the arbitral tribunal to determine whether the circumstances are exceptional or not. While the arbitral tribunal should generally indicate the reasons when extending the time frame, paragraph 2 does not require reasons so as to provide flexibility to the arbitral tribunal, particularly when the extended time period is rather short.

86. The second sentence of paragraph 2 provides that the maximum overall time frame, including any extended period, should be no longer than nine months from the date of the constitution of the arbitral tribunal. This responds to the expectations of the parties that an award would be rendered within a short time period, one of the key features of expedited arbitration. Paragraph 2, however, does not impose limitations on the number of extensions within the overall time frame. As parties are free to modify any time frame in the Expedited Rules, paragraph 2 also does not prevent the parties from agreeing on a time frame that is longer than nine months.

87. In case the arbitral tribunal considers that it is at risk of not rendering an award within the time frame provided for in paragraph 2, paragraph 3 provides a mechanism whereby that time period could be extended for one last time. This mechanism intends to address a situation where the arbitral tribunal is at risk of not being able to render an award within the time frame, for example, due to unusual circumstances arising near the end of the time frame or if only a short period of time beyond that time frame is required for rendering the award.

88. Parties and the arbitral tribunal should be mindful of the consequence when the time frame in paragraph 2 lapses without an award being rendered. Depending on the applicable law, this may result in the termination of the proceedings or the award rendered subsequently being the subject of possible annulment. In some jurisdictions, such an award might also be refused enforcement. To avoid such situations, paragraph 3 permits the arbitral tribunal to propose to the parties a final extended time limit, stating the reasons for the proposal. In so doing, it must also fix a time period within which the parties should express their views on the proposal. The proposed extension would only be permitted when all parties agree to the extension within the fixed time period. It will be the responsibility of the arbitral tribunal to ascertain that the agreement to its

proposal is expressed without ambiguity. For example, if in response to the proposal, a party agrees only to a time frame shorter than that proposed by the arbitral tribunal, the arbitral tribunal may invite the other parties to express their agreement to such shorter time frame. In addition, if one party agrees to the proposal within the fixed time period and the other party agrees after the time period has lapsed, the arbitral tribunal may wish to consult the parties to confirm whether it could assume that there was agreement by the parties, thus avoiding a possible application of paragraph 4.

89. Paragraph 3 does not set a maximum time frame that can be proposed by the arbitral tribunal. Nonetheless, to obtain the agreement of the parties, the extended time frame requested by the arbitral tribunal should be reasonable taking into account any concerns of the parties, and be sufficient for the arbitral tribunal to render the award.

90. Considering that in certain jurisdictions, extension of the time frame could only be granted upon the agreement or consent of the parties or by an entity other than the arbitral tribunal, paragraphs 2 and 3 underline that parties, by agreeing to the application of the Expedited Rules, are granting the arbitral tribunal the authority to extend the time period established in paragraphs 1 and 2.

91. Paragraph 4 alerts the parties and the arbitral tribunal to the mechanism provided for in article 2(2) of the Expedited Rules in case there is no agreement by the parties to the extension proposed by the arbitral tribunal. In such a case, any party may make a request to the arbitral tribunal that the Expedited Rules no longer apply to the arbitration. Indeed, the arbitral tribunal may wish to suggest this possibility along with its proposal to extend the time period in accordance with paragraph 3 as the consequence. Doing so could avoid a situation where none of the parties makes the request under paragraph 4 despite there being no agreement by the parties on the extension. Paragraph 4 could be particularly useful if one of the parties intentionally delays the proceedings as well as the issuance of the award within the time frame and does not agree to the extension.

92. After inviting the parties to express their views, the arbitral tribunal may determine that the Expedited Rules shall no longer apply to the arbitration, which in effect lifts any time limit for rendering the award in the Expedited Rules including those agreed by the parties. As the arbitral tribunal would have stated the reasons in proposing the extension under paragraph 3, the

arbitral tribunal might consider that exceptional circumstances exist as required under article 2(2) and would not need to repeat the reasons when determining that the Expedited Rules shall not longer apply. Should the arbitral tribunal make the determination under paragraph 4, the arbitral tribunal will remain in place and continue to conduct the arbitration but will do so in accordance with the UARs.

93. Article 16 should also be read together with articles 37 and 38 of the UARs, which respectively provide that the interpretation and the correction form part of the award. If the final award was made within the time frame in article 16, any subsequent interpretation or correction to that award after the lapse of the time frame shall not affect the timeliness of the final award for the purposes of article 16. Similarly, an additional award made in accordance with article 39 of the UARs after the lapse of the time frame in article 16 shall not affect the timeliness of the award made within that time frame.

It should be noted that article 16 does not aim to address 94. the instances of de jure or de facto impossibility of the arbitrator to perform his or her functions. In such a situation, articles 12(3), 13 and 14 of the UARs will likely lead to the termination of the arbitrator's services and his or her replacement. In the case of replacement, article 15 of the UARs provides that the proceedings shall resume at the stage where the arbitrator who was replaced ceased to perform his or her functions. In practice, this would have the effect of suspending the time period in article 16 of the Expedited Rules from the time the replaced arbitrator ceased to perform his or her functions to the date of replacement. If the new arbitrator considers that the remaining time would not be sufficient to render an award, he or she could rely on article 16 to extend the time frame. Also, if an arbitrator is temporarily unable to perform his or her functions and is not replaced, the arbitrator as well as the parties could rely on article 16 to extend the time frame and cope with any delay that may have occurred during such a period.

95. A similar solution applies if the arbitral tribunal suspends the proceedings in accordance with article 43(4) of the UARs, due to non-payment of the required deposits. In that case, the time period in article 16 would cease to run during the suspension.

96. Article 16 should be read together with article 34 of the UARs, in particular paragraph 3, which provides that the parties may agree that no reasons need to be given in the award. This could reduce the time required by the arbitral tribunal in

rendering the award and allow the arbitral tribunal to meet the time frame in the Expedited Rules. However, unless the parties have agreed that no reasons are to be given, arbitral tribunals in expedited arbitration shall state the reasons upon which the award is based. Requiring the arbitral tribunal to provide a reasoned award can assist its decision-making and provide comfort to the parties as they will find that their arguments have been duly considered and would be aware of the basis upon which the award was rendered. The absence of reasoning in an award could have an impact on the control mechanism and its scope, as such reasoning might be necessary for the court or any other competent authority to consider whether some of the grounds for setting aside the award or refusing its recognition and enforcement exist.

O. Model arbitration clause for expedited arbitration

97. The annex to the Expedited Rules contains a model arbitration clause for parties to agree to expedited arbitration under the Expedited Rules. The model arbitration clause notes that the parties should consider adding the appointing authority, the place and the language of arbitration.

98. When considering whether to refer a dispute that has arisen or could arise in the future to arbitration under the Expedited Rules, the parties should take into account, among others, the following elements:

- The urgency of resolving the dispute;

- The complexity of the transactions and the number of parties involved;

- The anticipated complexity of the dispute;
- The anticipated amount of the dispute;

- The financial resources available to the party in proportion to the expected cost of the arbitration;

- The possibility of joinder or consolidation; and

- The likelihood of an award being rendered within the time frames provided in article 16 of the Expedited Rules.

P. The Expedited Rules and the Transparency Rules

99. The suitability of the Expedited Rules for investment arbitration is a question left to the disputing parties, as express consent of the parties is required for the Expedited Rules to apply (see paras. 2, 4 and 5 above). States could refer to and consent to the Expedited Rules in their respective investment treaty, based on which an investor claimant may consent to refer a dispute under the Expedited Rules. However, a reference to the UARs in investment treaties (regardless of whether the reference was included prior to or after the effective date of the Expedited Rules) should not be construed as consent by the State Parties to the Expedited Rules as express consent is necessary for the application of the Expedited Rules.

100. According to article 1(4) of the UARs (as adopted in 2013), the UNCITRAL Rules on Transparency in Treaty-based Investor-State Arbitration ("Transparency Rules") form part of the UARs. Article 1 of the Transparency Rules addresses the applicability of the Transparency Rules to "investor-State arbitration initiated under the UNCITRAL Arbitration Rules". As the Expedited Rules are presented as an appendix to the UARs, an investor-State arbitration initiated under the Expedited Rules should be considered as being initiated under the UARs and therefore, the Transparency Rules could apply.

101. If the investor-State arbitration is initiated pursuant to an investment treaty concluded before 1 April 2014, the Transparency Rules would only apply when the disputing parties have agreed to their application or the States Parties to the treaty have agreed to their application after 1 April 2014. Therefore, even if the disputing parties agree to the application of the Expedited Rules, the proceedings would not be subject to the Transparency Rules unless above-mentioned conditions are met.

102. If the investor-State arbitration is initiated pursuant to an investment treaty concluded on or after 1 April 2014, the Transparency Rules would apply unless the States Parties to the treaty have agreed otherwise. In other words, if States Parties to the treaty have not agreed otherwise and the disputing parties agree to the application of the Expedited Rules, the proceedings would be subject to the Transparency Rules.

103. Parties that have agreed to refer an investor-State dispute to arbitration under the Expedited Rules may agree that the Transparency Rules shall not apply to the arbitration. For example, States could include a reference to the Expedited Rules in their investment treaties, while opting out of the Transparency Rules, for example, by making a reference to (i) the 2010 version of the UARs as modified by the Expedited Rules or (ii) the Expedited Rules without article 1(4) of the UARs.

104. However, the flexibility for the disputing parties to opt out of the Transparency Rules in investor-State arbitration initiated pursuant to an investment treaty concluded on or after 1 April 2014 which includes a reference to the UARs will be restricted, if the States Parties to that treaty have not opted out of the Transparency Rules. For example, if two States conclude a treaty after 1 April 2014 allowing an investor to refer a dispute to the UARs and the States have not opted out of the Transparency Rules, it would not be possible for a claimant investor and the respondent State to agree to the Expedited Rules without being subject to the Transparency Rules.

Q. Time frames in the Expedited Rules

105. The following provides an overview of the different time frames in the Expedited Rules. In the "time frame" column, "A + number (days(d)/months(m))" indicates "within" the number of days/months from stage A (in certain cases, receipt thereof).

	Time frame	Stages of the proceedings and procedural actions	Relevant articles
A		Notice of arbitration (including a proposal for the designation of an appointing authority (A1) and the appointment of a sole arbitrator (A2)) to the respondent	Expedited Rules 4(1)
	A+0d	Statement of claim to the respondent	Expedited Rules 4(2)
В	A+15d	Response to the notice of arbitration (including responses to A1 and A2) to the claimant	Expedited Rules 5(1)
U	15d after A1 or any proposal	If no agreement on the appointing authority, any party may request the Secretary-General of PCA to designate appointing authority or to serve as appointing authority	Expedited Rules 6(1)
D	15d after A2 or any proposal	If no agreement on the arbitrator, any party may request the appointing authority to appoint the sole arbitrator \Rightarrow Appointing authority to appoint as promptly as possible	Expedited Rules 8(2)
ы		Constitution of the arbitral tribunal	Expedited Rules 8; UARs 8 & 9
	E+0d	Claimant to communicate its notice of arbitration & statement of claim to the arbitral tribunal (as soon as it is constituted)	Expedited Rules 4(3)
	E+15d	Consultation with the parties through a case management conference or otherwise (promptly after and within 15 days)	Expedited Rules 9
		Establishment of a provisional timetable (as soon as practicable)	UARs 17(2)
ц	E+15d	Respondent to communicate its statement of defence to the claimant and the arbitral tribunal (possible extension)	Expedited Rules 5(2); 10
	F+0d	Counterclaim or a claim for purposes of set-off to be included in the statement of defence (permitted at a later stage, if tribunal considers that it appropriate)	Expedited Rules 12
U	G E+6m	Making of the award	Expedited Rules 16(1)
	E+9m	Possible extension of the time period for making of the award (exceptional circumstances)	Expedited Rules 16(2)
	E+9m+final extended time limit	Possible extension of the time period for making of the award (at risk of not rendering an award within nine months)	Expedited Rules 16(3)



www.vismoot.org

